

This item was submitted to Loughborough's Institutional Repository (<u>https://dspace.lboro.ac.uk/</u>) by the author and is made available under the following Creative Commons Licence conditions.



For the full text of this licence, please go to: http://creativecommons.org/licenses/by-nc-nd/2.5/

The Nature of Goal Orientations and their Relationships with Performance, Mental Effort and Self-Efficacy

by

Deborah Naudi

Doctoral thesis Submitted in partial fulfillment of the requirements for the award of Doctor of Philosophy of Loughborough University

June 2011

© by Deborah Naudi (2011)

Abstract

The concept of Goal Orientations (GOs), which was initially developed in the area of educational psychology, is becoming more and more popular in organisational psychology. Although research on GOs has come a long way since the 1970s there are still a number of conceptual and operational issues which have yet to be addressed. These include issues with the definition, dimensionality, stability and specificity of GOs. It is considered essential to address these concerns because they greatly influence the reliability, validity and accuracy of GO research. Consequently, one aim of this study was to examine the dimensionality, stability and specificity of GOs. It is believed that once enough evidence regarding these issues is gathered this could be used to develop a comprehensive definition of GOs. This study aims to contribute towards gathering such evidence. GOs have been examined both in terms of profiles as well as in terms of individual GO scales (non-profile perspective) in the past. This study uses both perspectives in order to attempt to provide as much information as possible regarding the issues being investigated. Another aim of this study was to investigate the relationships between GOs and performance, mental effort and self-efficacy. Further knowledge of these relationships is considered to be of benefit to organisations. So as to achieve these aims a longitudinal study, consisting of a survey and an experimental study, was carried out. The survey sample consisted of 641 participants whilst the experimental sample was made up of 73 participants. The participants were mainly Loughborough University students. However, there were also a number of employed and retired individuals participating in the study. Quantitative analysis was deemed to be the most appropriate method of analysis to achieve the aims of this study. Data analyses were carried out using SPSS and Latent Gold software packages. The results point towards GOs not being as general and stable as initially assumed. There is a strong possibility of GOs being domain-specific. Consistent with a number of past studies, the approach GOs were found to be more strongly related to self-efficacy, mental effort and performance on tasks than the avoidance GOs. Moreover, the relationships between GOs and these variables seem to be moderated by task characteristics. The profile analyses results revealed that a number of GO profiles obtained in this study were very similar to those obtained in other studies. A closer look at these common GO profiles indicated that some profiles were consistently more strongly related to self-efficacy, mental effort and performance than others. This study provides a foundation for future research studies to build on in order to better understand the nature of GOs and their relationships with selfefficacy, mental effort and performance.

Key Words: Goal Orientations, Achievement Motivation, Stability, Dimensionality, Specificity, Self-Efficacy, Mental Effort, Performance, Goal Orientation Profiles.

Acknowledgements

I am extremely grateful for being given the opportunity to do this PhD. It has been an incredible experience during which I have had the enormous pleasure of meeting and working with some wonderful people.

A heartfelt thank you goes to Professor John Arnold for his continuous guidance and support and for making this PhD such a positive (and bearable) experience. I would also like to thank John Loan-Clarke, as my director of research, and Professor Kevin Daniels who provided me with invaluable advice and support.

Thank you to all the participants who dedicated their valuable time to participate in this study. Without you this study would not have been possible. I greatly appreciate the help and support given to me by a number of staff at (and associated with) the School of Business and Economics, namely, Dr Cheryl Travers, Dr Alistair Cheyne, Dr Keith Pond, Professor Xiaohui Liu, Michelle Fischer, Dr Amon Chizema, Jill Wells, Dr Fiona Ellis-Chadwick and Ian Herbert.

I would definitely not have managed to complete this PhD without the endless support and encouragement from my mum, dad, Tommy and my friends. Thank you!

I would like to dedicate this thesis to Joe who was right behind me the whole way, through all the ups and downs. Not only did you put up with me when I was stressed but when I felt as though it was all too much you gave me the strength I needed to carry on.

List of Figures x
List of Tables xi
Chapter 1 : Introduction 1
1.0. Motivation at Work1
1.1. Motivation Research1
1.2. Dissertation Structure2
1.2.1. Chapter 22
1.2.2. Chapter 3
1.2.3. Chapter 43
1.2.4. Chapter 54
1.2.5. Chapter 64
1.2.6. Chapter 75
1.2.7. Chapter 85
1.2.8. Chapter 96
1.2.9. Chapter 107
1.2.10. Chapter 117
Chapter 2 : The Development of Goal Orientations8
2.0. The Development of the concept of Goal Orientations8
2.1. How did the concept of Goal Orientations develop?8
2.1.1. Dweck's Conceptualisation of Goal Orientations8
2.1.2. Nicholls' Conceptualisation of Goal Orientations13
2.1.3. Comparison of Dweck's and Nicholls' conceptualisations of Goal Orientations .15
2.2. When was the concept of Goal Orientations introduced into the Organisational
setting?16
2.3. Why is Goal Orientation research in organisations so important?
2.4. Synopsis19
Chapter 3 : Conceptual Inconsistencies in Goal Orientation Research: Definition and
Dimensionality of Goal Orientations

Contents

3.0. Research Issues with regards to Goal Orientations	20
3.1. Definition	20
3.1.1. Different terms, same concepts?	20
3.1.2. What is the definition of Goal Orientations?	23
3.2. The Dimensionality of Goal Orientations	26
3.2.1. The Two-Factor Model of Goal Orientations	27
3.2.2. The Three-Factor Model of Goal Orientations	27
3.2.3. The Four-Factor Model of Goal Orientations (2x2 approach)	31
3.3. Multiple Goal Perspective	41
3.4. Synopsis	49
Chapter 4 : Conceptual Inconsistencies in Goal Orientation Research: S	stability and
Specificity of Goal Orientations	50
4.0. Further Concerns in Goal Orientation Research	50
4.1. The Stability of Goal Orientations	50
4.1.1. Goal Orientations: Traits or States?	51
4.1.2. Goal Orientations: Traits and States?	60
4.2. Generality vs. Specificity of Goal Orientations	69
4.3. Synopsis	76
Chapter 5 : Variables related to Goal Orientations	77
5.0. Introduction	77
5.1. Goal Orientations and Perceptions of Ability	77
5.2. Goal Orientations and Self-Efficacy	78
5.2.1. Self-efficacy as an Antecedent of Goal Orientations	79
5.2.2. Self-efficacy as a Mediator of Goal Orientations and Performance	80
5.2.3. Self-efficacy as a Consequence of Goal Orientations	84
5.2.4. Are the designs and data analytical techniques used in the studies a	bove enough
to infer causality between Self-Efficacy and Goal Orientations?	84
5.2.5. Self-efficacy as a Correlate of Goal Orientations	86
5.2.6. Self-efficacy, Perceptions of Ability and Goal Orientations	89

5.3. Goal Orientations and Motivation	91
5.4. Goal Orientations and Performance	94
5.4.1. Goal Orientations as direct predictors of Performance	95
5.4.2. Goal Orientations as Indirect Predictors of Performance	101
5.4.3. Multiple Goal Orientations as direct predictors of Performance	101
5.5. Synopsis	105
Chapter 6 : Summary of Literature and Research Questions	106
6.0. Introduction	106
6.1. Summary of Literature Review	106
6.2. Summary of Hypotheses and Research Questions for the current study	108
6.2.1. Hypotheses proposed for the current study	108
6.2.2. Research Questions proposed for the current study	108
Chapter 7 : Methodology	112
7.0. Introduction	112
7.1. Panel Survey	112
7.1.1. Sample Characteristics	112
7.1.2. Survey Characteristics	113
7.1.3. Description of the Survey Data Collection Procedure	115
7.1.4. Survey Pilot Study	118
7.1.5. Description of Measures used in the Survey	119
7.2. Experimental Study	124
7.2.1. Sample Characteristics	125
7.2.2. Experiment Characteristics	125
7.2.3. Description of the Experiment Data Collection Procedure	131
7.2.4. Pilot Study for Experiment	133
7.2.5. Description of Measures used in the Experiments	133
7.2.5.1. Verbal and Numerical Test Goal Orientations	134
7.2.5.2. Task Experience	136
7.2.5.3. Practice	136

	7.2.5.4. Self-efficacy	137
	7.2.5.5. Mental Effort	138
	7.2.5.6. Verbal and Numerical Test Reliability and Validity	138
	7.3. Support for the research methods used in this study	140
	7.3.1. Quantitative versus Qualitative Methods	140
	7.3.2. Questionnaires	141
	7.3.3. Experiments	141
	7.3.4. Longitudinal Study	142
	7.4. Ethical Considerations	142
	7.5. Overview of Study	143
	7.6. Synopsis	148
Cl	hapter 8 : Exploratory Analyses	149
	8.0. Introduction	149
	8.1. Descriptive Statistics and Reliability Analyses of General Goal Orientation	is and
	Hypothetical Task Goal Orientations at Time 1 and Time 2	151
	8.2. Descriptive Statistics and Reliability Analyses of Verbal and Numerical Tes	t Goal
	Orientations at Time 1 and Time 2	153
	8.3. Descriptive Statistics and Reliability Analyses of Self-efficacy, Mental Effort,	Task-
	experience, Practice, and Performance, for Verbal and Numerical Tests at Time	1 and
	Time 2	158
	8.4. Exploratory and Confirmatory Factor Analyses of General and Hypothetical Tas	k Goal
	Orientations	164
	8.5. Latent Class Analyses	173
	8.5.1. LCA of General Goal Orientations at Time 1 and Time 2	176
	8.5.2. LCA of Hypothetical Task Goal Orientations at Time 1 and Time 2	181
	8.5.3. LCA of Verbal Test Goal Orientations at Time 1 and Time 2	185
	8.5.4. LCA of Numerical Test Goal Orientations at Time 1 and Time 2	189
	8.6. Intercorrelational Analyses	193
	8.6.1. Correlations between the four Goal Orientations	202

8.6.2. Correlations between Verbal and Numerical test task-experience (at Time 1) and
practice (at Time 2) with self-efficacy, and performance
8.6.3. Correlations between GOs (General, Verbal test, and Numerical test) and
Practice
8.7. Research Questions and Hypotheses addressed by the Correlation Matrices206
8.7.1. Analyses of the relationships between GOs and Self-efficacy
8.7.2. Analyses of the relationships between GOs and Mental Effort
8.7.3. Analyses of the relationships between GOs and Performance
8.8. Synopsis225
Chapter 9 : Further Analyses 227
9.0. Introduction
9.1. Assessing the Stability of General and Task-Specific Goal Orientations over time227
9.1.1. Assessing the Stability of General GOs228
9.1.2. Assessing the Stability of Task-Specific GOs236
9.1.2.4. Comparing the Stability of General Goal Orientations to that of Task-Specific
Goal Orientations245
9.2. Assessing Generality and/or Task-Specificity of Goal Orientations
9.2.1. Assessing the Task-Specificity of Goal Orientation Profiles
9.2.2. Assessing the Task-Specificity of Goal Orientations
9.2.3. Examining Differences between General and Task-Specific Goal Orientation
Profiles
9.2.4. Examining Differences between General and Task-Specific Goal Orientations.268
9.3. Examining the Effects (if any) of the Interactions between Trait and Induced GOs on
Participant Performance272
9.4. Summary of Results277
Chapter 10 : Discussion 289
10.0. Introduction
10.1. Latent Class Analysis of General Goal Orientations and choosing between the 2x2
and 3-factor models of Goal Orientations for the profile analyses

Appendices	63
References	44
11.4. Concluding Remarks	43
11.3. Recommendations for Future Research	41
11.2. Limitations of the Current Study33	37
11.1.2. Practical Implications	35
11.1.1. Theoretical Implications	33
11.1. Theoretical and Practical Implications33	33
11.0. Introduction	33
Chapter 11 : Conclusion 33	33
10.11. Synopsis	31
Orientations on Participants' Performance32	29
10.10. Examining the Effects of the Interactions between Trait and Induced Go	bal
practice?32	28
10.9. Is it best to use Goal Orientations or Goal Orientation profiles in research ar	nd
Goal Orientations32	26
10.8.2. Comparing Participants' General Goal Orientations with their Task-Specif	fic
10.8.1. Assessing the Task-Specificity of Goal Orientations	24
10.8. Assessing the Generality and/or Task-Specificity of Goal Orientations	24
10.7. Examining the Stability of Task-Specific Goal Orientations over Time32	22
10.6. Examining the Stability of General Goal Orientations over Time	19
10.5. Is one Goal Orientation or Goal Orientation Profile better than the others?31	16
10.4. The relationships between Goal Orientations and Performance	09
10.3. The relationships between Goal Orientations and Mental Effort	02
10.2.2. The Relationship between Goal Orientation Profiles and Self-Efficacy30	01
Perspective29	94
10.2.1. Relationships between Goal Orientations and Self-Efficacy from the Non-Profi	ile
10.2. The relationships between Goal Orientations and Self-Efficacy	94

List of Figures

12
33
. 147
. 178
. 180
. 182
. 184
. 186
. 188
. 190
. 192
. 241
. 276

List of Tables

TABLE 3.1. AN EXPLANATION OF THE FOUR TYPES OF GOAL ORIENTATIONS.	. 34
TABLE 3.2. GOAL ORIENTATION PROFILE RESULTS OF STUDIES USING THE 2x2 MODEL	. 45
TABLE 3.3. GOAL ORIENTATION PROFILE RESULTS OF STUDIES USING THE 3-FACTOR MODEL	. 46
TABLE 4.1. TASKS CARRIED OUT AT EACH TIME POINT	. 55
TABLE 4.2. COMPARISONS OF GOAL ORIENTATIONS OVER TIME	. 56
TABLE 4.3. PERCENTAGES OF INDIVIDUALS WHO SHOWED RELIABLE CHANGE (INCREASES/DECREASES) OVER TIME	. 56
TABLE 4.4. COMPARISONS OF GOAL ORIENTATIONS ACROSS TASKS	. 73
TABLE 4.5. PERCENTAGES OF INDIVIDUALS WHO SHOWED RELIABLE CHANGE (INCREASES/DECREASES) ACROSS TASKS	. 74
TABLE 5.1. RESULTS OF STUDIES IN CATEGORY 1	. 81
TABLE 5.2. RESULTS OF STUDIES IN CATEGORY 2	. 83
TABLE 5.3. RESULTS OF STUDIES EXAMINING THE CORRELATIONS BETWEEN SELF-EFFICACY AND GOAL ORIENTATIONS	. 87
TABLE 5.4. SUMMARY OF RESULTS OBTAINED BY STUDIES INVESTIGATING THE RELATIONSHIPS BETWEEN THE 2-FACTOR MODEL C GOAL ORIENTATIONS AND PERFORMANCE. GOAL ORIENTATIONS AND PERFORMANCE.)F . 95
TABLE 5.5. SUMMARY OF RESULTS OBTAINED BY STUDIES INVESTIGATING THE RELATIONSHIPS BETWEEN THE 3-FACTOR MODEL OF	<u>.</u>
GOAL ORIENTATIONS AND PERFORMANCE.	. 96
TABLE 5.6. SUMMARY OF RESULTS OBTAINED BY STUDIES INVESTIGATING THE RELATIONSHIPS BETWEEN THE 4-FACTOR MODEL OF	<u>.</u>
GOAL ORIENTATIONS AND PERFORMANCE.	. 96
TABLE 5.7. SUMMARY OF RESULTS OBTAINED BY STUDIES INVESTIGATING THE CORRELATIONS BETWEEN GOAL ORIENTATIONS ANI	D
Performance.	. 98
TABLE 7.1. MEASURES COMPLETED AT EACH POINT IN TIME	119
TABLE 7.2. COMPARISON OF AGQ-R QUESTIONNAIRE ITEMS WITH ADAPTED GENERAL GOAL ORIENTATION ITEMS. 3	121
TABLE 7.3. COMPARISON OF AGQ-R QUESTIONNAIRE ITEMS WITH ADAPTED HYPOTHETICAL TASK GOAL ORIENTATION ITEMS FOR	R
Таѕк 1	122
TABLE 7.4. COMPARISON OF AGQ-R QUESTIONNAIRE ITEMS WITH ADAPTED HYPOTHETICAL TASK GOAL ORIENTATION ITEMS FOR	ł
Таѕк 2	123
TABLE 7.5. VERBAL AND NUMERICAL TEST INSTRUCTIONS GIVEN TO THE DIFFERENT EXPERIMENTAL GROUPS AT TIME E1 AND TH	ИE
E2	130
TABLE 7.6. MEASURES COMPLETED AT EACH POINT IN TIME	134
TABLE 7.7. COMPARISON OF AGQ-R QUESTIONNAIRE ITEMS WITH ADAPTED VERBAL AND NUMERICAL TEST GOAL ORIENTATION	
ITEMS	135
TABLE 7.8. SURVEY ONLY PARTICIPANT DEMOGRAPHICS AND DATA COLLECTION TIMES	145
TABLE 7.9. DEMOGRAPHICS AND DATA COLLECTION TIMES OF PARTICIPANTS WHO TOOK PART IN THE SURVEY AND EXPERIMENTA	۹L
Study	146
TABLE 8.1. GENERAL GOS AT TIME 1	151

TABLE 8.2. GENERAL GOS AT TIME 2	. 151
TABLE 8.3. HYPOTHETICAL TASK GOS AT TIME 1	. 152
TABLE 8.4. HYPOTHETICAL TASK GOS AT TIME 2	. 153
TABLE 8.5. VERBAL TEST GOS AT TIME 1	. 154
TABLE 8.6. VERBAL TEST GOS AT TIME 2	. 154
TABLE 8.7. NUMERICAL TEST GOS AT TIME 1	. 154
TABLE 8.8. NUMERICAL TEST GOS AT TIME 2	. 155
TABLE 8.9. COMPARISON OF DESCRIPTIVE STATISTICS OBTAINED IN THIS STUDY WITH THOSE OBTAINED IN THE STUDY BY ELLIOT	AND
Микауама (2008)	. 156
TABLE 8.10. SELF-EFFICACY ON VERBAL AND NUMERICAL TESTS AT TIME 1 AND TIME 2	. 158
TABLE 8.11. PAIRED SAMPLES T-TEST TO TEST FOR SIGNIFICANT DIFFERENCES IN SELF-EFFICACY FROM TIME 1 TO TIME 2 ON	
Verbal and Numerical Tests (N≈70)	. 158
TABLE 8.12. MEAN SELF-EFFICACY SCORES IN STUDIES USING THE MSLQ SELF-EFFICACY SUBSCALE	. 159
TABLE 8.13. MENTAL EFFORT ON VERBAL AND NUMERICAL TESTS AT TIME 1 AND TIME 2	. 160
TABLE 8.14. PAIRED SAMPLES T-TEST TO TEST FOR SIGNIFICANT DIFFERENCES IN MENTAL EFFORT ON VERBAL AND NUMERICA	۹L
Tests at Time 1 and Time 2 (N=71)	. 160
TABLE 8.15. TASK EXPERIENCE ON VERBAL AND NUMERICAL TESTS AT TIME 1 AND TIME 2	. 161
TABLE 8.16. PRACTICE ON VERBAL AND NUMERICAL TESTS BETWEEN TIME 1 AND TIME 2	. 162
TABLE 8.17. VERBAL AND NUMERICAL TEST PERFORMANCE AT TIME 1 AND TIME 2	. 162
TABLE 8.18. PAIRED SAMPLES T-TEST TO TEST FOR SIGNIFICANT DIFFERENCES IN VERBAL AND NUMERICAL TEST PERFORMANCE	E
OVER TIME (N=71)	. 163
TABLE 8.19. FACTOR LOADINGS FOR GENERAL GO ITEMS AT TIME 1 (N=628)	. 164
TABLE 8.20. FACTOR LOADINGS FOR HYPOTHETICAL TASK GO ITEMS AT TIME 1 (N=612)	. 165
TABLE 8.21. CONFIRMATORY FACTOR ANALYSES RESULTS FOR GENERAL GOAL ORIENTATIONS AT TIME 1 (N=621)	. 169
TABLE 8.22. CONFIRMATORY FACTOR ANALYSES RESULTS FOR HYPOTHETICAL TASK GOAL ORIENTATIONS AT TIME 1 (N=612)	. 169
TABLE 8.23. CFA VALIDITY RESULTS FOR THE 4-FACTOR MODEL OF GENERAL GOAL ORIENTATIONS AT TIME 1	. 170
TABLE 8.24. CFA RELIABILITY RESULTS FOR THE 4-FACTOR MODEL OF GENERAL GOAL ORIENTATIONS AT TIME 1	. 170
TABLE 8.25. CFA VALIDITY RESULTS FOR THE 4-FACTOR MODEL OF HYPOTHETICAL TASK GOAL ORIENTATIONS AT TIME 1	. 171
TABLE 8.26. CFA RELIABILITY RESULTS FOR THE 4-FACTOR MODEL OF HYPOTHETICAL TASK GOAL ORIENTATIONS AT TIME 1	. 171
TABLE 8.27. LCA OF GENERAL GOS AT TIME 1 (N=632)	. 177
TABLE 8.28. CONDITIONAL PROBABILITIES OF GENERAL GOS WITH MODEL 4	. 177
TABLE 8.29. LCA OF GENERAL GOS AT TIME 2 USING THE 3-FACTOR MODEL OF GOS (N=192)	. 179
TABLE 8.30. CONDITIONAL PROBABILITIES OF GENERAL GOS AT TIME 2 WITH MODEL 3	. 179
TABLE 8.31. LCA OF HYPOTHETICAL TASK GOS AT TIME 1 (N=623)	. 181
TABLE 8.32. CONDITIONAL PROBABILITIES OF HYPOTHETICAL TASK GOS WITH MODEL 4	. 181

TABLE 8.33. LCA OF HYPOTHETICAL TASK GOS AT TIME 2 (N=189)	. 183
TABLE 8.34. CONDITIONAL PROBABILITIES OF HYPOTHETICAL TASK GOS WITH MODEL 3	. 183
TABLE 8.35. LCA OF VERBAL TEST GOS AT TIME 1 (N=71)	. 185
TABLE 8.36. CONDITIONAL PROBABILITIES OF VERBAL TEST GOS WITH MODEL 3	. 186
TABLE 8.37. LCA OF VERBAL TEST GOS AT TIME 2 (N=68)	. 187
TABLE 8.38. CONDITIONAL PROBABILITIES OF VERBAL TEST GOS WITH MODEL 3	. 187
TABLE 8.39. LCA OF NUMERICAL TEST GOS TIME 1 (N=71)	. 189
TABLE 8.40. CONDITIONAL PROBABILITIES OF NUMERICAL TEST GOS WITH MODEL 4	. 189
TABLE 8.41. LCA OF NUMERICAL TEST GOS AT TIME 2 (N=71)	. 191
TABLE 8.42. CONDITIONAL PROBABILITIES OF NUMERICAL TEST GOS WITH MODEL 3	. 191
TABLE 8.43. INTERCORRELATIONAL ANALYSES OF SURVEY STUDY VARIABLES (N=187-631)	. 195
TABLE 8.44. INTERCORRELATIONAL ANALYSES OF EXPERIMENTAL STUDY VARIABLES (N=49-71)	. 196
TABLE 8.45. INTERCORRELATIONAL ANALYSES OF VARIABLES ACROSS SURVEY AND EXPERIMENT (N=49-71)	. 199
TABLE 8.46. TESTING FOR SIGNIFICANT DIFFERENCES BETWEEN CORRELATION COEFFICIENTS ON VERBAL AND NUMERICAL TEST	rs at
TIME 1 AND TIME 2	. 208
TABLE 8.47. TESTING FOR SIGNIFICANT DIFFERENCES BETWEEN CORRELATION COEFFICIENTS ON VERBAL AND NUMERICAL TEST	IS AT
TIME 1 AND TIME 2	. 210
TABLE 8.48. TESTING FOR SIGNIFICANT DIFFERENCES BETWEEN CORRELATION COEFFICIENTS ON VERBAL AND NUMERICAL TEST	rs at
TIME 1 AND TIME 2	. 212
TABLE 8.49. TESTING FOR SIGNIFICANT DIFFERENCES BETWEEN CORRELATION COEFFICIENTS AT TIME 1 AND TIME 2	. 213
TABLE 8.50. ONE-WAY ANOVA TO TEST FOR DIFFERENCES IN SELF-EFFICACY AS A RESULT OF VERBAL TEST GO PROFILES ADOP	TED
(N=71)	. 215
TABLE 8.51. ONE-WAY ANOVA TO TEST FOR DIFFERENCES IN SELF-EFFICACY AS A RESULT OF NUMERICAL TEST GO PROFILES	
ADOPTED (N=71)	. 215
TABLE 8.52. SELF-EFFICACY MEANS ON VERBAL AND NUMERICAL TESTS AT TIME 1 FOR PARTICIPANTS IN EACH GO CLUSTER	. 216
TABLE 8.53. ONE-WAY ANOVA TO TEST FOR DIFFERENCES IN MENTAL EFFORT AS A RESULT OF VERBAL TEST GO PROFILES ADD)PTED
(N=71)	. 219
TABLE 8.54. ONE-WAY ANOVA TO TEST FOR DIFFERENCES IN MENTAL EFFORT AS A RESULT OF NUMERICAL TEST GO PROFILES	
ADOPTED (N=71)	. 219
TABLE 8.55. MENTAL EFFORT MEANS ON VERBAL AND NUMERICAL TESTS AT TIME 1 FOR PARTICIPANTS IN EACH GO CLUSTER.	. 220
TABLE 8.56. ONE-WAY ANOVA TO DIFFERENCES IN PERFORMANCE AS A RESULT OF VERBAL TEST GO PROFILES ADOPTED (N=	71)
	. 224
TABLE 8.57. ONE-WAY ANOVA TO DIFFERENCES IN PERFORMANCE AS A RESULT OF NUMERICAL TEST GO PROFILES ADOPTED	
(N=71)	. 224

TABLE 8.58. MEAN PERFORMANCE SCORE ON VERBAL AND NUMERICAL TESTS AT TIME 1 FOR PARTICIPANTS IN EACH GO CLUSTER.
TABLE 9.1. FREQUENCY TABLE INDICATING THE GENERAL GOAL ORIENTATION PROFILES ADOPTED BY PARTICIPANTS AT TIME 1 AND
TIME 2. (N=183)
TABLE 9.2. OBSERVED FREQUENCIES AND PROPORTIONS EXPECTED TO REMAIN IN THE SAME CLUSTERS (FROM TIME 1 TO TIME 2)
BY CHANCE
TABLE 9.3. TOTAL NUMBERS OF PARTICIPANTS REMAINING IN THE SAME CLUSTER AND TOTAL NUMBER CHANGING CLUSTER FROM
Тіме 1 то Тіме 2
TABLE 9.4. ASSESSING MEAN LEVEL CHANGE OF GENERAL GOS FROM TIME 1 TO TIME 2 231
TABLE 9.5. CORRELATIONS BETWEEN GENERAL GOS AT TIME 1 AND TIME 2 FOR PARTICIPANTS WHO COMPLETED TIME $Q2$
QUESTIONNAIRE 4-8 WEEKS AFTER TIME Q1 QUESTIONNAIRE (N≈93)
TABLE 9.6. CORRELATIONS BETWEEN GENERAL GOS AT TIME 1 AND TIME 2 FOR PARTICIPANTS WHO COMPLETED TIME $Q2$
QUESTIONNAIRE 9-51 WEEKS AFTER TIME Q1 QUESTIONNAIRE (N≈94)
TABLE 9.7. TESTING FOR SIGNIFICANT DIFFERENCES BETWEEN CORRELATION COEFFICIENTS FOR GENERAL GOS AT TIME 1 AND TIME
2 (FOR DIFFERENT TIME INTERVALS BETWEEN TIME Q1 AND TIME Q2 QUESTIONNAIRES)
TABLE 9.8. ASSESSING MEAN LEVEL CHANGE FOR HYPOTHETICAL TASK GOS AT TIME 1 AND TIME 2 (N≈186)
TABLE 9.9. CORRELATIONS BETWEEN HYPOTHETICAL TASK GOS AT TIME 1 AND TIME 2 FOR PARTICIPANTS WHO COMPLETED TIME
Q2 QUESTIONNAIRE 4-8 WEEKS AFTER TIME Q1 QUESTIONNAIRE (N≈99)
TABLE 9.10. CORRELATIONS BETWEEN HYPOTHETICAL TASK GOS AT TIME 1 AND TIME 2 FOR PARTICIPANTS WHO COMPLETED TIME
Q2 questionnaire 9-51 weeks after Time Q1 Questionnaire (N \approx 98)
TABLE 9.11. REPEATED MEASURES ANOVA TO EXAMINE MEAN-LEVEL CHANGE OF VERBAL TEST PERFORMANCE-AVOIDANCE GOS
OVER TIME (N=71)
TABLE 9.12. TESTING FOR SIGNIFICANT DIFFERENCES BETWEEN THE STABILITY COEFFICIENTS OF GENERAL AND TASK-SPECIFIC GOS
and between Hypothetical Task and Aptitude Test GOs
TABLE 9.13. FREQUENCY TABLE INDICATING THE HYPOTHETICAL TASK GOAL ORIENTATION PROFILES ADOPTED BY PARTICIPANTS AT
TIME 1 AND TIME 2. (N=187)
TABLE 9.14. OBSERVED FREQUENCIES AND PROPORTIONS EXPECTED TO REMAIN IN THE SAME CLUSTERS (FROM TIME 1 TO TIME 2)
BY CHANCE
TABLE 9.15. TOTAL NUMBERS OF PARTICIPANTS REMAINING IN THE SAME CLUSTER AND TOTAL NUMBER CHANGING CLUSTER
FROM TIME 1 TO TIME 2
TABLE 9.16. FREQUENCY TABLE INDICATING THE VERBAL TEST GO PROFILES ADOPTED BY PARTICIPANTS AT TIME 1 AND TIME 2.
(N=68)
TABLE 9.17. OBSERVED FREQUENCIES AND PROPORTIONS EXPECTED TO REMAIN IN THE SAME CLUSTERS (FROM TIME 1 TO TIME 2)
BY CHANCE

TABLE 9.18. TOTAL NUMBERS OF PARTICIPANTS REMAINING IN THE SAME CLUSTER AND TOTAL NUMBER CHANGING CLUSTER
FROM TIME 1 TO TIME 2
TABLE 9.19. FREQUENCY TABLE INDICATING THE NUMERICAL TEST GOAL ORIENTATION PROFILES ADOPTED BY PARTICIPANTS AT
TIME 1 AND TIME 2. (N=71)
TABLE 9.20. OBSERVED FREQUENCIES AND PROPORTIONS EXPECTED TO REMAIN IN THE SAME CLUSTERS (FROM TIME 1 TO TIME 2)
BY CHANCE
TABLE 9.21. TOTAL NUMBERS OF PARTICIPANTS REMAINING IN THE SAME CLUSTER AND TOTAL NUMBER CHANGING CLUSTER
FROM TIME 1 TO TIME 2
TABLE 9.22. FREQUENCY TABLE INDICATING THE HYPOTHETICAL TASK AND VERBAL TEST GOAL ORIENTATION PROFILES ADOPTED BY
Participants at Time 1 (N=67)
TABLE 9.23. OBSERVED FREQUENCIES AND PROPORTIONS EXPECTED TO REMAIN IN THE SAME CLUSTERS ACROSS THE HYPOTHETICAL
TASK AND VERBAL TEST BY CHANCE
TABLE 9.24. TOTAL NUMBERS OF PARTICIPANTS REMAINING IN THE SAME CLUSTER AND TOTAL NUMBER CHANGING CLUSTER
ACROSS THE HYPOTHETICAL TASK AND VERBAL TEST
TABLE 9.25. FREQUENCY TABLE INDICATING THE HYPOTHETICAL TASK AND NUMERICAL TEST GOAL ORIENTATION PROFILES
Adopted by Participants at Time 1 (N=67)254
TABLE 9.26. OBSERVED FREQUENCIES AND PROPORTIONS EXPECTED TO REMAIN IN THE SAME CLUSTERS ACROSS THE HYPOTHETICAL
TASK AND NUMERICAL TEST BY CHANCE
TABLE 9.27. TOTAL NUMBERS OF PARTICIPANTS REMAINING IN THE SAME CLUSTER AND TOTAL NUMBER CHANGING CLUSTER
ACROSS THE HYPOTHETICAL TASK AND NUMERICAL TEST
TABLE 9.28. FREQUENCY TABLE INDICATING THE VERBAL AND NUMERICAL TEST GOAL ORIENTATION PROFILES ADOPTED BY
Participants at Time 1 (N=71)
TABLE 9.29. OBSERVED FREQUENCIES AND PROPORTIONS EXPECTED TO REMAIN IN THE SAME CLUSTERS ACROSS THE VERBAL AND
NUMERICAL TESTS BY CHANCE
TABLE 9.30. TOTAL NUMBERS OF PARTICIPANTS REMAINING IN THE SAME CLUSTER AND TOTAL NUMBER CHANGING CLUSTER
ACROSS THE VERBAL AND NUMERICAL TESTS
TABLE 9.31. TESTING FOR SIGNIFICANT DIFFERENCES IN THE CORRELATION COEFFICIENTS OF THE TASK-SPECIFIC GOS
TABLE 9.32. ASSESSING MEAN LEVEL CHANGE OF MASTERY-APPROACH GOS ACROSS THE HYPOTHETICAL TASK AND VERBAL TEST
(N=69)
TABLE 9.33. ASSESSING MEAN LEVEL CHANGE OF PERFORMANCE-APPROACH GO ACROSS THE HYPOTHETICAL TASK AND VERBAL
Теят (N=68)
TABLE 9.34. ASSESSING MEAN LEVEL CHANGE OF PERFORMANCE-AVOIDANCE GO ACROSS THE HYPOTHETICAL TASK AND VERBAL
Теят (N=68)
TABLE 9.35. ASSESSING MEAN LEVEL CHANGE OF MASTERY-APPROACH GO ACROSS THE HYPOTHETICAL TASK AND NUMERICAL
Test (N=69)

TABLE 9.36. ASSESSING MEAN LEVEL CHANGE OF MASTERY-AVOIDANCE GO ACROSS THE HYPOTHETICAL TASK AND NUMERICAL
Test (N=67)
TABLE 9.37. ASSESSING MEAN LEVEL CHANGE OF PERFORMANCE-APPROACH GO ACROSS THE HYPOTHETICAL TASK AND
NUMERICAL TEST (N=68)
TABLE 9.38. ASSESSING MEAN LEVEL CHANGE OF PERFORMANCE-AVOIDANCE GO ACROSS THE HYPOTHETICAL TASK AND
NUMERICAL TEST (N=68)
TABLE 9.39. FREQUENCY TABLE INDICATING THE GENERAL AND HYPOTHETICAL TASK GOAL ORIENTATION PROFILES ADOPTED BY
Participants at Time 1 (N=617)
TABLE 9.40. OBSERVED FREQUENCIES AND PROPORTIONS EXPECTED TO REMAIN IN THE SAME GENERAL AND HYPOTHETICAL TASK
GO CLUSTERS BY CHANCE
TABLE 9.41. TOTAL NUMBERS OF PARTICIPANTS REMAINING IN THE SAME CLUSTER AND TOTAL NUMBER CHANGING CLUSTER
GENERAL AND HYPOTHETICAL TASK GOS
TABLE 9.42. FREQUENCY TABLE INDICATING THE GENERAL AND VERBAL TEST GOAL ORIENTATION PROFILES ADOPTED BY
Participants at Time 1 (N=67)
TABLE 9.43. OBSERVED FREQUENCIES AND PROPORTIONS EXPECTED TO REMAIN IN THE SAME GENERAL AND VERBAL TEST GO
Clusters by Chance
TABLE 9.44. TOTAL NUMBERS OF PARTICIPANTS REMAINING IN THE SAME CLUSTER AND TOTAL NUMBER CHANGING CLUSTER
GENERAL AND VERBAL TEST GOS
GENERAL AND VERBAL TEST GOS 266 TABLE 9.45. FREQUENCY TABLE INDICATING THE GENERAL AND NUMERICAL TEST GOAL ORIENTATION PROFILES ADOPTED BY 266 PARTICIPANTS AT TIME 1 (N=71) 266 TABLE 9.46. OBSERVED FREQUENCIES AND PROPORTIONS EXPECTED TO REMAIN IN THE SAME GENERAL AND NUMERICAL TEST GO 266 TABLE 9.47. TOTAL NUMBERS OF PARTICIPANTS REMAINING IN THE SAME CLUSTER AND TOTAL NUMBER CHANGING CLUSTER 267
GENERAL AND VERBAL TEST GOS 266 TABLE 9.45. FREQUENCY TABLE INDICATING THE GENERAL AND NUMERICAL TEST GOAL ORIENTATION PROFILES ADOPTED BY 266 PARTICIPANTS AT TIME 1 (N=71) 266 TABLE 9.46. OBSERVED FREQUENCIES AND PROPORTIONS EXPECTED TO REMAIN IN THE SAME GENERAL AND NUMERICAL TEST GO 266 TABLE 9.46. OBSERVED FREQUENCIES AND PROPORTIONS EXPECTED TO REMAIN IN THE SAME GENERAL AND NUMERICAL TEST GO 266 TABLE 9.47. TOTAL NUMBERS OF PARTICIPANTS REMAINING IN THE SAME CLUSTER AND TOTAL NUMBER CHANGING CLUSTER 267 TABLE 9.48. ASSESSING MEAN LEVEL CHANGE FOR PERFORMANCE-APPROACH ACROSS THE GENERAL GOS MEASURE AND VERBAL 267
GENERAL AND VERBAL TEST GOS 266 TABLE 9.45. FREQUENCY TABLE INDICATING THE GENERAL AND NUMERICAL TEST GOAL ORIENTATION PROFILES ADOPTED BY 266 PARTICIPANTS AT TIME 1 (N=71) 266 TABLE 9.46. OBSERVED FREQUENCIES AND PROPORTIONS EXPECTED TO REMAIN IN THE SAME GENERAL AND NUMERICAL TEST GO 266 TABLE 9.46. OBSERVED FREQUENCIES AND PROPORTIONS EXPECTED TO REMAIN IN THE SAME GENERAL AND NUMERICAL TEST GO 266 TABLE 9.47. TOTAL NUMBERS OF PARTICIPANTS REMAINING IN THE SAME CLUSTER AND TOTAL NUMBER CHANGING CLUSTER 267 TABLE 9.48. ASSESSING MEAN LEVEL CHANGE FOR PERFORMANCE-APPROACH ACROSS THE GENERAL GOS MEASURE AND VERBAL 269
GENERAL AND VERBAL TEST GOS 266 TABLE 9.45. FREQUENCY TABLE INDICATING THE GENERAL AND NUMERICAL TEST GOAL ORIENTATION PROFILES ADOPTED BY 266 PARTICIPANTS AT TIME 1 (N=71) 266 TABLE 9.46. OBSERVED FREQUENCIES AND PROPORTIONS EXPECTED TO REMAIN IN THE SAME GENERAL AND NUMERICAL TEST GO 266 CLUSTERS BY CHANCE 266 TABLE 9.47. TOTAL NUMBERS OF PARTICIPANTS REMAINING IN THE SAME CLUSTER AND TOTAL NUMBER CHANGING CLUSTER 266 TABLE 9.47. TOTAL NUMBERS OF PARTICIPANTS REMAINING IN THE SAME CLUSTER AND TOTAL NUMBER CHANGING CLUSTER 267 TABLE 9.48. ASSESSING MEAN LEVEL CHANGE FOR PERFORMANCE-APPROACH ACROSS THE GENERAL GOS MEASURE AND VERBAL 269 TABLE 9.49. ASSESSING MEAN LEVEL CHANGE FOR PERFORMANCE-AVOIDANCE ACROSS THE GENERAL GOS MEASURE AND VERBAL 269
GENERAL AND VERBAL TEST GOS 266 TABLE 9.45. FREQUENCY TABLE INDICATING THE GENERAL AND NUMERICAL TEST GOAL ORIENTATION PROFILES ADOPTED BY 266 TABLE 9.46. OBSERVED FREQUENCIES AND PROPORTIONS EXPECTED TO REMAIN IN THE SAME GENERAL AND NUMERICAL TEST GO 266 TABLE 9.46. OBSERVED FREQUENCIES AND PROPORTIONS EXPECTED TO REMAIN IN THE SAME GENERAL AND NUMERICAL TEST GO 266 TABLE 9.47. TOTAL NUMBERS OF PARTICIPANTS REMAINING IN THE SAME CLUSTER AND TOTAL NUMBER CHANGING CLUSTER 267 TABLE 9.47. TOTAL NUMBERS OF PARTICIPANTS REMAINING IN THE SAME CLUSTER AND TOTAL NUMBER CHANGING CLUSTER 267 TABLE 9.48. ASSESSING MEAN LEVEL CHANGE FOR PERFORMANCE-APPROACH ACROSS THE GENERAL GOS MEASURE AND VERBAL 269 TABLE 9.49. ASSESSING MEAN LEVEL CHANGE FOR PERFORMANCE-AVOIDANCE ACROSS THE GENERAL GOS MEASURE AND VERBAL 269 TABLE 9.49. ASSESSING MEAN LEVEL CHANGE FOR PERFORMANCE-AVOIDANCE ACROSS THE GENERAL GOS MEASURE AND VERBAL 269 TABLE 9.49. ASSESSING MEAN LEVEL CHANGE FOR PERFORMANCE-AVOIDANCE ACROSS THE GENERAL GOS MEASURE AND VERBAL 269
GENERAL AND VERBAL TEST GOS 266 TABLE 9.45. FREQUENCY TABLE INDICATING THE GENERAL AND NUMERICAL TEST GOAL ORIENTATION PROFILES ADOPTED BY 266 TABLE 9.46. OBSERVED FREQUENCIES AND PROPORTIONS EXPECTED TO REMAIN IN THE SAME GENERAL AND NUMERICAL TEST GO 266 TABLE 9.46. OBSERVED FREQUENCIES AND PROPORTIONS EXPECTED TO REMAIN IN THE SAME GENERAL AND NUMERICAL TEST GO 266 TABLE 9.47. TOTAL NUMBERS OF PARTICIPANTS REMAINING IN THE SAME CLUSTER AND TOTAL NUMBER CHANGING CLUSTER 267 TABLE 9.48. ASSESSING MEAN LEVEL CHANGE FOR PERFORMANCE-APPROACH ACROSS THE GENERAL GOS MEASURE AND VERBAL 269 TABLE 9.49. ASSESSING MEAN LEVEL CHANGE FOR PERFORMANCE-AVOIDANCE ACROSS THE GENERAL GOS MEASURE AND VERBAL 269 TABLE 9.49. ASSESSING MEAN LEVEL CHANGE FOR PERFORMANCE-AVOIDANCE ACROSS THE GENERAL GOS MEASURE AND VERBAL 269 TABLE 9.50. ASSESSING MEAN LEVEL CHANGE FOR MASTERY-APPROACH ACROSS THE GENERAL GOS MEASURE AND VERBAL 269 TABLE 9.50. ASSESSING MEAN LEVEL CHANGE FOR MASTERY-APPROACH ACROSS THE GENERAL GOS MEASURE AND NUMERICAL 269
GENERAL AND VERBAL TEST GOS. 266 TABLE 9.45. FREQUENCY TABLE INDICATING THE GENERAL AND NUMERICAL TEST GOAL ORIENTATION PROFILES ADOPTED BY 266 TABLE 9.46. OBSERVED FREQUENCIES AND PROPORTIONS EXPECTED TO REMAIN IN THE SAME GENERAL AND NUMERICAL TEST GO 266 TABLE 9.46. OBSERVED FREQUENCIES AND PROPORTIONS EXPECTED TO REMAIN IN THE SAME GENERAL AND NUMERICAL TEST GO 266 TABLE 9.47. TOTAL NUMBERS OF PARTICIPANTS REMAINING IN THE SAME CLUSTER AND TOTAL NUMBER CHANGING CLUSTER 267 TABLE 9.47. TOTAL NUMBERS OF PARTICIPANTS REMAINING IN THE SAME CLUSTER AND TOTAL NUMBER CHANGING CLUSTER 267 TABLE 9.48. ASSESSING MEAN LEVEL CHANGE FOR PERFORMANCE-APPROACH ACROSS THE GENERAL GOS MEASURE AND VERBAL 269 TABLE 9.49. ASSESSING MEAN LEVEL CHANGE FOR PERFORMANCE-AVOIDANCE ACROSS THE GENERAL GOS MEASURE AND VERBAL 269 TABLE 9.49. ASSESSING MEAN LEVEL CHANGE FOR PERFORMANCE-AVOIDANCE ACROSS THE GENERAL GOS MEASURE AND VERBAL 269 TABLE 9.50. ASSESSING MEAN LEVEL CHANGE FOR MASTERY-APPROACH ACROSS THE GENERAL GOS MEASURE AND NUMERICAL 269 TABLE 9.50. ASSESSING MEAN LEVEL CHANGE FOR MASTERY-APPROACH ACROSS THE GENERAL GOS MEASURE AND NUMERICAL 269
GENERAL AND VERBAL TEST GOS 266 TABLE 9.45. FREQUENCY TABLE INDICATING THE GENERAL AND NUMERICAL TEST GOAL ORIENTATION PROFILES ADOPTED BY 266 TABLE 9.46. OBSERVED FREQUENCIES AND PROPORTIONS EXPECTED TO REMAIN IN THE SAME GENERAL AND NUMERICAL TEST GO 266 CLUSTERS BY CHANCE 266 TABLE 9.47. TOTAL NUMBERS OF PARTICIPANTS REMAINING IN THE SAME CLUSTER AND TOTAL NUMBER CHANGING CLUSTER 266 GENERAL AND NUMERICAL TEST GOS 267 TABLE 9.48. ASSESSING MEAN LEVEL CHANGE FOR PERFORMANCE-APPROACH ACROSS THE GENERAL GOS MEASURE AND VERBAL 269 TABLE 9.49. ASSESSING MEAN LEVEL CHANGE FOR PERFORMANCE-AVOIDANCE ACROSS THE GENERAL GOS MEASURE AND VERBAL 269 TABLE 9.49. ASSESSING MEAN LEVEL CHANGE FOR PERFORMANCE-AVOIDANCE ACROSS THE GENERAL GOS MEASURE AND VERBAL 269 TABLE 9.50. ASSESSING MEAN LEVEL CHANGE FOR MASTERY-APPROACH ACROSS THE GENERAL GOS MEASURE AND VERBAL 269 TABLE 9.50. ASSESSING MEAN LEVEL CHANGE FOR MASTERY-APPROACH ACROSS THE GENERAL GOS MEASURE AND NUMERICAL 270 TABLE 9.51. ASSESSING MEAN LEVEL CHANGE FOR MASTERY-APPROACH ACROSS THE GENERAL GOS MEASURE AND NUMERICAL 270
GENERAL AND VERBAL TEST GOS. 266 TABLE 9.45. FREQUENCY TABLE INDICATING THE GENERAL AND NUMERICAL TEST GOAL ORIENTATION PROFILES ADOPTED BY 266 TABLE 9.46. OBSERVED FREQUENCIES AND PROPORTIONS EXPECTED TO REMAIN IN THE SAME GENERAL AND NUMERICAL TEST GO 266 CLUSTERS BY CHANCE 266 TABLE 9.47. TOTAL NUMBERS OF PARTICIPANTS REMAINING IN THE SAME CLUSTER AND TOTAL NUMBER CHANGING CLUSTER 266 GENERAL AND NUMERICAL TEST GOS 267 TABLE 9.48. ASSESSING MEAN LEVEL CHANGE FOR PERFORMANCE-APPROACH ACROSS THE GENERAL GOS MEASURE AND VERBAL 269 TABLE 9.49. ASSESSING MEAN LEVEL CHANGE FOR PERFORMANCE-AVOIDANCE ACROSS THE GENERAL GOS MEASURE AND VERBAL 269 TABLE 9.50. ASSESSING MEAN LEVEL CHANGE FOR MASTERY-APPROACH ACROSS THE GENERAL GOS MEASURE AND VERBAL 269 TABLE 9.50. ASSESSING MEAN LEVEL CHANGE FOR MASTERY-APPROACH ACROSS THE GENERAL GOS MEASURE AND NUMERICAL 270 TABLE 9.51. ASSESSING MEAN LEVEL CHANGE FOR MASTERY-APPROACH ACROSS THE GENERAL GOS MEASURE AND NUMERICAL 270
GENERAL AND VERBAL TEST GOS 266 TABLE 9.45. FREQUENCY TABLE INDICATING THE GENERAL AND NUMERICAL TEST GOAL ORIENTATION PROFILES ADOPTED BY 266 TABLE 9.46. OBSERVED FREQUENCIES AND PROPORTIONS EXPECTED TO REMAIN IN THE SAME GENERAL AND NUMERICAL TEST GO 266 TABLE 9.46. OBSERVED FREQUENCIES AND PROPORTIONS EXPECTED TO REMAIN IN THE SAME GENERAL AND NUMERICAL TEST GO 266 TABLE 9.47. TOTAL NUMBERS OF PARTICIPANTS REMAINING IN THE SAME CLUSTER AND TOTAL NUMBER CHANGING CLUSTER 267 TABLE 9.48. ASSESSING MEAN LEVEL CHANGE FOR PERFORMANCE-APPROACH ACROSS THE GENERAL GOS MEASURE AND VERBAL 269 TABLE 9.49. ASSESSING MEAN LEVEL CHANGE FOR PERFORMANCE-AVOIDANCE ACROSS THE GENERAL GOS MEASURE AND VERBAL 269 TABLE 9.49. ASSESSING MEAN LEVEL CHANGE FOR PERFORMANCE-AVOIDANCE ACROSS THE GENERAL GOS MEASURE AND VERBAL 269 TABLE 9.50. ASSESSING MEAN LEVEL CHANGE FOR MASTERY-APPROACH ACROSS THE GENERAL GOS MEASURE AND NUMERICAL 270 TABLE 9.51. ASSESSING MEAN LEVEL CHANGE FOR MASTERY-APPROACH ACROSS THE GENERAL GOS MEASURE AND NUMERICAL 270 TABLE 9.51. ASSESSING MEAN LEVEL CHANGE FOR MASTERY-AVOIDANCE ACROSS THE GENERAL GOS MEASURE AND NUMERICAL 270 TABLE 9.52. ASSESSING MEAN LEVEL CHANGE FOR PERFORMANCE-APPROACH ACROSS THE GENERAL GOS MEASURE AND NUMERICAL 270 TABLE 9.52. ASSESSING MEAN LEVEL CHANGE FOR PERFORMANCE-APPROACH ACROSS THE GENERAL GOS MEASURE AND NUMERICAL 270

TABLE 9.53	3. Assessing Mean Level Change for Performance-Avoidance across the General GOs measure and	
Nu	MERICAL TEST (N=67)	271
TABLE 9.54	4. Assessing Mean Level Change in GOs across the General GO measures and the Hypothetical Task (N ppi	≈65)
		271
TABLE 9.5	5. Repeated Measures ANOVA to test for changes in Numerical Test Performance from Time 1 to Time	2
DEP	ending on the Experimental Condition participants were assigned to (N=67)	275
TABLE 9.5	6. Summary of Results	278
TABLE 10.	1. COMPARISON OF GO PROFILES OBTAINED IN THIS STUDY WITH THOSE OBTAINED IN PREVIOUS STUDIES	291
TABLE 10.2	2. COMPARISON ACROSS STUDIES FOR RELATIONSHIPS BETWEEN GOAL ORIENTATIONS AND SELF-EFFICACY	295
TABLE 10.3	3. COMPARISON ACROSS STUDIES FOR RELATIONSHIPS BETWEEN GOAL ORIENTATIONS AND MENTAL EFFORT	304
TABLE 10.4	4. COMPARISON ACROSS STUDIES FOR RELATIONSHIPS BETWEEN GOAL ORIENTATIONS AND PERFORMANCE	310
TABLE 10.	5. Relationships between Goal Orientation Profiles and Performance	315

Chapter 1: Introduction

1.0. Motivation at Work

I decided to do this PhD because I was interested in discovering more about motivation at work. I used to work in a call centre where there was a lot of competition to sell and do better than others. Unfortunately, I did not feel motivated to compete and sell more than others because I am not a competitive person. I was more interested in helping customers as best I could. I actually felt uncomfortable working for this organisation because of the strong competitive environment. I knew that there were other employees in this organisation who were not motivated by the competition. This experience led me to believe that if only organisations could be flexible in the way that they motivate employees, both the employees and the organisation would benefit so much more. As a result I was determined to carry out research on employee motivation.

1.1. Motivation Research

At the start of this PhD I read about various concepts being researched in relation to employee motivation including, intrinsic and extrinsic motivation (e.g. Ryan & Deci, 2000; Eccles & Wigfield, 2002), personality and motivation (e.g. Furnham et al., 1999; Judge & Ilies, 2002; Barrick et al., 2002) and the core self-evaluations (e.g. Judge et al., 1997; Erez & Judge, 2001; Judge et al., 1998). The concept of goal orientations however, was the one that interested me most. This is due to the fact that I could really relate to it because of my previous work experience. According to Elliot and Thrash (2001:144) a goal orientation (GO) is "a cognitive representation of a competence-based possibility that an individual seeks to attain", or more simply, goal orientations (GOs) are aims or goals that people have in situations involving competence. The basic model of GOs, which will be discussed in further detail in Chapter 2, suggests that people may adopt mastery or performance

GOs. The focus of the former is on learning and improving whilst the focus of the latter is on doing better than others. I realised that in the call centre they were promoting a performance GO whilst I had more of a mastery GO and this is probably a major reason why I did not feel comfortable working in that environment.

As I read more about the concept of GOs, a number of gaps in research became apparent. The one that bothered me the most was that many researchers assumed that GOs were general, stable traits. However, very little research, if any, explicitly investigated the stability and generality of GOs. A number of other inconsistencies in research also became evident. These included inconsistencies regarding the relationships between GOs and other variables, the definition of GOs, questions regarding the dimensionality of GOs and whether GOs should be investigated as individual GO scales or in terms of profiles. As a result I decided to focus my research on the concept of GOs and investigate these gaps in knowledge (and inconsistencies in research) in order to take knowledge on GOs a step further. In order to maintain clarity and provide the reader with a clear understanding of the structure of this dissertation a summary of each chapter is presented below.

1.2. Dissertation Structure

1.2.1. Chapter 2

This chapter provides an introduction to the concept of GOs. A description of how the GO concept developed is presented and the different perspectives of early GO researchers are reviewed. This chapter also addresses how the concept of GOs may be useful to organisations as well as the importance of GOs in organisational research.

1.2.2. Chapter 3

Chapter 3 illustrates the inconsistencies in research with respect to the definition and dimensionality of GOs. Over the years researchers proposed a number of definitions of GOs and have also used different terms when referring to the concept of GOs (for example, achievement goals and goal orientations). In this chapter a description of the different definitions and terms used is provided and the ones used in the current study are presented.

The dimensionality of GOs refers to how many types of GOs there are. As will be described in further detail in Chapter 3, the model of GOs developed from a 2-factor model (having 2 types of GOs), to a 3-factor model (having 3 types of GOs), to a 2x2 model (having 4 types of GOs). A description of each of these models is provided and research investigating the utility and accuracy of the three models is reviewed. A discussion regarding which model will be used in the current study (and why) is also provided in this chapter.

The notion of multiple GOs is also discussed in this chapter. It has been argued that people can and do adopt more than one GO simultaneously. Consequently there has been debate in literature as to whether GOs should be investigated as individual scales or in terms of GO profiles. Research using the profile perspective is described and a Research Question indicating how this perspective will be investigated further in this study is presented.

1.2.3. Chapter 4

This chapter focuses on inconsistencies in research regarding the stability and specificity of GOs. Some researchers consider GOs to be stable traits whilst others attempt to induce

3

GOs or view them as being more changeable. Also, a number of researchers investigate the interaction between trait and state GOs. These research studies are discussed and relevant Research Questions and Hypotheses for the current study are presented.

Another aspect of the 'lack of stability' of GOs is the idea of GOs being task-specific. A limited number of research studies have been carried out in order to investigate whether participants adopt the same or different GOs on different tasks. These studies are reviewed and, since there are still uncertainties regarding the task-specificity of GOs, Research Questions and Hypotheses are presented thus proposing how the current study will further investigate the task-specificity of GOs.

1.2.4. Chapter 5

In Chapter 5 the inconsistencies in research with respect to the relationships between GOs and self-efficacy, effort and performance are described. Since a number of studies reviewed attempt to investigate GOs as antecedents, moderators, mediators, and consequences of these variables, the problem of determining the direction of causality is discussed. Moreover, Research Questions and Hypotheses are put forward with the aim of clarifying current inconsistencies in research with respect to the relationships between GOs and self-efficacy, mental effort and performance.

1.2.5. Chapter 6

This is a short summary chapter which provides an outline of the issues discussed in the previous four chapters as well as a list of the Research Questions and Hypotheses being addressed in the current study.

1.2.6. Chapter 7

This is the methodology chapter. The research methods used in this study are described in detail and the reasons for using these methods are also provided. A longitudinal study, consisting of a survey and experiments, was considered to be the most appropriate means of data collection for testing the Hypotheses and answering the Research Questions proposed in the current study. Survey participants completed a Time 1 and Time 2 questionnaire whilst experimental participants completed the Time 1 and Time 2 questionnaires and attended two experimental sessions. 641 participants took part in the survey and 73 of these took part in the experiment. A detailed description of the data collection procedure is presented in this chapter along with a description of the measures used and the ethical considerations for the current study.

1.2.7. Chapter 8

Chapter 8 is the first of two results chapters. The main focus of this chapter is to provide information regarding the psychometric properties of the measurement scales used. In order to do this descriptive statistics, reliability analyses, exploratory and confirmatory factor analyses results are presented. These were used to decide which model of GOs should be used in the current study. In order to make a decision regarding model choice for GO **profiles**, Latent Class Analyses, which is similar to cluster analysis, were carried out and the results are presented in this chapter. So as to provide a good basis for understanding the relationships between the variables measured in this study two correlation matrices, one for the non-experimental participants and one for the experimental participants, are presented. A number of correlations of interest, that do not directly relate to any of the Research Questions or Hypotheses but which may provide a better understanding of these, are discussed in this Chapter.

Due to the complexity of the study, in order to make the results chapters more reader friendly and understandable, it was thought best to group the Research Questions and Hypotheses according to their common themes. Five main themes were chosen:

a) Types and characteristics of General GO profiles and model choice with respect to GO profiles

b) The relationships between GOs and other variables (self-efficacy, effort and performance)

c) The stability of GOs over time

d) The task-specificity of GOs

e) Interactions between state and trait GOs.

The results relating to the first theme are provided in this chapter. The correlation matrices presented in this chapter provided the answers to a number of Research Questions and Hypotheses regarding the relationships between GOs and other variables (Theme B). Although these were discussed last in the literature review, in order not to have too much repetition and so as to keep the Research Questions and Hypotheses in close proximity to relevant results, these are discussed in Chapter 8 too. Moreover, although some of the Research Questions and Hypotheses relating to Theme B required some further analyses these were still presented in Chapter 8 (along with the additional analyses) in order to maintain thematic consistency.

1.2.8. Chapter 9

The results of Research Questions and Hypotheses relating to the themes of stability over time, task-specificity, and interactions between state and trait GOs (Themes C, D and E) were presented in this Chapter. Statistical techniques used to obtain these results included Repeated Measures Analyses of Variance, Paired Samples t-tests, Fisher z tests,

6

Correlational Analyses and Chi-square tests. The theme of stability over time is addressed first and is followed by the presentation of results relating to the theme of task-specificity. Finally, the results concerning the interactions between state and trait GOs are presented.

1.2.9. Chapter 10

This is the discussion chapter. In this chapter the results obtained from the current study are discussed and compared with those of previous research studies. The same order of themes as that in Chapters 8 and 9 is maintained in this Chapter. Therefore, the types and characteristics of GO profiles are discussed first. Subsequently, the relationships between GOs and self-efficacy, mental effort and performance are discussed. Next, the theme of stability of GOs over time is reviewed. This is followed by a discussion of the theme of task-specificity. The findings regarding the effects of the interactions between state and trait GOs on performance are then examined. In conclusion, a summary of the main findings of this study is presented.

1.2.10. Chapter 11

The conclusion chapter to this dissertation firstly addresses the theoretical and practical implications of the study. Following this, the limitations of the current study are discussed and recommendations for future research are made. This chapter ends with some concluding remarks regarding GO research and some reflective comments about this research project.

Chapter 2: The Development of Goal Orientations

2.0. The Development of the concept of Goal Orientations

The concept of GOs initially emerged in educational research. However, it quickly expanded to other areas of psychology, predominantly, sports and organisational psychology. The first part of this chapter provides a brief overview of the development of the GO concept. This is followed by a description of the development of GOs in organisational psychology. Subsequently, a section outlining the importance of GOs in organisational research is presented.

2.1. How did the concept of Goal Orientations develop?

According to Elliot (2005), the concept of GOs (which the author refers to as 'achievement goals') developed as a result of "independent and collaborative work by Carol Ames, Carol Dweck, Marty Maehr, and John Nicholls" (Elliot, 2005:53). Ames, Dweck, Maehr, and Nicholls conducted research on achievement motivation in an educational setting. The most prominent research on GOs was developed by Dweck and Nicholls, who produced a number of influential studies on GOs. Dweck and Nicholls eventually went their different ways and proposed two different conceptualisations of GOs. The following sub-section (2.1.1.) provides a description of the GO conceptualisation developed by Dweck. Subsequently, there is a description of the one developed by Nicholls (Section 2.1.2). Following this, a comparison of the ideas suggested by Dweck and Nicholls is made (Section 2.1.3).

2.1.1. Dweck's Conceptualisation of Goal Orientations

Dweck and Elliott (1983: 644) describe how "Intellectual performance and achievement have long been prized by our society.....However, too often we have assumed that the key

to understanding achievement lies in defining intelligence, assessing its level, and charting its course of development."

Dweck and Elliott (1983) discuss how intelligence is not the only factor influencing performance in achievement situations but that *motivation* plays an extremely important part in determining the outcomes of such situations. Since they were working in an educational setting, Dweck and Elliott (1983) were specifically interested in a particular type of motivation, that is, 'achievement motivation'. They were predominantly interested in the concept of achievement motivation because they believed that this type of motivation, independent of a child's ability, profoundly influences a child's performance and achievement in the classroom (Dweck & Elliott, 1983).

In 1983, Dweck and Elliott wrote a chapter on Achievement Motivation. In this chapter they reviewed four major theories of achievement motivation and used these to create an integrated model of achievement motivation. The rest of this sub-section will focus on the chapter written by Dweck and Elliott (1983) since it provides an excellent explanation of Dweck's conceptualisation of GOs. The chapter also provides a good basis for understanding the differences and similarities between Dweck's and Nicholls' conceptualisations of GOs (which will be described in further detail in section 2.1.3.). Dweck and Elliott (1983) used the following definitions of motivation and achievement motivation as a guide to their work:

Dweck and Elliott (1983:645) defined motivation as: "referring to the contemporaneous, dynamic psychological factors that influence such phenomena as the choice, initiation, direction, magnitude, persistence, resumption, and quality of goal-directed (including cognitive) activity."

9

They also proposed that achievement motivation "may be viewed as involving goals relating to competence – increases in competence and judgements of competence." (Dweck & Elliot, 1983:645).

According to Dweck and Elliot (1983), in 1979 Nicholls and Dweck wrote a manuscript which was not published. In this manuscript Nicholls and Dweck (1979) identified three goals relating to competence: a learning goal and two performance goals. The learning goal "involves seeking to acquire knowledge or skills, to master or understand something new" (Dweck & Elliot, 1983: 645) whilst the performance goals involve goals "to obtain favourable judgements of one's competence, and to avoid unfavourable judgements of one's competence, and to avoid unfavourable judgements of one's competence."

As will be emphasised in later chapters of this thesis, it is crucial to note that Dweck and Elliott (1983) specifically state that *it is possible for all three of the above goals to coexist*. However, in order to be considered an 'achievement situation', at least one of the above-mentioned goals must be present. Dweck and Elliott (1983) suggest that the three goals proposed above integrate previous views of achievement motivation. They put forward that the definition of motivation as proposed by Heckhausen (1967) actually represents the learning goal, whilst both performance goals are embodied by the definition proposed by Crandall et al. (1960). In order to provide a foundation stone on which to build their chapter, Dweck and Elliott (1983:646) defined the study of achievement motivation as "the study of psychological factors (other than ability) that affect the adoption and pursuit of these goals – that affect whether an achievement goal is pursued, how vigorously it is pursued, and how long it is pursued."

Keeping the above definition in mind, the rest of their chapter focuses on identifying the 'psychological factors' that affect the adoption and pursuit of achievement goals. Dweck and Elliott (1983) focus on three main factors: cognitive, affective, and value-related

factors. In discussing the cognitive factors Dweck and Elliot (1983) included their own work about 'Theories of Intelligence' which is described in further detail below.

Prior to suggesting the 'Theory of Intelligence', Dweck and Reppucci (1973), Dweck (1975), and Diener and Dweck (1978) carried out research in order to determine children's behavioural responses to failure. They discovered that although children had the same levels of ability some children did not persist in the face of failure whilst others did. According to Dweck and colleagues (Dweck & Reppucci, 1973; Dweck, 1975; & Diener & Dweck, 1978) some children responded to failure on achievement tasks with a 'mastery-oriented' pattern of behaviour. They attributed failure to insufficient effort. On the other hand, some children responded to failure with a maladaptive 'helpless' response in which they attributed failure to lack of ability. According to Dweck and Elliott (1983) these studies led Dweck and Elliott (1981) and Dweck and Bandura (1981) were unpublished manuscripts and the first publication on the Theory of Intelligence seems to be that by Dweck and Elliott (1983).

According to the 'Theory of Intelligence' (Dweck & Elliott, 1981; Dweck & Bandura, 1981) individuals may hold two different perceptions of competence. Individuals may either view intelligence as "a fixed, general, judgable entity" or they may view intelligence as "an ever-growing repertoire of skills and knowledge" (Dweck & Elliott, 1983:654). The former view of intelligence is referred to as the 'entity' theory of intelligence, whilst the latter is referred to as the 'incremental' theory of intelligence.

Dweck and Elliott (1983) describe how different individuals seem to favour different theories of intelligence and how individuals who favour different intelligence theories also seem to favour different achievement goals. It was found that individuals making use of an entity theory of intelligence seem to adopt performance goals in achievement situations. Conversely, individuals holding an incremental theory of intelligence seem to adopt learning goals in achievement situations. Beliefs about theories of intelligence as well as the types of achievement goals adopted were found to influence consequences such as quality of performance, task choice, effort, and persistence on tasks. Thus, Dweck (1986) proposed that these achievement goals interacted with perceived ability in order to create certain behavioural patterns in achievement situations.

In 1986, Dweck used the 'Theory of Intelligence' to create a model of achievement motivation (please refer to Figure 2.1.). In this model Dweck (1986) proposed that the theory of intelligence adopted by an individual influences the type of achievement goal, which Dweck (1986) refers to as a 'goal orientation', adopted. In turn, the GO adopted, along with the person's perception of ability elicit a particular behavioural response pattern.

Theory of Intelligence	Goal Orientation	Confidence in present abil	ity	Behaviour Pattern
Entity Theory $ ightarrow$	Performance Goal	If High	\rightarrow	Mastery-oriented
		But		
		If Low	\rightarrow	Helpless
Incremental \rightarrow	Learning	If High	\rightarrow	Mastery-oriented
(Dwock 1086:1041)	Goul	Low	\rightarrow	Mastery-oriented

Figure 2.1. Dweck's Model of Achievement Motivation

(Dweck, 1986:1041)

The review of literature carried out for the purposes of this study indicated that the term 'Goal Orientation' was used for the first time in the figure above drawn up by Dweck (1986). As shown in Figure 2.1., the term 'Goal Orientation' was used in order to refer to

the competence-related goals that individuals adopted in achievement contexts, that is, when referring to achievement goals. The term 'Goal Orientation' was used more and more often in literature from this point on (that is, from 1986 onwards). A more detailed discussion regarding the term 'goal orientation' will be provided in Chapter 3.

2.1.2. Nicholls' Conceptualisation of Goal Orientations

Nicholls (1984) also carried out work relating to GOs in an educational setting. In his research, Nicholls (1984) assumes (what he refers to as) an "intentional view of behaviour". This view of behaviour maintains that "individuals' actions serve to achieve purposes efficiently or economically" (Nicholls, 1984: 328).

According to Nicholls (1984), in order to be able to predict behaviour, one must first determine an individual's goals and then discover whether this goal will influence behaviour. Nicholls (1984) was interested in achievement behaviour, which he defines as "that behaviour in which the goal is to develop or demonstrate – to self or to others – high ability, or to avoid demonstrating low ability." (Nicholls, 1984:328). Thus, according to Nicholls (1984) achievement behaviour is characterised by the goal of competence (or at least perception of competence). Nicholls (1984) suggests that competence (or ability) may be interpreted in different ways by individuals. When he carried out developmental research on conceptions of ability he found that young children judge their ability with reference to their own previous level of performance (self-referenced). Therefore, higher performance than their previous performance would indicate higher ability. According to Nicholls (1984:909) "Ability and effort are imperfectly differentiated by young children." This is because mastery of a task using high effort signifies more gains in mastery and higher ability than mastery of a task using low effort. On the other hand, for adolescents (and adults) evidence of learning/mastery alone does not provide sufficient evidence to justify high ability. They believe that to be judged as having high ability they must learn more than others with equivalent effort or achieve the same level of performance as others with less effort. Therefore, adolescents use an 'other-referenced' conception of ability which is less subjective than that of young children.

Since Nicholls (1984) suggests that young children are not yet able to differentiate between self-referenced and other-referenced conceptions of ability, he refers to young children's conceptions of ability as the *'less differentiated'* and adolescents' and adults conceptions of ability as the *'more differentiated'*.

Nicholls and Jagacinski (1984) proposed that although adults are able to use the 'more differentiated' conception of ability, they may use either the more or the less differentiated conceptions of ability in different achievement settings (depending on their goals). When people use the less differentiated conception of ability, they compare their learning and performance to their own previous learning and performance (that is, perceptions of ability and performance are *self-referenced*). On the other hand, when people use the more differentiated conception of ability, they compare their ability and performance to the ability and performance of a reference group (Nicholls, 1984). Therefore, the two conceptions of ability create different frameworks for evaluating one's ability and performance. In the case of a less differentiated conception of ability, individuals consider tasks to be difficult if they expect to fail on them. In this case, increased mastery of the task indicates higher ability. In addition, since effort leads to more learning (which is an indication of high ability in this case) then effort is associated with high ability. On the other hand, in the case of the more differentiated conception of ability, both ability and task difficulty are specified by the performance of others. In this case a task is considered difficult if other people fail on it. High ability is evident if one uses less effort than others in completing a task (and vice versa for low ability).

Nicholls (1984) makes use of the term 'task-involvement' to indicate the state when an individual uses the less differentiated conception of ability in evaluating ability and performance. The term 'ego-involvement' refers to the state when an individual uses the more differentiated conception of ability for evaluating ability and performance.

2.1.3. Comparison of Dweck's and Nicholls' conceptualisations of Goal Orientations

From the above descriptions of Dweck's and Nicholls' work on achievement motivation it is evident that they agree that individuals adopt particular goals (which they refer to as 'achievement goals' or 'goal orientations') in an achievement situation. In addition, both Dweck and Nicholls agreed that in order to predict behaviour in an achievement situation one must look at an individual's achievement goals. Although initially they seemed to have suggested that there are one learning and two performance achievement goals in their early work (unpublished manuscript 1979), each seems to have later used different terms for achievement goals: Dweck used the terms 'learning' and 'performance' GOs whilst Nicholls used the terms' 'task' and 'ego' orientations. A discussion regarding whether the previously mentioned terms actually refer to the same constructs (or not) is provided in Chapter 3 (Section 3.1.1.).

The main difference between Dweck's and Nicholls' conceptualisations is that each developed a different theory about predicting individuals' adoption of the different achievement goals. Dweck proposed the Theory of Intelligence (Dweck & Elliott, 1981; Dweck & Bandura, 1981) whereas Nicholls proposed the 'conceptions of ability' theory (1984).

As a result of their different theories of ability, Nicholls (1984) and Dweck (1986) differ in terms of their understanding about effort and ability. Dweck (1986) seems to assume that beliefs about effort and ability are quite stable individual characteristics. Conversely,

Jagacinski and Nicholls (1984) proposed that individuals have different perceptions about ability and effort depending on the situation. According to Jagacinski and Nicholls (1984: 910), "In previous work on attributions and affects it has been implicitly assumed that the meanings of effort and ability are fixed." Jagacinski and Nicholls (1984) challenged this belief by proposing that the understanding of effort and ability may change according to the situation. The results of their research indicate that in competitive situations, individuals tend to adopt an ego-orientation and judge their ability as being high only when effort was low in comparison to others. On the contrary, when individuals adopt a task-orientation (in a non-competitive environment), they judge their ability as being high when effort is high. The results of this study indicate that conceptions of ability and effort are not as stable as previous researchers imagined them to be. In the 1990s, the model of GOs was further developed. However, the various models of GOs shall be described in further detail in Chapter 3.

2.2. When was the concept of Goal Orientations introduced into the Organisational setting?

In 1989, Wood and Bandura conducted a research study "governing performance in a simulated organisation" (Wood & Bandura, 1989). They made use of the concepts of effort and ability proposed by Dweck and Elliott (1983). It was suggested that people who adopt an incremental perspective of ability tend to take on a learning GO whereas individuals who adopt an entity perspective of ability are more likely to take on a performance GO. However, in their research, Wood and Bandura (1989) propose that these perceptions of ability are changeable and may be induced by situational characteristics (e.g. if an instructor emphasises that a certain task is improvable with practice, then it is more likely that individuals will adopt an incremental perspective of ability). They found that when individuals assumed an entity conception of ability they showed decreased perceived self-efficacy, set low organisational goals and used less

efficient analytic strategies than they did at the beginning of the experiment. In contrast, participants who assumed an incremental perspective of ability maintained the same level of perceived self-efficacy (as they did at the start of the experiment), set challenging organisational goals and used effective analytic strategies.

Wood and Bandura (1989) attempted to induce mastery and performance GOs in their organisational research. Consequently, they seem to have been two of the first researchers to make use of the concept of GOs in the organisational setting.

In 1993, Farr et al. (p.194) stated that the GO approach is "not well integrated at present into the I/O literature". During the 1990s studies on GOs in the organisational setting became more frequent. Initially, studies focused on how GOs relate to sales performance (e.g. Sujan et al., 1994; Kohli et al., 1998; VandeWalle et al., 1999). Eventually studies began to focus on GOs with respect to other areas of organisational behaviour such as training (e.g. Ford et al., 1998; Hertenstein, 2001; Kozlowski et al., 2001; Rogers & Spitzmueller, 2009), performance other than sales performance (e.g. Seijts et al., 2004; Dragoni, 2005; Mehta et al., 2009; Steele-Johnson et al., 2000) and feedback-seeking behaviour (e.g. Janssen & Prins, 2007) amongst others.

In general, the studies carried out in an organisational setting focused on determining the relationships between GOs and performance (e.g. sales performance, extent of learning and transfer resulting from training courses, team performance, and feedback-seeking behaviour leading to improvements in performance). Additionally, most of these studies used a two-factor model of GOs (consisting of mastery and performance GOs). In the next Chapter it will become evident how a two-factor model is considered to be inadequate for measuring GOs since there are more differentiated models available which were found to provide a better understanding of GOs. Therefore, although GO research has been carried out within an organisational setting, the majority of GO research is still carried out in the

educational sector. Further research examining GOs in the organisational setting is definitely required since it has been quite limited so far and, as described next, there are numerous benefits of carrying out such research.

2.3. Why is Goal Orientation research in organisations so important?

DeShon and Gillespie (2005) provide an important argument which offers a suitable answer to the question posed above. They describe how GO research has been able to provide "at least a partial answer" (DeShon and Gillespie, 2005:1096) to important questions that managers within organisations quite often pose. One such question is why certain individuals are continuously attempting to strive for self-improvement whilst others are happy to experience life using the same set of skills and knowledge. Another question relates to why some individuals are happy to take on challenges whilst others avoid challenges or make use of self-handicapping behaviour in order to avoid challenging tasks. GO research has been somewhat successful in providing answers to such questions. Research on GOs has also provided answers as to why some people are interested in performing better than others whilst other people are not as concerned about their performance relative to others (Nicholls et al., 1989).

The utilisation of GOs in organisational research has enabled researchers to better understand, as well as make predictions about, learning and behaviour in a wide array of organisational contexts. These include training, sales performance, feedback seeking, performance adaptability (DeShon and Gillespie, 2005) as well as cross-cultural adjustment (Gong, 2003). These organisational contexts are, beyond doubt, important for businesses in order to be able to maintain a competitive edge.

Thus, not only has GO research proved to be beneficial in enhancing knowledge by promoting a better understanding of human behaviour but it has also been successful in

18
providing recommendations to businesses in order for them to be able to increase their productivity. For example, as a result of their research regarding the relationship between GOs and performance, Lee et al. (2006:495) recommend that "when a performance goal orientation is dominant, managers should structure employee tasks so that they are specific, quantifiable and easy to master". They also recommend that managers should delegate tasks that promote or require a future orientation (e.g. strategic planning) to employees having a dominant mastery GO.

In addition to the use of GOs in the previously mentioned organisational contexts, it is possible that GOs may also be used in recruitment and selection. If one attempts to determine which GOs are valued by an organisation then it is possible to find suitable employees by matching these GOs with those adopted by prospective employees. This might also benefit prospective employees because they may feel more comfortable working in an organisation with a culture that matches their personal GOs. However, this would only be possible if GOs are stable dispositional traits. As will become evident in Chapter 4 (Section 4.1.) there is currently not enough evidence to make any claims regarding the stability of GOs. Therefore, the viability of using GOs in recruitment and selection is currently an unexplored issue. Consequently, as argued by Farr et al. (1993: 194) "We believe that I/O psychology theory and research would be well served by consideration of these constructs."

2.4. Synopsis

In this chapter a summary of the development of the concept of GOs and the importance of researching and further understanding this concept in organisational settings was provided. The next chapter (Chapter 3) focuses on the issues present in GO literature. It is crucial to understand these issues since this will provide a basis for understanding the aims and objectives of this research study.

19

Chapter 3: Conceptual Inconsistencies in Goal Orientation Research: Definition and Dimensionality of Goal Orientations

3.0. Research Issues with regards to Goal Orientations

Although research on GOs has come a long way since the 1970s, there are still many conceptual and operational issues which have not been addressed. These ambiguities in GO research have resulted in a large number of inconsistent research results which are extremely hard to reconcile (DeShon & Gillespie, 2005).

According to DeShon and Gillespie (2005) there are three main conceptual inconsistencies in GO research. These include issues with definition, dimensionality, and stability. There are also questions regarding the specificity of GOs, that is, whether GOs are general or more task-specific. This chapter will focus on the definition and dimensionality issues. The stability and specificity issues will be discussed in the next chapter.

3.1. Definition

The review of literature (carried out for the purposes of this study) indicated that there is a lot of confusion over the definition of GOs. This confusion is compounded by the fact that different researchers use different terms when referring to GOs. In this section the concerns surrounding the use of different terms in GO literature are first described. Following this, the problems with regards to the definition of GO are discussed.

3.1.1. Different terms, same concepts?

GO literature has been rather confusing from the start since even the earliest studies on GOs have been laced with different terms and definitions of the concept. This is evident in

Dweck's (1986) study when she uses the terms 'achievement goals' and 'goal orientations' interchangeably. When Dweck (1986) first suggested the terms 'learning' and 'performance', she classified these as *achievement goals*. In the same article she uses the term 'goal orientation' to refer to the type of achievement goals that individuals are oriented towards (Dweck, 1986: 1041). This issue of different terms extends to the **types** of achievement goals too. As discussed in Chapter 2 (Section 2.1.1.), initially Dweck and Nicholls seem to have proposed three achievement goals: one learning and two performance goals. However, as seen in later articles both researchers seem to have abandoned the idea of three achievement goals and focused on two. As a result of their different theories about predicting the adoption of GOs they also used different terms for the GOs. Dweck seems to have stuck with the original 'learning goal orientation' and 'performance goal orientation' whilst Nicholls (1984) suggested the terms 'task-involvement' and 'ego-involvement'.

An additional problem creating confusion in GO literature results from the fact that many researchers make use of the term 'mastery orientation' when referring to a 'learning' GO (e.g. Meece & Holt, 1993; Harackiewicz & Elliot, 1993, and Senko & Harackiewicz, 2005, amongst others). The term 'mastery' seems to have been proposed by Diener and Dweck (1978). However, contrary to its use in many GO articles, Dweck uses the term 'mastery' oriented' when referring to a behavioural response NOT a GO. As mentioned in Chapter 2 (Section 2.1.1.) Dweck makes use of the term 'learning goal' when referring to the GO. A *mastery-oriented response* refers to the positive behaviours exhibited by an individual who adopts a learning GO (e.g. persistence on tasks). The *learning GO* refers to the goal that the individual is pursuing in that achievement situation, that is, the goal of seeking to increase competence. The use of the terms 'mastery-oriented' and 'learning goal' as used by Dweck (1986) are evident in her article (Dweck, 1986:1041).

As will be discussed later on in this chapter, researchers eventually proposed that there are three types of GOs. The 'performance' GO was bifurcated into two GOs which are generally referred to as 'performance-approach' and performance-avoidance' GOs. However, they are also referred to as 'proving' and 'avoiding' GOs respectively, (e.g. in VandeWalle et al., 2001) or as 'self-enhancing ego orientation' and 'self-defeating ego orientation' respectively (Skaalvik, 1997).

Pintrich and Schunk (2002) discuss how there is still some debate between researchers as to whether these different terms actually refer to the same constructs. Pintrich (2000a:96) suggests that it is important for researchers to

maintain distinctions in terms or labels when they reflect important and real differences in the terms, theories, and supporting empirical data, but not to let terms proliferate when they signify distinctions without any real theoretical or empirical differences.

The problem with this solution is that although there seem to be slight differences between the different terms used, there is also a great deal of overlap. In addition, it seems as though many researchers acknowledge that the terms refer to slightly different conceptualisations of GOs but have still used them interchangeably in literature. This has resulted in a great deal of confusion and misunderstanding. At this point, I believe that the solution proposed by Wigfield and Cumbria (2010) is by far the most appealing due to its simplicity in solving the matter. They suggest that although they acknowledge that the different terms do in fact reflect some distinctions in the conceptualisations of these GOs, they also "believe that the similarities are stronger than the distinctions between them" (Wigfield & Cumbria, 2010: 7). They decided to use the terms 'mastery' and 'performance' goals in their study. The terms 'mastery' and 'performance' will also be used in this study. These terms were chosen due to the fact that the measure of GOs used in this study.

makes use of them and, in addition, according to Pintrich et al. (2003) they are the most commonly used terms in GO literature.

3.1.2. What is the definition of Goal Orientations?

Elliot and Thrash (2001) describe how there is no single widely accepted definition of GOs. They explain how theorists generally make use of one of two basic approaches in defining achievement goals. The first approach describes achievement goals "in terms of the *purpose* for which a person engages in achievement behaviour" (Elliot & Thrash, 2001:140) whilst the second approach involves characterising "achievement goals as a network or integrated pattern of variables that together create an *orientation* toward achievement tasks" (Elliot & Thrash, 2001: 140).

Elliot and Thrash (2001) explain how theorists using the first definitional approach understand achievement goals as being a combination of a) the reason for behaviour and b) the outcome an individual would like to attain in an achievement setting. Conversely, theorists using the second definitional approach view GOs as the general orientations adopted on tasks which result from numerous beliefs about success and ability, effort, and standards for evaluation. These beliefs, effort, and standards of evaluation are thought to be interrelated and thus provide a framework for approaching achievement settings.

Elliot and Thrash (2001) offer a number of limitations for the two definitional approaches described above. For example, they argue that the definitions above view achievement goals as being a combination of variables which make it difficult to determine which of the variables should be considered to be responsible for any hypothesised or observed effects. In addition, they state that there are no guidelines available for determining how

many characteristics of the achievement goal must be present before a person may be described as having adopted that achievement goal.

In light of these limitations, Elliot and Thrash (2001) proposed that achievement goals could be defined as "a cognitive representation of a competence-based possibility that an individual seeks to attain" (Elliot and Thrash, 2001:144). This definition is different from the first definitional approach (described above) in that the reason for achievement behaviour and the aim of achievement behaviour are kept separate. Additionally, it is different from the second definitional approach since the processes associated with the aim are kept separate from the aim itself.

Another study which paid significant attention to the concern of defining GOs was carried out by DeShon and Gillespie (2005). They reviewed 88 GO studies, and came up with five categories of definitions: goals, traits, quasi-traits, mental frameworks, and beliefs. According to DeShon and Gillespie (2005:1097) the 'goals' definition of GOs is the most commonly used (29 studies). Researchers using this approach view "goal orientation as the adoption and pursuit of specific goals in achievement contexts". The second category proposed is that of 'traits'; 26 studies fit into this category. Researchers using this definition "view goal orientation as a trait or disposition that is responsible for individual differences in behavior" (DeShon & Gillespie, 2005:1097). The third definitional category proposed is that of 'quasi-traits' (9 studies in this category). In this definitional approach, GOs are viewed "as a somewhat stable trait that can be modified by the appropriate situational characteristics" (DeShon & Gillespie, 2005:1101). The fourth category proposed is the 'mental framework' category (9 studies in this category). Researchers with studies in this category view GOs as "a mental framework consisting of a wide variety of beliefs, affects, goals, and cognitions that covary in achievement contexts and result in achievement related behaviour" (DeShon & Gillespie, 2005:1101). This category is somewhat similar to the second definitional approach described by Elliot and Thrash

(2001). The final category proposed is that of 'beliefs' (9 studies in this category). Researchers having studies in this category view GOs as "following from an individual's beliefs or implicit theories concerning the malleability of ability" (DeShon & Gillespie, 2005:1101). There were three studies which DeShon and Gillespie (2005) could not classify due to insufficient detail. In addition, they had a category which they referred to as 'other' in which they placed two studies that did not clearly fit into any of their five categories.

Although DeShon and Gillespie (2005) seem to have clear-cut categories of the different understandings of GOs it is not necessarily as easy as it seems to determine which definition of GOs researchers adopt in their studies. As DeShon and Gillespie (2005) mention themselves, a significant amount of judgement was required when classifying the studies. Some of the categories are questionable. For example, it is debatable whether the 'trait' and 'quasi-trait' categories are significantly different from each other. Nowadays, many researchers using the trait approach tend to accept that situational factors may influence dispositional traits and the resulting behaviour (e.g. Button et al., 1996). Moreover, there is the difficulty of the three studies having insufficient detail as well as the 'other' category. Additionally, as DeShon and Gillespie (2005) mention, a number of researchers tend to use different definitions of GOs in the same study (e.g. Brett & Atwater, 2001; Yeo & Neal, 2004).

After their review of GO studies, DeShon and Gillespie (2005) propose a model of goaloriented behaviour which they refer to as the 'Motivated Action Theory'. According to their model, "Goal orientation is a label used to describe the pattern of cognition and action that results from pursuing a mastery-approach, performance-approach, or performance-avoid goal at a particular point in time in a specific achievement situation." (DeShon & Gillespie, 2005:1114). Since this definition states that GO is *the result of* mastery-approach, performance-approach, and performance-avoidance it is not consistent with what is meant by GOs in the current study (or most of GO literature!). Consequently, this definition will not be used for the present study. After careful consideration, it was decided that the definition proposed by Elliot and Thrash (2001) will be used as a guide for this study.

Therefore, in this study GOs are viewed as cognitive representations of competencebased possibilities that individuals want to achieve. This definition was chosen since it is the most comprehensive and well-researched definition of GOs found in literature. It is the result of an in-depth investigation into the various definitions of GOs and an assessment of their strengths and limitations. The authors who proposed this definition clearly indicate an in-depth understanding of the history and development of GOs. Moreover, this definition was used as a guide for developing the GO questionnaire used in this study, that is, the Achievement Goal Questionnaire-Revised (please refer to Section 3.2.3. for an explanation of the choice of questionnaire). Consequently, it was considered appropriate to use this definition in order to maintain definitional consistency throughout this study.

3.2. The Dimensionality of Goal Orientations

As described in the introduction to this chapter the problem of dimensionality of GOs is another area of great debate within GO literature. Researchers have used different models of GOs in order to guide their research and to date there is still disagreement regarding which model of GOs is best. The development of the different models of GOs as well as the reasoning behind their use is described in this section.

3.2.1. The Two-Factor Model of Goal Orientations

As discussed in Chapter 2 (Section 2.1.1.), initially, Dweck and Nicholls (1979) proposed that there are three types of achievement goals: one learning goal and two performance goals. However, eventually they both focused on two main achievement goals: a learning/task GO and a performance/ego GO. This understanding of GOs is referred to as the 'two-factor model of goal orientations'. A substantial amount of research on GOs makes use of a two-factor model of GOs (e.g. Ames, 1992; VandeWalle et al., 1999; Anderman & Midgley, 1997; and Phillips & Gully, 1997, amongst others).

Although Dweck and Nicholls initially suggested two types of performance GOs "the concept of independent approach and avoidance goal orientations received very little theoretical and empirical attention and was soon overlooked by researchers" (Smith et al., 2002: 156). However, a number of years later, the idea of the presence of three types of GOs was rediscovered and developed further by Elliot and colleagues (Elliot & Harackiewicz, 1996; Elliot et al., 1999) as described in Section 3.2.2. Due to the fact that three GOs were proposed, the model developed by Elliot and Harackiewicz (1996) is referred to as the three-factor model of GOs.

3.2.2. The Three-Factor Model of Goal Orientations

Throughout their research Elliot and colleagues (e.g. Elliot & Harackiewicz, 1996; Elliot, 1999; Elliot & Thrash, 2001; and Elliot & Covington, 2001) describe how studies on achievement motivation have always incorporated an approach-avoidance distinction. For example, according to Elliot and Harackiewicz (1996), Lewin et al. (1944) as well as McClelland et al. (1953) suggested that there are two types of motivational orientations, one focusing on achieving success and the other on avoiding failure. Thus, the idea of approach and avoidance motivations seems to have been present quite early on in motivation research.

As a consequence of early motivation theorists (e.g. McClelland et al., 1953) emphasising the importance of the approach-avoidance distinction, Elliot and Harackiewicz (1996: 462) suggested "that the conventional achievement goal dichotomy be expanded in order to incorporate independent approach and avoidance components within the performance goal orientation". Thus, they proposed a three-factor model of GOs. This three-factor model consisted of a 'mastery' GO [which retained the same meaning as the learning GO described by Dweck (1986) in earlier research], and the performance GO was divided into what they refer to as a 'performance-approach' GO and a 'performance-avoidance' GO. The performance-approaqch GO consists of "attaining favourable judgements of competence" (Elliot & Harackiewicz, 1996:461), whilst the performance-avoidance GO entails "avoiding unfavourable judgements of competence" (Elliot & Harackiewicz, 1996: 461).

Apart from the fact that early motivation theorists emphasised the importance of the approach-avoidance distinction, it was evident from research that the performance GO was not being measured accurately enough. Research studies investigating the consequences of GOs produced very different results with regards to the performance GO and quite consistent results with respect to the mastery GO. Many studies using the two-factor model found that a mastery GO was found to be consistently positively related to persistence on difficult tasks, use of deep cognitive strategies, and attributing failure to lack of effort rather than lack of ability (e.g. Ames & Archer, 1988; Dweck & Leggett, 1988; Meece & Holt, 1993; Pintrich & Garcia, 1991; Wolters et al., 1996). On the other hand, the results of studies investigating the relationships between a performance GO and other variables were rather inconsistent. Some studies found significant positive relationships between a performance GO and positive outcomes such as self-efficacy (Bandalos et al., 2003) and task performance (Harackiewicz et al., 1997). Other studies found no significant relationships between a performance GO and performance GO and task performance (e.g. Schraw et al., 1995) and yet other studies found a performance GO to be negatively correlated with task

performance (e.g. Bell & Kozlowski, 2002). These inconsistencies in research regarding the performance GO provided Elliot and colleagues with another reason to include the approach-avoidance distinction in GO literature.

In order to test the utility of distinguishing between a performance-approach GO and a performance-avoidance GO, Elliot and Harackiewicz (1996) conducted a study investigating the effects of the two types of performance GOs on intrinsic motivation. By using sequential simultaneous regression they found that a performance-approach GO did not undermine intrinsic motivation but a performance-avoidance GO did. Elliot et al. (2005) carried out an experimental study to investigate the effects of mastery, performance-approach, and performance-avoidance GOs on performance. They used planned comparisons in order to analyse their data. The results of their study indicate that performance-approach and performance-avoidance GOs have different effects on performance. The performance-avoidance GO was found to undermine performance relative to performance-approach and mastery GOs.

Further support for the utility of the approach-avoidance distinction was obtained by Church et al. (2001) who indicated how performance-approach, performance-avoidance, and mastery GOs have different antecedents. By using hierarchical linear modelling they found that lecture engagement seems to positively predict a mastery GO (.34, p < .05); evaluation focus was found to positively predict performance-approach (.50, p < .05) as well as performance-avoidance GOs (.44, p < .05) and negatively predict a mastery GO (-.13, p < .05); and harsh evaluation (which refers to work being marked strictly) was found to positively predict a performance-avoidance GO (.35, p < .05) and negatively predict a mastery GO (-.23, p < .01). Thus, it seems as though the two types of performance GOs proposed have somewhat different antecedents.

A number of studies assessing GO measures provide further support for the three-factor model of GOs. For example, Elliot and Church (1997) developed and tested scales measuring performance-approach and performance-avoidance GOs. Factor analysis indicated that the performance-approach and performance-avoidance scales were different and the results indicated the presence of three separate factors (i.e. mastery, performance-approach, and performance-avoidance GOs).

Research carried out by Day et al. (2003) provides additional evidence for the use of a three-factor model. They found that a two-factor model "does not adequately explain the nature of the goal orientation instruments." (Day et al., 2003:448). They also found that measures of general performance GO are correlated with measures of both performance-approach and performance-avoidance GOs. Thus, they suggest that one should be cautious in making use of results of studies based on a two-factor model as opposed to a three-factor model of GOs.

Midgley et al. (1998) carried out research in order to validate scales measuring the three types of GOs proposed by Elliot and Harackiewicz (1996). The scales developed were found to have good convergent and discriminant validity and the confirmatory factor analysis provided additional support for a three-factor model of GOs. In another study on the three-factor model, Skaalvik (1997) attempted to create a measure for the three types of GOs as well as a measure for work avoidance. The results of his exploratory factor analysis revealed good factor loadings for items on each of the predicted four factors and low factor loadings for items on factors other than the predicted ones. These results indicated that the performance-approach and performance-avoidance GOs were found to be different factors. They were also found to be weakly correlated with each other.

It is interesting to note that although researchers show how the two-factor model is not accurate enough and provide reasonable support for using a 3-factor model (e.g. Elliot &

Harackiewicz, 1996; Midgley et al., 1998; Skaalvik, 1997; Elliot et al., 2005; Church et al., 2001; Day et al., 2003), a number of studies still make use of the 2-factor model of GOs in their research (e.g. Yeo & Neal, 2004; Kozlowski et al., 2001; Bereby-Meyer et al., 2010; Hanrahan & Cerin, 2009). There may be a number of possible explanations for this. For example, the two-factor model is the original model proposed by early GO theorists. However, a closer look at literature reveals that early GO theorists actually proposed three types of achievement goals initially (as mentioned in Section 3.2.1.). This provides further support for the use of a three-factor model as opposed to a two-factor model. Another possible explanation for the popular use of the two-factor model is that many researchers coming from an educational psychology background believed that a mastery GO is adaptive and a performance GO is non-adaptive (e.g. Meece & Holt, 1993; Schraw et al., 1995). Consequently, they were not really interested in the performance GO. These researchers were more focused on the mastery GO and finding ways in which to encourage students to adopt a mastery GO.

The prevailing belief of the mastery GO being adaptive and the performance GO being negative might have biased and stalled the development of GO literature. This belief was held by the majority of GO theorists until the late 1990s (e.g. Phillips & Gully, 1997; VandeWalle et al., 1999). As was made evident in this section, bifurcating the performance GO led researchers to discover that a performance-approach GO has quite positive outcomes whilst a performance-avoidance GO has rather negative outcomes. Therefore, promoting a performance-approach GO may be as important as promoting a mastery GO.

3.2.3. The Four-Factor Model of Goal Orientations (2x2 approach)

After proposing and testing the three-factor model of GOs, Elliot continued to investigate the dimensionality of GOs and, as a result, he further developed the model of GOs. In

1999, Elliot proposed that even the mastery GO should be bifurcated into approach and avoidance components because "a full 2x2 crossing of the performance-mastery and approach-avoidance distinctions seems necessary to account for the broad spectrum of competence-based strivings." (Elliot, 1999:181). Pintrich (2000b) also discussed the idea of having a 2x2 model of GOs. He stated that although (at the time) there was no empirical evidence supporting the presence of a mastery-avoidance GO, research should be carried out in order to investigate this concept further. The research proposed by Pintrich (2000b) was carried out by Elliot and McGregor (2001) who described how competence is "at the conceptual core of the achievement goal construct" (Elliot & McGregor, 2001:501). They suggest that achievement goals (and competence) may be distinguished along to two basic dimensions: **definition** and **valence**.

Definition: According to Elliot and McGregor (2001) achievement goals may be defined in terms of the standards used in performance evaluation. They propose that three types of standards are used in performance evaluation: absolute, intrapersonal, and normative standards. Absolute standards refer to the requirements of the task itself; intrapersonal standards refer to evaluation compared with an individual's past performance or maximum potential; whilst normative standards refer to comparing one's performance to that of others. Elliot and McGregor (2001) describe how absolute and intrapersonal standards are very similar and often hard to distinguish, therefore they collapsed these into one category. Consequently, they explain how achievement goals can be distinguished depending on whether performance is defined as self-referenced (mastery GO) or other-referenced (performance GO).

Valence: According to Elliot and McGregor (2001: 502) "Competence is valenced in that it is either construed in terms of a positive, desirable possibility (i.e. success) or a negative, undesirable possibility (i.e. failure)." It seems as though people tend to process most stimuli in terms of valence and this seems to be done unconsciously. This processing is, in

32

turn, alleged to induce approach and avoidance behavioural predispositions. Thus, they suggest that GOs should also be distinguished according to whether they have approach or avoidance valences.

The result of distinguishing GOs along definition and valence is a four-factor model, or as it is better known, a 2x2 factor model. Achievement goals are distinguished along two basic dimensions (definition and valence) both of which have two ends of a continuum (*definition*: self-referenced vs. other-referenced; and *valence*: approach vs. avoidance) as illustrated in Figure 3.1.

Figure 3.1.The 2x2 model of Goal Orientations



As indicated in the model above, Elliot and McGregor (2001) proposed that there are four main types of GOs: mastery-approach, mastery-avoidance, performance-approach, and performance-avoidance GOs. Pintrich and Schunk (2002) provide an extremely clear understanding of what is meant by these terms as presented in the table below.

	Approach Focus	Avoidance Focus
Mastery orientation	 Focus on mastering task, learning, and understanding. 	 Focus on avoiding misunderstanding, avoiding not learning or not mastering task.
	 Use of standards of self- improvement, progress, deep understanding of task. 	 Use of standards of not being wrong, not doing task incorrectly.
Performance orientation	 Focus on being superior, being the smartest, best at task in comparison to others. 	 Focus on avoiding inferiority, not looking stupid or dumb in comparison to others.
	 Use of normative standards such as getting best or highest grades, being top or best performer in class. 	 Use of normative standards of not getting the worst grades, being lowest performer in class.

Table 3.1. An Explanation of the Four Types of Goal Orientations

Adapted from Pintrich & Schunk (2002: 219).

As illustrated in Section 3.2.2., Elliot and Harackiewicz (1996) already provided support for the performance GO being divided into performance-approach and performanceavoidance GOs. In a research study carried out in 2001, in which they proposed the 2x2 model, Elliot and McGregor provided support for this model by designing a measure (the Achievement Goal Questionnaire [AGQ]) assessing the four types of GOs and determining its validity and reliability. They provide evidence for the existence of four GOs by the use of factor analysis. Their results indicate how each of the four factors were found to be empirically distinct from one other and were therefore not measuring the same construct. In addition, Elliot and McGregor (2001) offered further support for the 2x2 model of GOs by indicating how the four GOs each have different antecedents and different consequences. Their regression analyses indicated that need for achievement was an antecedent of a mastery-approach GO whilst fear of failure was an antecedent of a performance-approach GO whilst fear of failure was found to be a predictor of a performance-approach GO whilst fear of failure was found to predict both performanceapproach and performance-avoidance GOs. In addition, self-determination was found to positively predict a mastery-approach GO; negatively predict mastery-avoidance and performance-avoidance GOs; and not relate to a performance-approach GO. The antecedents and consequences of GOs will be discussed in further detail in Chapter 4.

With regards to the consequences of the four types of GOs, Elliot and McGregor (2001) examined the relationships between GOs and the type of cognitive strategies used. Cognitive strategies are generally categorised into surface strategies and deep strategies. Surface strategies refer to cognitive processes in which information is remembered as a result of repetition and rehearsal whilst deep strategies refer to cognitive processes such as elaboration and organisation of new information with previous information (Lyke & Kelaher, 2006).

Elliot and McGregor (2001) found that a mastery-approach GO significantly predicted deep processing; a performance-avoidance GO marginally negatively predicted deep processing; whilst mastery-avoidance and performance-approach GOs were not significantly related to deep processing. Furthermore, a performance-avoidance GO was found to positively predict surface processing whilst a performance-approach GO was found to marginally predict surface processing. Mastery-approach and mastery-avoidance GOs were found *not* to be significantly related to surface processing. The results presented above provide further evidence indicating that the four GOs are significantly different from each other since they are clearly related differently to various variables.

Finney et al. (2004) conducted a study to test the AGQ. They describe how the AGQ is very course-oriented and therefore adapted it slightly in order to test it at a domain-level of specificity (a discussion regarding levels of specificity of GOs is provided in Chapter 4 Section 4.1.3.). They aimed to assess the AGQ in terms of model structure, reliability and discriminant validity in a domain-specific context (academic) as opposed to a course-

specific context. They conducted a confirmatory factor analysis (CFA) and found that the four-factor model fit significantly better than the alternative (three- and two-factor) models tested (CFI=0.95; RMSEA= 0.066; SRMR= 0.048). Moreover, the results of the CFA indicated that all the scales had high reliabilities (greater than 0.70) except the performance-avoidance scale which had a reliability of 0.68. This lower reliability was the result of a problem item (please refer to Chapter 7 Section 7.1.5.1.) for a more detailed description of the properties of the AGQ and its revision). The results of their analyses also provided evidence of discriminant validity since the intercorrelations between the four GOs were low to moderate (intercorrelations ranged between 0.11 and 0.51 with four out of the six intercorrelations being lower than 0.30). Since Finney et al. (2004) used a much larger sample size (2,111 participants) than Elliot and McGregor (2001), who used three different samples of 180, 148, and 182, the results obtained by the former are an important addition to the evidence providing support for the 2x2 model of GOs.

Additional support for the presence of four GOs was provided by Janssen and Prins (2007). Prior to examining the differences in feedback-seeking behaviour depending on the type of GO adopted they conducted a principal component analysis. This indicated the presence of four factors, with items loading highly on the predicted factors. Furthermore, the scales had high reliability levels (Cronbach's alpha between 0.71 and 0.91). Their results indicate that each of the four GOs is related to different types of feedback-seeking behaviour. With respect to the seeking of self-improvement information, mastery-approach and performance-avoidance GOs were found to be significantly positively correlated whilst a performance-approach GO was found to be negatively related. No significant relationship was found between a mastery-avoidance GO and seeking of self-improvement information. With regards to the relationships between GOs and seeking of self-validation information, a performance-avoidance GO was found to be positively related whilst a mastery-approach GO was found to be negatively related. No significant relationships were found between seeking of self-validation information, a performance-avoidance GO was found to be positively related whilst a mastery-approach GO was found to be negatively related. No significant

performance-approach and mastery-avoidance GOs. The different relationships between the four types of GOs and feedback-seeking behaviour provide further support for the four-factor model.

In a later study, Elliot and Murayama (2008) further developed the 2x2 measure of GOs. They also assessed the utility of 2x2 model in greater detail. They discovered that each of the four GOs had a high degree of internal consistency (mastery-approach, masteryavoidance, performance-approach, and performance-avoidance had Cronbach's α of: 0.84, 0.88, 0.92, and 0.94, respectively). The 2x2 model was also found to provide the best fit compared with the two- and three-factor models of GOs. Elliot and Murayama (2008) applied a multiple-indicator correlated trait-correlated method model (MI CT-CM model) to their data (using maximum likelihood estimation) in order to test the theory that GOs may be distinguished along two basic dimensions (i.e. competence and valence). This model provided a good fit to the data: $X^{2}(49, N=229) = 78.54$, $\rho < 0.01$, $X^{2}/df = 1.60$, CFI= 0.99, IFI= 0.99, RMSEA= 0.051, and all path coefficients were significant. Additionally, when compared to alternative models (e.g. 3-factor model consisting of masteryapproach, mastery-avoidance and performance GOs and a 3-factor model consisting of mastery, performance-approach and performance-avoidance GOs) the MI CT-CM model still provided the best fit thus supporting the theory that GOs may be distinguished along two basic dimensions.

Further support for the 2x2 model of GOs is provided by Conroy et al. (2003) who carried out a research study in order to create an adapted version of the AGQ for the sports context (AGQ-S). Confirmatory factor analyses were conducted in order to test the 2x2 model and six other models. The results of the analyses indicate that the 2x2 model provided an acceptable fit to the data with goodness-of-fit indices (NFI, NNFI, CFI) of 0.92 to 0.94. These indices exceeded the minimum conventional criteria of 0.90 for acceptable fit. Moreover, the 2x2 model was found to have a superior fit to the data than the other six models tested. Each item was also found to significantly load onto its hypothesised latent variable.

Radosevich et al. (2007) carried out a research study to investigate whether a three- or four-factor model provides a better conceptualisation of GOs. They used structural equation modelling in order to determine model fit. Their research results indicate that the four-factor model seems to provide a better conceptualisation of GOs over the three factor model (3-factor model goodness of fit statistics: CFI=0.75, SRMR=0.09, Chi Square=118.68 with df=16; 4-factor model goodness of fit statistics: CFI=0.83, SRMR=0.07, Chi Square=92.10 with df=16). However, as indicated by the goodness of fit statistics, *the four-factor model still does not provide an excellent fit*. In addition, there is no evidence that the four-factor model fit is *significantly* better than the three-factor model fit.

A study published in 2007 by Pastor et al. focused on determining which model of GOs helps to provide the best understanding of GOs from a multiple goal perspective (this perspective will be discussed in further detail in the next section). They used Latent Class Analysis (which is a similar to cluster analysis) to analyse their data. Their results indicate that, *to a certain extent*, the four-factor model of GOs is better than the two- and three-factor models since it included a cluster of participants that would not have been identified by the two- and three-factor models. Although this cluster is quite small (2% of their sample) the authors describe how the more complex model of GOs (i.e. the four-factor model) could help to better differentiate among students (since their study was based in educational psychology). This cluster was different from the other clusters obtained since it was characterised by individuals having high performance-avoidance and mastery-avoidance GOs in addition to a high mastery-approach GO and somewhat lower performance-approach GO. Therefore, it was different from the other clusters since it is high on all four GOs. However, what is especially notable about this cluster is that both avoidance GOs are high. Therefore, it may be that the cluster is so small because people

adopting high avoidance GOs might not be willing to indicate that they adopt avoidance GOs due to the negative connotations that this may imply. Another reason for the cluster being so small is that people adopting avoidance GOs may be less likely to participate in such studies for a number of reasons. For example, it was found that a performanceavoidance GO is predicted by fear of failure. This may cause individuals to be less likely to participate in research studies because of the stress that this fear of failure might cause. Therefore, although the additional cluster is a small cluster, it might be a very important one.

The 2x2 model of GOs is becoming more widely accepted in research on GOs and more and more researchers are beginning to make use of it in their studies (e.g. Narayan, & Steele Johnson, 2007; Van Yperen & Renkema, 2008; Karabenick, 2004; Van Yperen, 2006; Liu et al., 2009; Schantz & Conroy, 2009; Warburton & Spray, 2009; Putwain & Daniels, 2010, amongst others) since as Van Yperen and Renkema (2008: 260) stated "Elliot...introduced the most sophisticated contemporary achievement goal model." Although there has been an increase in research studies using the 2x2 model of GOs there are still many research studies which acknowledge the existence of the 2x2 model but proceed to use the three-factor model (e.g. Wolters, 2004) and even the two-factor model in their research (e.g. Diefendorff, 2004; Anderson & Dixon, 2009; Beckmann, et al., 2009; Hanrahan & Cerin, 2009; Bereby-Meyer et al., 2010; amongst others). Van Yperen et al. (2009) suggest that some researchers are reluctant to use the 2x2 model of GOs as a result of the fact that firstly, the concept of a mastery-avoidance GO is a relatively recent concept and secondly because

an avoidance component of mastery-based goals is more difficult to envision than an avoidance component of performance-based goals and empirical evidence to date has provided no evidence of a link between mastery-avoidance goals and indicators of actual performance. (Van Yperen et al., 2009:941).

39

Since there is still debate regarding whether the 3-factor or 2x2 model of GOs is the more comprehensive, a final decision regarding which model of GOs will be used in the current study will be made after the initial data analyses are carried out. As described earlier, the main difference between the 3-factor model and the 2x2 model of GOs is the mastery GO. In the 2x2 model it is divided into mastery-approach and mastery-avoidance GOs whilst in the 3-factor model there is a combined mastery GO. An examination of measures of the 3factor model of GOs indicated that, in measuring a mastery GO, these questionnaires tend to use items measuring a mastery-approach GO only rather than mastery-approach and mastery-avoidance GOs (e.g. Liem et al., 2008; Alkharusi, 2008). Consequently, the inclusion of the mastery-avoidance GO (to create the 2x2 model) does not change the structure of the 3-factor model, but rather, it simply adds a new dimension to it. Therefore, in essence, the 2x2 model is the 3-factor model plus 1 (the mastery-avoidance GO). As a result, in order to be able to decide whether to use the 3-factor or 2x2 model of GOs in the current study it was thought best to use a measure of the 2x2 model of GOs. Therefore, if the 3-factor model is thought to be more appropriate the mastery-avoidance scale will be omitted from the analyses. However, if the 2x2 model is found to be better, the mastery-avoidance scale will be included in the analyses.

An adapted version of the AGQ-R developed by Elliot and Murayama (2008) will be used in this study since this is the most recently developed measure using the 2x2 model of GOs and there seems to be sufficient validity and reliability information for use (refer to Chapter 7 Sections 7.1.5.1. and 7.2.5.1. for a description of the GO questionnaire used in this study). Moreover, the AGQ-R makes it possible to be flexible regarding model choice since it allows individuals to score high or low on one or more of the GOs. Therefore, if an individual scores highly on the mastery-avoidance GO items, this will not influence his/her scores on the other three GOs thus making it possible to omit the mastery-avoidance GO scale without having any influence on the other three GOs.

3.3. Multiple Goal Perspective

Some researchers (e.g. VandeWalle, 1997; Darnon et al., 2010) seem to assume that early GO researchers viewed mastery and performance GO as being at opposite ends of the same continuum. However, Dweck and Elliott (1983:645) explain how GOs are not mutually exclusive and, in fact, they explicitly stated that "all three goals can coexist, but the presence of one is sufficient (also necessary) to define a situation as an achievement situation." Hence, although in their early studies on GOs Dweck and Elliott (1983) did not explicitly make use of multiple goal measures they certainly did not assume that learning and performance goals are mutually exclusive.

As described in Chapter 2 (Section 2.1.2.) Nicholls (1984) made use of the terms 'taskorientation' and 'ego-orientation'. He too did not assume that these two orientations are mutually exclusive. In fact, Nicholls et al. (1989:69) state that "There is not, as some imply, a bipolar dimension of task- versus ego-orientation. Students are approximately as likely to be high or low on both dimensions as they are to be high on one and low on the other." In another study Nicholls et al. (1990:110) describe how a number of research studies (e.g. Maehr & Braskamp, 1986; Nicholls et al., 1985; Nolen, 1988; and Thorkildsen, 1988) found task- and ego-orientations to be "virtually independent of one another". According to Urdan and Maehr (1995:215) as a result of the fact that a number of researchers found goal orientations "to be weakly related or even orthogonal (Nicholls, 1992), recent research has begun to examine the effects of pursuing multiple goals". Both the terms 'profile perspective' (e.g. DeShon & Gillespie, 2005) and 'multiple goal perspective' (e.g. Barron & Harackiewicz, 2001; Pastor et al., 2007; and Midgley et al., 2001) refer to the view of those researchers who accept that individuals may be high or low on one or more GOs. Several researchers (e.g. Van Yperen, 2006) acknowledge the multiple goal perspective. For example Janssen and Prins (2007:237) state that "several achievement goal orientations can coexist in a person, so that, for example, trying to develop competency is not necessarily inconsistent with striving to demonstrate competency." Although the multiple goal perspective seems to be more accepted by researchers in recent years (since, as will be discussed below, a number of research studies have been carried out in order to investigate it) DeShon and Gillespie (2005:1102) describe how "goal orientation researchers are not commonly using this more complex goal orientation perspective." This may be because use of the multiple goal perspective may complicate matters in attempting to understand the causes and consequences of GOs. Since most researchers are interested in determining the causes and consequences of GOs they may decide NOT to adopt the multiple goal perspective in order to be able to understand the causes and consequences of each GO more clearly.

However, if individuals do in fact adopt multiple GOs, choosing to examine the relationships between each GO and other variables separately may provide an incorrect picture of how GOs relate to different variables. Barron and Harackiewicz (2001: 707) emphasise this in their research when they state that "Given the possibility that individuals can and do pursue multiple goals, it is critical to test the simultaneous effects of mastery and performance goals, as well as test whether mastery and performance goals interact." They describe how many GO studies do not test for these interaction effects. Thus, in their research they attempted to determine whether adopting multiple GOs influences performance. More specifically, they wanted to determine whether adopting a high mastery and a high performance GO simultaneously would result in better performance than adopting only a high mastery GO.

Barron and Harackiewicz (2001) suggested four different ways in which adopting multiple GOs may enhance performance as opposed to adopting a single GO.

42

In the *additive goal hypothesis*, they propose that both mastery and performance GOs will have positive and independent effects on a particular achievement outcome. Thus, the two GOs together will have a greater positive effect on achievement as opposed to either of the GOs alone.

As the name suggests, in the *interactive goal hypothesis*, the mastery and performance GOs are expected to interact so that individuals adopting both goal orientations are at an advantage over those adopting a single GO.

In the *specialised goal hypothesis*, they suggest that mastery and performance GOs have positive effects on different outcomes (e.g. mastery GO increases interest whilst performance GO increases performance), thus adopting both GOs simultaneously will be of benefit overall.

Finally, in the *selective goal hypothesis*, individuals are assumed to be able to choose between pursuing different GOs and are thus able to adopt the GO that best suits the situation at hand.

As a result of their research, Barron and Harackiewicz (2001) found support for the specialised goal hypothesis. They used the two-factor model of GOs and found that mastery and performance GOs each had a positive influence on different achievement outcomes. The mastery GO was found to be a predictor of interest (but not performance), whilst the performance GO was found to positively influence performance (but have no effect on interest). Wolters (2004) also investigated the influence of multiple goals on behaviour, and found partial support for the specialised goal hypothesis in that adopting a performance-avoidance GO along with other GOs was found to have a negative influence on task-involvement. In another research study, Harackiewicz et al. (2002) found further support for the specialised hypothesis. Their results indicate the positive consequences of adopting both mastery and performance-approach GOs in college courses. Their data

provide evidence that mastery and performance-approach GOs "have positive and complementary consequences for motivation and performance in college courses over the course of students' academic careers" (Harackiewicz et al., 2002: 574). Harackiewicz et al. (2000) also provided evidence for the specialised hypothesis when they found that a mastery GO positively predicted course interest (but not performance) whilst a performance GO positively predicted performance (but not interest).

However, the results of a study conducted by Pintrich (2000b) do not seem to support the specialised goal hypothesis. He found that adopting high mastery and high performance GOs simultaneously did not result in better performance than adopting high mastery and low performance GOs. The contradictory research results obtained may be a consequence of the two-factor model of GOs being used in the study by Pintrich (2000b) as opposed to the more differentiated model which was used in all the other research studies mentioned (that investigated the multiple goal perspective). As was discussed in the previous section (Section 3.2.) studies that did not differentiate between performance-approach and performance-avoidance GOs (i.e. studies that used a 2-factor model of GOs) obtained inconsistent results. This provides a possible explanation for the inconsistent results obtained across studies using different GO models.

So far, it seems as though only two studies adopting the multiple goal perspective use the 2x2 model of GOs. These are the study by Pastor et al. (2007) and that by Cano and Berben (2009). These studies examined the types of GO profiles that participants adopt. Two of the studies reviewed, those by Pastor et al. (2007) and Fortunato and Goldblatt (2006), used the 3-factor model to investigate the types of GO profiles that participants adopt. Pastor et al. (2007) used both the 3-factor and the 2x2 models of GOs in their research since they were interested in examining which model provides the best understanding of GOs in terms of the multiple goal perspective. Since a decision regarding model choice will be made following preliminary analyses of data, it was thought useful to provide an overview of the GO profiles obtained by studies using the 3-factor model as

well as the 2x2 model of GOs. Please refer to Table 3.2. and Table 3.3. for an overview of the results of studies using the 3-factor and 2x2 models of GOs, respectively. The percentages shown in brackets refer to the percentages of participants who were found to adopt this GO profile.

	Cano & Berben (2009)	Pastor et al. (2007)
	(27.21%) MAP : Low	(13%) MAP : High
Cluster 1	MAV : Low	MAV : Moderately Low
	PAP : Moderately Low	PAP : High
	PAV : Moderately High	PAV : Moderately High
	(20.44%)	(11%)
	MAP : Moderately High	MAP : High
Cluster 2	MAV : Moderately Low	MAV : Moderately Low
	PAP : Moderately Low	PAP : High
	PAV : Low	PAV : Moderately Low
	(28.09%)	(30%)
	MAP : Moderately High	MAP : Moderately High
Cluster 3	MAV : High	MAV : Moderately Low
	PAP : Low	PAP : Moderately High
	PAV : Moderately High	PAV : Moderately High
	(24.26%)	(24%)
	MAP : Moderately High	MAP : Moderately High
Cluster 4	MAV : High	MAV : Low
	PAP : High	PAP : Moderately High
	PAV : High	PAV : Moderately Low
		(20%)
Cluster 5		MAP : Moderately High
		MAV : Moderately Low
		PAP : Moderately Low
		PAV : Moderately Low
		(2%)
		MAP : High
Cluster 6		MAV : High
		PAP : High
		PAV : High

Table 3.2. Goal Orientation Profile Results of Studies Using the 2x2 Model

Key: MAP = Mastery-Approach Goal Orientation; MAV = Mastery-Avoidance Goal Orientation; PAP = Performance-Approach Goal Orientation; PAV = Performance-Avoidance Goal Orientation. Cano & Berben (2009) N=680; Pastor et al. (2007) N=1868

	Pastor et al. (2007) N=1868	Fortunato & Goldblatt (2006) N=311
Cluster 1	(12%) MAP: High PAP: High PAV: Moderate	(32%) MGO : Low PAP : Moderate PAV : Moderate
Cluster 2	(9%) MAP: High PAP: High PAV: Mod Low	(18%) MGO : Moderate PAP : Low PAV : Low
Cluster 3	(25%) MAP: Mod High PAP: Mod High PAV: Moderate	(28%) MGO : Moderate PAP : High PAV : High
Cluster 4	(44%) MAP: Mod High PAP: Moderate PAV: Mod Low	(22%) MGO : High PAP : Moderate PAV : Low
Cluster 5	(10%) MAP: Moderate PAP: Low PAV: Mod Low	

Table 3.3. Goal Orientation Profile Results of Studies Using the 3-factor Model

Key: MAP = Mastery-Approach Goal Orientation; MAV = Mastery-Avoidance Goal Orientation; PAP = Performance-Approach Goal Orientation; PAV = Performance-Avoidance Goal Orientation.

Cano and Berben (2009) and Fortunato and Goldblatt (2006) used cluster analysis in order to examine GO profiles whilst Pastor et al. (2007) used Latent Class Analysis (LCA). LCA is similar to cluster analysis but it is model-based unlike cluster-analysis (Pastor et al., 2007). An advantage of LCA over other types of cluster analysis is that there are more rigorous criteria available in order to decide on the final model (Pastor et al., 2007). Yeo et al. (2008:298) argue that it has been difficult to interpret findings with regards to the multiple goal perspective "because most past research has used analytic techniques such as median splits or cluster analysis which prevent a powerful test of interactions". As a result, it seems necessary to use more advanced statistical tools such as LCA in order to advance knowledge on GO profiles. A more detailed description of LCA will be provided in the results chapter (Chapter 8 Section 8.5).

A comparison of the GO profiles of the two studies presented in Table 3.2. revealed that Cluster 2 and Cluster 4 of Cano and Berben (2009) are quite similar to Cluster 5 and Cluster 4, respectively, of Pastor et al. (2007). A comparison of the GO profiles presented in Table 3.3. indicates that Cluster 5 in the study by Pastor et al. (2007) and Cluster 2 obtained by Fortunato and Goldblatt (2006) are very similar in that participants seem to be adopting a higher mastery-approach GO than performance-approach and performance-avoidance GOs. A second notable commonality is that Cluster 3 from the study by Pastor et al. (2007) and Cluster 3 from the study by Fortunato and Goldblatt (2006) seem to have moderate to high GOs for all three GOs in the profiles. Moreover, Clusters 2 and 4 from the studies by Pastor et al (2007), Fortunato and Goldblatt (2006), respectively, may be argued to be similar since the PAV GO in these profiles is lower than the MAP and PAP GOs. Consequently, participants adopting this particular GO profile seem to have high approach and low avoidance GOs.

As a result of their study, Pastor et al. (2007) encouraged researchers to carry out further analyses of GOs using the multiple goal perspective in order to determine whether GO profiles are replicated or not. This is necessary in order to provide evidence of the consistency and validity of the GO profiles. The similarities across studies indicate that further research into GO profiles might result in the determination of patterns of GO profiles. Therefore, the recommendation by Pastor et al. (2007) seems to be a very sensible one. The latter also recommended that research should be carried out in order to attempt to better understand the relationships between the different GO profiles and outcomes of GOs. The present study takes up these recommendations made by Pastor et al. (2007) and aims to help to provide a better understanding of the GO profiles that people adopt. This study will improve on the study by Pastor et al. (2007) by using a more recent and better developed questionnaire in order to assess GOs using the multiple goal perspective (based on the 2x2 model of GOs). The questionnaire used by Pastor et al. (2007) was the AGQ developed by Elliot & McGregor (2001). As described in Section 3.2.3. Elliot and Murayama (2008) developed the questionnaire by Elliot and McGregor (2001) further by making some amendments to the items in order for the questionnaire to better allow for the measurement of GOs using the multiple goal perspective. Similarly to the study of Pastor et al. (2007) this study will use Latent Class Analysis (LCA) to determine the number, size, and type of GO profiles that individuals may adopt.

Although there is support for the 2x2 model of GOs, there is still debate regarding whether the 3- or 2x2 factor models of GOs is more comprehensive. Pastor et. al (2007) compared the 3- versus the 2x2 models of GOs in terms of GO profiles in their study. However, their results do not provide a clear-cut conclusion as to which model should be used in future research. Consequently, in the current study, the multiple goal perspective will also be used to assess the utility of the 3- versus the 2x2 factor models of GOs. Therefore, the first Research Question for this study is:

Research Question 1: Using LCA as a method of clustering goal orientations, how many different types of goal orientation profiles are there and what are the characteristics of each goal orientation profile? Does the 2x2 model significantly improve on the 3-factor model in terms of identifying goal orientation profiles?

Once the second part of the Research Question 1 has been answered, a decision will be made regarding whether to use the 3- or the 2x2 factor model of GOs in the rest of the analyses using the multiple goal perspective in the current study.

Following one of the recommendations made by Pastor et al. (2007) and as a result of the contradictory evidence found in research so far (e.g. Barron & Harackiewicz, 2001 and Pintrich, 2000b) this study will also investigate the relationships between GO profiles and variables such as self-efficacy, performance, and mental effort. However, this will be discussed in further detail in Chapter 5 which focuses on the causes, consequences, and correlates of GOs.

3.4. Synopsis

The focus of this chapter was on the definitional concerns, dimensionality issues and utility of the multiple goal perspective. As mentioned in the introduction to this chapter there are also concerns regarding the stability and specificity of GOs. These will be discussed in depth in the next chapter.

Chapter 4: Conceptual Inconsistencies in Goal Orientation Research: Stability and Specificity of Goal Orientations

4.0. Further Concerns in Goal Orientation Research

The previous chapter focused on concerns regarding the definition and dimensionality of GOs as well as the utility of the multiple goal perspective. However, these are not the only concerns in GO literature. Additional issues include the stability and specificity of GOs. These are discussed in this chapter.

Although GO *profiles* are being investigated in the current study (as discussed in Chapter 3), the nature of individual GO scales will also be examined since this may provide further insight into the results obtained for GO profiles. From here onwards, so as to avoid confusion, investigation using individual GO scales (as opposed to GO profiles) will be referred to as the 'non-profile perspective'.

4.1. The Stability of Goal Orientations

Over the years, running in parallel to the concern of the dimensionality of GOs was the concern regarding the stability of GOs; that is, whether GOs are stable personality traits or whether they are state-like. Button et al. (1996:27) maintain that "past research has been inconsistent on this issue; some researchers treat it as an individual trait, whilst others have manipulated it as if it were a characteristic of the setting or situation". Another dilemma in GO research is whether GOs are general, domain-specific, task-specific, or situation-specific. The issues of stability and generality are likely to be deeply intertwined: if GOs are found to be stable personality traits then they are more likely to be state-like, they are more likely to be domain/task/situation-specific.

Although the two issues are closely related first the problem of stability of GOs will be addressed. Following this (Section 4.1.3.) the issues concerning the generality (or specificity) of GOs will be discussed. This decision to separate the two issues was made because they are in fact separate concerns and do not *necessarily* influence each other. For example, although it would be very likely that if GOs were found to be stable then they would also be general, there is a possibility that GOs are stable over time on particular tasks. Since it is still currently debatable whether GOs are stable and/or general, it was thought best to focus on these aspects of GOs separately.

4.1.1. Goal Orientations: Traits or States?

DeShon and Gillespie (2005:1115) describe how the problem of stability in GO literature "has been one of the most vexing problems". They describe how "Rather than face the issue head on, researchers appear to operationalize the construct in whatever manner is convenient for the adopted research methodology and then provide a passing acknowledgement that other perspectives exist." (DeShon & Gillespie, 2005: 1115). Payne et al. (2007) also highlight the problem of the stability of GOs in their research and state that the stability of GOs still needs to be determined.

There was always a great deal of confusion with regards to the nature of GOs. When the concept of GOs was first introduced, Dweck (1978) suggested that children respond to achievement situations according to their beliefs about ability and effort. These beliefs were assumed to be relatively stable over time (Dweck, 1978) thus implying that the resulting GOs should also be relatively stable. Nevertheless, in later research, Dweck and Leggett (1988:6) "induced experimentally" performance and learning GOs, thus implying that GOs are not as stable as initially thought to be. However, even though different GOs may be induced according to the situation this does not necessarily imply that they may

not be defined as traits. In fact, trait theorists do not rule out that people's behaviour may be influenced by situations and acknowledge the interaction between person and situation (Pervin, 1989). Dweck and Leggett (1988:269) clearly describe this idea regarding dispositions and situational influences as follows:

dispositions are seen as individual difference variables that determine the a priori probability of adopting a particular goal and displaying a particular behaviour pattern, and situational factors are seen as potentially altering these probabilities.

In addition, they argue that different GOs may be present in different situations. For example, in achievement situations achievement goals are not the sole predictors of behaviour. There may be other goals present such as social goals. Therefore, for example, a person might not adopt a performance-approach GO (which might indicate competitiveness) with a new group of friends owing to the fact that he/she may wish to appear agreeable since this may seem to be more socially acceptable in that particular situation.

As discussed in Chapter 2 (Section 2.1.2.) Nicholls (1984) proposed that the GOs adopted by individuals depend on whether they adopt a more or less differentiated conception of ability. Nicholls and Jagacinski (1984) describe how adults are able to adopt either conception of ability. They also suggest that certain situations induce people to adopt either the more or the less differentiated conception of ability. For example playing basketball in a sports competition would probably induce people to adopt a more differentiated conception of ability whilst playing basketball in training session would be more likely to induce a less differentiated conception of ability. Consequently, it seems as though Nicholls (1984) and Nicholls and Jagacinski (1984) believe that GOs are situationspecific and therefore, more state-like rather than trait-like. As a result of the fact that early researchers did not directly address the matter of GOs being trait-like or state-like many GO researchers seem to have assumed the views adopted by the early GO researchers unquestioningly. Nicholls' (1984) research seems to be used more in the area of sports psychology whilst Dweck's (1984) research is more prominent in the area of educational psychology. Since a lot more GO research was carried out in educational psychology than in sports psychology, the idea that GOs are stable traits has become a very prominent view in GO research. With regards to organisational psychology, there seem to be mixed views regarding the nature of GOs. A number of studies in the organisational setting draw on Dweck's early theories of achievement goals e.g. Rogers and Spitzmueller (2009), Hertenstein (2001) and Ford et al. (1998). Some organisational psychology researchers carrying out research on GOs view them as individual differences which may be influenced by situational factors (e.g. Kozlowski et al., 2001; Sujan et al., 1994; Kohli et al., 1998). There are also some researchers (in organisational psychology) who distinguish between trait and state GOs (e.g. Dragoni, 2005; VandeWalle, 1999; Steele-Johnson et al., 2008).

The review of GO literature carried out for the purposes of this study indicated that a number of researchers simply assume that GOs are stable dispositional traits (e.g. Cunningham & Xiang, 2008; Colquitt & Simmering, 1998; Kozlowski et al., 2001; Chen et al., 2000; Phillips & Gully, 1997). Other researchers acknowledge that there is an ongoing debate regarding the stability of GOs but nonetheless assume that GOs are traits that can be influenced by situational cues (e.g. Bettencourt, 2004; van Hooft & Noordzij, 2009; VandeWalle, 2003; Roberson & Alsua, 2002; VandeWalle & Cummings, 1997; Harris et al., 2008; Horvath et al., 2006). However, some researchers attempted to assess the stability of GOs. A description of these studies is provided below. This is followed by a discussion outlining why further research is required in order to arrive at an adequate conclusion regarding the stability of GOs.

Anderman & Midgley (1997) carried out a study in order to investigate the stability of children's GOs over a one year time period. The results of their study indicate quite low stability over time with correlation coefficients ranging between 0.28 and 0.47 (ρ <0.001). Elliot and McGregor (2001) used simultaneous regression analyses to assess the stability of GOs. They found evidence of stability for all the four GOs which were found to predict subsequent GOs.

In 2007, Fryer and Elliot assessed changes in the GOs of individuals (for academic courses) at three points in time (approximately 5 weeks apart). They found evidence for both stability and change over time. When assessing change (based on the 2x2 model) using differential continuity (that is, correlations over time) they found GOs to be quite stable over time with correlations between 0.57 and 0.78 (ρ <0.001). When assessing stability using mean-level change (that is, the average amount of change in a construct over time) they found mastery-approach and performance-avoidance GOs to significantly increase and decrease, respectively, from Time 1 to Time 2. No significant changes were found for mastery-avoidance and performance-approach GOs for this time period. None of the GOs changed significantly from Time 2 to Time 3 whilst from Time 1 to Time 3 only the mastery-approach GO changed significantly (increased). They also assessed individuallevel change of GOs over time. This "represents the magnitude of increase or decrease in a construct over time exhibited by an individual...individual-level change examines stability and change at the level of the single person within the sample." (Fryer & Elliot, 2007:702). This was assessed by means of the reliable change index (RCI) and the results indicated changes in all four GOs over time. Finally, they assessed stability of GOs in terms of ipsative continuity. This "represents the level of stability and change exhibited in an individual's configuration of constructs over time" (Fryer & Elliot, 2007:702). In order to do this, they correlated each participant's scores on all four GOs over the three time points in order to estimate profile consistency coefficients. Their results indicated withinperson stability over time.
Muis and Edwards (2009) carried out a study to investigate the stability of GOs over time and the task-specificity of GOs. They examined the GOs of students for the same academic subject over four time points (using a three-factor model of GOs). At two time points participants were required to complete an exam for the course whilst at the other two time points participants were required to complete an assignment for the course. The table below provides a brief description of the tasks undertaken at each time point.

Time Point at which GOs were measured	Task being carried out at this time point
Time 1	Assignment 1
Time 2	Exam 1
Time 3	Assignment 2
Time 4	Exam 2

Table 4.1. Tasks carried out at each time point

The GOs were measured across a semester. However, it was not evident how long the gaps between the time points were. The different assessment types were considered to constitute different tasks. Consequently, stability of GOs over time was assessed by examining changes from Time 1 to Time 3 and from Time 2 to Time 4. On the other hand task-specificity (stability across tasks) was assessed by examining changes from Time 1 to Time 4 and Time 1 to Time 4. The task-specificity of GOs will be discussed in further detail in Section 4.2.

Stability and task-specificity were assessed by means of differential continuity, mean-level change, and individual-level change. In addition, Muis and Edwards (2009) examined individuals profile consistency with regards to the GOs adopted. In order to do this, they used the dominant GOs of participants to examine whether participants switched GOs across tasks (to assess task-specificity) or over time to assess stability (e.g. if they scored highest on a mastery-approach GO on Exam 1 did they also score highest on Exam 2 or did

they switch to a high performance-approach GO?). Thus, although they refer to it as 'profile consistency' they did not use the multiple goal perspective.

Their differential continuity results indicated a moderate to high level of stability for the three GOs with correlations ranging from 0.32 to 0.78 (significance level not provided). When assessing stability in terms of mean-level change they found some evidence of stability and some evidence of change (refer to Table 4.2.). RCIs were used to examine individual-level change. RCI values less than -1.96 or more than 1.96 are considered to be evidence of reliable change (refer to Table 4.3. for the percentages of individuals who showed reliable change over time).

Table 4.2. Comparisons of Go	al Orientations over time
------------------------------	---------------------------

	Comparison Between Assignment 1 and Assignment 2	Comparison Between Exam 1 and Exam 2
МАР	No change	Increased (ρ<0.001)
РАР	No change	No change
PAV	No change	Decreased (p<0.002)

Key: MAP = Mastery-Approach Goal Orientation; PAP = Performance-Approach Goal Orientation; PAV = Performance-Avoidance Goal Orientation.

Table 4.3. Percentages	of Individuals who sh	nowed reliable change	(increases/decreases)
		0	

over time

	A1 to A2	E1 to E2
MAP	50%	72.4%
PAP	70%	78.5%
PAV	80.6%	78.5%

Key: A1= Assignment 1; A2 = Assignment 2; E1 = Exam 1; E2 = Exam 2; MAP = Mastery-Approach Goal Orientation; PAP = Performance-Approach Goal Orientation; PAV = Performance-Avoidance Goal Orientation.

Finally, when they assessed profile consistency, it was found that 76.7% of participants did not engage in goal switching on the assignments. However, only 37% of participants did not switch goals from Exam 1 to Exam 2. Goal switching was found to be predominantly from a mastery-approach to a performance-avoidance GO or vice versa. Overall, the results obtained indicate evidence of both stability and change over time.

Payne et al. (2007) conducted a meta-analysis of GO studies. One of their aims in this study was to assess the stability of trait GOs. Their results indicate that trait GOs

were quite stable over the short term.....However, the longer the time interval, the weaker the coefficient of stability, undermining the stability of trait GOs. That said, few studies have examined the stability of trait GOs beyond the length of one college semester, so the long-term stability of trait GOs remains unclear (Payne et al., 2007, pp139).

Their meta-analysis included 178 independent samples from 141 studies examining trait GOs. One outlier was found for the test of temporal stability, that is, the study by Amabile et al. (1994) which was removed from the analysis. In order to assess temporal stability Payne et al. (2007) examined the sample-weighted mean correlations for each dimension. Their results indicated moderate stability over time from 1 to 14 weeks (M=7.01, SD=3.89). The sample weighted mean, r, was 0.66 (total number of effect sizes included in the analysis, k, =20) for a mastery GO; r=0.70 (k=16) for a performance-approach GO; r=0.73 (k=4) for a performance-avoidance GO. Since this meta-analysis was based on the three-factor model of GOs, those studies which were conducted using a two-factor model had the performance GO results included with the performance-approach GO dimension for the analyses. This decision was made since Payne et al. (2007) thought that the performance-approach GO than the performance-avoidance GO.

In addition, Payne et al. (2007) examined the length of time between administrations as a continuous moderator of the test-retest relationships. The results indicated that the

relationships between the time interval and the coefficients of stability were negative for all three GOs (for a mastery GO r=-0.20, for a performance-approach GO r=-0.29, and for a performance-avoidance GO r=-0.74). This led them to the conclusion that the longer the time frames between the administrations, the smaller the coefficients of stability. Thus, the stability of GOs seems to decrease over time. Although the results of this metaanalysis offer a good indication of the stability of GOs there are a number of limitations. Firstly, as suggested by the authors themselves, the administrations of GO measures might have coincided with the beginning and end of a semester. Therefore, changes in GOs may have been due to changes in situation e.g. exams or assignments due in at the end of a semester may have led to a stronger performance-approach GO. Secondly, as mentioned earlier, although the three-factor model of GOs was used in this study some of the studies included in the analyses used a two-factor model (and their performance GO being classified as a performance-approach GO) which may limit the accuracy of the results. Thirdly, the studies used in this meta-analysis did not use the multiple goal perspective. Different results may be obtained if GO profiles are assessed as opposed to GOs from the non-profile perspective. Finally, the use of meta-analysis for testing the stability of GOs is questionable due to all the inconsistencies in GO research. The major concern in this case would be that the studies included in the meta-analysis used different measures of GOs. As a result of GO researchers using different definitions of GOs in order to guide their research, the GO scales used in the different studies are not necessarily measuring the same concepts. Consequently, the results of the meta-analysis should be interpreted with caution.

Although a few research studies have investigated the stability of GOs there is still a lot more evidence required in order to be able to draw sound conclusions based on empirical evidence regarding the stability of GOs. Results from the research studies described above indicate evidence of both stability and change of GOs. However, the results are inconclusive for a number of reasons. The study by Anderman and Midgley (1997) was carried out using children as participants. It is very likely that traits are not yet stable in children. For example, personality traits are thought to become stable after thirty years of age (Costa & McCrae, 1988). Therefore, the changes in GOs found in the study by Anderman and Midgley (1997) may be a consequence of age rather than GOs having low stability. Elliot and McGregor (2001) used correlational analyses in order to assess stability over time. Although they found a high degree of consistency for all four GOs over time, correlational analysis alone is not sufficient to assess the stability of GOs.

As described above, Fryer and Elliot (2007) assessed stability and change of GOs by means of differential continuity, mean-level change, individual-level change, and ipsative continuity. Since they closely examine the stability of GOs using the 2x2 model of GOs, their study provides a good understanding of the stability and change of GOs. Muis and Edwards (2009) also closely examined the stability of GOs over time by assessing differential continuity, mean-level change and individual-level change. However, both studies do not provide evidence regarding the stability of general GOs since task-specific measures of GOs were used in their study. (The problem of generality versus taskspecificity will be described in further detail in Section 4.1.3. below). With respect to assessing the stability of GOs it might be useful to assess the stability of general GOs before assessing the stability of task-specific GOs.

Muis and Edwards (2009) assessed what they refer to as 'profile consistency'. However, they only measured changes in participants' dominant GOs. Consequently, this does not provide much information regarding the stability of multiple GOs. On the other hand, Fryer and Elliot (2007) assessed the stability of GO profiles by means of ipsative continuity. This provides within-person correlations for the participants' GO profiles over time thus showing whether each participant's profile changes over time. In order to assess the stability of GO profiles from a broader perspective (that is, at the sample level as opposed to the individual level) it was thought to be appropriate to use LCA in the

current study. LCA provides clusters/classes indicating the types of GO profiles that participants adopt. It therefore allows us to investigate whether, **overall**, participants' tend to change their GO profiles over time or not.

As a result of the limitations of the research studies discussed above and the limited research on the stability of GO profiles, it is still inconclusive whether general GOs (and GO profiles) are stable over time or not. Therefore, the current study aims to further develop knowledge on the stability of GOs by using LCA to assess the stability of general GO *profiles* of an adult sample over time as well as further investigate the stability of GOs from the non-profile perspective. The following Research Questions were therefore proposed.

Research Question 2a: Do individuals' General goal orientation profiles change over time? **Research Question 2b:** Do individuals' General goal orientations change significantly over time?

4.1.2. Goal Orientations: Traits and States?

As a result of the debate regarding the stability of GOs some researchers have discussed the possibility of GOs existing as both traits and states. For example, Pintrich and Schunk (2002:237) maintain that

It may be that the solution to this matter regarding the stability of goal orientations will involve the adoption of a strategy used in social and personality psychology that assumes that both situational and personal conceptualizations are important and the issue is to specify how they interact

Harwood and Swain (1998) conducted a study using both dispositional (trait) and situational (state) GO measures. This study was conducted in a sports psychology context

using a two-factor model of GOs. They found that trait GOs did not predict the state GOs. Therefore, they suggest that both trait and state GO measures should be used in GO research. These findings may be due to different measures being used to measure trait and state GOs. Results might have differed had an adapted version of the trait measure been used to assess state GOs.

In another study, Button et al. (1996) attempted to investigate whether GOs are dispositional traits which may be influenced by situational characteristics or not. They concluded that a GO may be described as "a somewhat stable individual difference that may be influenced by situational characteristics" (Button et al., 1996: 28). Therefore, individuals are predisposed to have particular response patterns. However, their response patterns may vary considerably as a result of situational influences (Button et al., 1996). It seems as though they found that GOs may be both traits and states. This is evident from their correlational analyses in which dispositional and situational GOs were found to be related but distinguishable. Button et al. (1996:40) themselves state that "dispositional and situational goal orientations were distinguishable." However, as with Harwood and Swain (1998), this distinction may have resulted from the fact that they used two different questionnaires to measure trait and state GOs. It is debatable whether it is best to use adapted or different measures to examine whether GOs exist as both traits and states. 'Adapted measures' refers to using questionnaires which have only very slight differences in wording to assess trait and state GOs (e.g. by keeping the item wording as similar as possible but for the trait measure to ask about participants' GOs in general and for the state measure to ask about participants' GOs for that particular point in time). 'Different measures' refers to questionnaires developed specifically for assessing trait or state GOs which have items with very different wording from each other (e.g. Harwood & Swain, 1998, used the 'Task and Ego Orientation in Sport Questionnaire' to measure dispositional GOs and the 'Match Context Questionnaire' to measure state GOs).

On the one hand, if different measures are used it may be argued that any differences found may be a result of different constructs being measured. On the other hand, if an adapted measure is used and no differences are found, it may be argued that this is a result of common method variance. However, if an adapted measure is used and significant differences are found, this would clearly indicate that there are differences between trait and state GOs. Since studies have already found significant differences when assessing GOs using different measures for trait and state GOs, adapted measures will be used in the current study. If significant differences are found, then this will provide additional support for the existence of trait (general) and state (specific) GOs.

This research into trait and state GOs led other researchers to examine these concepts. For the purposes of this review studies assessing state GOs have been placed into three categories. The first category consists of studies attempting to prove that trait GOs influence state GOs. The second category consists of studies that only examine state GOs and their consequences. Studies in the third category focus on assessing the interactions between trait and state GOs. The studies shall be discussed in the order mentioned above. For the purpose of this review 'induced' (e.g. Steele-Johnson et al., 2008), 'manipulated' (e.g. Barker et al., 2002), or 'assigned' GOs (e.g. Van Yperen, 2003) are considered to be state GOs.

Category 1: Breland and Donovan (2005) suggest that dispositional GOs influence state GOs which in turn influence self-efficacy and performance. Their research study focused on attempting to prove the relationships between trait GOs, state GOs and self-efficacy using structural equation modelling. From their study, it was evident that dispositional GOs do in fact influence the corresponding state GOs.

Apart from assessing the stability of GOs (as described earlier) in their meta-analysis Payne et al. (2007) also investigated the antecedents and proximal consequences of GOs. They suggest that state GOs are proximal consequences of trait GOs. 178 independent samples from 141 studies were used for the analyses of trait GOs whilst 19 independent samples derived from 16 studies were used for the analyses of state GOs. A three-factor model of GOs was used for the analyses. Trait and state GOs were found to be positively correlated with one another (estimated true mean correlations = 0.55, 0.58, and 0.55). However, since this was a meta-analysis, it cannot be determined whether the trait GOs were actually **antecedents** of state GOs or not.

Category 2: Six studies were found that examined the consequences of state GOs. These were, the studies by Steele-Johnson et al. (2008), Steele-Johnson et al. (2000), Loraas and Diaz (2009), Barker et al. (2002), Van Yperen (2003) and Elliot et al. (2005). Steele-Johnson et al. (2008) found that inducing a state mastery GO is related to participants reporting higher levels of perceived challenge than when no state mastery GO is induced. They also hypothesised that individuals with an induced state mastery GO will initially demonstrate higher levels of task performance than those without an induced state mastery GO. However, they failed to find support for this hypothesis. They found support for the hypothesis that for individuals having high cognitive ability, inducing a mastery GO demonstrates greater improvement in task performance over time than when no mastery GO is induced. However, they did not find support for the hypothesis that a state mastery GO relates to higher satisfaction. Finally, they found that those individuals with an induced performance GO reported higher levels of perceived effort required. The results obtained by Steele-Johnson et al. (2008) should be interpreted with caution. This is mainly due to the way in which the mastery GO was induced. In order to induce the mastery GO Steele-Johnson et al. (2008) informed participants that performance could be increased by developing strategies for learning. However, participants having a mastery GO would not necessarily want to improve performance but rather their focus would be on increasing learning. This reference to performance in the GO induction is not consistent with inducing a mastery GO but rather it is consistent with inducing a performance GO. As

a result, the conclusions derived from this study might not be accurate in describing the effects of inducing a mastery GO.

Steele-Johnson et al. (2000) also conducted a study (using the two-factor model of GOs) in order to investigate the influence of state GOs on performance. They assessed the influence of state GOs on performance for two tasks: a simple task and a difficult task. Their results indicate that for the simple task, participants having an induced performance GO performed better than participants having an induced mastery GO. On the other hand, for the difficult task there were no differences in performance between participants having an induced mastery or performance GO. Performance on the difficult task was lower than performance on the simple task for both groups of participants.

Loraas and Diaz (2009) found that participants with a state mastery GO indicated intent to learn how to use new technology regardless of perceived difficulty whilst participants with a state performance GO indicated intent to learn how to use new technology depending on how easy it was to learn.

Barker et al. (2002) carried out a study in which they induced mastery, performanceapproach and performance-avoidance GOs in order to assess the recall of verbal information. Their results indicate that participants in the performance-approach and performance-avoidance induction conditions had higher levels of recall than participants in the mastery induction and control conditions. The manipulations in this study are also questionable. The mastery GO inductions refer to improving performance which, as mentioned earlier is not entirely accurate since individuals with a mastery GO would want to improve their learning and not necessarily their performance. In addition, the performance-avoidance GO induction was not entirely accurate and seemed quite unethical, especially since the experiments were conducted with kindergarten and primary school students! In the performance-avoidance manipulation the emphasis was on not appearing silly in front of the class as opposed to trying not to be one of the worst performers: "Answer the following questions to this test with the correct answers so your class don't think you are silly or stupid." (Barker et al., 2002: 578). Consequently, the results of this study should also be interpreted with caution.

Van Yperen (2003) induced mastery-approach, mastery-avoidance, performanceapproach and performance-avoidance GOs in his study. His study focused on determining whether GOs that are incongruent with task interest (which he considers to be all the GOs except for mastery-approach) undermine the positive effect of task interest on performance. The results of his study support his hypotheses in that individuals were found to have higher performance when they showed high task interest. Moreover, he found that assigned mastery-avoidance, performance-approach and performanceavoidance GOs undermined the positive effect of task interest on performance.

Elliot et al. (2005) induced mastery, performance-approach and performance-avoidance GOs. They found that a performance-avoidance GO was found to undermine performance relative MGOs or PAP GOs. According to Elliot et al. (2005) the performance-avoidance manipulations in the studies by Barker et al. (2002) and Van Yperen (2003) were not entirely accurate since participants in this condition (in both studies) were asked to focus on getting answer right as opposed to not getting answers wrong. GO manipulations have not been consistent across studies and, as mentioned above, not all manipulations are entirely accurate. Consequently, it is advisable to keep the way in which GOs were manipulated in mind when interpreting research results.

Although some of the manipulations are questionable, the results described above indicate how examining state GOs might provide valuable information regarding their consequences. However, the studies described above do not take trait GOs into account.

If trait GOs do in fact exist, it may be wise to investigate the interactions between trait and state GOs. This is what researchers in Category 3 did.

Category 3: Three studies of GOs, those by Chen and Mathieu (2008); Gerhardt and Luzadis (2009) and Jagacinski, Madden, and Reider (2001), focused on investigating the interactions between trait and state GOs and the influence of these interactions on performance. In these studies participants' trait GOs were measured and state GOs were induced, the effects on performance were then measured. All three studies used a two-factor model of GOs. The study by Chen and Mathieu (2008) indicated that a mastery GO predicted a more positive performance trajectory when combined with an induced performance GO than when combined with an induced mastery GO. There were no significant interaction effects between trait performance and state (mastery and performance) GOs. The study carried out by Gerhardt and Luzadis (2009) only measured a mastery GO. This study was focused on determining whether a trait mastery GO coupled with an assigned mastery GO would result in a higher level of performance than if a trait mastery GO is coupled with an assigned performance GO. The authors did not find any support for their hypothesis.

In their study, Jagacinski et al. (2001) attempted to determine whether inducing mastery and performance GOs influenced the performance of individuals having either mastery or performance trait GOs. It was found that inducing a mastery GO had a *nearly* significant positive effect on performance for participants having a trait mastery GO. They did not find support for their hypothesis that inducing a performance GO would have more positive effects on the performance of those individuals having a performance trait GO than those individuals having a mastery trait GO.

Although it seems as though there is evidence for the presence of trait and state GOs, further research is definitely required. Additional assessment of the interaction between

trait and state GOs is also necessary. The studies carried out by Chen and Mathieu (2008) and by Gerhardt and Luzadis (2009) have a number of limitations. With respect to the study by Chen and Mathieu (2008), the trait GOs and state GOs were measured in very close proximity. Therefore, the measurement of trait GOs might have had a carryover effect on the measurement of state GOs. The study by Gerhardt and Luzadis (2009) only measured a trait mastery GO. Therefore, it does not provide sufficient information regarding the interactions of assigned GOs with trait GOs. Finally, Jagacinski et al. (2001) themselves state that the observed effects of their study were not strong and that future investigation into the effects of the interactions between trait and state GOs is necessary. The current study aims to improve knowledge on trait and state GOs by investigating the interactions between dispositional (trait) and induced (state) GOs from the profile and non-profile perspectives.

The state GOs in the current study were limited to mastery-approach and performanceapproach GOs since it was thought to be rather unethical to induce avoidance GOs as a result of the negative implications that these inductions might have. I do not think it is appropriate to ask participants to focus on not having the worst performance (performance-avoidance induction) or to make sure that they do not misunderstand the task (mastery-avoidance induction) since this might make some individuals feel uncomfortable or insulted. Consequently, the following Research Questions and Hypothesis were proposed:

Research Question 3a: How do trait goal orientation profiles interact with induced mastery-approach and performance-approach goal orientations in order to influence task performance of participants?

Hypothesis 1. Individuals holding a trait mastery-approach goal orientation are expected to perform significantly better when a performance-approach goal orientation is induced as opposed to when a mastery-approach goal orientation is induced.

The Hypothesis above was proposed since the results of the study by Chen and Mathieu (2008) indicate that there is a significant positive effect on performance of individuals having a mastery trait GO when a performance GO is induced. A possible explanation for this result is the 'specialised goal hypothesis' (described earlier) proposed by Barron and Harackiewicz (2001) in which mastery and performance GOs have beneficial effects on different outcomes. Consequently, the adoption of both GOs would have an overall more beneficial effect than the adoption of just one of the GOs. Barron and Harackiewicz (2001) suggested this hypothesis (and found support for it) in the context of multiple goal adoption (as opposed to that of the trait-state GO interactions being discussed here). However, since the results of Chen and Mathieu (2008) are remarkably similar to those obtained by Barron and Harackiewicz (2001) it is possible that, regardless of whether the GOs being measures are traits or states, the presence of both mastery and performance GOs may be more beneficial than the presence of one GO alone.

Since there is not sufficient evidence for the interaction effects of trait masteryavoidance, performance-approach and performance-avoidance GOs with state masteryapproach and performance-approach GOs on performance these relationships will not be hypothesised but investigated as Research Questions.

Research Question 3b. What effects will the relationships between a trait masteryavoidance goal orientation and induced mastery-approach and induced performanceapproach goal orientations have on the performance of participants? This Research Question will only be answered if a 2x2 model of goal orientations is chosen as an appropriate model. **Research Question 3c.** What effects will the relationships between a trait performanceapproach goal orientation and induced mastery-approach and induced performanceapproach goal orientations have on the performance of participants?

Research Question 3d. What effects will the relationships between a trait performanceavoidance goal orientation and induced mastery-approach and induced performanceapproach goal orientations have on the performance of participants?

4.2. Generality vs. Specificity of Goal Orientations

In addition to the problem concerning GOs being traits or states, another concern raised in recent literature focuses on the specificity of GOs. Besides the assumption that GOs are stable, many researchers also assume that GOs are general (e.g. Rheinberg et al., 2000). Although they are assumed to be general, some studies investigating 'general' GOs use task-specific measures (e.g. Fan et al., 2008; Lin et al., 2009). This is a great cause for concern due to the fact that if GOs are truly general dispositions then task-specific measures should not be required. Researchers using trait measures do not usually adapt their questionnaires to the task/situation at hand. The fact that researchers have been using task-specific GO questionnaires might be an indication of the concern regarding the generality of GOs.

DeShon and Gillespie (2005) brought up the matter of generality vs. specificity of GOs in their review of GO literature. They indicate how studies have been inconsistent with regards to whether they conceptualise GOs as general, domain-specific, or situationspecific (these different terms are described below). From their review of GO studies, they found that 46.6% of the studies viewed GOs as being stable, general characteristics of the person. 26.1% of the studies viewed GOs as a combination of personal and situational factors whilst 4.5% had a view of GOs being highly unstable and situationally induced. Finally, 12.5% of the studies reviewed conceptualised GOs as being domain-specific. As mentioned earlier, the issue of generality is deeply intertwined with the issue of stability of GOs. The categories discussed by DeShon and Gillespie (2005) seem to address both issues simultaneously. However, in order to clarify each and every concern regarding the nature of GOs it is necessary to distinguish between the issues of generality vs. specificity and stability (trait vs. state) of GOs. With respect to the problem of generality, in order to maintain conceptual clarity, it is important to distinguish between: a) domain-specificity, b) task-specificity, and c) situational-specificity. A description of each term is provided below.

The dictionary definition of the word 'domain' is "a sphere of activity or knowledge" (Oxford English Dictionary, 2010). Examples may include: academic domain, sports domain, or social domain. However, it is not clear how specific domains are *or* what constitutes a domain. A task is defined as "A piece of work imposed, exacted, or undertaken as a duty or the like" (Oxford English Dictionary, 2010). Therefore, a 'task' refers to a specific piece of work e.g. a verbal aptitude test. 'Situation' refers to the context in which a task is carried out. Therefore, a person may be working on an essay with no time limit and another time under exam conditions. In such a case the task is the same but the situation (context) is different, thus it is situation-specific.

In 1997 Allan Wigfield wrote a paper highlighting the need to examine the domainspecificity of motivational constructs. The concept of GOs is one of the motivational constructs discussed. Wigfield (1997) describes how most research on GOs has used general measures. He states how research concerning competence and self-efficacy beliefs have provided strong evidence regarding the domain-specificity of these constructs but not much is known about the domain-specificity of GOs. Van Yperen (2006) also discusses the stability of GOs across time and across contexts. He suggests that future research should be carried out in order to investigate whether individuals have certain GOs anytime and anywhere or whether these differ as a function of time and domain.

Some studies (e.g. VandeWalle et al., 2001; Dweck, 1999; Yeo et al., 2009) assume that GOs are domain-specific, whilst other studies have tested this (e.g. Anderman & Midgley, 1997; Green et al., 2007; Magson et al., 2008; Bong, 2001; Nicholls et al., 1990). A description of the studies assessing the domain-specificity of GOs is provided below along with an outline of the results obtained. Following this there is a discussion regarding why further investigation into the generality vs. specificity of GOs is required and how this will be addressed in the current study.

Duda and Nicholls (1992) examined whether GOs were generalisable across academic and sport domains. They found that there was considerable generality across domains for GOs. The results of a study carried out by Anderman and Midgley (1997) contradict the results obtained by Duda and Nicholls (1992). The former found that overall participants reported higher performance goals in Maths than in English. This difference across the two studies may be a result of the domains assessed in the two studies being very different from each other. Duda and Nicholls (1992) used very broad domains (academic and sports) whilst Anderman and Midgley (1997) used two different academic subjects as different domains. Therefore, the domains used by the latter were much more specific than those used by the former. Bong (2001) highlights the broadness of the domains used by Duda and Nicholls (1992), and argues that they are too broad to understand much about the associations of GOs across domains. Therefore, she assessed GOs across four academic subjects: Korean, English, Science and Mathematics using the three-factor model of GOs. The results of her study indicate that performance-approach and performance-avoidance GOs were significantly consistent across academic subjects but the mastery GO was found to be subject-specific.

71

Green et al. (2007) also evaluated the domain-specificity of motivation. Unfortunately, the study is not focused solely on the domain-specificity of GOs and only a mastery GO was assessed. The results of this study indicate that students' mastery GOs were not identical across subjects therefore, there seems to be some evidence for domain-specificity. However, since only a mastery GO was assessed the usefulness of this study in supporting the domain-specificity of GOs is rather limited. In another study assessing the domainspecificity of GOs, Magson et al. (2008) address the need to discover whether GOs are general or domain-specific. They focus on testing the domain-specificity of mastery and performance GOs. Their results are very similar to those obtained by Bong (2001), in that, the performance GO appears to be generalised across subjects whereas the mastery GO appeared to be more subject-specific. One possible reason for a mastery GO being subject-specific is interest in the subject at hand. If a person is interested in the subject being studied, they are possibly more likely to adopt a mastery GO in which they are focused on learning as much as possible about the subject as opposed to simply trying to perform better than others in class (performance GO). However, this possibility would need to be investigated further. Since motivation and performance (which are of primary interest to organisations) are the main variables included in this study (as will be described in the next Chapter), interest will not be measured in the current study.

It is evident from the results of the studies described above that evidence regarding the domain-specificity of GOs is inconclusive. Some studies provide support for the generality of GOs whilst others provide support for the generality of some GOs but not others. There are a number of reasons which may explain the inconsistencies in the results of the above studies. First of all, all the studies described above used a two-factor model of GOs with the exception of Green et al. (2007) who only assessed a mastery GO. As described in Chapter 3 (Section 3.2.) the two-factor model of GOs is not differentiated enough. This may account for differences in the results across studies. Moreover, although all of the above studies assessing the specificity of GOs are said to be measuring 'domain-

specificity' it is arguable that all the studies assessing different academic subjects are in fact assessing 'task-specificity' within the academic domain. For the purposes of the current study, the studies assessing differences in GOs across **academic subjects** will be grouped together as studies assessing the task-specificity of GOs.

As mentioned earlier Muis and Edwards (2009) examined the task-specificity of GOs. When assessing this in terms of differential continuity they found moderate to high levels of stability in GOs across tasks (correlation coefficients ranged from 0.27 to 0.92, significance level not provided). Table 4.4. shows the results obtained when assessing task-specificity in terms of mean-level change.

Comparison	Assignment 1	Assignment 1	Assignment 2	Assignment 2
Between:	and Exam 1	and Exam 2	and Exam 1	and Exam 1
MAD	Decreased	No chango	Increased	No chango
IVIAP	(p<0.001)	No change	(p<0.001)	No change
DAD	No chango	No chango	Decreased	No change
PAP	No change	e No change	(ρ<0.002)	NO change
DAV	Increased	No chango	Decreased	No chango
PAV	(p<0.001)	No change	(p<0.001)	No change

Table 4.4. Comparisons of Goal Orientations across Tasks

Key: MAP = Mastery-Approach Goal Orientation; PAP = Performance-Approach Goal Orientation; PAV = Performance-Avoidance Goal Orientation.

Table 4.4. indicates both stability and change in GOs across tasks. When GOs were assessed in terms of individual-level change (refer to results in Table 4.5.) the percentages of participants showing reliable change seem to be quite high. Moreover, when the individual-level change results were aggregated to the group level each GO showed changes across tasks.

	A1 to E1	A1 to E2	E1 to A2	A2 to E2
MAP	77.4%	79.3%	73.3%	76.7%
PAP	76.7%	72.4%	72.4%	58.6%
PAV	80.6%	67.9%	76.7%	60.7%

Table 4.5. Percentages of Individuals who showed reliable change (increases/decreases) across tasks

Key: A1= Assignment 1; A2 = Assignment 2; E1 = Exam 1; E2 = Exam 2; MAP = Mastery-Approach Goal Orientation; PAP = Performance-Approach Goal Orientation; PAV = Performance-Avoidance Goal Orientation.

When assessing stability of GOs in terms of what Muis and Edwards (2009) refer to as 'profile consistency' (refer to Section 4.1. for an explanation of this) they found that the majority of participants did not switch goals from assignment 1 to exam 2 and from assignment 2 to exam 2. However, the majority of participants did seem to engage in goal switching from assignment 2 to exam 1 and assignment 1 to exam 1.

The results of the study described above indicate that as yet, no definite conclusions can be drawn with respect to the task-specificity of GOs because there seems to be evidence of both stability and change in GOs across tasks. Some results provide support for GOs not changing across tasks. For example some of the correlation coefficients obtained for GOs across tasks were quite high (e.g. 0.91 and 0.92). Additionally, there were no mean-level changes in GOs across assignment 1 and exam 2 and across assignment 2 and exam 1. Moreover, the majority of participants did not engage in goal switching across assignment 1 and exam 2 and across assignment 2 and exam 2. However, a number of other results obtained in this study provide support for GOs being task-specific. For example, some of the correlation coefficients obtained when assessing differential continuity were found to be as low as 0.27. Moreover, as indicated in Table 4.4. some of the mean-level change results show significant changes in GOs across tasks and the percentages of participants who showed reliable increases or decreases in GOs across tasks were quite high (refer to Table 4.5.). In addition, the majority of participants seemed to engage in goal switching across assignment 1 and exam 1 as well as across exam 1 and assignment 2. Since the results obtained do not provide consistent evidence regarding the task-specificity of GOs further investigation is necessary.

As described in Section 4.1., Muis and Edwards (2009) used the same academic subject with different assessment types (exams and assignments). These different assessment types were considered to constitute different tasks. Although different skills are required for the different types of assessment, the academic subject is the same. Consequently, it is debatable whether different assessment types could be said to constitute different tasks. Nevertheless, Muis and Edwards (2009) found some evidence for GOs changing significantly across tasks. The results of their study indicate the importance of investigating the task-specificity of GOs. Since they used the same academic subject without adopting the multiple goal perspective, this study will improve on the study conducted by Muis and Edwards (2009) by using different tasks to assess the task-specificity of GOs from the profile and the non-profile perspectives. Since evidence regarding the stability and change of GOs is inconclusive, it was not possible to develop Hypotheses for the current study. Instead, a number of Research Questions were raised:

Research Question 4a: Do participants adopt different goal orientation profiles across tasks?

Research Question 4b: Are participants' task-specific goal orientation profiles different from their General GO profiles?

Research Question 4c: Do participants' adopt different goal orientations across tasks? **Research Question 4d:** Are participants' task-specific goal orientations significantly different from their General goal orientations?

Research Question 5a: Are task-specific goal orientation **profiles** stable over time? **Research Question 5b:** Are task-specific goal orientations stable over time? A number of studies mention the idea of GOs being situation-specific rather than domainspecific or task-specific. In a review of GOs, Elliot and Dweck (2005:66) address the matter of generality vs. situational-specificity when they state that "although the achievement goal construct can be utilized at both dispositional and situation-specific level of analysis, conceptual and empirical considerations seem to suggest that it may be best suited for the situation-specific level." Although it is plausible that GOs are influenced by situations (e.g. time limit vs. no time limit), since not much is known about the task-specificity of GOs, it is thought to be ideal to first determine whether GOs vary according to task AND whether task-specific GOs are stable over time prior to investigating situational-specificity of GOs. This study focuses on investigating the former two.

4.3. Synopsis

In Chapter 3 (Section 3.1.) it was discussed how there is currently no widely accepted definition of GOs. As a result, GO researchers have been using different models, measures and perspectives of GOs. All this diversity in research has caused a great amount of confusion in literature on GOs. The concept of GOs cannot be researched properly if there is no consensus about what GOs are and how they should be measured and assessed. Button et al. (1996) stress the importance of discovering the true nature of goal orientations in order for future research regarding GOs in organisations to be more accurate and reliable than past research has been. This suggestion should be taken into serious consideration. It is hoped that the current study will improve knowledge on the nature of GOs in order for future research to come a step closer towards being able to provide organisations with accurate information regarding GOs. The next chapter focuses on the causes, consequences, and correlates of GOs.

Chapter 5: Variables related to Goal Orientations

5.0. Introduction

This chapter focuses on the variables related to GOs. Past GO studies have examined the relationships between GOs and a number of variables (e.g. culture, gender, personality, cognitive strategies). However, since the current study is being carried out from an organisational psychology perspective only those variables which are considered to be especially relevant to organisations will be discussed. These include perceptions of ability and self-efficacy (since these have been associated with a number of positive outcomes such as increased performance e.g. Pajares, 1997; Bouffard-Bouchard, 1990), motivation, and performance. Past studies investigating the relationships between GOs and perceptions of ability, self-efficacy, motivation, and performance are discussed below. Each section of this chapter includes a description regarding how the current study will further develop knowledge on the relationships between GOs (and GO profiles) and each of these variables.

5.1. Goal Orientations and Perceptions of Ability

Early GO theorists included perceptions of ability in their research on GOs. Dweck (1986) proposed that GO adoption resulted from the type of 'theory of intelligence' adopted by individuals. Individuals holding an incremental theory of intelligence adopted a mastery GO whilst individuals holding an entity theory of ability adopted a performance GO. However, according to Dweck (1986) the type of behavioural response resulting from the two GOs depended on the perceptions of ability held by the individual. She proposed that if an individual adopted a mastery GO he/she would exhibit mastery-oriented behavioural responses. However, the behavioural responses exhibited by individuals adopting a performance GO depended on the perceptions of ability of the individual. According to Dweck (1986) if a person having a performance GO had high perceptions of ability they

would still exhibit mastery-oriented behavioural responses. In contrast, if a person having a performance GO had low perceptions of ability they would exhibit a helpless behaviour pattern. Nicholls (1984) also suggested that when individuals are ego-involved (performance GO), low perceptions of ability produce lower levels of performance whilst high perceptions of ability produce higher levels of performance. In addition, he too (like Dweck, 1986) maintained that perceptions of ability do not influence the performance of individuals who are task-involved (mastery GO).

The GO research described above positions perceptions of ability as moderating the relationship between GOs and performance. A number of studies on GOs adopted this perspective and tested for moderation effects (e.g. Kaplan & Midgley, 1997; Miller, et al., 1993) of perceptions of ability between GOs and performance. These studies found little or no support for this moderation effect.

A number of researchers (e.g. Greene & Miller, 1996) noted that in literature on GOs the terms 'perceptions of ability/competence', 'self-concept of ability' and 'self-efficacy' were used interchangeably. In fact, Anderman and Midgley (1997:271) state that "Expectancy beliefs have been conceptualized in different ways, including expectancies for success, self-efficacy, and self-perceptions of competence." Since studies have used the terms 'perceptions of ability/competence' and 'self-efficacy' interchangeably it was decided that GO studies focusing on perceptions of competence and self-efficacy should be considered together. The next section focuses on studies investigating the relationship between GOs and self-efficacy/perceptions of competence.

5.2. Goal Orientations and Self-Efficacy

Albert Bandura (1986) is well-known for his theory of personality referred to as the 'Social-Cognitive Theory'. According to this theory individuals are active beings and not

just passive respondents to their environment. Bandura (1986) describes how individuals actively monitor their behaviour, assess their progress, and learn by means of observation (Larsen & Buss, 2006). One of Bandura's (1986) key concepts in the Social Cognitive Theory is that of self-efficacy. Self-efficacy may be defined as "an individuals' beliefs about their capabilities to produce designated levels of performance" (Judge et al., 2007:107). Thus, according to Bandura's Social Cognitive Theory, if a person believes that he/she can perform a certain task, then he/she has high self-efficacy beliefs on that task.

Some studies have examined self-efficacy as being an antecedent, consequence, correlate or mediator of GOs. Studies investigating self-efficacy as antecedents of GOs will be described first. Subsequently, an outline of studies examining self-efficacy as a mediator between GOs and performance will be provided. Following this, a description of studies assessing self-efficacy as a consequence of GOs will be presented. Prior to discussing the studies examining the correlations between GOs and self-efficacy, a subsection will be devoted to discussing the research designs and analytical techniques used in the studies inferring causality between GOs and self-efficacy. Following this a description of the studies examining the correlations between GOs and self-efficacy will be provided. Finally, a discussion regarding how this study aims to advance knowledge with respect to the relationship between self-efficacy and GOs will be presented.

5.2.1. Self-efficacy as an Antecedent of Goal Orientations

A number of studies have focused on determining whether self-efficacy is an antecedent of GOs (e.g. Payne et al., 2007; Liem et al., 2008; Lau et al., 2008,). The results of these research studies are consistent with respect to mastery GOs but not completely so with respect to performance GOs. Elliot and Church (1997), Liem et al. (2008), and Payne et al. (2007) used two- or threefactor models of GOs. They found self-efficacy to positively predict a mastery GO. The study by Lau et al. (2008) used a four-factor model of GOs. They found self-efficacy to significantly positively predict a mastery-approach GO but not a mastery-avoidance GO. As discussed in Chapter 3 Section 3.2. measures of the three-factor model of GOs tend to use mastery-approach GO items rather than mastery-avoidance GO items (e.g. Liem et al., 2008; Alkharusi, 2008) for the mastery GO scale. Therefore, the results of the studies described are consistent in suggesting that self-efficacy positively predicts a masteryapproach GO. Since only one study (that by Lau et al., 2008) used a four-factor model of GOs, further investigation regarding whether self-efficacy is a predictor of a masteryavoidance GO is required.

With regards to self-efficacy and performance GOs, the studies by Lau et al. (2008), Morris and Kavussanu (2008), Liem et al. (2008) and Elliot and Church (1997) found self-efficacy to positively predict a performance-approach GO. However, Payne et al. (2007) did not find a significant relationship between task-specific self-efficacy and a performanceapproach GO in their meta-analysis. Payne et al. (2007), Liem et al. (2008), Elliot and Church (1997), and Lau et al. (2008) found self-efficacy to negatively predict a performance-avoidance GO. However, Morris and Kavussanu (2008) did not find a significant relationship. Since the results for the relationships between self-efficacy and performance-approach and performance-avoidance GOs are inconsistent, further research is required in order to better understand these relationships. Moreover, since only one study investigated the relationships between self-efficacy and a masteryavoidance GO further investigation of this is also required.

5.2.2. Self-efficacy as a Mediator of Goal Orientations and Performance

Studies examining self-efficacy as a mediator between GOs and performance may be divided into two main categories. These are:

Category 1:

Goal Orientations	\rightarrow	Self-efficacy	\rightarrow	Performance	
Category 2:					
Goal Orientations	\rightarrow	Self-efficacy	\rightarrow	Self-set Goals \rightarrow	Performance

Seven studies falling into Category 1 were found from the review of literature carried out for this study. Please refer to Table 5.1. for the results of these studies.

	Goal	Effect of Goal	Effect of Self-	
Authors of Study	Orientation	Orientations on Self-	Efficacy on	
	model used	efficacy	Performance	
Ford at al. (1998)	Two factor	MGO: Positive	Positivo	
Ford et al. (1998)	TWO-TACIOT	PGO: Negative	POSITIVE	
Bandalos et al. (2002)	Two factor	MGO: Positive	Positivo	
Bandalos et al. (2003)	TWO-TACIOT	PGO: Positive	POSITIVE	
Kazlowski at al. (2001)	Two-factor	MGO: Positive	Positivo	
		PGO: Not Tested	FUSILIVE	
South of al (2004)*	Two-factor	MGO: Positive	Positivo	
		PGO: Not Tested	FUSILIVE	
		MGO: Positive		
VandeWalle et al. (2001)	Three-factor	PAP: Not Significant	Positive	
		PAV: Negative		
		MGO: Positive		
Tanaka (2007)	Three-factor	PAP: Positive	Positive	
		PAV: Negative		
		MAP: Not Significant		
Crimpon et al. (2000)	Four factor	MAV: Not Significant	Not Significant	
Crippen et al. (2009)	FOUT-TACLOF	PAP: Not Tested	Not Significant	
		PAV: Not Tested		

Table 5.1. Results of studies in Category 1

Key: *Effect of state GOs assessed in this study; MGO=Mastery Goal Orientation; PGO=Performance Goal Orientation; MAP=Mastery-Approach Goal Orientation; MAV=Mastery-Avoidance Goal Orientation; PAP=Performance-Approach Goal Orientation; PAV=Performance-Avoidance Goal Orientation.

In all the studies using a two- or three-factor model summarised in the table above, a mastery GO was found to positively influence self-efficacy. Similarly, self-efficacy was found to positively influence performance. As mentioned previously, studies using the two- and three-factor models tend to use mastery-approach items in their measures as opposed to mastery-avoidance items. Consequently, the mastery GO results of the studies using two- and three-factor models would be expected to be similar to the mastery-approach GO results for studies using a 2x2 model of GOs. However, as indicated in the table above, the study by Crippen et al. (2009) did not find a significant relationship between self-efficacy and a mastery-approach GO. The only study using a 2x2 model of GOs (that by Crippen et al., 2009), found a non-significant relationship between a MAV GO and self-efficacy.

With respect to performance GOs, only two studies investigated the relationship between a performance-approach GO and self-efficacy and they obtained very inconsistent results. Vandewalle et al. (2001) found a non-significant relationship whilst Tanaka (2007) found a significant positive relationship between a performance-approach GO and self-efficacy. With respect to the performance-avoidance GO, it seems as though this negatively influences self-efficacy.

Four of the studies reviewed fall into Category 2. All four studies used a two-factor model of GOs. Please refer to Table 5.2. for the results of studies in Category 2.

Table 5.2.	Results	of	studies	in	Category	/ 2	
------------	---------	----	---------	----	----------	-----	--

Authors of Study	Effect of Goal Orientations on Self- efficacy	Effect of Self- efficacy on Self- set Goals	Effect of Self-Set goals on Performance
Phillips & Gully (1997)	MGO: Positive PGO: Negative	Positive	Positive
Diefendorff (2004)	MGO: Positive PGO: Negative ¹	Positive	Positive ¹
Vrugt et al. (2002)	MGO: Positive ² PGO: Not Tested	Positive	Positive
Breland & Donovan (2005)*	MGO: Positive PGO: Negative ¹	Positive	Positive ³

Key: * Effect of state GOs assessed; ¹ For 1 out of 3 data sets; ² For 1 out of 2 data sets; ³ For 2 out of 3 data sets; MGO=Mastery Goal Orientation; PGO=Performance Goal Orientation

In addition to the results presented in the table above, Phillips and Gully (1997) and Breland and Donovan (2005) also found a direct significant positive relationship between self-efficacy and performance. The results from Table 5.2. clearly indicate that a mastery GO positively influences self-efficacy. Additionally, self-efficacy was found to positively influence self-set goals which, in turn, positively influenced performance.

If the two categories of studies are combined, it becomes evident that a mastery GO positively influences self-efficacy. The results are not entirely consistent for performance-approach and performance-avoidance GOs across studies. Most of the studies using a two-factor model of GOs, found the performance GO to negatively influence self-efficacy. However, when a more differentiated model was used, the performance-approach GO was found to positively influence or not relate to self-efficacy. Since only one of the studies used the 2x2 model of GOs and because the results for the performance-approach GO are not entirely consistent further investigation of the relationships between GOs and self-efficacy is required. As mentioned in the introduction to this section, a discussion regarding the causality inferred in the studies described above will be provided in Section

5.2.4. However, first the results of studies investigating self-efficacy as a consequence of GOs are discussed.

5.2.3. Self-efficacy as a Consequence of Goal Orientations

Three of the studies reviewed (Elliot & Sheldon, 1997; Wolters et al., 1996; Linnenbrink, 2005) investigated self-efficacy as a consequence of GOs. Wolters et al. (1996) and Linnenbrink (2005) found self-efficacy to be a consequence of a mastery GO. Elliot & Sheldon (1997) did not examine the relationship between a mastery-approach GO and self-efficacy. However, they found a significant negative relationship between a mastery-avoidance GO and self-efficacy. The study by Wolters et al. (1996) only measured a mastery GO whilst Linnenbrink (2005) measured mastery and performance-approach GOs. Linnenbrink (2005) did not find any significant relationship between a performance-approach GO and self-efficacy. Elliot & Sheldon (1997) did not examine the relationship between a performance-approach GO and self-efficacy. The study by Holters et al. (1996) only measured a mastery GO whilst Linnenbrink (2005) measured mastery and performance-approach GOs. Linnenbrink (2005) did not find any significant relationship between a performance-approach GO and self-efficacy. Elliot & Sheldon (1997) did not examine the relationship between a performance-approach GO and self-efficacy. The correctness of inferring causality in these three studies (as well as in the studies described in sub-sections 5.2.1. and 5.2.2.) is discussed in the next sub-section.

5.2.4. Are the designs and data analytical techniques used in the studies above enough to infer causality between Self-Efficacy and Goal Orientations?

In order to infer causality, multivariate data analytical techniques such as structural equation modelling are used. These are normally combined with the use of experimental designs in order to provide support for the hypothesised direction of causality being inferred. Experimental designs are necessary in order to infer causality due to the fact that the statistical techniques alone cannot establish the direction of causality (Tabachnik & Fidell, 1996:715).

Referring to self-efficacy as an 'antecedent', 'mediator', or 'consequence' of GOs means inferring causality. However, most of the studies described above that infer causality did not use experimental designs (e.g. Liem et al., 2008; Lau et al., 2008; VandeWalle et al., 2001; Bandalos et al, 2003; Crippen et al., 2009; Vrugt et al., 2002). As a result, their inferences about the direction of causality are still quite subjective. Although some of the studies described above measured self-efficacy on multiple occasions, it is still uncertain whether the GOs influence self-efficacy or vice versa. This matter is compounded by the fact that it is still not certain whether GOs are stable or not. If GOs are found to be stable, then it would be highly unlikely that self-efficacy is an antecedent of GOs since self-efficacy is known to be changeable (Bandura, 1982). If on the other hand, GOs are found to change over time (or across tasks) then there is a greater possibility of GOs being influenced by self-efficacy. Since the stability and task-specificity of GOs are being tested in this study, the causality of the relationship between GOs and self-efficacy will not be addressed. However, the direction and magnitude of *correlation* between self-efficacy and GOs (and GO profiles) will be examined.

The studies described above that used experimental designs (e.g. Breland & Donovan, 2005 and Seijts et al., 2004) used the two-factor model of GOs. As mentioned a number of times throughout this review of literature, a two-factor model is not ideal for examining GOs since it is not differentiated enough. This is especially the case for the performance GO which was found to produce very inconsistent results when it was not divided into approach and avoidance orientations. Moreover, both Breland and Donovan (2005) and Seijts et al. (2004) examined the relationships between **state** GOs and self-efficacy in their studies. These difficulties make it extremely tricky to combine the results of studies carried out so far in order to determine the direction of causality for the relationships between GOs and self-efficacy. Although an experimental design is being used in this study it was thought best to examine the correlations between GOs and self-efficacy for

the reasons described in the previous paragraph (regarding the stability and task-specificity of GOs).

Before providing an overall summary of the results of all the studies described in the subsections above, a short description of the studies that examined the correlations between GOs and self-efficacy is provided next.

5.2.5. Self-efficacy as a Correlate of Goal Orientations

A number of studies did not attempt to infer causality with regards to the relationships between self-efficacy and GOs but rather they assessed the correlations between them. The results of these studies are presented in Table 5.3. As discussed in the previous section, the majority of studies attempting to infer causality did so by using statistical techniques such as structural equation modelling but not experimental designs. Consequently, the main difference between studies investigating the direction of causality and those that do not are the types of statistical techniques used. A number of studies (e.g. Liem et al., 2008; and Lau et al., 2008) examined self-efficacy as an antecedent, mediator or consequence of GOs (hence attempting to determine the direction of causality) by using statistical techniques such as structural equation modelling but they also provided correlational analyses results as part of their descriptive statistics. Consequently, the correlational analyses results of these studies will also be included in Table 5.3.

Table 5.3. Results of studies examining the correlations between self-efficacy and Goal Orientations

Authors of Study	Goal Orientation Model Used	Correlations between GOs and Self-Efficacy
Ford et al. (1998)	Two-factor	MGO: Positive PGO: Negative
Diefendorff (2004)	Two-factor	MGO: Positive PGO: Non-significant
Breland & Donovan (2005)	Two-factor	MGO: Non-significant for Trait, Positive for State PGO: Non-significant
Chen et al. (2000)	Two-factor	MGO: Positive PGO: Positive ⁶ ; Non-significant ⁷
Sins et al. (2008)	Four-factor (but only 2 factors measured)	MAP: Positive PAV: Non-significant
Wolters et al. (1996)	Only MGO measured	MAP: Positive
Liem et al. (2008) Three-factor		MAP: Positive PAP: Positive PAV: Negative
Bong (2001) Three-factor		MGO: Positive PAP: Positive PAV: Positive ⁸ ; Non-significant ⁹
VandeWalle et al. (2001) Three-factor		MGO: Positive PAP: Non-significant PAV: Negative
Elliot & Church (1997) Three-factor		MAP: Positive PAP: Positive PAV: Negative
Tanaka (2007)	Three-factor	MAP: Positive PAP: Positive PAV: Non-significant
Linnenbrink (2005) Three-factor		MAP: Positive PAP: Positive PAV: Not tested

Table 5.3. continued on next page.

Table 5.3. continued

Bong (2009)	Four-factor	MAP: Positive
		MAV: Positive ¹ , Non-significant ² , Negative ³
		PAP: Positive
		PAV: Positive ¹ , Non-significant ⁴ , Negative ⁵
Radosevich et al. (2007)	Four-Factor	MAP: Positive
		MAV: Negative
		PAP: Positive
		PAV: Negative
Lau et al. (2008)	Four-factor	MAP: Positive
		MAV: Negative
		PAP: Positive
		PAV: Negative
Morris & Kavussanu (2008)	Four-factor	MAP: Positive
		MAV: Non-significant
		PAP: Positive
		PAV: Non-significant

Key: ¹ For lower elementary; ² For Middle Elementary and Middle School; ³ For Upper Elementary; ⁴ For Middle and Upper Elementary; ⁵ For Middle School; ⁶ Prior to initial performance; ⁷ Following initial performance; ⁸ For English, Maths and Science in Middle school and for Science in High School; ⁹ For Korean, English & Maths in High School and Korean & Science in Middle School; MGO=Mastery Goal Orientation; PGO=Performance Goal Orientation; MAP=Mastery-Approach Goal Orientation; MAV=Mastery-Avoidance Goal Orientation; PAP=Performance-Approach Goal Orientation; PAV=Performance-Avoidance Goal Orientation.

The results presented in Table 5.3. indicate that a mastery GO (for those studies using a two-factor model) and a mastery-approach GO (for those studies using the three- or four-factor model) are significantly positively related to self-efficacy with the exception of Breland and Donovan (2005) for the trait mastery GO. The results for a mastery-avoidance GO are not clear since two studies (Radosevich et al., 2007; and Lau et al., 2008) found significant negative correlations between a mastery-avoidance GO and self-efficacy whilst Bong (2009) found positive, negative and non-significant relationships depending on the age of participants. Morris and Kavussanu (2008) found a non-significant relationship between a mastery-avoidance GO and self-efficacy.

Some of the studies using the two-factor model of GOs found significant negative relationships between a performance GO and self-efficacy whilst others found non-

significant relationships. Moreover, Chen et al. (2001) found a significant positive relationship between a performance GO and self-efficacy prior to initial performance and a non-significant relationship following initial performance. All the studies using the threeor four-factor models found significant positive relationships between a performanceapproach GO and self-efficacy with the exception of VandeWalle et al. (2001). The results were not so consistent for a performance-avoidance GO since some studies found nonsignificant (e.g. Sins et al., 2008; Morris & Kavussanu, 2008), significant negative (e.g. Radosevich et al., 2007; VandeWalle et al., 2001) and significant positive (e.g. Bong, 2001, for English, Maths and Science in Middle school and Science in High School) relationships between a performance-avoidance GO and self-efficacy. The inconsistent results obtained for the avoidance GOs (for the studies presented in Table 5.3.) reflect the inconsistencies found in studies in the other sub-sections.

Since the studies examining the relationships between GOs and self-efficacy were discussed in different sub-sections (according to their hypothesised relationships) an overview of the results of all the studies is provided below. This is followed by the Research Questions and Hypotheses proposed for the current study with regards to the relationships between GOs and self-efficacy.

5.2.6. Self-efficacy, Perceptions of Ability and Goal Orientations

After summing up the results of the studies described above, it seems as though the majority of studies found significant positive relationships between a mastery-approach (or mastery) GO and self-efficacy. However, two studies (those by Crippen et al, 2009, and Breland & Donovan, 2005) found non-significant relationships. Therefore, although there is a strong indication that a mastery/mastery-approach GO is significantly positively related to self-efficacy, further investigation into why two of the studies did not obtain this result would be useful. Possible explanations for the differences in results include

different measures of self-efficacy being used (e.g. Crippen et al., 2009, developed their own self-efficacy scale which seems to be quite different from the self-efficacy measures used in other studies) or general GOs being used to measure self-efficacy on tasks. With respect to a mastery-avoidance GO, the results of the studies described above are not entirely consistent since negative, non-significant and positive relationships were found between a mastery-avoidance GO and self-efficacy.

Studies using a two-factor model of GOs found self-efficacy to be predominantly negatively related to a performance GO. However, two studies found a significant positive relationship (Bandalos et al., 2003; and Chen et al., 2001, prior to initial performance) and some other studies (e.g. Breland & Donovan, 2005; Diefendorff, 2004) found no significant relationship between self-efficacy and a performance GO. When the three- and four-factor models were used, a performance-approach GO tended to be significantly positively related to self-efficacy. However, in a few cases no significant relationship was found. The results for a performance-avoidance GO with self-efficacy tended to be quite inconsistent with studies finding significant negative, positive and non-significant relationships with self-efficacy. After taking into consideration all the results of studies described in Section 5.2. the following Hypotheses and Research Questions were proposed for the current study:

Research Question 6a: Do the different task-specific GO profiles score significantly differently on self-efficacy?

Research Question 6b: How do task-specific mastery-avoidance goal orientations correlate with self-efficacy? This Research Question will only be answered if a 2x2 model of goal orientations is chosen as an appropriate model.

Research Question 6c: How do task-specific performance-avoidance goal orientations correlate with self-efficacy?
Hypothesis 2a: Task-specific mastery-approach goal orientations are expected to be significantly positively correlated with self-efficacy.

Hypothesis 2b: Task-specific performance-approach goal orientations are expected to be significantly positively correlated with self-efficacy.

In the current study, it was decided that self-efficacy should be assessed rather than perceived competence since "A perceived competency could be defined as generalized self-efficacy" (Gist, 1987: 479). Since task-specificity is important in the current study (as described in Chapter 4) it was thought to be more appropriate to measure self-efficacy of participants rather than perceived competence.

5.3. Goal Orientations and Motivation

Motivation may be seen as made up of three major components: direction, effort, and persistence (Arnold et al., 2005). *Direction* refers to what the person is trying to achieve. *Effort* refers to the amount of energy an individual is putting into achieving what he/she wants to achieve. *Persistence* may be defined as the length of time that a person continues trying to achieve what they want to achieve (Arnold, et al., 2005).

A number of studies have examined the relationships between GOs, effort and persistence. Early GO research carried out by Dweck and her colleagues (e.g. Dweck & Reppucci, 1973; Diener & Dweck, 1978) focused on determining why some children persist in the face of failure whilst others do not (even though they have the same level of ability). Their research indicated that the type of GO that children adopted led them to adopt one of two behavioural responses: a mastery-oriented response or a helpless response. According to Dweck and Reppucci (1973), children exhibiting a mastery-oriented response persisted in the face of failure and attributed failure to lack of effort. On the other hand, children who exhibited a helpless response pattern did not persist in

the face of failure and attributed failure to lack of ability. From their research, it became evident that children adopting a mastery GO tended to exhibit a mastery-oriented behavioural response whilst children adopting a performance GO tended to exhibit a helpless behavioural response. However, they found that it was only those children with low perceived ability who exhibited a helpless response when adopting a performance GO. Children with high perceived ability who adopted a performance GO still tended to exhibit a mastery-oriented behavioural response.

As GO research evolved a few authors revisited the ideas suggested by Dweck and Reppucci (1973) in order to investigate whether the GOs that individuals adopt are related to their effort and persistence on tasks. Out of the studies reviewed, three investigated the relationships between both effort and persistence with GOs (Wolters, 2004; Agbuga & Xiang, 2008; and Elliot et al., 1999) whilst a fourth study investigated the relationship between effort and GOs (Phan, 2009). As a result of his analyses, Wolters (2004) found that a mastery GO was positively related to effort and persistence whilst a performanceavoidance GO was negatively related. A performance-approach GO was found to be significantly positively related to effort but not significantly related to persistence. Agbuga and Xiang (2008) found positive correlations between all three GOs investigated (mastery, performance-approach, and performance-avoidance GOs) and effort and persistence. The results of their regression analysis indicated positive relationships between a mastery GO and persistence and effort. There was also a positive relationship between a performance-approach GO and persistence and effort. However, no relationship was found between a performance-avoidance GO and persistence and effort. Elliot et al. (1999), found significant positive correlations between mastery as well as performanceapproach GOs with effort and persistence. However, no significant relationships were found between a performance-avoidance GO and effort and persistence. The results of their study also indicated that persistence and effort mediated the relationships between a mastery GO and exam performance as well as between a performance-approach GO

and exam performance. However, no mediation effects were found for the relationships between a performance-avoidance GO and exam performance.

Phan (2009) used structural equation modelling in his analysis. His results indicated that all three GOs were positive predictors of effort. Although the four studies described above have rather large sample sizes (N= 525, 229, 275 and 179 for the studies by Wolters, 2004; Agbuga & Xiang, 2008; Phan, 2009; and Elliot et al., 1999, respectively), and used multilevel modelling and multiple regression analysis in order to conduct their data analysis, they did not use experimental designs. All four studies conducted surveys. Therefore, again, one cannot be certain about the direction of causality.

The results of the studies discussed above are quite consistent with respect to masteryapproach and performance-approach GOs. However, this is not the case for the relationships between a performance-avoidance GO and effort. Since these are quite inconsistent and because there is not enough research examining the relationship between a mastery-avoidance GO and effort it is necessary to further investigate the relationships between GOs and motivational factors. Due to the nature of the tasks which will be used to assess GOs in this study it is not possible to examine the relationship between GOs and persistence or direction. However, the relationships between GOs and effort will be investigated in the current study. Since the tasks used in this study are aptitude tests (described in further detail in Chapter 7) the mental effort of participants on these tasks will be measured. Once again, since the multiple goal perspective is being used in this study, the relationship between GOs and effort will be assessed from both the profile and non-profile perspectives.

Although this study makes use of an experimental design no attempts to make inferences regarding the direction of causality (for the relationships between GOs and effort) will be made. This was thought to be appropriate because in order to design an experiment to

test the direction of causality one should have a reasonable theory regarding the direction of causality. For example if one wanted to test whether the sun rising causes roosters to crow they would have a good theory supporting this direction of causality (based on what is known about the sun rising and roosters crowing) since it is extremely unlikely that the direction of causality is in reverse. With respect to mental effort and GOs the direction of causality may be hypothesised to go in either direction (mental effort being an antecedent OR consequence of GOs). Since little is known about the relationships between GOs and mental effort (and the direction of causality may be hypothesised to go either way) it was thought best to assess the correlations between these variables before attempting to make inferences regarding the direction of causality. The following Research Questions were therefore proposed:

Research Question 7a: Do the different task-specific GO profiles score significantly differently on mental effort?

Research Question 7b: How do the different task-specific goal orientations correlate with mental effort (if at all)?

5.4. Goal Orientations and Performance

Beyond doubt the relationships between GOs and performance has been one of the most researched relationships in GO literature. From an educational, sports, as well as organisational perspective, performance is an outcome of extreme interest. If one or more GOs are found to influence performance positively, then it might be possible to induce these beneficial GOs (in all three settings) in order to increase performance. Research on the relationships between GOs has been carried out in various ways. In order to summarise the results of past research in this area, studies investigating the relationship between GOs and performance will be discussed in three subsections. The first subsection will describe the results of research investigating GOs as direct predictors

of performance as well as studies investigating the direct correlations between GOs and performance. In the second subsection, the results of studies investigating the indirect relationship between GOs and performance will be described. The third subsection will focus on those studies investigating the relationships between GOs and performance from a multiple goals perspective.

5.4.1. Goal Orientations as direct predictors of Performance

Since a large number of studies have investigated GOs as direct predictors of performance as well as the direct correlations between GOs and performance it was thought best to present the results in table form. The former will be presented first followed by the latter.

Different studies assessing GOs as direct predictors of performance used the two-, three-, and four-factor models of GOs. Consequently, three summary tables are presented below, one each for studies using the two-, three-, and four-factor models, respectively. Following this, the results of studies which provided the correlational analyses results for the relationships between GOs and performance are provided.

Table	5.4.	Summary	of	Results	obtained	by	studies	investigating	the	relationships
between the 2-factor model of Goal Orientations and performance.										

Authors of Study	Relationship between MGO and Performance	Relationship between PGO and Performance	
Schraw et al. (1995)	Positive	Non-significant	
Bell & Kozlowski (2002)	Positive	Negative	
Harackiewicz et al. (1997)	Non-significant	Positive	
Harackiewicz et al. (2002)	Non-significant ¹ ; Negative ²	Positive	
Harackiewicz et al. (2000)	Non-significant	Positive	

Key: MGO = Mastery Goal Orientation; PGO = Performance Goal Orientation; ¹For Final Grade; ²For Semester GPA.

The studies by Harackiewicz et al. conducted in 2000 and 2002 focused on assessing the relationships between multiple GOs and their outcomes. However, since they assessed the outcomes of each GO separately rather than as GO profiles they have been included in the table above as opposed to the section on multiple GOs.

Table 5.5. Summary of Results obtained by studies investigating the relationshipsbetween the 3-factor model of Goal Orientations and Performance.

Authors of Study	MGO	РАР	PAV
Payne et al. (2007)	Non-significant ¹ ; Positive ²	Non-significant ³	Non-significant ⁴ ; Negative⁵; Not Tested ⁶
Elliot & Church (1997)	Non-significant	Positive	Negative
Elliot et al. (1999)	Non-significant	Positive	Direct relationship not tested
Church et al. (2001)	Positive	Positive	Negative
Elliot et al. (2005)	Positive ⁷	Positive ⁷	Negative ⁷
Durik et al. (2009)	Non-significant	Positive	Negative

Key: MGO = Mastery Goal Orientation; PAP = Performance-approach Goal Orientation; PAV = Performanceavoidance Goal Orientation; ¹ For Task Performance and Job Performance; ²For academic performance; ³For academic, task and job performance; ⁴For academic performance; ⁵For task performance; ⁶For job performance.

Table 5.6. Summary of Results obtained by studies investigating the relationships between the 4-factor model of Goal Orientations and Performance.

Authors of Study	ΜΑΡ	MAV	ΡΑΡ	PAV
Elliot & McGregor (2001)	Non-significant	Non-significant	Positive	Negative
Elliot & Murayama (2008)	Not Tested	Not Tested	Positive	Negative

Key: MAP = Mastery Approach Goal Orientation; MAV = Mastery Avoidance Goal Orientation; PAP = Performance-approach Goal Orientation; PAV = Performance-avoidance Goal Orientation.

The studies presented in Tables 5.4. to 5.6. assess whether GOs are predictors of performance by measuring GOs prior to measuring performance (e.g. Elliot & McGregor, 2001, and Elliot & Murayama, 2008, measure GOs 1 week prior to task performance) and using statistical techniques such as regression analyses and structural equation modelling. In some studies a mastery/mastery-approach GO was found to significantly positively predict performance whilst in others it did not significantly predict performance.

Moreover, one study (that by Harackiewicz et al., 2002) found a mastery GO to negatively predict performance for Semester GPA. These results indicate that further investigation into the relationship between a mastery-approach GO and performance is definitely required. The relationship between a mastery-avoidance GO and performance was only tested in one study, that by Elliot and McGregor (2001), who found a non-significant relationship.

Three out of the five studies reviewed that used the 2-factor model of GOs found a performance GO to positively predict performance. However one study (Bell & Kozlowski, 2002) found a performance GO to negatively predict performance and the other (Schraw et al., 1995) found no significant relationship. When the three- and four-factor models were used a performance-approach GO was found to positively predict performance in all studies except one (the meta-analysis by Payne et al., 2007). On the other hand, a performance-avoidance GO was found to negatively predict performance in all studies except that by Payne et al. (2007) for academic performance where a non-significant relationship was found.

Although the studies above state that GOs are predictors of performance it is difficult to determine the direction of causality. If GOs were stable, it would seem plausible to conclude that GOs influence performance rather than vice versa. However, since the stability of GOs is still being questioned, determining the causality of the relationships between GOs and performance is quite tricky. A number of studies investigated the correlations between GOs and performance. The results of these studies are presented in Table 5.7. Some of the studies investigating GOs as direct predictors of performance (presented in Tables 5.4. to 5.6.) examined the correlations between GOs and performance are also presented in Table 5.7.

97

Table 5.7. Summary of Results obtained by studies investigating the correlationsbetween Goal Orientations and Performance.

Authors of Study	Goal Orientation Model Used	Correlations between GOs and Performance
Bell & Kozlowski (2002)	Two-factor	MGO: Non-significant PGO: Non-significant
Harackiewicz et al. (1997)	Two -factor	MGO: Non-significant PGO: Positive
Harackiewicz et al. (2000)	Two -factor	MGO: Non-significant PGO: Positive
Harackiewicz et al. (2002)	Two -factor	MGO: Non-significant PGO: Positive ¹ ; Non-significant ²
Jagacinski et al. (2001)	Two-factor	MGO: Non-significant PGO: Non-significant
Elliot et al. (1999)	Three-factor	MGO: Positive ³ ; Non-significant ⁴ PAP: Positive ³ ; Non-significant ⁴ PAV: Negative
Church et al. (2001)	Three-factor	MGO: Non-significant ⁵ ; Positive ⁶ PAP: Non-significant ⁵ ; Positive ⁶ PAV: Negative
Elliot & McGregor (2001)	Four-factor	MAP: Non-significant MAV: Non-significant PAP: Non-significant ⁷ ; Positive ⁸ PAV: Non-significant ⁸ ; Negative ⁷
Yeo et al. (2009)	Four-factor	MAP: Positive ⁹ , Non-significant ¹⁰ MAV:Non-significant ¹⁰ PAP: Positive ⁹ ; Non-significant ¹⁰ PAV:Non-significant ⁹ ; Negative ¹⁰
Bong (2009)	Four-factor	MAP: Positive ¹¹ , Non-significant ¹² MAV: Negative ¹³ , Non-significant ¹⁴ PAP: Positive ¹⁵ , Non-significant ¹⁶ PAV: Negative ¹⁷ , Non-significant ¹⁸

Key: MGO = Mastery Goal Orientation; PGO = Performance Goal Orientation; MAP = Mastery Approach Goal Orientation; MAV = Mastery Avoidance Goal Orientation; PAP = Performance-approach Goal Orientation; PAV = Performance-avoidance Goal Orientation; ¹For final grade; ²For semester GPA, Psychology GPA and Subsequent GPA; ³In study 1; ⁴In study 2; ⁵For SAT Scores; ⁶For Graded Performance; ⁷For Study 3; ⁸For Study 2; ⁹On Air Traffic Control Task; ¹⁰On Exam; ¹¹For Middle Elementary, Upper Elementary and Middle School students; ¹²For Lower Elementary; ¹³For Middle Elementary; ¹⁴For Lower Elementary, Middle School and Upper Elementary; ¹⁵For Upper Elementary and Middle School; ¹⁶For Lower and Middle Elementary; ¹⁷For Middle and Upper Elementary and Middle School; ¹⁸For Lower Elementary.

Table 5.7. indicates that the majority of studies found a mastery/mastery-approach GO not to be related to performance. However, a number of studies (e.g. Yeo et al., 2009 and Bong, 2009) found a significant positive relationship between a mastery/mastery-approach GO and performance. There were no consistent distinguishing differences between those studies that found significant positive and non-significant relationships. For example Elliot and McGregor (2001) found a *non-significant* relationship between a mastery-approach GO and exam performance on an introductory psychology course. On the other hand Elliot et al. (1999) found a *significant positive* relationship between a mastery GO and exam performance on an introductory psychology course. Consequently, it is unclear why these differences were found. Two out of the three studies using the four-factor model of GOs found non-significant relationships between a mastery-avoidance GO and performance. However, Bong (2009) also found a significant negative relationship for participants in Middle Elementary School.

With respect to the relationships between performance GOs and performance, those studies using a 2-factor model found significant positive and non-significant relationships. Studies using the three- and four-factor models of GOs found both positive and non-significant correlations between a performance-approach GO and performance. The relationships between a performance-avoidance GO and performance were also quite inconsistent with some studies finding significant negative relationships and others finding non-significant relationships.

Overall, the results for the relationships between GOs and performance are quite inconsistent with different studies obtaining different results. Moreover, a number of studies (e.g. Elliot et al., 1999; Church et al., 2001) found inconsistent results for the relationships between GOs and performance across tasks, studies (e.g. Study 1 and Study 2), or for participants of different ages. These inconsistencies in research indicate that

further investigation of the relationships between GOs and performance is definitely required.

As was mentioned earlier (Section 5.3.), in order to determine the direction of causality for the relationships between GOs and performance one should have a sufficiently sound theory to test. At this point in time it is quite difficult to form a theory regarding whether GOs predict performance or vice versa. Although it is more likely that GOs predict performance there is no evidence to show that the opposite is not true. For example, it is possible that past performance on a task influences the types of GOs adopted on that task. Consequently, it was thought that testing the direction of causality in this study would not be appropriate. Therefore, in the current study, the strength and direction of the *correlations* between GOs and performance will be assessed. Based on the results obtained from the studies described above, the following Research Questions were proposed for the current study:

Research Question 8a: How do task-specific mastery-approach goal orientations correlate with performance on tasks (if at all)?

Research Question 8b: How do task-specific mastery-avoidance goal orientations correlate with performance on tasks (if at all)? This Research Question will only be answered if a 2x2 model of goal orientations is chosen as an appropriate model.

Research Question 8c: How do task-specific performance-approach goal orientations correlate with performance on tasks (if at all)?

Research Question 8d: How do task-specific performance- avoidance goal orientations correlate with performance on tasks (if at all)?

5.4.2. Goal Orientations as Indirect Predictors of Performance

As mentioned earlier some researchers focused on examining the direct relationships between GOs and performance whilst others (e.g. Bandalos et al., 2003; VandeWalle et al., 2001; Elliot & Sheldon, 1997) examined how GOs influence performance indirectly e.g. with self-efficacy, effort, and cognitive strategies as mediators. It was decided that the indirect relationships between GOs and performance will not be discussed in the current study. This decision was made as a result of the fact that there are still inconsistencies in research with respect to the direct relationships between GOs and performance and (as will become evident in the next section) not much is known about the relationships between **GO profiles** and performance. Therefore, it was thought necessary to focus on further investigating the direct relationships between GOs and performance as well as examining whether different GO profiles have different relationships with performance prior to investigating the processes by which GOs influence performance (or vice versa).

5.4.3. Multiple Goal Orientations as direct predictors of Performance

Studies using the multiple goals perspective have obtained varied results with respect to the types of GO profiles and their effects on performance. A number of studies investigated the relationships between GO profiles and performance. These are described briefly below. Following this, the necessity to investigate these relationships further is discussed.

Ng (2009) assessed GO profiles by measuring the mastery, performance-approach, and work-avoidance orientations of participants. He obtained three GO profiles from his cluster analysis. **Cluster 1** was named 'performance-focused learners'. This cluster included participants having high a performance-approach GO and low mastery and work avoidance orientations. **Cluster 2**, which was referred to as 'work-avoidant learners', was

characterised by participants having moderate mastery and performance-approach GOs and slightly higher work avoidance orientations. In contrast, Cluster 3, referred to as 'multiple-goal learners' was characterised by participants having high mastery and performance GOs and low work avoidance GOs. His sample consisted of 441 adult learners in distance learning programs at a university in Hong Kong. Their performance on a compulsory academic essay was used as an indicator as performance. As a result of his analysis he found that multiple-goal learners performed significantly better than workavoidant learners. However, there were no significant differences in performance between multiple-goal learners and performance-focused learners. One concern regarding the design of this study is that different essays were used as indicators of performance since students from different courses were recruited for the study. This difference might introduce bias into the study since different essays may have had different levels of difficulty and slightly different requirements. In addition, grading of the essays may have been quite subjective as these are not marked on an all-or-none basis. These factors do not seem to have been taken into account during the study since no mention of them was made. Consequently, these concerns should be taken into consideration when evaluating the results of this study.

Meece and Holt (1993) also carried out a study investigating the relationships between multiple GOs and performance. They used a two-factor model of GOs and their cluster analyses revealed three GO profiles. **Cluster 1**, labelled 'high mastery' was characterised by participants having high mastery and low performance GOs. **Cluster 2**, which was referred to as 'combined mastery-ego', was characterised by participants who had high mastery and high performance GOs. Finally, **Cluster 3**, which was labelled 'low mastery-ego' consisted of those participants who endorsed low mastery and low performance GOs. Their sample consisted of 257 5th and 6th graders whose performance in science was assessed by means of their grade at the end of the year and a self-report measure of strategy use. The results of the analyses indicated that participants in the high mastery

cluster (Cluster 1) performed significantly better than participants in the other two clusters. There were no differences in performance for participants in Clusters 2 and 3. Participants in Clusters 1 and 2 reported greater use of active learning strategies than those in Cluster 3. Moreover, participants in Clusters 2 and 3 reported greater use of superficial engagement (effort minimising strategies) than those in Cluster 1. Consequently, it seems as though the greater use of active learning strategies and lower use of superficial engagement may be contributing to the higher performance of participants in Cluster 1.

Pintrich (2000) used the multiple goals perspective in order to assess the relationships between multiple GO adoption and performance in Maths. A sample of 150 8th and 9th graders participated in his study. Maths grades were used as an indicator of performance. He investigated four categories of GO profiles: High mastery/Low performance; High mastery/High performance; Low mastery/High performance; Low mastery/Low performance; Low mastery/Low performance. The results of his study indicated no significant relationships between GO profiles and performance.

Daniels et al. (2008) found four types of GO profiles as a result of their cluster analyses. The four clusters characterised individuals with high mastery/high performance GOs (referred to as multiple goals); dominant mastery GO; dominant performance GO; and low mastery/low performance GOs (referred to as low motivation). Their sample consisted of 1002 undergraduates enrolled in an Introductory Psychology course. Their final grade on this course was used as an indicator of performance. The analyses revealed that there were no significant differences in the performance of individuals in the multiple goals, dominant mastery, and dominant performance clusters. However, participants in the low motivation cluster performed significantly worse than participants in the other three clusters.

Out of the four studies described above, three of them (those by Meece & Holt, 1993; Pintrich, 2000; and Daniels et al., 2008) used a two-factor model of GOs. The other study (by Ng, 2009) only measured a performance-approach GO. The studies described in Chapter 3 (Section 3.3.) used the 3-factor model (Table 3.2.) and the 2x2 model (Table 3.3.) of GOs in examining GO profiles. These studies also investigated the relationships between GO profiles and performance. A comparison of these studies indicated that for the studies investigating the 3-factor model (that is, the studies by Pastor et al., 2007, and Fortunato & Goldblatt, 2006), participants adopting a 'High Approach, Low Avoidance' profile performed better than participants adopting a 'High Mastery, Low Performance' profile or a 'High Performance-Avoidance' profile (the latter profile was only found in the study by Fortunato & Goldblatt, 2006). Pastor et al. (2007) and Cano and Berben (2009) used the 2x2 model of GOs. The results obtained when using a 2x2 model are not as consistent as those obtained when using the 3-factor model. For example, Cano and Berben (2009) found that participants who scored highly on all four GOs had higher performance than participants in other clusters with the exception of participants in the 'High Mastery-Approach' cluster (participants who adopted high mastery-approach and low mastery-avoidance, performance-approach and performance-avoidance GOs) who obtained similar mean performance scores. On the other hand, Pastor et al. (2007) found that the performance of participants who scored high on all four GOs was average whilst that of participants in the 'High Mastery-Approach' cluster was low. Since further research is definitely required before any conclusions may be drawn regarding the relationships between GO profiles and performance the current study aims to investigate these relationships further. Therefore, the following Research Question was proposed for the current study:

Research Question 9: Do the different task-specific GO profiles score significantly differently on task performance?

5.5. Synopsis

The focus of this chapter was on variables related to GOs that are considered to be important to organisations, namely, self-efficacy, motivation, and performance. As described in the introduction chapter, the next chapter provides an overview of the literature review and a list of the Hypotheses and Research Questions for the current study.

Chapter 6: Summary of Literature and Research Questions

6.0. Introduction

The aim of this chapter is to provide an outline of the previous four chapters (the literature review chapters) and a summary of the Research Questions and Hypotheses which will be addressed in the current study. This will facilitate the understanding of the next chapter in which the research methods used in the current study are described.

6.1. Summary of Literature Review

In Chapter 2 it was described how the concept of GOs, which initially developed in educational psychology, is being applied to organisational research. Throughout this chapter it became evident how the use of this concept in organisational research may benefit organisations.

In Chapter 3 it became clear how there is currently no widely accepted definition of GOs and researchers are using different definitions to guide their research. Different terms (when referring to GOs) are also being used and it is not entirely clear whether these different terms refer to the same concepts or not. The definition and terms to be used in the current study and the reasons for this choice were presented in this Chapter. The problem of dimensionality was also discussed in this Chapter. Following a description of the different models of GOs it was discussed how the 2-factor model was not differentiated enough. There is still not enough evidence indicating whether the 3- or 4-factor model provides the best understanding of GOs. Consequently, a decision was made to investigate this further in the current study and choose the model to be used in this study following the initial analyses. Finally, the multiple goals perspective was discussed. Following a description of the studies investigating this perspective it was described how

the current study will further investigate this in order to promote knowledge on the multiple goals perspective.

In Chapter 4 a discussion of the concerns regarding the stability of GOs led to the conclusion that further research is required in order to establish whether GOs are stable traits or not. A number of studies investigated state GOs. These studies were grouped into three categories according to whether (a) they assessed if trait GOs predict state GOs, (b) they examined the consequences of state GOs or (c) they investigated the interactions between trait and state GOs.

After describing these studies it was concluded that since GOs may exist as traits and states, the interaction effects between trait and state GOs on performance would be examined in this study. Since the existence of trait GOs is also being investigated (by assessing the stability of GOs) the results of the latter will provide a better understanding of the interaction effects. Following this, the issue of task-specificity was discussed. Researchers have used both general and task-specific measures of GOs. However, little research has been carried out in order to investigate whether GOs are general or task-specific. Therefore, the current study will investigate this. The conclusion to this chapter described the importance of investigating the stability and task-specificity of GOs since these greatly influence research designs and the accuracy of research outcomes.

Finally, in Chapter 5 studies investigating the relationships between GOs and self-efficacy, effort, and performance were described. This description made it clear that there is still a lot of inconsistency in research with respect to these relationships. Since it is not entirely clear how these variables relate to GOs the current study will examine these relationships (both in terms of the profile and non-profile perspective).

A number of Research Questions and Hypotheses were proposed for the current study throughout the literature review. These are presented Section 6.2.

6.2. Summary of Hypotheses and Research Questions for the current study

6.2.1. Hypotheses proposed for the current study

Hypothesis 1. Individuals holding a trait mastery-approach goal orientation are expected to perform significantly better when a performance-approach goal orientation is induced as opposed to when a mastery-approach goal orientation is induced.

Hypothesis 2a: Task-specific mastery-approach goal orientations are expected to be significantly positively correlated with self-efficacy.

Hypothesis 2b: Task-specific performance-approach goal orientations are expected to be significantly positively correlated with self-efficacy.

6.2.2. Research Questions proposed for the current study

Research Question 1: Using LCA as a method of clustering goal orientations, how many different types of goal orientation profiles are there and what are the characteristics of each goal orientation profile? Does the 2x2 model significantly improve on the 3-factor model in terms of identifying goal orientation profiles?

Research Question 2a: Do individuals' General goal orientation profiles change over time?

Research Question 2b: Do individuals' General goal orientations change significantly over time?

Research Question 3a: How do trait goal orientation profiles interact with induced mastery-approach and performance-approach goal orientations in order to influence task performance of participants?

Research Question 3b. What effects will the relationships between a trait masteryavoidance goal orientation and induced mastery-approach and induced performanceapproach goal orientations have on the performance of participants? This Research Question will only be answered if a 2x2 model of goal orientations is chosen as an appropriate model.

Research Question 3c. What effects will the relationships between a trait performanceapproach goal orientation and induced mastery-approach and induced performanceapproach goal orientations have on the performance of participants?

Research Question 3d. What effects will the relationships between a trait performanceavoidance goal orientation and induced mastery-approach and induced performanceapproach goal orientations have on the performance of participants?

Research Question 4a: Do participants adopt different goal orientation profiles across tasks?

Research Question 4b: Are participants' task-specific goal orientation profiles different from their General GO profiles?

Research Question 4c: Do participants' adopt different goal orientations across tasks?

Research Question 4d: Are participants' task-specific goal orientations significantly different from their General goal orientations?

Research Question 5a: Are task-specific goal orientation profiles stable over time?

Research Question 5b: Are task-specific goal orientations stable over time?

Research Question 6a: Do the different task-specific GO profiles score significantly differently on self-efficacy?

Research Question 6b: How do task-specific mastery-avoidance goal orientations correlate with self-efficacy? This Research Question will only be answered if a 2x2 model of goal orientations is chosen as an appropriate model.

Research Question 6c: How do task-specific performance-avoidance goal orientations correlate with self-efficacy?

Research Question 7a: Do the different task-specific GO profiles score significantly differently on mental effort?

Research Question 7b: How do the different task-specific goal orientations correlate with mental effort (if at all)?

Research Question 8a: How do task-specific mastery-approach goal orientations correlate with performance on tasks (if at all)?

Research Question 8b: How do task-specific mastery-avoidance goal orientations correlate with performance on tasks (if at all)?This Research Question will only be answered if a 2x2 model of goal orientations is chosen as an appropriate model.

Research Question 8c: How do task-specific performance-approach goal orientations correlate with performance on tasks (if at all)?

Research Question 8d: How do task-specific performance- avoidance goal orientations correlate with performance on tasks (if at all)?

Research Question 9: Do the different task-specific GO profiles score significantly differently on task performance?

A careful review of literature on GOs led to the proposal of the Research Questions and Hypotheses presented above. A description of the research methods used in the current study to address these Research Questions and Hypotheses is provided in the next chapter.

Chapter 7: Methodology

7.0. Introduction

This chapter focuses on the research methods used in this study. This study was a longitudinal one which was carried out over a period of one year. It was carried out using two main research designs: a panel survey and an experimental study. Consequently, the large part of this chapter is split into two main sections: one describing the survey (Section 7.1.) and the other describing the experiments (Section 7.2.). In each of the two sections, firstly, there is an overall description of the research design and the reasoning behind the use of such a design. This is followed by descriptions of the sample and survey (in Section 7.1.) or experiment (in Section 7.2.) characteristics. Next, the data collection procedure is presented and details of the pilot study are provided. Finally, a description of each of the measures used as well as support for the use of these measures is provided.

Following the two main sections on the survey and experiments, an overall explanation of the rationale for using these data collection methods is provided. This is followed by a description of the ethical issues relating to this study and the ways in which these were addressed. In conclusion, two summary tables providing information on the demographics and data collection times for the overall study are presented along with a diagram indicating which aspects of the research relate to each of the research questions and hypotheses.

7.1. Panel Survey

7.1.1. Sample Characteristics

The panel survey data were required in order to provide answers to a number of the Research Questions and Hypotheses (e.g. Hypothesis 1 and Research Questions 1, 2, 3 and 4). Participants were asked to complete questionnaires at two points in time (Time Q1

and Time Q2). At Time Q1 participants were asked to indicate whether they would be interested in participating in the experimental study. Therefore, a sub-sample of the panel survey participants also participated in the experimental study. Further details of this are provided in Section 7.2.

The majority of participants consisted of Loughborough University School of Business and Economics students. However, there were also a number of employed and retired individuals. Participant numbers were assigned to individuals so as to be able to match their responses at different Time points (e.g Time Q1 and Time Q2). The sample size at Time Q1 consisted of 641 participants whilst the sample size at Time Q2 consisted of 202 participants.

7.1.2. Survey Characteristics

At Time Q1, participants were asked to complete a measure of General GOs, and a short demographics questionnaire¹. The questionnaire at Time Q1 also contained a Hypothetical Task GO measure. For this, participants were provided with one of two task scenarios (refer to Appendix A for copies of the Time Q1 questionnaires) and asked to complete the Hypothetical Task GO measure keeping in mind the way they would approach the task provided in the task scenario. Participants were randomly assigned to task scenarios. Two task scenarios were provided in order to ensure that any differences between General and Hypothetical Task GOs (as well as between Hypothetical Task and Verbal and Numerical Test GOs) would be the result of task-specificity per se and not some unknown bias resulting from the nature of the hypothetical task chosen.

¹ Participants were also asked to complete a personality measure at Time Q1 (refer to Appendix A). However, it was later decided that the relationships between GOs and personality would not be included in this thesis. At Time Q2 the questionnaire consisted of a General GO measure and a Hypothetical Task GO measure (refer to Appendix B for copies of the Time Q2 questionnaires). At Time Q2 participants completed the Hypothetical Task GO measure with the same task scenario that they had at Time Q1.

The time interval between Time Q1 and Time Q2 needed to be long enough to reduce familiarity with the questionnaire but not too long due to the time constraints of the study. Since a number of participants were also taking part in the experimental study it was decided that there should be two different time intervals (one for survey only participants and one for survey participants who also took part in the experiments). In this way the stability of General and Hypothetical Task GOs over shorter and longer time intervals could be compared. The plan was to have a four week gap between Time Q1 and Time Q2 for survey only participants and a twelve week gap (with the experimental sessions in between) for survey participants who also took part in the experimental sessions in between) for survey participants who also took part in the experiments. Unfortunately, things did not always go according to plan and the interval between Time Q1 and Time Q2 ranged between 5 and 51 weeks for different participants (refer to Section 7.1.3. for further details).

The questionnaires were handed out personally in order to obtain as high a response rate as possible (refer to Section 7.1.3. for further details of the data collection procedure). The response rate for postal surveys is not very high (Frankfort-Nachmias & Nachmias, 1996). The same can be said for e-mail surveys. Therefore, by personally asking participants to complete the questionnaire a higher response rate was hoped for. Since some participants were not present when the Time 2 questionnaires were handed out in person, an e-mail (with the Questionnaire attached) was sent out to the participants who did not respond at Time Q2 (refer to Appendix C for a copy of the e-mail).

7.1.3. Description of the Survey Data Collection Procedure

This section provides a brief description of how the survey data collection was carried out including the original timeline for data collection and the reasons for deviating from this timeline. As described in Section 7.1.2. it was decided that there should be two different time intervals between the administration of Time Q1 and Time Q2 questionnaires for the survey only participants and for the survey participants who also took part in the experiments.

Survey Only Participants. The participants who only took part in the survey (and not the experiment) mainly consisted of full-time Loughborough University students and a group of employed individuals (most of whom were attending a part-time management course at Loughborough University). The plan was to have a four week time interval between Time Q1 and Time Q2 for these participants.

In order to increase the survey response rate it was decided that the questionnaires should be handed out personally. A number of lecturers at the School of Business and Economics at Loughborough University provided the opportunity for questionnaires to be handed out during their lectures. Each lecturer allocated ten to fifteen minutes of two of their lectures (once for the Time Q1 questionnaire and approximately 4 weeks later for the Time Q2 questionnaire) for data collection purposes. During this time students were provided with a brief summary of the study and of the questionnaire. Moreover, the questionnaires were handed out it was made clear that *only participants who were not participating in the experiment* were being asked to complete the questionnaires being handed out. The survey participants who also took part in the experiments were informed that they would be asked to complete the questionnaire at a later stage.

The part-time students were all employed by the same company and were attending a part-time course at Loughborough University. These participants had two lectures based at Loughborough University (during the data collection phase of this study). Consequently, the timings of the Time Q1 and Time Q2 questionnaires needed to coincide with the dates of these two lectures. Again, ten minutes were provided during each of these two lectures to provide a brief summary of the study, hand out the questionnaires and collect the completed ones. The MSc students and employed participants were people the researcher knew and personally invited to participate in the study. These people were invited to participate in order to increase the sample size as much as possible (especially for the experimental study).

At Time Q1 participants were invited to take part in the experimental study by providing them with a brief explanation of what this consisted of and asking them to indicate on their questionnaires (which at Time Q1 had a short section inviting participants to take part in the experimental study) whether they were interested in participating or not (refer to Appendix A for a copy of the Time Q1 questionnaire). Participants who did not indicate whether or not they were interested in participating in the experimental study were assumed not to be interested and were asked to complete the Time Q2 questionnaire 4 weeks after completing the Time Q1 questionnaire.

It was assumed that, for various reasons (e.g. illness) a number of participants might not have been present when the Time Q2 questionnaire was handed out. Consequently, an email with the Time Q2 questionnaire attached was sent out to those participants who had completed the Time Q1 questionnaire but not the Time Q2 questionnaire (refer to Appendix C for a copy of this e-mail).

The variations in the time interval for the survey only participants came about for three main reasons:

- 1. It was not always possible to attend lectures exactly 4 weeks apart.
- Some participants who were contacted by e-mail took much longer to respond than others (and than the participants who completed the Time Q2 questionnaire during the lectures).
- 3. Only two of the part-time student lectures were being carried out at Loughborough University. Therefore, the Time Q1 and Time Q2 questionnaires were handed out during these two lectures. Unfortunately, there was a change in their timetable and the second lecture was postponed to a much later date. Consequently, completion of the Time Q2 questionnaires by these participants was delayed by a few months.

Survey participants who also took part in the experiment. These consisted of Loughborough University students, a group of employed individuals from Malta, and a group of retired and employed individuals from the UK. As mentioned in Section 7.1.2. a 12 week interval between the Time 1 and Time 2 questionnaires of these participants was planned.

The Time Q1 questionnaire was handed out personally to all participants. The Time Q1 questionnaires were handed out during lectures, as described above, for the Loughborough University students. The employed participants from Malta as well as the retired and employed participants in the UK consisted of family, friends, and friends of friends and family. The Time Q1 questionnaires were also handed out personally to these participants. All participants taking part in the experimental study (apart from the retired participants) received an e-mail with the Time Q2 questionnaire attached a few weeks after their second (Time E2) experimental session (refer to Appendix J for a copy of the e-mail). The retired participants expressed during their experimental sessions that they preferred to receive their second questionnaire by post. Subsequently, their Time Q2 questionnaire was posted to them four weeks after their second experimental session.

The variations in the time interval (between Time Q1 and Time Q2) for these participants came about for two main reasons:

- There were a number of delays in carrying out the experimental sessions (described in further detail in Section 7.2.3.) and the Time Q2 questionnaires were sent out after the Time E2 experiments.
- Some participants took longer than others to send back the Time Q2 questionnaires by e-mail and post.

7.1.4. Survey Pilot Study

A pilot study was conducted prior to the main data collection in order to ensure that there were no problem items in the questionnaires. Participants in the survey pilot study were asked to complete and provide feedback on the Time Q1 Questionnaire. The survey pilot study consisted of 6 participants. These were all Loughborough University School of Business and Economics PhD students.

The feedback provided indicated that there were a few problems with the masteryavoidance items. Some of the pilot study participants either did not understand these items properly or assumed that the mastery-avoidance GO items were measuring a mastery-approach GO in different terms. This became evident when one participant asked for one of the mastery-avoidance items to be clarified and another participant provided an explanation which showed that he understood the mastery-avoidance item to have the same meaning as the mastery-approach items but to be simply worded differently. This prompted some other participants to point out that they found the mastery-avoidance items difficult to understand. The problem items were modified and pilot study participants were asked to evaluate the changes. Once it was agreed that the items were understandable the survey data collection for the study commenced.

7.1.5. Description of Measures used in the Survey

In this section the measures used in the survey are described. The choice of measures is justified and relevant reliability and validity statistics (from previous studies using these measures) are provided for each of the measures used. The reliability results for the measures in the current study are presented in the next chapter. A summary of the measures completed by survey participants at each time point are presented in Table 7.1.

Table 7.1. Measures completed at each point in time

Time	Measures completed			
Time Q1	 Demographics Measure General GOs Measure Hypothetical Task GOs Measure 			
Time Q2	 General GOs Measure Hypothetical Task GOs Measure 			

7.1.5.1. General and Hypothetical Task Goal Orientations

The measures of General and Hypothetical Task GOs were adapted from the Achievement Goal Questionnaire–Revised (AGQ-R) developed by Elliot and Murayama (2008). As described in Chapter 3 (Section 3.2.3.) the AGQ-R (Elliot & Murayama, 2008) is a revised version of the AGQ developed by Elliot and McGregor (2001). On careful examination of the AGQ, Elliot & Murayama (2008) found a number of issues with it, including pitting one goal against another, thus not allowing for the multiple goal perspective (Elliot & Murayama, p615). As a result they revised it and developed the AGQ-R. This was found to have reliability coefficients for the four scales ranging between .84 and .94 (including the performance avoidance scale which had a lower reliability in the AGQ). Moreover, the intercorrelational analyses results ranged from non-significant to .68 (p<.01) with higher correlations for GOs sharing the same competence dimension (mastery vs. performance) than those that did not.

Elliot and Murayama (2008) also conducted a confirmatory factor analysis in order to test whether the AGQ-R items loaded onto the respective GO scales. Their results indicated that all the items loaded onto the scales they were meant to be measuring (the factor loadings ranged from 0.73 to 0.93). Moreover, they found good model fit for the fourfactor model (χ^2 (48, *N*=229)= 78.32; *p* <.01, χ^2/df =1.63, CFI=0.99, IF=0.99, RMSEA=0.053). They also tested six alternative models (including a 3-factor model with the MAP and MAV items loading onto one scale and a 3-factor model with the PAP and PAV items loading onto one scale). The fit indexes for these alternative models indicated that none of the alternative models provided a good fit to the data (Elliot & Murayama, 2008, pp619).

Since this study makes use of the multiple goal perspective, and because the AGQ-R improved upon the AGQ in a number of ways, the GO measures in the current study were adapted from the AGQ-R. The GO measures needed to be adapted because the AGQ-R was originally developed for measuring students' GOs on university courses e.g. "I am striving to understand the content of this course as thoroughly as possible" (Elliot & Murayama, 2008). For the purposes of the survey, the AGQ-R was adapted in order to make the items relevant to the General and Hypothetical Task GOs. Items were presented in the present and future tense for the General and Hypothetical Task GOs measures, respectively. Further adaptations to the questionnaires were made as a result of the feedback obtained from the pilot study regarding the unclarity of the mastery-avoidance items. The mastery-avoidance items were modified in order to try and make them as clear as possible. However, overall, the adaptations were kept as minor as possible. Since the confirmatory factor analysis results indicated that all the items in the AGQ-R loaded onto the appropriate scales and had good reliabilities, all the AGQ-R items were included in the questionnaires and the same order of items was maintained.

Table 7.2. indicates the items in the original AGQ-R and the adapted items in the General GO questionnaire. Tables 7.3. and 7.4. indicate the items in the original AGQ-R and the adapted items in the Hypothetical Task GO questionnaires for Task 1 and Task 2, respectively.

ltem number	AGQ-R Item	General GO Questionnaire Item
1	My aim is to completely master the material presented in this class.	My aim is to completely master everything I do.
2	I am striving to do well compared to other students.	I strive to do well compared with others in everything I do.
3	My goal is to learn as much as possible.	My goal is to learn as much as possible about the things I do.
4	My aim is to perform well relative to other students.	My aim is to perform well relative to others in everything I do.
5	My aim is to avoid learning less than I possibly could.	My aim is to ensure I do not miss out on new learning opportunities.
6	My goal is to avoid performing poorly compared to others.	My goal is to avoid performing poorly compared to others in everything that I do.
7	I am striving to understand the content of this course as thoroughly as possible.	I strive to understand the content of the things I do as much as possible.
8	My goal is to perform better than the other students	My goal is to perform better than others in everything I do.
9	My goal is to avoid learning less than it is possible to learn.	My goal is to avoid missing opportunities to fully understand an activity.
10	I am striving to avoid performing worse than others.	I strive to avoid performing worse than others in everything that I do.
11	I am striving to avoid an incomplete understanding of the course material.	I strive to avoid an incomplete understanding of the things I do.
12	My aim is to avoid doing worse than other students.	My aim is to avoid doing worse than others in everything that I do.

Table 7.2. Comparison of AGQ-R questionnaire items with adapted General Goal Orientation items.

Key: Items measuring MAP GO: 1, 3 and 7; Items measuring MAV GO: 5, 9 and 11; Items measuring PAP GO: 4, 2 and 8; Items measuring PAV GO: 6, 10 and 12.

ltem number	AGQ-R Item	Hypothetical Task GO Questionnaire Item (for Task 1)
1	My aim is to completely master the material presented in this class.	I will aim to completely master selling the new carpets.
2	I am striving to do well compared to other students.	I will strive to do better at selling the new carpets compared to my colleagues.
3	My goal is to learn as much as possible.	My goal will be to learn as much as possible regarding how to sell these new carpets.
4	My aim is to perform well relative to other students.	My aim will be to sell more new carpets relative to my colleagues.
5	My aim is to avoid learning less than I possibly could.	My aim will be to avoid learning less than I possibly could about selling these new carpets.
6	My goal is to avoid performing poorly compared to others.	My goal will be to avoid selling fewer new carpets compared to my colleagues.
7	I am striving to understand the content of this course as thoroughly as possible.	I will be striving to understand how to sell these new carpets as thoroughly as possible.
8	My goal is to perform better than the other students	My goal will be to perform better than my colleagues at selling these new carpets.
9	My goal is to avoid learning less than it is possible to learn.	My goal will be to avoid learning less than it is possible to learn about selling new carpets.
10	I am striving to avoid performing worse than others.	My goal will be to avoid missing opportunities to fully understand how to sell the new carpets.
11	I am striving to avoid an incomplete understanding of the course material.	I will be striving to avoid an incomplete understanding about how to sell the new carpets.
12	My aim is to avoid doing worse than other students.	My aim will be to avoid doing worse than my colleagues when selling the new carpets.

Table 7.3. Comparison of AGQ-R questionnaire items with adapted Hypothetical Task Goal Orientation items for Task 1

Key: Items measuring MAP GO: 1, 3 and 7; Items measuring MAV GO: 5, 9 and 11; Items measuring PAP GO: 4, 2 and 8; Items measuring PAV GO: 6, 10 and 12.

ltem number	AGQ-R Item	Hypothetical Task GO Questionnaire Item (for Task 2)
1	My aim is to completely master the material presented in this class.	My aim will be to completely master the new computer program.
2	I am striving to do well compared to other students.	I will strive to do better in training compared to my colleagues.
3	My goal is to learn as much as possible.	My goal will be to learn as much as possible about this new computer program.
4	My aim is to perform well relative to other students.	My aim will be to perform better during training relative to my colleagues.
5	My aim is to avoid learning less than I possibly could.	My aim will be to avoid learning less than I possibly could about this new computer program.
6	My goal is to avoid performing poorly compared to others.	My goal will be to avoid performing poorly during training compared to my colleagues.
7	I am striving to understand the content of this course as thoroughly as possible.	I will be striving to understand the content of this computer program as thoroughly as possible.
8	My goal is to perform better than the other students	My goal will be to perform better than my colleagues during training.
9	My goal is to avoid learning less than it is possible to learn.	My goal will be to avoid learning less than it is possible to learn about this new computer program.
10	I am striving to avoid performing worse than others.	My goal will be to avoid missing opportunities to fully understand the new computer program.
11	I am striving to avoid an incomplete understanding of the course material.	I will be striving to avoid an incomplete understanding of this new computer program.
12	My aim is to avoid doing worse than other students.	My aim will be to avoid doing worse than my colleagues during training.

Table 7.4. Comparison of AGQ-R questionnaire items with adapted Hypothetical Task Goal Orientation items for Task 2

Key: Items measuring MAP GO: 1, 3 and 7; Items measuring MAV GO: 5, 9 and 11; Items measuring PAP GO: 4, 2 and 8; Items measuring PAV GO: 6, 10 and 12.

7.1.5.2. Demographics Questionnaire

At Time Q1 and Time Q2 participants were asked to complete a short demographics questionnaire. This included items asking about age and current occupation amongst others. The demographics questionnaire was included in the panel survey for two main reasons. Firstly, due to the fact that the study was longitudinal (and the stability of a number of variables over time was being examined) it was essential to be able to match participants' questionnaires at Time Q1 and Q2. Participants' full name, age, and occupation were requested for matching purposes. Secondly, it was considered important to be able to assess the presence of any significant differences in GOs as a result of differences in age and occupation, if necessary. Therefore, it was essential to ask for this information in order to be able to perform the necessary analyses if required. Additionally, participants' e-mail addresses were requested as part of this questionnaire. This was required in order to be able to contact them at a later date regarding the experiment (in the case of participants who were interested in participating in the experimental study), or (in the case of all other participants) to send them the follow-up questionnaire if required (refer to Appendix A and Appendix B for copies of the demographics questionnaires at Time Q1 and Time Q2, respectively).

7.2. Experimental Study

One purpose of the experiments was to examine the interaction effects of General and Induced GOs on performance. An additional purpose was to examine whether taskspecific GOs adopted on different tasks are different from each other and from General GOs. The experiments were also carried out in order to examine the relationships between task-specific GOs and mental effort, self-efficacy and performance on tasks.

7.2.1. Sample Characteristics

A sub-sample from the panel survey was used for the experiments (rather than an independent sample) due to the fact that data concerning the General GOs of participants taking part in the experimental study were required in order to answer some of the Research Questions and Hypotheses. Rather than measure the General GOs of an independent sample (as well as of the survey participants), it was thought to be more practical to ask participants taking part in the panel survey to participate in the experimental study since they had already completed the General GOs and demographics questionnaires. Experimental study participants were asked to attend two experimental sessions (Time E1 and Time E2; further details of the data collection procedure are provided in Section 7.2.3.). The sample size at Time E1 was 73 participants, of whom 71 attended the experimental session at Time E2.

7.2.2. Experiment Characteristics

The experimental study consisted of two 1 hour sessions. Participants were asked to attend the experimental sessions in groups (mainly due to time constraints). The fact that participants attended in groups as opposed to individually was not thought to have any influence on their performance since participants could not see what other participants were doing and complete confidentiality was maintained with regards to questionnaire and test results. Moreover, it is not uncommon to ask participants to attend GO experimental sessions in groups. This was done in a number of other GO studies such as those by Chen and Mathieu (2008), Loraas and Diaz (2009), and Jagacinski et al. (2001), amongst others. Participants were randomly assigned to groups according to their availabilities.

During the experimental sessions participants were asked to complete a verbal and a numerical aptitude test as well as a number of questionnaires. The exact content of each session is provided Sections 7.2.2.3. and 7.2.2.4.

7.2.2.1. Use of Aptitude Tests

Since this study is aimed at developing knowledge of GOs in organisations it was thought appropriate to use aptitude tests as the experimental tasks. Aptitude tests were chosen due to the fact that they are used in employment and selection (and therefore relevant to organisations). Moreover, although they are both components of General Mental Ability, they are quite different tasks. Verbal and Numerical skills are considered to be quite different from each other and participants having one set of skills do not necessarily have the other.

Furthermore, aptitude tests are NOT all-or-none tasks in which individuals either win or lose. On the contrary, it is possible for participants to obtain different levels of performance on the two tasks. This was considered to be important so that even small differences in performance (as a result of different GOs) could be determined. Aptitude tests also provide a link between university and work. Since the majority of the sample consisted of university students and because this research was carried out in order to provide insight into the use of the concept of GOs in organisations, aptitude tests were considered to be ideal since they are work-related but can be administered to university students.

An additional benefit of the use of aptitude tests was for the participants themselves. By participating in the experiments students had the opportunity to practise aptitude tests which are widely used in employee selection. I was able to give these students feedback on their performance as well as tips on how to improve on aptitude tests. The fact that students were given an opportunity to practise aptitude tests also helped me obtain a
larger sample size. Thus, the use of aptitude tests provided a win-win situation. Finally, aptitude tests may easily be administered in a classroom. Consequently, they were chosen for practical reasons too.

A decision was made to use aptitude tests developed by test publishers rather than develop aptitude tests for the purposes of this study. This decision was based on the fact that published aptitude tests have been tried and tested and they have good levels of reliability and validity as well as norm groups for comparison (details of the reliability and validity of the aptitude tests are provided in Section 7.2.5.6.).

7.2.2.2. Incentives

Participants were offered the opportunity to take part in a £50 cash prize draw (three £50 prizes were offered) as an incentive to participate in the experimental study. This prize draw took place on the 10th May 2010 once all the experiments were completed. However, the main incentive for participation in the experimental study was the opportunity to practise Verbal and Numerical aptitude tests and obtain feedback as well as hints and tips regarding how to improve on these tests. This was thought to be useful for participants since nowadays a large number of employers make use of aptitude tests in their selection process. In fact, one of the many reasons for choosing aptitude tests as the experimental tasks was that it would provide participants with a good opportunity to practise these tests in order to increase their chances of success in the selection process when applying for jobs or placements. The feedback was sent out to participants approximately 2 months after their Time E2 session (and, for those who completed the Time Q2 questionnaires, approximately 1 month after completing these). Please refer to Appendix M for a copy of the feedback form template.

7.2.2.3. Experimental Session 1

Prior to attending the first experimental session all the necessary information was sent out to participants by e-mail (Refer to Appendix D for a copy of the e-mail sent out to participants).

On arrival at the first experimental session, participants were given a description of what the session would entail and were provided with an information sheet and consent form to sign (Refer to Appendix E for a copy of the information sheet and consent form). Following this, participants were provided with an example question of the first test which they were asked to complete. The answer to this example question was provided and subsequently, participants were asked to complete a self-efficacy questionnaire in order to rate their perceived self-efficacy on the test they were about to complete (refer to Section 7.2.5.4. for details on the self-efficacy questionnaire and Appendix F for a copy of the self-efficacy questionnaire). On completion of the self-efficacy questionnaire, participants were provided with task instructions and the first aptitude test was handed out (refer to Appendix G for a copy of the instructions provided throughout the experiment at Time E1).

The order of the tests was randomised for the different participant groups so that any carryover effects would be distributed evenly through the sample. Therefore, some participants were asked to complete the verbal test first followed by the numerical test whilst others were asked to complete the numerical test first.

On completion of the first aptitude test, participants were asked to complete a questionnaire (referred to as the post-test questionnaire, Appendix H) consisting of measures of their GOs on the Verbal/Numerical test (depending on which test they had just completed), as well as measures of mental effort and task-experience for the Verbal/Numerical test.

The only two GO studies which assessed state GOs (Harwood & Swain, 2001, and Breland & Donovan, 2005) did not manipulate GOs. In these studies the state GO questionnaire was handed out to participants prior to the task. In other studies (e.g. Loraas and Diaz, 2009; Kozlowski et al., 2001; Steele-Johnson et al.,2008; Jagacinski et al., 2001) goal commitment was measured to check whether the manipulations worked. However, state GOs were not specifically assessed. In this study it was thought best to assess the GOs being adopted on the Verbal and Numerical Tests rather than goal commitment. These Verbal and Numerical Test GOs were not 'state' GOs since at Time E1 no manipulations occurred. However, they were considered to be 'task-specific' GOs. The Verbal and Numerical test GOs were measured on completion of the test rather than before. It was thought that participants might be able to understand the questionnaire better and provide more accurate answers, after completing the test since the thoughts they had during test completion, as well as the approach that they adopted, would still be fresh in their minds.

The same process (questionnaire – experiment - questionnaire) was repeated for the second test. In order to minimise external factors influencing GOs, self-efficacy, mental effort and performance on tests, the test instructions were strictly adhered to and the experimental sessions were kept as consistent as possible.

7.2.2.4. Experimental Session 2

Prior to the second experiment, participants were divided into three groups: Group A, Group B, and Group C. All three groups completed the same measures and the same tests as they did during the first experimental session. However, the three groups were given *different test instructions* during the second experimental session. Group A were the control group. Group B were given instructions intended to induce a performance-approach GO whilst Group C were given instructions intended to induce a mastery-approach GO (please refer to Appendix I for the Time 2 test instructions).

Although it would have been ideal to assign participants randomly to these groups this was not entirely possible because of the different participant backgrounds. The mastery-approach GO induction relied on participants benefitting from improving their understanding of the aptitude tests. Consequently, this induction was not relevant to the employed and retired participants. As a result, the majority of these were assigned to the control group (Group A) and a few were assigned to the performance-approach induction group (Group B). All other experimental participants were randomly assigned to the three groups. An overview of the Verbal and Numerical test instructions given to the different experimental groups at Time E1 and Time E2 is provided in Table 7.5.

Table 7.5. Verbal and Numerical Test Instructions given to the Different ExperimentalGroups at Time E1 and Time E2

Group	Time 1	Time 2			
А	Neutral Instructions	Neutral Instructions			
В	Neutral Instructions	Instructions inducing a PAP GO			
С	Neutral Instructions	Instructions inducing a MAP GO			

Key: PAP GO = Performance-approach goal orientation; MAP GO = Mastery-approach goal orientation.

In order to maximise the probability that the GO manipulations would be successful, other GO studies which successfully manipulated GOs were examined. These included the studies by Loraas and Diaz (2009), Chen and Mathieu (2008), Jagacinski et al. (2001), Gerhardt and Luzadis (2009), Kozlowski et al. (2001), and Steele-Johnson et al. (2008). It was noted that in these studies the mastery-approach manipulation emphasised the importance of learning and stressed that the purpose of the session was to learn. Moreover, there were no references to comparison with others, but rather, the emphasis was on performance and the comparison of participants' performance with that of others was stressed. Manipulation checks performed in a number these studies indicated that the GO manipulations were successful. Therefore, the GO manipulations for the current

study were based on the GO manipulations of the previously mentioned studies. The following instructions were provided for the mastery-approach GO induction:

"It is important that you keep in mind that this session is a chance for you to practise aptitude tests for when you apply for employment. So it is important that you use this opportunity to learn more about what it is like to take a verbal/numerical aptitude test. The results on these tests are not as important as the opportunity to learn more about aptitude tests."

As in the other studies, the focus was on learning and on emphasising that the purpose of the experimental session was for them to learn more about aptitude tests. The instructions provided for inducing a performance-approach GO consisted of the following:

"This time I shall be comparing your performance on the test to the test norms and I shall also be comparing your performance on this test with the performance of other participants of my research. I would like to remind you that the results of this test will ONLY be used for the purpose of this research and will not have any influence whatsoever on your course since the results will be kept in very strict confidentiality."

Similarly to the other studies in which performance-approach GOs were manipulated the emphasis was on performance and on comparing participants' performance to that of other individuals (other-referenced).

7.2.3. Description of the Experiment Data Collection Procedure

This section provides a brief description of how the experimental data collection was carried out including the original timeline for data collection and the reasons for deviating from this timeline. Experimental participants were contacted by e-mail prior to each experimental session in order to remind them what the study consisted of and how the experimental sessions would be conducted (refer to Appendix D for a copy of the e-mail sent out to participants). This e-mail was only sent out to those participants who indicated (on the Time Q1 questionnaire) that they were interested in participating in the experiments.

Participants were provided with a list of possible dates and times for the experimental sessions which they could sign up for. However, they were also informed that extra sessions could be arranged if they could not make any of the proposed dates and times. At the end of the first experimental session participants were informed that they would be contacted by e-mail regarding the second experimental session. Once again, an e-mail was sent out to experimental participants prior to the second experimental sessions (Time E2) in order to make arrangements (refer to Appendix K for a copy of the e-mail inviting participants to attend the Time E2 experimental session).

The initial plan was to allow a four week time interval between each questionnaire and experimental session and between the two experimental sessions. Therefore, 4 weeks between Q1 and E1; between E1 and E2; and between E2 and Q2. These time intervals were required in order to diminish the possibility of participants becoming too familiar with the tasks and measures which might in turn influence their performance, responses to the questionnaires, and/or GOs that they adopted. These time intervals were chosen because they were thought to be an optimal compromise between allowing sufficient time between measures to reduce familiarity and not spending too much time collecting data due to the time constraints of the study. Unfortunately, the data collection did not proceed perfectly according to plan. There were three main reasons for the deviations from the planned timeline:

1. There were problems gaining access to the aptitude tests. This delayed the experimental sessions by 2 months.

- 2. Summer holidays, Christmas holidays, and time periods during which there were exams made it difficult to stick to a 4 week time interval (for students).
- It was not always possible to go to Malta and other places in the UK at exact 4 week intervals to carry out the experimental sessions for the Maltese, employed, and retired participants.

7.2.4. Pilot Study for Experiment

Similarly to the survey, the pilot study was carried out purely to ensure that there were no problem items in any of the questionnaires and that participants would not have any problems understanding the test instructions. Participants in the pilot study were asked to attend an experimental session having the same content and instructions as the Time E1 experimental session. Four participants attended the experiment pilot study; three of these had participated in the survey pilot study. All four participants were Loughborough University School of Business and Economics PhD students. The mastery-avoidance GO items on the questionnaires had been modified following feedback from the survey pilot study. Participants in the experiment pilot study seemed to fully understand the questionnaire items and test instructions. Consequently, experimental data collection commenced.

7.2.5. Description of Measures used in the Experiments

In this section the measures used in the experiments are described. The choice of measures is justified and relevant reliability and validity statistics (from previous studies using these measures) are provided for each of the measures used. The reliability results for the measures in the current study are presented in the next chapter. A summary of the measures completed by experimental participants during each of the experimental sessions is presented in Table 7.6.

Time	Measures completed by Experimental participants
Time E1	 Verbal and Numerical Tests Verbal and Numerical Test GOs Measures Verbal and Numerical Test measures of task experience Verbal and Numerical Test measures of perceived self-efficacy Verbal and Numerical Test measures of mental effort
Time E2	 Verbal and Numerical Tests Verbal and Numerical Test GOs Measures Verbal and Numerical Test measures of task experience Verbal and Numerical Test measures of perceived self-efficacy Verbal and Numerical Test measures of mental effort

Table 7.6. Measures completed at each point in time

7.2.5.1. Verbal and Numerical Test Goal Orientations

For the reasons described in Section 7.1.5.1. the AGQ-R was adapted in order to measure participants' GOs in this study. This time adaptations were required to make the questionnaires relevant to the Verbal and Numerical Tests. Moreover, the survey pilot study feedback was kept in mind when adapting the questionnaires for the Verbal and Numerical tests in order to make the MAV items as clear and understandable as possible. However, as with the General and Hypothetical Task GO questionnaires, the adaptations were kept as minor as possible. Table 7.7. provides a comparison of the original AGQ-R questionnaire items and the Verbal and Numerical Test GO questionnaire items.

ltem number	AGQ-R Item	Verbal/Numerical Test GO Questionnaire Item
1	My aim is to completely master the material presented in this class.	My aim was to completely master this task.
2	I am striving to do well compared to other students.	I was striving to do well on this task compared to the other participants.
3	My goal is to learn as much as possible.	My goal was to learn as much as possible.
4	My aim is to perform well relative to other students.	My aim was to perform well on this task relative to the other participants.
5	My aim is to avoid learning less than I possibly could.	My aim was to avoid learning less than the maximum possible about this task.
6	My goal is to avoid performing poorly compared to others.	My goal was to avoid performing poorly compared to others on this task.
7	I am striving to understand the content of this course as thoroughly as possible.	I was striving to understand the content of this task as thoroughly as possible.
8	My goal is to perform better than the other students	My goal was to perform better on this task than other participants.
9	My goal is to avoid learning less than it is possible to learn.	My goal was to avoid learning less than it was possible to learn about this task.
10	I am striving to avoid performing worse than others.	I was striving to avoid performing worse than others on this task.
11	I am striving to avoid an incomplete understanding of the course material.	I was striving to avoid an incomplete understanding of this task.
12	My aim is to avoid doing worse than other students.	My aim was to avoid doing worse than the other participants on this task.

Table 7.7. Comparison of AGQ-R questionnaire items with adapted Verbal and Numerical Test Goal Orientation items

Key: Items measuring MAP GO: 1, 3 and 7; Items measuring MAV GO: 5, 9 and 11; Items measuring PAP GO: 4, 2 and 8; Items measuring PAV GO: 6, 10 and 12.

7.2.5.2. Task Experience

It was thought that experience of Verbal and Numerical aptitude tests or regular practice of the skills required to do well on these tests might influence the types of GOs adopted. For example, if a participant was very familiar with Numerical tests or regularly practised the skills required to do well on a numerical test they might be more likely to adopt a performance-approach GO than a participant who is not familiar with the test or who did not usually practise numerical tasks. A measure of task experience was therefore drawn up in order to be able to control for familiarity with the tests and regular practice of the associated skills (refer to Appendix H for a copy of the Task Experience measure).

7.2.5.3. Practice

Some participants may have chosen to practise Verbal and Numerical tests between Time E1 and Time E2. There were two main reasons for choosing to ask participants whether they practised Verbal and Numerical tests between Time E1 and Time E2. Firstly, there was a possibility that participants who chose to practise between Time E1 and Time E2 had certain GOs in common. For example, if a participant with a performance-avoidance GO felt that he/she might have performed the worst on one of the tests at Time E1, he/she might decide to practise so as not perform the worst on that particular test at Time E2. Therefore, the first reason for choosing to measure practice was to be able to test whether having a particular GO motivated participants to practise between Time 1 and Time 2. Secondly, it was thought necessary to be able to control for practice in the data analyses in order to ensure that any of the observed effects were not a result of practice. Consequently, an item measuring whether participants practised the aptitude tests before the Time E2 experiment was introduced into the task experience measure at Time 2 (refer to Appendix L for a copy of the task experience measure at Time E2). This item replaced item 1 (of the Time E1 task experience measure) since this was no longer relevant at Time E2. Thus, the item, 'Have you ever completed a verbal (or numerical) aptitude test?' was replaced by the item, 'Have you completed any other verbal (or

numerical) aptitude test (other than that in the first part of this experiment)?' Participants who answered 'No' to Item 1 on the Time E1 Task Experience Questionnaire and 'Yes' to Item 1 on the Time E2 Task Experience Questionnaire were assumed to have practised the aptitude tests between Time E1 and Time E2. Although the new item 1 formed part of the task experience measure at Time E2, it was used as a measure of practice on its own in the data analyses (e.g. numerical test practice consisted of the answer to item 1 of the Task Experience measure at Time E2).

7.2.5.4. Self-efficacy

The self-efficacy subscale of the Motivated Strategies for Learning Questionnaire (MSLQ, Pintrich, 1993) was used as a measure of self-efficacy. This subscale was chosen due to its psychometric properties as well as the fact that it has been frequently used in GO research. The fact that a subscale of the MSLQ was used as opposed to the whole questionnaire did not compromise the psychometric properties of the measure. Duncan and McKeachie (2005:119) state that "The 15 scales on the MSLQ can be used together or singly. The scales are designed to be modular and can be used to fit the needs of the researcher." Pintrich et al. (1993) carried out a study in order to test the reliability and predictive validity of the MSLQ. Their factor analysis results indicated 6 latent variables, each one of which consisted of the items associated with the subscale it was meant to be measuring (that is, Intrinsic GO, Extrinsic GO, Task Value, Control beliefs about Learning, Self-Efficacy, and Test Anxiety). The self-efficacy subscale of the MSLQ was found to have a Coefficient Alpha of 0.93 indicating good reliability and internal consistency. The MSLQ self-efficacy subscale was also found to have good predictive validity with respect to academic performance (r=.41, ρ =.05).

As mentioned previously, a number of studies assessing the relationship between self-efficacy and GOs used the self-efficacy subscale of the MSLQ. These include Bandalos et al. (2003), Bong (2001), Green (2001), Lau et al. (2008), and Karabenick (2004), amongst others. The MSLQ selfefficacy subscale was adapted for use in the experiments. This was necessary because the original version was aimed for assessing students' self-efficacy on university courses. The adaptations, which were kept to a bare minimum, were made to allow the measure to be relevant to the experimental tasks. Therefore, for example the original item "I'm certain that I can understand the most difficult material presented in the readings for my courses" was changed to "I'm certain that I can understand the most difficult items presented on this task." Refer to Appendix F for a copy of the self-efficacy measure on the verbal and numerical tests at Time E1 and Time E2 (the same questionnaire was used in all cases).

7.2.5.5. Mental Effort

The measure of mental effort used in the current study is one developed by Paas (1992), which was a modification of the perceived task difficulty scale developed by Bratfisch et al. (1972). In a study conducted by Paas et al. (1994) this scale was found to have good reliability (Cronbach's alpha=.90) and sensitivity (mental effort was found to change significantly in different training conditions). Moreover, the scale has been used in a number of other studies including Paas (1992), Pas and van Merrienboer (1994), De Crook et al. (1998), and De Crook and van Merrienboer (2007) and was found to have reliability coefficients of 0.90, 0.82, 0.98, and 0.88, respectively (refer to Appendices H and L for copies of the measures of Mental Effort at Time 1 and Time 2, respectively).

7.2.5.6. Verbal and Numerical Test Reliability and Validity

As mentioned earlier (Section 7.2.2.1.) the Verbal and Numerical aptitude tests were obtained from test publishers. A number of decisions regarding the type of aptitude tests required for this study were made. Firstly, the aptitude tests needed to have a constant level of difficulty as opposed to tests that increased in difficulty as the test progressed. This was necessary because it was necessary to split each test in half for participants to complete the first half at Time E1 and the second half at Time E2. The alternative to splitting the tests would have been to obtain two versions of the same test. However, this would have been far too costly and was decided against. Secondly, paper and pencil tests were required since it would have been problematic to book computer labs for the experimental sessions. Moreover, it would have been problematic to use online tests for the employed and retired participants since the experimental sessions for these participants were not carried out at the university and there would not have been enough computers for the sessions to be carried out in groups. Thirdly, the tests were required not to be too easy or too technical for the experimental purposes but rather aptitude tests which were developed for use at a University level standard of education were required. Finally, tests that were long enough to be able to split into half and still have a good number of items in each half were needed. These criteria drastically limited the number of appropriate aptitude tests for use in the experimental study. In the end access to Verbal and Numerical Tests from ERAS test publishers was gained. The tests used were the Managerial and Professional Series Verbal and Numerical Reasoning Tests. These tests were designed to test people having "at least some significant attainments in secondary or possibly tertiary education" (etm103, 2007).

The Verbal test was designed to be 30 minutes long and consisted of 13 stems having 4 items each (52 items in all). Participants were asked to complete 6 stems (24 items) at Time E1 and another 6 stems at Time E2. At Time E1 the example question provided on the test booklet was used as an example question whilst at Time E2 the extra stem was used as the example question. Participants were given 14 minutes to complete the Verbal Test at Time E1 and Time E2. The Numerical Test was designed to be 35 minutes long and consisted of 5 stems having 6 items each (30 items in all). Participants were asked to complete the first 2 stems (12 items) at Time E1 and the last 2 stems at Time E2. Similarly to the Verbal test, the example question provided on the booklet was used as an example question. Participants were used as the Time E2 example question. Participants were given 15 minutes to complete the Numerical Test at Time E1 and Time E2 and Time E1 and Time E1 and Time E1 and the last 2 stems at Time E2. Similarly to the Verbal test, the example question provided on the booklet was used as an example question. Participants were given 15 minutes to complete the Numerical Test at Time E2 example question. Participants were given 15 minutes to complete the Numerical Test at Time E1 and Time E2.

The psychometric properties of these tests were established and tested during the test development phase by the test publishers. The Verbal and Numerical tests had good levels of reliability (Alpha=.89 on Verbal test and .88 on Numerical test). The tests also had good content

validity since they were reviewed by chartered occupational psychologists in order to assess whether they met a number of criteria (e.g. they were related to/drawn from the world of work). Any items that were thought to be problematic were amended or removed. With respect to criterion-related validity, the verbal and numerical tests were significantly correlated to a number of competencies such as leadership, teamwork, and planning and organisation, amongst others. The test publishers also obtained a correlation of 0.53 when correlating the Verbal and Numerical tests. This indicates good construct validity and shows that the tests are measuring a similar underlying construct (probably general reasoning ability).

7.3. Support for the research methods used in this study

As described previously, this study entails the use of questionnaires and experiments. In addition, it is a longitudinal study. In this section the rationale for choosing quantitative methods over qualitative ones is provided. Furthermore, the advantages of using questionnaires and experiments in the current study as opposed to other data collection methods are described.

7.3.1. Quantitative versus Qualitative Methods

The main reason for choosing to use quantitative methods over qualitative ones was due to the nature of the Research Questions and Hypotheses of the current study. Examining the stability of GOs, determining the different types of GO profiles, as well as assessing the relationships between GOs and other variables such as self-efficacy, performance, and mental effort all require the use of quantitative methods. Most GO studies have made use of quantitative methods of data collection and analyses in the past. Although it is possible to use qualitative methods in order to discover more about the concept of GOs this was not thought to be appropriate for the purposes of the current study.

7.3.2. Questionnaires

In this study questionnaires are used a number of times. At Time Q1 and Time Q2, the only other option available for collecting the data required would be to use interviews. The preference of questionnaires over interviews at Time Q1 and Time Q2 was the result of a number of reasons. Firstly, questionnaires are much less time consuming than interviews and participants are able to complete them at their own leisure. Since a sub-sample from the survey study were asked to attend two experimental sessions it would have been rather excessive to ask them to attend two interviews too (one before and one after the experimental sessions). A second reason for using questionnaires rather than interviews is that these allow the participants to take as long as they need to answer them and there is much less pressure to answer in socially desirable ways in comparison with interviews (where one is face to face with the interviewer). Since the mastery-avoidance and performance-avoidance GOs may have negative connotations it was thought that participants may answer more truthfully if they completed a questionnaire rather than if they attended an interview. Thirdly, questionnaires are much less costly than interviews and much easier to record. Finally, questionnaires are more objective than interviews as they are not subjected to interviewer bias. Thus, they tend to have greater reliability and validity than interviews. For these reasons it was thought that the use of questionnaires would be more advantageous in order to achieve the aims of this study.

7.3.3. Experiments

Rather than carrying out an experimental study, it would have been possible to send out a survey asking participants to think back to different tasks and state whether they adopted different GOs according to the task at hand. This would have enabled the use of a much larger sample and would have been much less time consuming and less costly than carrying out an experiment. Interviewing people would have been another alternative. Participants could have been asked how they would approach particular tasks in certain settings. However, experiments were carried out for three main reasons. Firstly, they made it possible to measure

performance, mental effort and self-efficacy under controlled conditions. Secondly, they allowed for the manipulation of GOs on tasks and, thirdly, they reduced the possibility of memory biases. These would not have been easy to achieve had a survey or interviews been used.

7.3.4. Longitudinal Study

In order to assess the stability of a number of variables over time (including General GOs, Hypothetical Task GOs, and Verbal and Numerical test GOs) a longitudinal study was necessary. By measuring these variables a number of times and at different time periods it was possible to examine their stability over time.

A second important reason for using a longitudinal design was to reduce practice effects on the experimental task. If participants were asked to perform the tasks after a very short interval there would have been an increased chance of performance improvement as a result of recent practice. Therefore, it was essential to have as long a time interval as possible between the experimental sessions so as to reduce any practice effects. Since participants may have practised in between the two experimental sessions in order to improve their tests scores, the task experience questionnaire at Time E2 included a measure of practice. Practice on tests was taken into account when carrying out the data analyses. Finally, it was considered essential to measure the variables over a period of time rather than in quick succession so as to prevent (as much as possible) one measure influencing participants' answers (and performance) on the other measures (and tests).

7.4. Ethical Considerations

Ethical clearance was obtained from the Loughborough University Ethical Advisory Committee. In order to comply with the ethical advisory committee codes of practice a number of ethical considerations were made for the current study. With respect to the survey, participants were informed that participation in the study was completely voluntary, and they had the right to withdraw from the study at any point in time without any negative consequences whatsoever. Moreover, participants were given a brief description of the study and informed that the information provided in their questionnaires would be kept in very strict confidentiality at all times. Finally, participants were given the researcher's e-mail address in order to be able to contact the researcher if they had any questions regarding the study (refer to Appendices A and B for copies of the information provided to participants at Time Q1 and Time Q2, respectively).

With regards to the experiments, at Time E1, participants were provided with an information sheet (which was for them to keep) which contained details regarding what the experimental sessions would entail, the length of the experimental sessions, and that they would be asked to attend a second experimental session. The information sheet also provided information regarding their eligibility to participate in the prize draw, their right to withdraw from the study, the fact that participation in the study was completely voluntary, and that the information provided by them throughout the experimental sessions would be kept in very strict confidentiality. Participants were also provided with the researcher's e-mail address so as to be able to contact the researcher if they had any queries or if they wanted to ask for further information regarding the study (refer to Appendix E for a copy of the Information Sheet given to participants of the experimental study). Participants were given time to read the Information Sheet at the beginning of the Time E1 experimental session. Following this they were asked to complete an Informed Consent Form (as advised by the Loughborough University ethical advisory committee, refer to Appendix E for a copy of the consent form).

7.5. Overview of Study

This section aims to summarise the information provided throughout the chapter by providing information regarding the data collection times and the sample characteristics for the survey and experiments. Table 7.8. provides a summary of the characteristics of the survey only participants as well as the median time intervals between the Time Q1 and Time Q2

questionnaires. Table 7.9. provides a summary of the demographic characteristics of the participant who took part in the survey and experimental study, as well as the median time intervals between the questionnaires and experiments. Following this, in order to provide an overview of how the questionnaire survey and experiment are related to each of the research questions and hypotheses (presented in Chapter 6 Section 6.2.) a diagram of research (Figure 7.1.) is presented.

Category	Number of participants at Time 1	Gender (Time 1 participants)	Number of participants at Time 2	Gender (Time 2 participants)	Median Time between Q1 and Q2 (in weeks)	Age Range
Full-Time 1st Year	131	60 male 69 female 2 missing	14	4 male 10 female	6	17-40
Full-Time 2 nd Year*	50	25 male 24 female 1 missing	8	2 male 6 female	36	19-22
Full-Time 3 rd Year	354	195 male 123 female 36 missing	108	61 male 47 female	7	20-32
MSc	2	2 male	1	1 male	19	24
Employed	31	24 male 7 female	18	17 male 1 female	38	19-45
Total	568	306 male 223 female 39 missing	149			17-45

* These participants were going on placement the following academic year. Those participants who consented to me having their placement e-mail address were sent the Time Q2 questionnaire approximately 36 weeks later (when they were on placement). I sent the Time Q2 questionnaire to all other participants approximately 8 weeks after they completed the Time Q1 questionnaire.

Category	Number of participants at Time Q1 (Gender)	Number of participants at Time Q2 (Gender)	Number of participants at Time E1 (Gender)	Number of participants at Time E2 (Gender)	Median Time between Q1 and Q2*	Median Time between Q1 and E2*	Median Time between E1 and E2*	Median Time between E2 and Q2*	Age Range	Experimental Group
Full-Time 1st Year	21 (9 male 12 female)	15 (5 male 10 female)	21 (9 male 12 female)	21 (9 male 12 female)	10	3	3	4	18-21	MAP: 10 PAP: 10 1 did not attend at Time E2
Full-Time 3 rd Year	18 (8 male 8 female 2 missing)	12 (4 male 8 female)	18 (8 male 8 female 2 missing)	17 (8 male 8 female 1 missing)	10	2	2	6	20 - 22	Control: 6 MAP: 7 PAP: 5
MSc	9 (6 male 3 female)	9 (6 male 3 female)	9 (6 male 3 female)	9 (6 male 3 female)	19	5	5	7	22 -29	Control: 4 MAP: 5
Employed	19 (8 male 10 female 1 missing)	14 (5 male 9 female)	19 (8 male 10 female 1 missing)	18 (8 male 10 female)	17	6	4	5	18 – 57	Control: 9 MAP: 3 PAP: 6 1 did not attend at Time E2
Retired	6 (2 male 4 female)	3 (1 male 2 female)	6 (2 male 4 female)	6 (2 male 4 female)	32	5	30	12	59-64	Control: 6
Total	73 (33 male 37 female 3 missing)	53 (21 male 32 female)	73 (33 male 37 female 3 missing)	71 (33 male 37 female 1 missing)					18-64	Control: 25 MAP: 25 PAP: 21 2 did not attend at Time E2

Table 7.9. Demographics and Data Collection Times of Participants who took part in the Survey and Experimental Study

Note: The year of study of participants was based on when they completed the Time Q1 questionnaire e.g. those participants who completed the Time Q1 questionnaire in their 1^{st} year and Time E1 experiment in their 2^{nd} year were placed in the ' 1^{st} year' category. * In weeks **Key:** MAP = mastery-approach induction condition; PAP = performance-approach induction condition.

Figure 7.1. Diagram indicating which aspects of the research relate to each of the research questions and hypotheses



As will be discussed in the introduction to the next chapter, the results and discussion chapters are presented according to theme. Five main themes were chosen for this study (refer to Chapter 8 Section 8.0. for further details). Each box on the right-handside represents a theme and includes all the research questions and hypotheses associated with that theme. The two boxes on the left-hand-side indicate the two research designs used in the study and the arrows indicate which data were used in order to answer the research questions and hypotheses for each of the themes.

7.6. Synopsis

Data collection methods and procedures for the current study were presented in this chapter. Furthermore, a description regarding how the ethical considerations were addressed in the current study was provided. In the next chapter the results of the exploratory analyses of the data are presented. Moreover, a number of Research Questions are answered and Hypotheses tested.

Chapter 8: Exploratory Analyses

8.0. Introduction

The literature review was structured in a way to make it "reader friendly" by first introducing the concept of GOs and how this developed, then creating an awareness of the problems with its conceptualisation and finally discussing how these problems have created inconsistencies with respect to the relationships between GOs and other variables (relevant to organisations). Although this structure was considered to be the most suitable for the literature review, the order in which these themes are discussed in the results chapters has been modified in line with the order in which the analyses are most sensibly reported. These modifications were made so that the results chapters would be more reader friendly and understandable.

The aim of this chapter is to provide information regarding the psychometric properties of the measurement scales used in this study. Descriptive statistics, reliability analyses, and factor analyses results are presented. Moreover, the results of the LCAs are provided. These analyses will provide the reader with a good understanding of the nature of the scales used as well as the reasoning behind the choice of GO model used in the current study.

In conducting these exploratory analyses a number of Hypotheses and Research Questions were answered. These are therefore addressed in this chapter so as not to have unnecessary repetition of these results in the next chapter. Therefore, although the Hypotheses and Research Questions regarding the relationships between GOs and other variables were discussed last in the literature review, these Hypotheses and Research Questions will be discussed first in the results chapters.

Since a number of Research Questions and Hypotheses are closely related, in order for the results to be described in a meaningful way, the Research Questions and Hypotheses that are closely related were grouped together according to their common themes. The themes chosen for this study included: a) Types and characteristics of General GO profiles and model choice with respect to GO profiles

b) The relationships between GOs and other variables

c) The stability of GOs over time

d) The task-specificity of GOs

e) Interactions between state and trait GOs.

A consequence of this thematic grouping is that some Research Questions were included in this chapter even though their results were not directly answered by the exploratory analyses. For example Hypotheses 2a and 2b and Research Questions 6b and 6c address the relationships between GOs and self-efficacy. These were all tested or answered by means of the exploratory analyses and were therefore included in the present chapter. Research Question 6a also addresses the relationships between GOs and self-efficacy. However, the exploratory analyses do not provide an answer to it. Nevertheless, this Research Question was included in the present chapter along with the other Research Questions and Hypotheses addressing the relationships between GOs and self-efficacy. Although a number of Research Questions and Hypotheses that are not answered or tested by the exploratory analyses are included in this chapter, the majority of these will be dealt with in Chapter 9. The themes addressed in this chapter include Theme A and Theme B. Themes C, D and E will be addressed in Chapter 9. Preceding the first section of the results is a brief outline of this chapter.

First the reliability analyses and descriptive statistics for all the scales used in the current study are presented (Sections 8.1., 8.2. and 8.3.). Next, there is a description of the results of the exploratory and confirmatory factor analyses (Section 8.4.). Following this, the results of the LCAs are presented along with the answer to the Research Question 1 (Section 8.5.). Subsequently, the correlation matrices for all the variables included in the analyses are provided and a few correlations of interest are discussed (Section 8.6.). Subsequently, the results of the Research Questions and Hypotheses addressed by the correlational analyses (and any Research Questions and

Hypotheses having the same themes as the latter Research Questions and Hypotheses) are then focused on (Section 8.7.).

8.1. Descriptive Statistics and Reliability Analyses of General Goal Orientations and Hypothetical Task Goal Orientations at Time 1 and Time 2

Tables 8.1. through to 8.4. show the descriptive statistics and reliability analyses for the General and Hypothetical Task GOs. These data were collected at Time Q1 and Time Q2 of the General Survey. The descriptive statistics obtained here and in Section 8.2. will be compared with those of other studies using the 2x2 model of GOs in Section 8.2. in order to examine any variation in the GOs adopted by participants across studies.

Table 8.1. General	GOs at Time	1
--------------------	-------------	---

	Mean	SD	N	Highest Possible Score	Lowest Possible Score	Number of Items	Cronbach's Alpha
MAP	3.88	0.56	636	5	1	3	0.59
MAV	3.75	0.64	634	5	1	3	0.70
PAP	3.95	0.67	636	5	1	3	0.78
PAV	3.96	0.76	635	5	1	3	0.83

Key: MAP=Mastery Approach Goal Orientation; MAV=Mastery Avoidance Goal Orientation; PAP=Performance Approach Goal Orientation; PAV=Performance Avoidance Goal Orientation

	Mean	SD	N	Highest Possible Score	Lowest Possible Score	Number of Items	Cronbach's Alpha
MAP	3.90	0.60	189	5	1	3	0.67
MAV	3.72	0.63	187	5	1	3	0.74
PAP	3.81	0.72	188	5	1	3	0.85
PAV	3.82	0.76	190	5	1	3	0.86

Table 8.2. General GOs at Time 2

Key: MAP=Mastery Approach Goal Orientation; MAV=Mastery Avoidance Goal Orientation; PAP=Performance Approach Goal Orientation; PAV=Performance Avoidance Goal Orientation.

The means and standard deviations of all four GOs at Time Q1 and Time Q2 seem to indicate that participants tend to adopt moderate to high GOs (on all four GOs). This will be picked up on later in the Chapter when describing the GO profiles adopted by

participants (Section 8.5.). There was quite a substantial decrease in sample size from Time 1 to Time 2. However, as this is nearly always the case in longitudinal studies, this attrition was expected.

The General GO scales at Time 1 and Time 2 seem to be quite reliable with masteryavoidance (MAV), performance-approach (PAP), and performance-avoidance (PAV) GOs all having a Cronbach's Alpha of 0.7 or more. However, the reliabilities of the mastery-approach (MAP) GO scales (both at Time 1 and at Time 2) are not as high as those of the other scales. When Item 1 was deleted the reliabilities increased slightly (Cronbach's Alpha = 0.64 at Time 1 and 0.69 at Time 2). However, this increase in reliability was not large enough to justify removal of Item 1 in either case. Moreover, when assessing a MAP GO on Verbal and Numerical tests, removal of item 1 caused a substantial decrease in reliability (e.g. Crobach's alpha decreased from .56 to .37 on the Verbal test MAP scale at Time 2 when item 1 was deleted). Therefore, it was decided that Item 1 should be kept on the MAP scale.

The mean scores presented in Table 8.1. and 8.2. show that participants' PAP and PAV General GOs seem to be decreasing from Time 1 to Time 2 whilst the MAP and MAV GOs seem to be stable for respondents as a whole. The differences in mean scores from Time 1 to Time 2 will be investigated further in Chapter 9 Section 9.1.1.

	Mean	SD	N	Highest Possible Score	Lowest Possible Score	Number of Items	Cronbach's Alpha
MAP	3.95	.66	631	1	5	3	.76
MAV	3.70	.70	630	1	5	3	.75
PAP	3.80	.76	633	1	5	3	.89
PAV	3.86	.63	629	1	5	3	.65

Table 8.3. Hypothetical Task GOs at Time 1

Key: MAP=Mastery Approach Goal Orientation; MAV=Mastery Avoidance Goal Orientation; PAP=Performance Approach Goal Orientation; PAV=Performance Avoidance Goal Orientation

	Mean	SD	N	Highest Possible Score	Lowest Possible Score	Number of Items	Cronbach's Alpha
MAP	3.93	.70	193	1	5	3	.81
MAV	3.62	.77	191	1	5	3	.85
PAP	3.71	.85	194	1	5	3	.93
PAV	3.73	.75	194	1	5	3	.77

Table 8.4. Hypothetical Task GOs at Time 2

Key: MAP=Mastery Approach Goal Orientation; MAV=Mastery Avoidance Goal Orientation; PAP=Performance Approach Goal Orientation; PAV=Performance Avoidance Goal Orientation

Similarly to the General GOs, participants seem to adopt moderate to high Hypothetical Task GOs on all four GOs. The mean scores indicate that participants' MAV, PAP and PAV GOs seem to be decreasing from Time 1 to Time 2 whilst the MAP GO seems to be rather stable for respondents as a whole. These changes over time will be investigated further in Chapter 9 Section 9.1.2.

With respect to reliability, the Cronbach's alpha of the Hypothetical Task GOs were all greater than 0.7 (indicating good levels of reliability) except for the PAV GO at Time 1 which had a Cronbach's Alpha of 0.65. One of the items on the PAV scale at Time 1 seemed to be a problem item. Cronbach's alpha increases to .74 if this item is deleted. However, since this item does not seem to be a problem in any of the other PAV scales it was not thought wise to delete it.

8.2. Descriptive Statistics and Reliability Analyses of Verbal and Numerical Test Goal Orientations at Time 1 and Time 2

Verbal and Numerical test GOs were assessed by means of a questionnaire following completion of the Verbal and Numerical Tests during the experimental sessions (Time E1 and Time E2). The descriptive statistics and reliability analyses of the Verbal and Numerical Test GOs are presented in Tables 8.5. through to 8.8.

	Mean	SD	N	Highest Possible Score	Lowest Possible Score	Number of Items	Cronbach's Alpha
MAP	3.85	0.66	71	5	1	3	0.42
MAV	3.55	0.72	71	5	1	3	0.72
PAP	3.51	0.86	71	5	1	3	0.93
PAV	3.59	0.92	71	5	1	3	0.91

Table 8.5. Verbal Test GOs at Time 1

Key: MAP=Mastery Approach Goal Orientation; MAV=Mastery Avoidance Goal Orientation; PAP=Performance Approach Goal Orientation; PAV=Performance Avoidance Goal Orientation

	Mean	SD	N	Highest Possible Score	Lowest Possible Score	Number of Items	Cronbach's Alpha
MAP	3.71	0.66	71	5	1	3	0.56
MAV	3.46	0.80	71	5	1	3	0.80
PAP	3.42	0.94	71	5	1	3	0.93
PAV	3.61	0.95	71	5	1	3	0.94

Table 8.6. Verbal Test GOs at Time 2

Key: MAP=Mastery Approach Goal Orientation; MAV=Mastery Avoidance Goal Orientation; PAP=Performance Approach Goal Orientation; PAV=Performance Avoidance Goal Orientation

Although participants seem to be adopting moderate to high GOs (on all four GOs) on the Verbal tests, these means are a little lower than the mean GO scores for General GOs. The statistical significance of differences in the mean scores of General and taskspecific GOs will be addressed in Chapter 9 Sections 9.2.3. and 9.2.4. With respect to the reliability analyses, there seems to be a problem with the MAP GO scale both at Time 1 and at Time 2. Unfortunately, deletion of any of the items on this scale does not seem to increase the reliability of the scale. Cronbach's alpha was greater than 0.7 for all the other scales, indicating good levels of reliability.

Table 8.7. Numerical Test GOs at Time 1

	Mean	SD	Ν	Highest Possible Score	Lowest Possible Score	Number of Items	Cronbach's Alpha
MAP	3.69	0.79	71	5	1	3	0.63
MAV	3.34	0.74	70	5	1	3	0.69
PAP	3.40	0.99	71	5	1	3	0.91
PAV	3.57	1.02	71	5	1	3	0.95

Key: MAP=Mastery Approach Goal Orientation; MAV=Mastery Avoidance Goal Orientation; PAP=Performance Approach Goal Orientation; PAV=Performance Avoidance Goal Orientation

	Mean	SD	Z	Highest Possible Score	Lowest Possible Score	Number of Items	Cronbach's Alpha
MAP	3.72	0.76	71	5	1	3	0.70
MAV	3.54	0.78	70	5	1	3	0.77
PAP	3.47	0.92	71	5	1	3	0.94
PAV	3.58	0.88	71	5	1	3	0.95

Table 8.8. Numerical Test GOs at Time 2

Key: MAP=Mastery Approach Goal Orientation; MAV=Mastery Avoidance Goal Orientation; PAP=Performance Approach Goal Orientation; PAV=Performance Avoidance Goal Orientation

The means and standard deviations once again indicate that participants seem to be adopting moderate to high GOs on all four GOs for the Numerical tests at Time 1 and Time 2 (but again these seem to be lower than the mean General GO scores). The MAP GO scale on the Numerical tests seems to have much better reliability than that on the Verbal test. In fact, at Time 2, the MAP GO scale had a Cronbach's alpha of 0.7. The MAP and MAV GO scales on the Numerical test at Time 1 do not have particularly good reliabilities. They are however, reasonable (and much better than the Verbal test MAP GO scale).

The studies reviewed which investigated GOs using the 2x2 model used different measures and scale types. For example Janssen and Prins (2007) used a measure of GOs developed by Biemond and Van Yperen (2001) and a 7-point scale. Elliot and McGregor (2001) used the AGQ with a 7-point scale whilst Elliot and Murayama (2008) used the AGQ-R with a 5-point scale. Since an adaptation of the AGQ-R was used in this study (with a 5-point scale) a decision was made to compare the means and reliabilities obtained in this study to those obtained by Elliot and Murayama (2008). This comparison is made in Table 8.9. For the purpose of this comparison the mean scores and reliabilities obtained for General, Hypothetical Task, Verbal Test and Numerical Test GOs at Time 1 are used.

Research Study	Mean GO Score				Scale Reliability			
	MAP	MAV	ΡΑΡ	PAV	MAP	MAV	ΡΑΡ	PAV
Elliot & Murayama (2008) N=229	4.23	3.61	4.05	3.83	0.84	0.88	0.92	0.94
Present Study – General GOs	3.88	3.75	3.95	3.96	0.59	0.70	0.78	0.83
Present Study – Hypothetical Task GOs	3.95	3.70	3.80	3.86	0.76	0.75	0.89	0.65
Present Study – Verbal Test GOs	3.85	3.55	3.51	3.59	0.42	0.72	0.93	0.91
Present Study – Numerical Test GOs	3.69	3.34	3.40	3.57	0.63	0.69	0.91	0.95

Table 8.9. Comparison of Descriptive Statistics obtained in this study with thoseobtained in the study by Elliot and Murayama (2008)

A comparison of the mean scores indicates that the mean MAP and PAP GO scores obtained in the study by Elliot and Murayama (2008) are higher than those obtained in this study, particularly for the Numerical Test. The mean MAV and PAV General and Hypothetical Task scores obtained are higher than those of Elliot and Murayama (2008). However, the mean Verbal and Numerical Test scores are lower than those of the Elliot and Murayama (2008). From the table, it is clear that the mean General and Hypothetical Task GO scores are higher than the mean Verbal and Numerical Test scores. Since there are variations in the mean scores across tasks within this study it was thought best to focus on these differences for the purposes of this study and not examine the differences in mean scores obtained in this study with those obtained in other studies. These differences in mean GO scores obtained in this study will be investigated in further detail in Chapter 9 Sections 9.2.3. and 9.2.4.

The reliability analyses comparison in Table 8.9. indicates that the MAP and MAV GO scale reliabilities obtained in this study are lower than those of Elliot and Murayama (2008). The PAP and PAV General GO scales and the PAV Hypothetical Task GO scale in this study also have lower reliabilities than those of Elliot and Murayama (2008). However, the PAP and PAV Verbal and Numerical Test GO scales and the PAP

Hypothetical Task GO scale have similar (and sometimes higher) reliabilities to those of Elliot and Murayama (2008).

The scale reliabilities obtained in this study were all good with the exception of the MAP General GO at Time 1 and the MAP Verbal Test GOs at Time 1 and Time 2. Moreover, the MAP General GO at Time 2, MAP Numerical Test GO at Time 1 and PAV Hypothetical Task GO at Time 1 were a little low. It is not entirely clear why the MAP and PAV GO scales had low reliabilities at times especially since these were found to be very good in some cases e.g. when measuring Numerical Test GOs at Time 2. Since the scale reliabilities were only a little low in a few instances and were very good in all other cases it was decided that, for the purpose of the current study, all the items should be kept on all the four GO scales. This decision will be discussed further in Section 8.4. (where the results of the exploratory and confirmatory factor analyses are presented).

With respect to model choice, as discussed in the literature review, the main difference between the 3-factor model and the 2x2 model of GOs is the mastery GO. In the 2x2 model it is divided into MAP and MAV GOs whilst in the 3-factor model there is a combined mastery GO (MGO) scale. However, on careful inspection of scales measuring a MGO it was found that these only include MAP GO items (e.g. Liem et al., 2008; Alkharusi, 2008) as opposed to items measuring both MAP and MAV GOs. Consequently, the 2x2 model of GOs may be considered to be the same as the 3-factor model plus the MAV GO. As a result, the usefulness and scale properties of the MAV GO will be examined in order to determine which GO model should be used in this research study.

With respect to the reliability of the MAV scales in this study, there was a high level of reliability in all cases except for Numerical Test GOs at Time 1. However, even in the latter case, Cronbach's alpha was found to be 0.69, which is an acceptable level of scale reliability. Consequently, in terms of reliability, the 2x2 factor model seems to be an appropriate model for assessing GOs. The results of the exploratory and confirmatory factor analyses presented in Section 8.4. of the current chapter will

provide further evidence as to whether the 3-factor or 2x2 model of GOs should be used in the current study.

8.3. Descriptive Statistics and Reliability Analyses of Self-efficacy, Mental Effort, Task-experience, Practice, and Performance, for Verbal and Numerical Tests at Time 1 and Time 2

The self-efficacy, practice, mental effort, task experience, and performance of participants on Verbal and Numerical tests were all measured during the experimental sessions at Time E1 and Time E2. The descriptive statistics and reliability analyses for these variables are presented in Table 8.10. through to Table 8.17.

	Mean	SD	Z	Highest Possible Score	Lowest Possible Score	Number of Items	Cronbach's Alpha
VT S.E. T1	3.67	0.61	71	5	1	7	0.88
VT S.E. T2	3.44	0.73	70	5	1	7	0.92
NT S.E. T1	3.54	0.88	71	5	1	7	0.95
NT S.E. T2	3.29	0.95	71	5	1	7	0.95

Table 8.10. Self-efficacy on Verbal and Numerical Tests at Time 1 and Time 2

Key: VT = Verbal Test; NT = Numerical Test; S.E. = Self-efficacy; T1= Time 1; T2 = Time 2;

Self-efficacy seems to be moderate to high on both Verbal and Numerical tests at Time 1 and Time 2. Moreover, the mean self-efficacy scores seem to decrease slightly from Time 1 to Time 2 on both the Verbal and the Numerical Tests. Paired Samples T-Tests were carried out in order to determine whether these decreases in self-efficacy from Time 1 to Time 2 were significant. The results are presented in Table 8.11.

Table 8.11.	Paired Samples	T-Test to Test	for Significant	Differences in	Self-Efficacy
from Time 1	to Time 2 on Ve	erbal and Num	erical Tests (N≈	70)	

Variables	Time 1 Mean	Time 1 SD	Time 2 Mean	Time 2 SD	Df	T-value	Sig.
VT S.E.	3.67	0.61	3.44	0.73	69	3.27	<.01
NT S.E.	3.54	0.88	3.29	0.95	70	3.16	<.01

Key: VT S.E. = Verbal Test Self-Efficacy; NT S.E. = Numerical Test Self-Efficacy

Table 8.11. shows that there is a significant decrease in participants' self-efficacy scores from Time 1 to Time 2 on both the Verbal and Numerical Tests. A possible reason for this may be that participants thought they could do well on the tests at Time 1 but found the tests difficult. Consequently their confidence regarding their performance on the tests decreased at Time 2. Although participants answered an example question before completing their self-efficacy questionnaire a number of participants disclosed that they found the example questions much easier to complete than the actual tests. This provides some support for the explanation above.

It was thought useful to compare the mean self-efficacy scores obtained in this study to those obtained in other studies in order to be able to evaluate better whether the mean scores obtained in this study were in fact moderate to high. The mean scores obtained in other studies using the MSLQ to measure self-efficacy are presented in Table 8.12.

Table 8.12. Mean Self-Efficacy Scores in Studies using the MSLQ Self-EfficacySubscale

Research Study	Mean Self-Efficacy Score			
Lau et al. (2008)	3.77			
Bong (2009)	3.36			
N=500	(for Middle School students)			
Bong & Hocevar (2002)	Scores ranging between 4.17 and 5.51 on 6			
N=358	different academic subjects			

All studies used a 5-point scale with the same labels ranging from Strongly Disagree to Strongly Agree.

The mean scores obtained in this study are very similar to the mean scores obtained in the studies by Lau et al. (2008) and Bong (2009). However, they are low compared to the mean scores obtained in the study by Bong and Hocevar (2002). Since the former two studies have the largest sample sizes it was considered reasonable to compare the mean scores obtained in this study with them. Consequently, it may be said that the mean self-efficacy scores obtained in this study are similar to those obtained in other studies. With respect to scale reliability, the self-efficacy scale indicates high levels of reliability with Cronbach's alpha being greater than 0.88 in all cases. Pintrich et al. (1993) carried out a study in order to test the reliability and predictive validity of the MSLQ. The self-efficacy subscale of the MSLQ was found to have a Coefficient Alpha of 0.93 indicating good reliability and internal consistency. The reliability coefficients indicated in Table 8.10. are very similar to that obtained by Pintrich et al. (1993).

	Mean	SD	Z	Highest Possible Score	Lowest Possible Score	Number of Items
VT M.E. T1	6.34	1.23	71	9	1	1
VT M.E. T2	6.31	1.25	71	9	1	1
NT M.E. T1	6.62	1.38	71	9	1	1
NT M.E. T2	6.72	1.28	71	9	1	1

Table 8.13. Mental Effort on Verbal and Numerical Tests at Time 1 and Time 2

Key: VT = Verbal Test; NT = Numerical Test; M.E.=Mental Effort; T1= Time 1; T2 = Time 2;

Since mental effort was a single item measure it was not possible to assess its internal reliability. However, the psychometric properties of this measure were assessed in a number of studies as described in Chapter 7 Section 7.2.5.5. The mean scores on this measure indicate that participants seem to adopt rather high mental effort on Verbal and Numerical tests. Moreover, it seems as though participants tended to adopt slightly higher mental effort on Numerical tests than they did on Verbal tests. Paired samples t-tests were carried out in order to assess whether these differences in mental effort were significant. The results are presented in Table 8.14. below.

Table 8.14. Paired	Samples T-Test to	o Test for	Significant	Differences	in Mental	Effort
on Verbal and Nu	merical Tests at Ti	me 1 and 1	Time 2 (N=7	71)		

Variables	VT Mental Effort Mean	VT Mental Effort SD	NT Mental Effort Mean	NT Mental Effort SD	Df	T-value	Sig.
Time 1	6.34	1.23	6.62	1.38	70	0.22	.07
Time 2	6.31	1.25	6.72	1.28	70	-0.63	<.01

Key: VT Mental Effort = Verbal Test Mental Effort; NT Mental Effort = Numerical Test Mental Effort

Table 8.14. indicates that at Time 1 there was a nearly significant difference and at Time 2 there was a significant difference in the mental effort of participants on the

Verbal and Numerical Tests. Therefore, it seems as though participants did in fact adopt higher levels of mental effort on the Numerical Tests than on the Verbal Tests. This is possibly because people tend to find Numerical tasks more challenging than Verbal tasks.

The task experience scales (please refer to Table 8.15. below) consisted of 3 items. In each case the Cronbach's alpha increases (to .76, .61, .73, and .61 for VT T1, VT T2, NT T1 and NT T2, respectively) if item 1 is deleted. This was probably due to the fact that item 1 was somewhat different from items 2 and 3. Items 2 and 3 focused on understanding how often participants practised Verbal and Numerical tasks whilst item 1 focused on determining whether they had completed similar aptitude tests before. Although item 1 had a slightly different emphasis to items 2 and 3 all three items were considered to be important for understanding how much task experience participants had on the Verbal and Numerical Tests.

When measuring a broad construct with relatively few items, the items should not necessarily be very highly correlated with each other because they should be tapping somewhat different parts of the broad construct. Therefore, item 1 was kept. With respect to the means, it seems as though participants had a moderate level of task experience on both the Verbal and Numerical aptitude tests at Time 1 and Time 2.

	Mean	SD	N	Highest Possible Score	Lowest Possible Score	Number of Items	Cronbach's Alpha
VT T.E. T1	2.25	0.83	71	4	1	3	.58
VT T.E. T2	2.23	0.83	71	4	1	3	.53
NT T.E. T1	2.16	0.75	71	4	1	3	.55
NT T.E. T2	2.06	0.80	71	4	1	3	.55

Table 8.15. Task Experience on Verbal and Numerical Tests at Time 1 and Time 2

Key: VT = Verbal Test; NT = Numerical Test; T1= Time 1; T2 = Time 2; T.E. = Task Experience

	Mean	SD	N	% Yes	% No	Number of Items
Practice on VT	1.53	1.59	70	47	53	1
Practice on NT	0.50	0.50	71	41	59	1

Table 8.16. Practice on Verbal and Numerical Tests between Time 1 and Time 2

Key: VT = Verbal Test; NT = Numerical Test.

As described in the Methodology Chapter (Chapter 7 Section 7.2.5.3.) the practice measure at Time 2 consisted of item 1 of the task experience scale. It was thought that the type of General or task-specific GOs adopted by individuals may have influenced participants' motivation to practise aptitude tests between Time E1 and Time E2. Moreover, practising tests may have an influence on the type of GOs adopted on the Verbal and Numerical Tests at Time E2. Therefore, practice was measured in order for it to be included as a covariate when examining changes in GOs over time as well as changes in mental effort, self-efficacy, and performance. Since this was a 1-item measure with a yes or no response, it was thought that the percentage of responses in each category might provide more useful information than the mean scores. Therefore, although the mean scores are provided, the percentages of participants who answered 'yes' and 'no' to the item "Have you completed any other verbal/numerical aptitude test (other than that in the first part of this experiment)?" are also provided in Table 8.16.

Table 8.16. indicates that just over half the experimental participants (53%) did **not** practise Verbal Tests between Time E1 and Time E2. With respect to the Numerical Test it seems as though a slightly larger percentage of participants (than that on the Verbal test) did not practise between Time E1 and Time E2 (59%).

Table 8.17. Verbal and	Numerical Test	Performance at	Time 1 and Time 2
------------------------	----------------	----------------	-------------------

	Mean	SD	N	Highest Possible Score	Lowest Possible Score
VT Performance T1	15.31	3.47	71	24	0
VT Performance T2	13.58	3.80	71	24	0
NT Performance T1	6.08	2.27	71	12	0
NT Performance T2	6.92	2.41	71	12	0

Key: VT = Verbal Test; NT = Numerical Test.
Since the aptitude tests were purchased from a test publisher, the reliabilities of the Verbal and Numerical aptitude tests were provided in Chapter 7 Section 7.2.5.1. The descriptive statistics in Table 8.17. indicate the mean performance scores of participants on the Verbal and Numerical tests and Time 1 and Time 2. The data presented in Table 8.17. were gathered during the experimental sessions (Time E1 and Time E2).

The mean scores indicate a decrease in Verbal Test performance from Time 1 to Time 2 and a slight increase in Numerical Test performance from Time 1 to Time 2. In addition, there seems to be more variation in performance on the Verbal tests than on the Numerical tests (as indicated by the standard deviations of the two tests at Time 1 and Time 2). Paired samples t-tests were carried out in order to test whether these observed differences were significant. The results are presented in Table 8.18.

 Table 8.18. Paired Samples t-test to Test for Significant Differences in Verbal and

 Numerical Test Performance over time (N=71)

Variables	Time 1 Mean	Time 1 SD	Time 2 Mean	Time 2 SD	Df	T-value	Sig.
VT Perf.	15.31	3.47	13.58	3.80	70	4.10	<.01
NT Perf.	6.08	2.27	6.92	2.41	70	-2.80	<.01

Key: VT Perf. = Verbal Test Performance; NT Perf. = Numerical Test Perf.

The table above indicates that Verbal Test performance decreases significantly from Time 1 to Time 2 whilst Numerical Test performance increases significantly over time. It is not entirely clear why these changes in performance occurred especially since more participants practised Verbal tests than Numerical Tests between Time 1 and Time 2. It may be the case that participants who practised the Numerical Tests did so to a much higher degree. Since the practice scale did not assess the number of hours or the extent to which participants practised it is not possible to be certain whether this might be the case. However, it is also possible that (although the test difficulty was meant to be constant throughout the test) there were slight differences in item difficulty between the items used at Time E1 and Time E2. Unfortunately, it is not possible to determine whether this was the case.

8.4. Exploratory and Confirmatory Factor Analyses of General and Hypothetical Task Goal Orientations

In order to assess the extent to which the scale items are actually measuring the same variable, exploratory factor analyses (EFAs) were carried out. General and Hypothetical Task GOs at Time 1 were chosen for the factor analyses since these had the largest sample sizes. The data for these factor analyses were collected during the survey at Time Q1.

EFAs were carried out using Principal Axis Factoring as a method for extraction and Direct Oblimin rotation. Maximum iterations for conversion were set to 999 and coefficients smaller than 0.30 were suppressed. The results of the factor analyses are presented in Tables 8.19. and 8.20. below.

General GO Item Number	GO Scale Item is assessing	Factor 1	Factor 2	Factor 3
1	MAP			.38
2	РАР			.73
3	MAP		.54	
4	РАР			.62
5	MAV		.64	
6	PAV	.59		
7	MAP		.68	
8	РАР	.43		.49
9	MAV		.71	
10	PAV	.80		
11	MAV		.58	
12	PAV	.81		

Table 8.19. Factor Loadings for General GO Items at Time 1 (N=628)

Key: Minimum loading of 0.30 required.

The results presented in Table 8.19. indicate the presence of three factors. The items loading onto Factor 1 and Factor 3 correspond with the items on the PAV scale and the PAP scale, respectively. However, Item 1 is an exception. This is meant to be measuring a MAP GO and therefore seems to be loading onto the wrong factor. Moreover, Item 8 seems to be a problem too since it is cross-loading onto Factor 1 and Factor 3. Most of the MAP and MAV items (with the exception of Item 1 which seems

to be loading onto the wrong factor) loaded onto one factor (Factor 2). Overall, these results support the 3-factor model of GOs with a mastery scale and PAP and PAV scales.

With respect to the two problem items, Item 1 seems to be a problem item since it is loading onto the wrong factor. In the reliability analyses of General GOs at Time 1 it also seemed to be a problem item and removing it would have increased the reliability of this scale somewhat. However, removal of this item would have greatly decreased the reliability of the MAP scale for the Verbal Test at Time 2. Since it is not appropriate to use different scales for measuring General and Verbal Test MAP GOs a decision was made to keep item 1. Moreover, as shown in Table 8.20. item 1 loads onto Factor 1 for the Hypothetical Task GOs at Time 1. Consequently, it was not thought to be reasonable to remove this item from the MAP GO scale. Item 8 also seems to be a problem item with respect to the EFA of General GOs at Time 1. However, removing this item would reduce the reliability of the MAP scale. Similarly to item 1, item 8 also loads onto the expected factor for Hypothetical Task GOs at Time 1 (as shown in Table 8.20.). Consequently, it was decided that this item should not be removed from the MAP GO scale.

General GO Item Number	GO Scale Item is assessing	Factor 1	Factor 2	Factor 3
1	MAP	.59		
2	РАР		79	
3	MAP	.76		
4	РАР		88	
5	MAV			.74
6	PAV		57	
7	MAP	.74		
8	РАР		84	
9	MAV			.73
10	PAV	.31		.42
11	MAV	.48		.33
12	PAV		71	

Table 8.20. Factor Loadings for Hypothetical Task GO Items at Time 1 (N=612)

Key: Minimum loading of 0.30 required.

Similarly to the EFA of General GO items at Time 1, three factors emerged from the EFA of the Hypothetical Task GO items. This time items 10 and 11 were found to be problem items since they were cross-loading onto factors 1 and 3. Item 10 was the cause of the reliability of the PAV Hypothetical Task GO being on the low side (Table 8.3.). In fact, if item 10 is removed, the reliability of the PAV Hypothetical Task GO increases from .65 to .74. However, item 10 did not negatively affect the reliability of the other PAV GO scales. Unfortunately, it is not possible to use different scales to measure PAV GOs on different tasks, because unwanted variation would be introduced when comparing the PAV GOs. Therefore, it was decided that item 10 should be kept. Item 11 also cross-loaded. However, since this item did not seem to be a problem in the General GO factor analysis and because the reliability analysis (Table 8.3.) indicated a good level of reliability with Item 11 included in the MAV scales it was decided that this item should be kept.

In contrast to the EFA of General GOs at Time 1, the Hypothetical Task GOs EFA at Time 1 shows that the items loading onto Factors 1 and 3 correspond to the items on the scales measuring a MAP and a MAV GO respectively, whilst the items loading onto factor 2 correspond to items on the PAP and PAV scales. In other words the PAP and PAV items combined to form one factor instead of the MAP and MAV items combining.

The fact that the EFA results for General and Hypothetical Task GOs indicate different 3-factor models (one having mastery, PAP, and PAV GOs and the other having performance, MAP, and MAV GOs) is quite puzzling. It is extremely unlikely that these differences are a result of the measures being adapted because item adaptations were kept to a bare minimum.

Since it is difficult to draw conclusions regarding the model structure from the EFA results Confirmatory Factor Analyses (CFA) were carried out in order to further test the measurement model and determine which model provides the best fit to the data. CFAs were also carried out using the General and Hypothetical Task GO data at Time 1. Since the EFAs indicated two different 3-factor models (one having mastery, PAP, and

PAV GOs and the other having performance, MAP, and MAV GOs) these two models were tested by means of a CFA and compared in order to assess which model provides the best fit. Moreover, since the EFAs indicated that for the different GOs (General and Hypothetical Task GOs) four factors emerged overall (MAP, MAV, PAP and PAV) the four-factor model was also tested and its model fit compared with that of the 3-factor models. The CFAs were carried out using LISREL 8.80. The analyses were carried out on covariance matrices using maximum-likelihood estimation. The results are presented in Tables 8.21 and 8.22 for General GOs and Hypothetical Task GOs, respectively.

In order to test the overall model fit a number of goodness-of-fit indices were used. These include the chi-square statistic (χ^2), Root Mean Square Error of Approximation (RMSEA), comparative fit index (CFI), non-normed fit index (NNFI), and the standardised root mean square residual (SRMR). The χ^2 statistic was included since it is a "traditional measure for evaluating overall model fit in covariance structure models" (Diamantopoulos & Siguaw, 2000:83) and it is convention to report this statistic. In order to show good model fit, the χ^2 probability value should be greater than or equal to 0.05. However, since the χ^2 tests *perfect* fit (Diamantopoulos & Siguaw, 2000) and is sensitive to sample size (Jöreskog, 1969), it is highly unlikely to obtain a significant result with sample sizes greater than 200 (Reinard, 2006). The sample sizes in the CFAs presented below are both over 600. Consequently, it is extremely unlikely for the χ^2 to indicate good model fit. Therefore in interpreting the results presented below, not much weight will be put on the χ^2 value but rather the other goodness-of-fit statistics will be relied on more heavily in assessing model fit. Hu and Bentler (1999) recommended the following cut-off values for the goodness-of-fit indices when carrying out CFA: RMSEA value lower than 0.06, CFI and NNFI values greater than 0.95 and SRMR value lower than 0.09. These cut-off values will be used when interpreting the CFA results below.

When all the scale items were used in the General GOs CFA, the results did not show good model fit. Since the exploratory factor analyses results indicated that Items 1 and 8 seemed to be problematic, these were removed and the CFAs were repeated without these two problem items. The results indicated a slightly better model fit when Items 1 and 8 were removed. Out of the three models tested, the 2x2 model showed the best fit to the data (RMSEA=0.07, CFI=0.97, NNFI=0.95, SRMR=0.04). However, due to the RMSEA value being slightly higher than the recommended cut-off value, this was still not an excellent fit.

With respect to the Hypothetical Task GOs, when all the scale items were included in the CFA none of the models fit the data adequately (refer to Table 8.22). Similarly to the General GOs, the problem items identified from the exploratory factor analyses (Items 10 and 11) were removed and the analyses were repeated. The results of the CFA with Items 10 and 11 excluded from the analyses indicated a much better fit to the data. Again, out of the three models tested, the 2x2 model showed the best fit to the data (RMSEA=0.07, CFI=0.98, NNFI=0.96, SRMR=0.05). However, similarly to the CFA for General GOs the fit was not excellent (again due to the RMSEA value being slightly higher than the recommended cut-off value). The CFA validity results for the 2x2 models of General and Hypothetical Task GOs (with the problem items removed) are provided in Tables 8.23 and 8.25, respectively, whilst the reliability results (R²) are provided in Tables 8.24 and 8.26, respectively.

Items in Scales	Model	χ²	df	RMSEA	CFI	NNFI	Standardised RMR
	4-factor model	251.67**	48	0.00	0.95	0.93	0.06
Scales including all items	3-factor model (Mastery, PAP, PAV)	319.56**	51	0.10	0.94	0.92	0.07
	3-factor model (MAP, MAV, Performance)	425.35**	51	0.12	0.91	0.88	0.08
Scales excluding	4-factor model	115.26**	29	0.07	0.97	0.95	0.04
problem items (i.e. items 1 and 8)	3-factor model (Mastery, PAP, PAV)	184.61**	32	0.09	0.95	0.93	0.06
	3-factor model (MAP, MAV, Performance)	256.93**	32	0.11	0.92	0.89	0.06

Table 8.21. Confirmatory Factor Analyses Results for General Goal Orientations at Time 1 (N=621)

Key: ** p<0.01

Table 8.22. Confirmatory Factor Analyses Results for Hypothetical Task Goal Orientations at Time 1 (N=612)

Items in Scales	Model	χ²	df	RMSEA	CFI	NNFI	Standardised RMR
	4-factor model	454.48**	48	0.12	0.93	0.90	0.11
Scales including all items	3-factor model (Mastery, PAP, PAV)	597.59**	51	0.13	0.90	0.87	0.11
	3-factor model (MAP, MAV, Performance)	563.72**	51	0.13	0.91	0.88	0.12
Scales excluding	4-factor model	121.90**	29	0.07	0.98	0.96	0.05
problem items (i.e. items 10 and 11)	3-factor model (Mastery, PAP, PAV)	349.74**	32	0.13	0.92	0.89	0.08
	3-factor model (MAP, MAV, Performance)	204.69**	32	0.10	0.96	0.94	0.06

Key: ** p<0.01

GO Scale Item is Assessing	Item Number	МАР	MAV	РАР	PAV
РАР	2			0.66	
МАР	3	0.66			
ΡΑΡ	4			0.87	
MAV	5		0.65		
PAV	6				0.69
МАР	7	0.73			
MAV	9		0.75		
PAV	10				0.84
MAV	11		0.62		
PAV	12				0.82

Table 8.23. CFA Validity Results for the 4-factor model of General Goal Orientations at Time 1

Table 8.24. CFA Reliability Results for the 4-factor model of General Goal Orientations at Time 1

GO Scale item is measuring	Item Number	R ²
РАР	2	0.44
МАР	3	0.43
РАР	4	0.76
MAV	5	0.42
PAV	6	0.48
МАР	7	0.53
MAV	9	0.56
PAV	10	0.71
MAV	11	0.39
PAV	12	0.67

Table 8.25. CFA Validity Results for the 4-factor model of Hypothetical Task Goal Orientations at Time1

GO Scale Item is Assessing	Item Number	МАР	MAV	РАР	PAV
МАР	1	0.59			
РАР	2			0.82	
МАР	3	0.81			
РАР	4			0.89	
MAV	5		0.83		
PAV	6				0.70
МАР	7	0.76			
РАР	8			0.84	
MAV	9		0.80		
PAV	12				0.83

Table 8.26. CFA Reliability Results for the 4-factor model of Hypothetical Task Goal Orientations at Time 1

GO Scale item is measuring	ltem Number	R ²
МАР	1	0.35
ΡΑΡ	2	0.67
МАР	3	0.66
ΡΑΡ	4	0.79
ΜΑν	5	0.68
PAV	6	0.49
МАР	7	0.57
ΡΑΡ	8	0.70
ΜΑν	9	0.64
PAV	12	0.70

The validity of the indicators is assessed in terms of the magnitude and significance of the paths between each latent variable and its indicators (Diamantopoulos & Siguaw,

2000). When using the completely standardised solution, factor loadings of 0.5 and higher show that the indicators are loading significantly onto the latent variable. The CFA validity results for the General GOs at Time 1 (Table 8.23) and for the Hypothetical Task GOs at Time 1 (Table 8.25) show that all the indicator loadings are significant thus indicating their validity.

The squared multiple correlations (R^2) provide an indication of the reliability of the indicators. A high R^2 is an indication of high reliability (Diamantopoulos & Siguaw, 2000). The reliability results for the General GOs at Time 1 (Table 8.24) do not indicate excellent reliabilities, particularly for Item 11. Item 11 was one of the problem items for the Hypothetical Task GO factor analysis. Therefore, there might possibly be an issue with this item. In contrast to the reliability results for the General GOs the General GOs, those for the Hypothetical Task GOs look much better with the exception of item 1. Since item 1 was one of the problem items for the General GOs this item may also be problematic.

Following the reliability analyses and factor analyses, it was necessary to make a number of decisions regarding which model of GOs to use in the current study and which items to keep on the scales measuring the GOs. Deciding on which items to keep on the GO scales was quite a difficult decision due to the fact that in some instances Items 1, 8, 10 and 11 seemed to be problematic but not in others. As discussed earlier, removal of these items would negatively affect the reliability of the scales measuring other GOs (e.g. Verbal and Numerical Test GOs). Since it is not appropriate to use different scales for measuring the different types of GOs (e.g. General and Verbal Test GOs), based on the results of the reliability analyses, and the principle of keeping as many of the items as possible (without having large negative consequences on the scale reliabilities) a decision was made to keep all the items on the four scales measuring GOs.

With respect to model choice the decision was also not that straightforward. Although both CFAs indicated the best model fit for the four-factor (2x2) model of GOs, the model fit for this was not outstanding. However, since the 2x2 model provided better fit than the 3-factor models, this model was chosen for the current study for the analyses of GOs from the *non-profile* perspective. With respect to the analyses using GO *profiles*, a decision as to whether to use the 3-factor or 2x2 model of GOs will be made in the next section.

8.5. Latent Class Analyses

In the previous section a decision was made regarding which model of GOs to use for the non-profile perspective analyses. Consequently, it is timely now to present the LCA results, indicating which model of GOs should be used in the GO profile analyses, at this point in the chapter. However, before presenting the LCA results a brief description of LCA and its advantages over traditional cluster analyses is presented. A description of the criteria used in choosing the final model is also provided.

"Latent class analysis provides a powerful, flexible approach to the analysis of categorically-scored data." (McCutcheon & Hagenaars, 1997:266). With LCA the investigator can use a number of observed variables, which represent the characteristics of people, in order to organise people into two or more meaningful groups (Collins & Lanza, 2010). LCA is used to organise participants into homogenous groups (referred to as latent classes) depending on the types of latent variables that they adopt (Collins & Lanza, 2010). In this study, LCA will be used to organise participants adopting similar profiles of GOs into groups or clusters (latent classes). Therefore, the goal of LCA is similar to that of cluster analysis. However, unlike traditional cluster analysis, LCA is model-based "This means that a statistical model is postulated for the population from which the sample under study is coming. More precisely, it is assumed that the data is generated by a mixture of underlying probability distributions." (Vermunt & Magidson, 2002:90). Consequently, it is less subjective than traditional cluster analysis. Moreover, with LCA it is relatively easy to deal with variables having different scale types and there are more formal or rigorous criteria one can use in order to make decisions about one's final model (Pastor et al., 2007). As a result of LCA having a number of advantages over standard cluster analysis, the former was used in the current research study. Prior to presenting the

173

LCA results of General GOs, a description of the criteria used in choosing the final model in the LCAs is presented.

As mentioned earlier there are more formal criteria in place for choosing one's final model when using LCA than traditional cluster analysis. These include the likelihood ratio chi-squared statistic L², the Akaike information criterion (AIC) and the Bayesian information criterion (BIC). The model L² indicates the amount of association among the variables that remains unexplained after estimating the model. Therefore, the lower the L² value, the better the model fit (Vermunt & Magidson, 2005). A model fits the data if the value of L² is sufficiently low to be attributable to chance with the normal statistical error limit of 0.05 (Magidson & Vermunt, 2004). Therefore, the model that tends to be chosen is the one with a p-value greater than 0.05. Additionally, it is ideal to choose the most parsimonious model, that is, the model with the fewest number of parameters (Vermunt & Magidson, 2005). The 'number of parameters' refers to (a) the sizes of the latent classes *and* (b) the probability of an individual in a particular latent class scoring high on an item. Consequently, the model with a p-value greater than 0.05 and having the lowest number of parameters is normally the model of choice.

The AIC and BIC are similar to the L² in that they compare the expected cell frequency count (which is provided by the estimated model) with the actual (observed) cell frequency count found in the sample data. However, unlike the L², the AIC and BIC take the number of parameters into consideration. More parameters yield a greater likelihood ratio. Consequently, the information criteria (AIC and BIC) penalise the likelihood ratio by reducing it as a function of the increased number of estimated parameters. A smaller value of the information criteria indicates a better balance of model fit and parsimony. Consequently, models having the lowest BIC and AIC tend to indicate a good fit of the data.

When the number of observed variables or the number of variable categories is large the data are usually sparse. This means that the total number of cells in the resulting frequency table will be large relative to the sample size which results in many empty cells (Magidson & Vermunt, 2004). In such cases it is not wise to use the chi-square distribution to compute the p-value since L² would not be a good approximation (Magidson & Vermunt, 2004). In the case of sparse data it is best to use the bootstrap procedure which provides a more precise estimate of the models by relaxing the assumption that the L² statistic follows a chi-square distribution (Vermunt & Magidson, 2005). This is done by means of 'resampling'. "Resampling methods aim at finding the distribution of a statistic by repeatedly drawing a sample, making use of the original sample" (Langeheine, Pannekoek and Van de Pol, 1996:493). Moreover, in such a case the BIC and AIC tend to become extremely useful in comparing models (Magidson & Vermunt, 2004). It is important to note that the AIC and BIC do not always indicate the same model as providing the best fit since they are computed differently (Collins & Lanza, 2010). The BIC is more widely used in LCA than the AIC (Magidson & Vermunt, 2004). However, these criteria are more useful in eliminating models and narrowing down the options rather than for making a final decision regarding model choice (Collins & Lanza, 2010). In the LCAs conducted during this study the BIC was used as an indicator. However, the final decisions regarding model choice were based on the number of parameters and the bootstrap p-value. The latter was used since in most cases the sample size was quite small and consequently the data were sparse.

In order to make a decision regarding model choice with respect to GO profiles, LCAs were carried out using both the 3-factor and 2x2 models of GOs. When the 2x2 model was used none of the resulting LCA models were found to provide adequate fit to the data for the General GOs at Time 1 or for the Numerical Test GOs at Time 1. Consequently, a decision was made to use the 3-factor model of GOs in all analyses involving GO profiles. This decision was made due to the fact that when using the bootstrap procedure, which has been described as "perhaps the best statistical criterion for determining the number of latent classes" (Uebersax, 2000) none of the models provided adequate fit to the data for General and Numerical Test GOs at Time 1 when using the 2x2 model of GOs. On the other hand, adequate model fit was obtained when using the 3-factor model of GOs. Therefore, the LCA results presented are those for the 3-factor model of GOs (consisting of MAP, PAP and PAV GOs).

The LCA results obtained for the General, Hypothetical Task, Verbal Test and Numerical Test GOs at Time 1 and Time 2 are presented in Sections 8.5.1., 8.5.2., 8.5.3., 8.5.4., respectively, below. The criteria described above were used in making decisions regarding LCA model choice.

8.5.1. LCA of General Goal Orientations at Time 1 and Time 2

The LCA results of the General GOs at Time 1 provide the answer to Research Question 1. The focus of this Research Question was on determining the type and number of GO profiles obtained when using LCA as a method of clustering as well as examining whether the 2x2 model significantly improved on the 3-factor model in terms of identifying goal orientation profiles. This Research Question will be addressed following the presentation of the LCA results. Additionally, although the LCAs partially answer a number of other Research Questions (e.g. Research Question 2a and Research Question 4b), a decision was made to address these in the next chapter since a substantial number of additional data analyses are required in order to provide the full answers to these Research Questions. Moreover, these Research Questions (which are partially answered by the LCAs presented in this chapter) have similar themes to a considerable number of other Research Questions which are not answered by the LCAs provided below. Thus, it was thought wise to address these Research Questions in the next chapter.

The LCA results for General GOs at Time 1 are presented first (Tables 8.27., 8.28. and Figure 8.1.) and are followed by the LCA results for the General GOs at Time 2. The data used in these analyses were collected at Time Q1 by means of the General GOs measure. As discussed in Section 8.2. it was assumed that the 3-factor model consisted of a MAP, PAP and PAV GOs since the MGO scale of the 3-factor model consists of MAP GO items only (as opposed to MAP **and** MAV GO items). Consequently, the MAP, PAP and PAV GOs were included in the LCA for the 3-factor model.

Model	Number of Clusters	ш	BIC	Npar	L²	df	p- value	Bootstrap p-value	Class. Err.
1	1	-1158.89	2356.48	6	172.81	20	<.01	<.01	<.01
2	2	-1102.65	2269.79	10	60.33	16	<.01	<.01	0.08
3	3	-1093.38	2277.04	14	41.78	12	<.01	<.01	0.09
4	4	-1081.57	2279.21	18	18.16	8	.02	.07	0.11
5	5	-1080.02	2301.91	22	15.06	4	<.01	.05	0.22

Table 8.27. LCA of General GOs at Time 1 (N=632)

In this case the BIC shows model 2 as providing the best fit to the data. The p-values indicate that none of the models provide an adequate fit since they all have p-values of less than 0.05. However, when the bootstrap procedure was used the bootstrap p-value points towards Model 4 as providing the best fit. This is due to the fact that Model 4 is the model with the lowest number of parameters and a bootstrap p-value of more than 0.05. The results presented in Table 8.28. and Figure 8.1. are those obtained for Model 4.

	Г	1		
	Cluster 1	Cluster 2	Cluster 3	Cluster 4
Cluster Size	67% (423)	29% (181)	2% (14)	2% (14)
GENMAP				
Low	<.01	.02	<.01	.53
Moderate	.15	.32	<.01	.42
High	.85	.66	.99	.04
GENPAP				
Low	<.01	<.01	.75	.85
Moderate	<.01	.60	.25	.15
High	1.00	.39	<.01	<.01
GENPAV				
Low	<.01	.08	.87	.13
Moderate	.09	.39	.12	.44
High	.91	.54	<.01	.44

Table 8.28. Conditional Probabilities of General GOs with Model 4

Key: GENMAP = General Mastery-Approach Goal Orientation; GENPAP = General Performance-Approach Goal Orientation; GENPAV = General Performance-Avoidance Goal Orientation; The numbers in brackets indicate the number of participants in each cluster.

Figure 8.1. Profile Plot for Model 4



Key: GENMAPPR = General Mastery-Approach Goal Orientation; GENPAPPR = General Performance-Approach Goal Orientation; GENPAVR = General Performance-Avoidance Goal Orientation; Cluster 1 = High Goal Orientations Cluster; Cluster 2 = High Mastery, Moderate Performance Cluster; Cluster 3 = High Mastery, Low Performance Cluster; Cluster 4 = High Performance-Avoidance Cluster.

Table 8.28. shows the conditional probabilities of individuals in each Cluster being Low, Moderate, or High, on each of the three GOs. For example, the table shows that the probability of a participant in Cluster 1 having a low General MAP GO is less than 1%. However, the probability of them having a high General MAP GO is 85%. Figure 8.1. presents the GO profiles of individuals in each Cluster. Therefore, individuals in Cluster 1 are likely to be high on all three GOs. This cluster will be referred to as 'High GOs' from this point onwards. Individuals within this cluster are likely to have very slightly lower MAP and PAV GOs than PAP GOs. As indicated in Table 8.28., 67% of the participants could be classified as being in Cluster 1. Cluster 2 contained 29% of the participants. Participants in this cluster are also likely to be high on all three GOs. However, their GOs are not as high as those of participants in Cluster 1 and unlike participants in Cluster 1 they are likely to have slightly higher MAP GOs than PAP and PAV GOs. This cluster will be referred to as 'High Mastery, Moderate Performance' from this point onwards. 2% of the sample were classified as being in Cluster 3. Participants in this cluster are likely to have high MAP and low PAP and PAV GOs. Consequently, this cluster will be referred to as 'High Mastery, Low Performance'. Finally, another 2% of participants were in Cluster 4. Individuals in Cluster 4 are likely to have low MAP and PAP GOs and a high PAV GO. This cluster will be referred to as the 'High Performance-Avoidance' cluster. The LCA results for the General GOs at Time 2 are presented next.

Model	Number of Clusters	LL	BIC	Npar	L²	df	p- value	Bootstrap p-value	Class. Err.
1	1	-393.59	818.54	6	113.86	20	<.01	<.01	<.01
2	2	-354.87	761.99	10	36.41	16	<.01	<.01	.06
3	3	-341.79	756.75	14	10.27	12	.59	.68	.06
4	4	-340.72	775.51	18	8.13	8	.42	.42	.15
5	5	-340.61	796.18	22	7.89	4	.10	.25	.25

Table 8.29. LCA of General GOs at Time 2 using the 3-factor model of GOs (N=192)

The BIC in Table 8.29. indicates that Model 3 provides the best fit to the data, since this has the lowest value. This time the p-value and the bootstrap p-value also point towards Model 3 as being the best fitting model since it has the lowest number of parameters out of all the models having a p-value (and bootstrap p-value) greater than 0.05. Consequently, the results in Table 8.30. and Figure 8.2. are those obtained for Model 3.

	Cluster 1	Cluster 2	Cluster 3
Cluster Size	60% (115)	35% (67)	5% (10)
GENMAP			
Low	<.01	.02	.20
Moderate	.16	.29	.54
High	.84	.69	.26
GENPAP			
Low	<.01	<.01	.84
Moderate	.01	.62	.16
High	.99	.38	<.01
GENPAV			
Low	<.01	.07	.73
Moderate	Moderate .01		.26
High	.99	.32	<.01

Table 8.30. Conditional Probabilities of General GOs at Time 2 with Model 3

Key: GENMAP = General Mastery-Approach Goal Orientation; GENPAP = General Performance-Approach Goal Orientation; GENPAV = General Performance-Avoidance Goal Orientation; The numbers in brackets indicate the number of participants in each cluster.



Figure 8.2. Profile Plot for Model 3

Key: GENMAPP2R = General Mastery-Approach Goal Orientation at Time 2; GENPAPP2R = General Performance-Approach Goal Orientation at Time 2; GENPAV2R = General Performance-Avoidance Goal Orientation at Time 2; Cluster 1= High Goal Orientations cluster; Cluster 2 = High Mastery, Moderate Performance Cluster; Cluster 3 = High Mastery, Low Performance Cluster.

The results presented in Table 8.30. and Figure 8.2. indicate that the three General GO profiles obtained at Time 2 are nearly identical to three of the four General GO profiles obtained at Time 1. It is possible that a fourth GO profile was not identified at Time 2 due to the much smaller sample size. Clusters 1, 2, and 3 presented in Table 8.30. and Figure 8.2. are very similar to the 'High GOs'; 'High Mastery, Moderate Performance'; and 'High Mastery, Low Performance' clusters, respectively, obtained for General GOs at Time 1.

Model	Number of Clusters	ш	BIC	Npar	L²	df	p- value	Bootstrap p-value	Class. Err.
1	1	-1245.94	2530.48	6	182.42	20	<.01	<.01	<.01
2	2	-1168.89	2402.14	10	28.34	16	.03	.04	.03
3	3	-1165.79	2421.66	14	22.12	12	.04	.04	.08
4	4	-1160.51	2436.84	18	11.57	8	.17	.39	.10
5	5	-1159.43	2460.41	22	9.40	4	.05	.31	.18

Table 8.31. LCA of Hypothetical Task GOs at Time 1 (N=623)

The data collected at Time Q1 were used in this LCA. The results presented in Table 8.31. indicate that Model 2 has the lowest BIC value and therefore seems to provide the best fit to the data. However, its p-value is less than 0.05 thus showing inadequate fit. The p-value and bootstrap p-value both indicate that Model 4 shows the best fit to the data since it has a p-value (and bootstrap p-value) greater than 0.05 and the lowest number of parameters. Consequently, Model 4 was chosen as the best-fitting model. Table 8.32. and Figure 8.3. indicate the results obtained for Model 4.

	Cluster 1	Cluster 2	Cluster 3	Cluster 4
Cluster Size	73% (452)	18% (110)	8% (49)	2% (12)
ΗΥΡΜΑΡ				
Low	<.01	.07	<.01	.78
Moderate	.15	.43	<.01	.21
High	.84	.50	.99	<.01
НҮРРАР				
Low	<.01	.10	.27	.46
Moderate	.18	.48	.54	.46
High	.81	.42	.20	.08
ΗΥΡΡΑΥ				
Low	<.01	<.01	.28	.27
Moderate	<.01	.75	.72	.72
High	1.00	.24	<.01	<.01

 Table 8.32. Conditional Probabilities of Hypothetical Task GOs with Model 4

Key: HYPMAP = Hypothetical Task Mastery-approach Goal Orientation; HYPPAP= Hypothetical Task Performance-approach Goal Orientation; HYPPAV= Hypothetical Task Performance-avoidance Goal Orientation; The numbers in brackets indicate the number of participants in each cluster.

The results presented in Table 8.32. show that Clusters 1, 2, 3, and 4 made up 73%, 18%, 8% and 2% of the sample respectively. The profiles obtained from the LCA

indicate that the four Hypothetical Task GOs clusters at Time 1 are quite similar to the General GO clusters at Time 1 (when using the 3-factor model). Hypothetical Task GO Cluster 1 is very similar to the General GOs 'High GOs' cluster. However, the Hypothetical Task PAV GO is slightly higher in this cluster than in the 'High GOs' General GOs cluster. Moreover, in this case the PAP GO is slightly lower than the PAP GO obtained in the General GOs 'High GOs' cluster. Cluster 2 is extremely similar to the 'High Mastery, Moderate Performance' General GOs Cluster whilst Cluster 3 is similar to the 'High Mastery, Low Performance' General GOs cluster. However, in Cluster 3 the PAP and PAV GOs are slightly higher than those in the corresponding General GOs cluster. Cluster 4 is quite similar to the General GOs 'High Performance-Avoidance' cluster. However, the MAP and PAV GOs of participants in this cluster seem to be lower than those of participants in the corresponding General GOs cluster whilst the PAP GO seems to be higher than that in the corresponding General GOs cluster. On the whole though, the Hypothetical Task GO profiles obtained are quite similar to the General GO profiles at Time 1 obtained using the 3-factor model. This comparison of General and Hypothetical Task GOs will be discussed in further detail in Section 9.2.3.



Figure 8.3. Profile Plot for Model 4

Key: TSPMAPPR = Hypothetical Task Mastery-Approach Goal Orientation; TSPPAPPR = Hypothetical Task Performance-Approach Goal Orientation; TSPPAVR=Hypothetical Task Performance-Avoidance Goal Orientation; Cluster 1=High Goal Orientations Cluster; Cluster 2=High Mastery, Moderate Performance Cluster; Cluster 3=High Mastery, Low Performance Cluster; Cluster 4=High Performance-Avoidance Cluster.

The data collected at Time Q2 were used for the LCA of Hypothetical Task GOs at Time 2. The results are presented in Tables 8.33. and 8.34. as well as Figure 8.4.

Model	Number of Clusters	ш	BIC	Npar	L²	df	p- value	Bootstrap p-value	Class . Err.
1	1	-441.74	914.93	6	121.04	20	<.01	<.01	<.01
2	2	-395.56	843.53	10	28.67	16	.03	.02	.04
3	3	-390.36	854.10	14	18.27	12	.11	.12	.04
4	4	-388.02	870.40	18	13.60	8	.09	.09	.09
5	5	-385.71	886.75	22	8.98	4	.06	.15	.12

Table 8.33. LCA of Hypothetical Task GOs at Time 2 (N=189)

The BIC points to Model 2 as providing the best fit to the data. However, the p-value shows Model 3 as providing the best fit. When the bootstrap procedure was carried out, Model 3 was again shown to provide the best fit to the data since it had the lowest number of parameters out of the models with bootstrap p-value greater than 0.05. Consequently, Model 3 was chosen as the best-fitting model. The results presented in Table 8.34. and Figure 8.4. are those obtained for Model 3.

	Cluster 1	Cluster 2	Cluster 3
Cluster Size	67% (127)	25% (47)	8% (15)
НҮРМАР			
Low	<.01	<.01	.42
Moderate	.15	.27	.53
High	.85	.72	.05
НҮРРАР			
Low	<.01	.16	.62
Moderate	.19	.55	.35
High	.80	.29	.03
HYPPAV			
Low	<.01	.11	.43
Moderate	Moderate <.01		.56
High	.99	.06	<.01

Table 8.34. Conditional Probabilities of Hypothetical Task GOs with Model 3

Key: HYPMAP =Hypothetical Task Mastery-approach Goal Orientation; HYPPAP= Hypothetical Task Performance-approach Goal Orientation; HYPPAV= Hypothetical Task Performance-avoidance Goal Orientation; The numbers in brackets indicate the number of participants in each cluster.

The conditional probabilities show that 67%, 25% and 8% of the sample had GO profiles described by Clusters 1, 2, and 3 respectively. Clusters 1 and 2 are very similar to the 'High GOs' and 'High Mastery, Moderate Performance' GO profiles, respectively, obtained for Time 1 Hypothetical Task GOs. Cluster 3 is quite similar to the 'High Performance-Avoidance' cluster for Hypothetical Task GOs at Time 1. However, at Time 2, participants in this cluster seem to have higher MAP GOs than in the corresponding Hypothetical Task GO cluster at Time 1. Moreover, the PAV GO of participants in this cluster is lower at Time 2 than at Time 1. One other difference between the Hypothetical Task GO profiles at Time 1 and Time 2 is that one of the Clusters obtained at Time 1 (Cluster 3) is missing at Time 2. This indicates that participants' GO profiles seem to be changing over time. These changes will be examined more closely in Chapter 9 Section 9.1.2.1. The LCA results for Verbal Test GOs at Time 1 and Time 2 are presented next.





Key: TSPMAP2R = Hypothetical Task Mastery-Approach Goal Orientation at Time 2; TSPPAP2R = Hypothetical Task Performance-Approach Goal Orientation at Time 2; TSPPAV2R = Hypothetical Task

Performance-Avoidance Goal Orientation at Time 2; Cluster 1 = High Goal Orientations Cluster; Cluster 2 = High Mastery, Moderate Performance Cluster; Cluster 3 = High Performance-Avoidance Cluster.

8.5.3. LCA of Verbal Test Goal Orientations at Time 1 and Time 2

The data collected at Time E1 and Time E2 were used for the LCAs of Verbal Test GOs at Time 1 and Time 2. The results of these LCAs are presented in Tables 8.35. through to 8.38. and Figures 8.5. and 8.6.

Model	Number of Clusters	ш	BIC	Npar	L²	df	p- value	Bootstrap p-value	Class. Err.
1	1	-188.83	403.24	6	83.46	20	<.01	<.01	<.01
2	2	-159.45	361.53	10	24.70	16	.08	.01	.04
3	3	152.20	364.08	14	10.20	12	.60	.20	.08
4	4	-149.61	375.96	18	5.02	8	.76	.49	.06
5	5	-149.27	392.31	22	4.33	4	.36	.46	.12

Table 8.35. LCA of Verbal Test GOs at Time 1 (N=71)

The BIC and p-values presented in Table 8.35. show Model 2 as providing the best fit to the data since it has the lowest BIC value and a p-value greater than 0.05 with the least number of parameters. However, when the bootstrap procedure was carried out this indicated Model 3 as providing the best fit to the data (since it is the model with a bootstrap p-value greater than 0.05 and the least number of parameters. The small sample size makes it even more important to use the bootstrap procedure in this case. Consequently, Model 3 was chosen as the best-fitting model. The results presented in Table 8.36. and Figure 8.5. are those obtained for Model 3.

	Cluster 1	Cluster 2	Cluster 3	
Cluster Size	48% (34)	29% (21)	23% (16)	
VTMAP				
Low	<.01	.09	.01	
Moderate	.09	.52	.27	
High	.91	.39	.72	
VTPAP				
Low	<.01	.02	.54	
Moderate	.05	.87	.46	
High	.95	.11	<.01	
VTPAV				
Low	<.01	.02	.72	
Moderate	Moderate .05		.28	
High	.95	.26	<.01	

Table 8.36. Conditional Probabilities of Verbal Test GOs with Model 3

Key: VTMAP = Verbal Test Mastery-approach Goal Orientation; VTPAP= Verbal Test Performanceapproach Goal Orientation; VTPAV= Verbal Test Performance-avoidance Goal Orientation; The numbers in brackets indicate the number of participants in each cluster.

The results indicate that Clusters 1, 2 and 3 are made up of 48%, 29% and 23% of the sample, respectively. Clusters 1, 2 and 3 are nearly identical to the 'High GOs', 'High Mastery, Moderate Performance', and 'High Mastery, Low Performance' clusters, respectively, obtained for General GOs at Time 1. The LCA of Verbal Test GOs at Time 2 are presented next.





Key: T1VTMAPPR = Time 1 Verbal Test Mastery-Approach Goal Orientation; T1VTPAPPR = Time 1 Verbal Test Performance-Approach Goal Orientation; T1VTPAVR = Time 1 Verbal Test Performance-Avoidance Goal Orientation; Cluster 1 = High Goal Orientations Cluster; Cluster 2 = High Mastery, Moderate Performance Cluster; Cluster 3 = High Mastery, Low Performance Cluster.

Model	Number of Clusters	ш	BIC	Npar	L²	df	p- value	Bootstrap p-value	Class. Err.
1	1	-184.16	393.64	6	88.46	20	<.01	<.01	<.01
2	2	-153.05	348.29	10	26.23	16	.05	.01	.01
3	3	-144.36	347.79	14	8.86	12	.72	.50	.05
4	4	-143.16	362.27	18	6.46	8	.60	.30	.14
5	5	-143.07	378.97	22	6.28	4	.18	.22	.25

Table 8.37. LCA of Verbal Test GOs at Time 2 (N=68)

The p-value obtained from this analysis shows Model 2 as providing the best fit to the data since it has a p-value of 0.05 and the least number of parameters. However, the BIC and bootstrap p-value indicate otherwise. According to these, Model 3 provides the best fit to the data since it has the lowest BIC value, and a bootstrap p-value greater than 0.05 with the least number of parameters. Similarly to the LCA for Verbal Test GOs at Time 1, the bootstrap procedure was necessary in this case due to the small sample size. As a result, Model 3 was chosen as the best-fitting model and the results presented in Table 8.38. and Figure 8.6. below are those obtained for Model 3.

	Cluster 1	Cluster 2	Cluster 3
Cluster Size	52% (35)	29% (21)	18%(12)
VTMAP			
Low	.01	.02	.09
Moderate	.28	.33	.52
High	.71	.65	.39
VTPAP			
Low	<.01	.02	.84
Moderate	.06	.95	.16
High	.94	.03	<.01
VTPAV			
Low	<.01	.06	.79
Moderate	<.01	.66	.21
High	.99	.28	<.01

 Table 8.38. Conditional Probabilities of Verbal Test GOs with Model 3

Key: VTMAP = Verbal Test Mastery-approach Goal Orientation; VTPAP= Verbal Test Performanceapproach Goal Orientation; VTPAV= Verbal Test Performance-avoidance Goal Orientation; The numbers in brackets indicate the number of participants in each cluster.

This time Clusters 1, 2 and 3 were made up of 52%, 29% and 18% of the sample respectively. The three GO profiles obtained (Clusters 1, 2 and 3) are very similar to the 'High GOs', 'High Mastery, Moderate Performance', and 'High Mastery, Low Performance' clusters obtained for General GOs at Time 1. They are also very similar to the three clusters obtained for Verbal Test GOs at Time 1. The LCA results for Numerical Test GOs at Time 1 and Time 2 are presented next.



Figure 8.6. Profile Plot for Model 3

Key: T2VTMAPPR = Time 2 Verbal Test Mastery-Approach Goal Orientation; T2VTPAPPR = Time 2 Verbal Test Performance-Approach Goal Orientation; T2VTPAVR = Time 2 Verbal Test Performance-Avoidance Goal Orientation; Cluster 1 = High Goal Orientations Cluster; Cluster 2 = High Mastery, Moderate Performance Cluster; Cluster 3 = High Mastery, Low Performance Cluster.

8.5.4. LCA of Numerical Test Goal Orientations at Time 1 and Time 2

The data collected at Time E1 and Time E2 were used in the following LCAs. These analyses were carried out in order to examine the types of GO profiles adopted by participants on the Numerical Test provided to them at Time 1 and at Time 2.

Model	Number of Clusters	LL	BIC	Npar	L²	df	p- value	Bootstrap p-value	Class. Err.
1	1	-198.51	422.60	6	81.80	20	<.01	<.01	<.01
2	2	-172.33	387.28	10	29.44	16	.02	.01	.03
3	3	-167.26	394.19	14	19.30	12	.08	.03	.07
4	4	-163.69	404.11	18	12.16	8	.14	.13	.05
5	5	-160.41	414.59	22	5.59	4	.23	.51	.05

Table 8.39. LCA of Numerical Test GOs Time 1 (N=71)

In the case of Numerical Test GOs at Time 1 the BIC and p-value show Model 2 and Model 3, respectively, as providing the best fit to the data. However, the bootstrap p-value shows Model 4 as providing the best fit because it has a bootstrap p-value greater than 0.05 and the least number of parameters. Therefore, this was chosen as the best-fitting Model. The conditional probabilities and profile plot for Model 4 are presented below (Table 8.40. and Figure 8.7.).

	Cluster 1	Cluster 2	Cluster 3	Cluster 4
Cluster Size	56% (40)	19% (14)	18% (13)	6 % (4)
NTMAP				
Low	<.01	.12	.04	.40
Moderate	.14	.63	.49	.55
High	.86	.24	.48	.06
NTPAP				
Low	<.01	.05	.80	.62
Moderate	.13	.83	.20	.37
High	.87	.12	<.01	<.01
NTPAV				
Low	<.01	.03	.80	<.01
Moderate	.09	.87	.20	.06
High	.91	.10	<.01	.94

Table 8.40. Conditional Probabilities of Numerical Test GOs with Model 4

Key: NTMAP = Numerical Test Mastery-approach Goal Orientation; NTPAP= Numerical Test Performance-approach Goal Orientation; NTPAV= Numerical Test Performance-avoidance Goal Orientation; The numbers in brackets indicate the number of participants in each cluster.



Figure 8.7. Profile Plot for Model 4

Key: T1NTMAPPR = Time 1 Numerical Test Mastery-Approach Goal Orientation; T1NTPAPPR = Time 1 Numerical Test Performance-Approach Goal Orientation; T1NTPAVR = Time 1 Numerical Test Performance-Avoidance Goal Orientation; Cluster 1 = High Goal Orientations Cluster; Cluster 2 = High Mastery, Moderate Performance Cluster; Cluster 3 = High Mastery, Low Performance Cluster; Cluster 4 = High Performance-Avoidance Cluster.

Clusters 1, 2, 3 and 4 were made up of 56%, 19%, 18% and 6% of the sample respectively. The GO profiles obtained here are very similar to the GO profiles obtained from the LCA of General GOs at Time 1. Clusters 1, 2, 3 and 4 are very similar to the 'High GOs', 'High Mastery, Moderate Performance', 'High Mastery, Low Performance' and 'High Performance-Avoidance' clusters (obtained for General GOs at Time 1), respectively. A more detailed comparison of the Numerical Test and General GO profiles will be presented in Chapter 9 Section 9.2.3. The LCA for Numerical Test GOs at Time 2 is presented next.

Model	Number of Clusters	ш	BIC	Npar	L²	df	p-value	Bootstrap p-value	Class. Err.
1	1	-185.39	396.36	6	97.97	20	<.01	<.01	<.01
2	2	-158.60	359.83	10	44.39	16	<.01	<.01	.03
3	3	-144.42	348.51	14	16.02	12	.19	.05	.04
4	4	-139.69	356.11	18	6.57	8	.58	.48	.05
5	5	-139.29	372.37	22	5.78	4	.22	.35	.18

Table 8.41. LCA of Numerical Test GOs at Time 2 (N=71)

In the case of Numerical Test GOs at Time 2, the BIC, p-value and bootstrap p-value all point towards Model 3 providing the best fit to the data. This model has the lowest BIC value and it is the model having a p-value and bootstrap p-value greater than 0.05 with the lowest number of parameters. The conditional probabilities and profile plot for Model 3 are presented below.

	Cluster 1	Cluster 2	Cluster 3
Cluster Size	54% (38)	33% (24)	13% (9)
NTMAP			
Low	<.01	.03	.11
Moderate	.22	.37	.54
High	.77	.61	.35
NTPAP			
Low	<.01	.05	.95
Moderate	.05	.83	.05
High	.95	.11	<.01
NTPAV			
Low	<.01	.01	.94
Moderate	.02	.72	.06
High	.98	.27	<.01

Table 8.42. Conditional Probabilities of Numerical Test GOs with Model 3

Key: NTMAP = Numerical Test Mastery-approach Goal Orientation; NTPAP= Numerical Test Performance-approach Goal Orientation; NTPAV= Numerical Test Performance-avoidance Goal Orientation; The numbers in brackets indicate the number of participants in each cluster.

Figure 8.8. Profile Plot for Model 3



Key: T2NTMAPPR = Time 2 Numerical Test Mastery-Approach Goal Orientation; T2NTPAPPR = Time 2 Numerical Test Performance-Approach Goal Orientation; T2NTPAVR = Time 2 Numerical Test Performance-Avoidance Goal Orientation; Cluster 1 = High Goal Orientations Cluster; Cluster 2 = High Mastery, Moderate Performance Cluster; Cluster 3 = High Mastery, Low Performance Cluster; Cluster 4 = High Performance-Avoidance Cluster.

The three clusters obtained from this LCA are very similar to three out of the four clusters obtained for General and Numerical Test GOs at Time 1. Clusters 1, 2, and 3 were made up of 54%, 33% and 13% of the sample, respectively. Due to the similarities with the General GO clusters, Clusters 1, 2 and 3 may also be referred to as 'High GOs', 'High Mastery, Moderate Performance' and 'High Mastery, Low Performance', respectively.

Since the analyses presented above provide an answer to Research Question 1, it was thought appropriate to discuss this Research Question next.

Research Question 1. Using LCA as a method of clustering goal orientations, how many different types of goal orientation profiles are there and what are the characteristics of each goal orientation profile? Does the 2x2 model of goal

orientations significantly improve on the 3-factor model in terms of identifying goal orientation profiles?

In answer to the Research Question 1, four main types of GO profiles emerged from the LCA of General GOs: 'High GOs', 'High Mastery, Moderate Performance', 'High Mastery, Low Performance' and 'High Performance-Avoidance' profiles. Moreover, the LCAs for the Hypothetical Task, Verbal and Numerical Test GO profiles obtained were very similar to the General GO profiles. Consequently, they support the results obtained for the types of General GO profiles. As discussed in Section 8.5., when the LCAs were carried out using the 2x2 and 3-factor models of GOs no adequate fit was found for the General and Numerical Test GOs at Time 1 when using the 2x2 model of GOs. On the other hand, when using the 3-factor model adequate fit was found for all the GOs at both Time 1 and Time 2. Consequently, it seems as though overall the 2x2 model of GOs does not necessarily improve on the 3-factor model in terms of GO profiles.

Thus far some of the psychometric properties of the scales and variables used in the current study were presented. Moreover, decisions were made as to whether the 3-factor or 2x2 models of GOs should be used (for the purposes of the present study) in analysing GOs from the profile and non-profile perspectives. The next section provides further information about the variables used in the current study by providing the correlational analyses results of all the variables used.

8.6. Intercorrelational Analyses

Three correlation matrices are presented in this section. The first one (Table 8.43.) provides the correlational analyses results for the variables in the survey study. The data used in this correlational analysis were collected at Time Q1 and Time Q2. Since all participants were included in this analysis the sample size varies from 187 to 631 (depending on whether participants completed the Time Q1 or Time Q2 questionnaire or both).

The second correlation matrix (Table 8.44.) provides the correlational analyses results for variables in the experimental study. Therefore, the data used for this analysis were collected at Time E1 and Time E2. Since only experimental participants were included in this analysis, the sample size varies from between 49 to 71.

The third correlation matrix (Table 8.44) indicates the correlational analysis results for variables across the survey and experiments for experimental participants only. This matrix was included since some of the correlations presented in it are used to answer a few of the research questions and hypotheses (e.g. Research Question 4c). Similarly to the second correlation matrix only experimental participants were included in this analysis therefore, the sample size varies from between 49 to 71 too.

Due to the large number of correlations tested there is a larger possibility of Type 1 error occurring. Corrections such as the Bonferroni control for Type 1 error. However, in doing so they increase the probability of Type II error (Perneger, 1998). In order to take Type I error into consideration without increasing the chances of Type II error a decision was made not to use a statistical correction but rather to discuss the possibility of Type I error whenever correlations are reported (in answer to any of the research questions and hypotheses).

	1	2	3	4	5	6	7	8	9	10	11	12	13	14	15	16
1. GEN MAP T1	1															
2. GEN MAV T1	.54**	1														
3. GEN PAP T1	.27**	.23**	1													
4. GEN PAV T1	.13**	.29**	.60**	1												
5. HYP MAP T1	.42**	.39**	.12**	.14**	1											
6. HYP MAV T1	.30**	.41**	.11**	.23**	.55**	1										
7. HYP PAP T1	.22**	.22**	.54**	.48**	.29**	.26**	1									
8. HYP PAV T1	.19 ^{**}	.32**	.41**	.52**	.37**	.52**	.62**	1								
9. GEN MAP T2	.56**	.34**	.19**	.08	.38**	.21**	.26**	.22**	1							
10. GEN MAV T2	.35**	.47**	.07	.09	.28**	.37**	.08	.28**	.59**	1						
11. GEN PAP T2	.08	.22**	.54**	.46**	.23**	.21**	.49**	.41**	.35**	.25**	1					
12. GEN PAV T2	<01	.21**	.38**	.47**	.14	.28**	.37**	.42**	.25**	.38**	.69**	1				
13. HYP MAP T2	.40**	.41**	.11	.14	.57**	.35**	.27**	.30**	.54**	.50**	.27**	.28**	1			
14. HYP MAV T2	.15 [*]	.29**	.07	.07	.36**	.54**	.11	.25**	.27**	.55**	.19 [*]	.38**	.51**	1		
15. HYP PAP T2	.11	.20**	.43**	.37**	.18 [*]	.20**	.63**	.45**	.26**	.21**	.63**	.56**	.34**	.20**	1	
16. HYP PAV T2	.09	.27**	.27**	.35**	.24**	.34**	.40**	.44**	.25**	.41**	.52**	.69**	.47**	.54**	.68**	1

Table 8.43. Intercorrelational Analyses of Survey Study Variables (N=187-631)

Key: Gen=General; HT=Hypothetical Task; T1=Time 1; T2=Time 2; MAP=Mastery Approach Goal Orientation; PAP=Performance Approach Goal Orientation; PAV=Performance Avoidance Goal Orientation *p<0.01 + p<0.05

	1	2	3	4	5	6	7	8	9	10	11	12	13	14	15	16
1. VT M.E. T1	1															
2. NT M.E. T1	.51**	1														
3. VT Perf. T1	20	18	1													
4. NT Perf. T1	.10	<.01	.38**	1												
5. VT MAP T1	.24*	.27 [*]	16	.25 [*]	1											
6. VT MAV T1	04	.02	02	.22	.54**	1										
7. VT PAP T1	.13	.276 [*]	<.01	.12	.48**	.34**	1									
8. VT PAV T1	.10	.21	.09	.15	.30 [*]	.21	.84**	1								
9. NT MAP T1	.16	.36**	07	.28 [*]	.74**	.49**	.39**	.27 [*]	1							
10. NT MAV T1	.07	.30 [*]	16	03	.46**	.68**	.35**	.21	.64**	1						
11. NT PAP T1	.07	.17	<.01	.19	.39**	.43**	.75**	.71**	.51**	.48**	1					
12. NT PAV T1	.04	.18	.02	.15	.37**	.33**	.84**	.81**	.37**	.36**	.79**	1				
13. VT S. E. T1	06	15	.08	02	.25*	.20	.28*	.18	.13	.09	<.01	.18	1			
14. NT S. E. T1	.09	02	.08	.38**	.25 [*]	.33**	.32**	.30 [*]	.40**	.27 [*]	.47**	.39**	.21	1		
15. VT M.E. T2	.61**	.51**	26 [*]	.02	.24 [*]	.06	.14	.08	.26*	.18	.09	.12	13	<01	1	
16. VT Perf. T2	<.01	.03	.52**	.25*	17	07	.11	.15	07	06	<.01	.15	.16	<.01	10	1
17. NT M.E. T2	.49**	.52**	25 [*]	12	.03	19	<01	02	.05	06	09	11	05	12	.54**	06
18. NT Perf. T2	14	21	.34**	.43**	02	.20	.16	.14	.13	.13	.31**	.17	03	.49**	28 [*]	.23

Table 8.44. Intercorrelational Analyses of Experimental Study Variables (N=49-71)

Key: S.E.=Self-efficacy; M.E.=Mental Effort; T.E.=Task Experience; Perf.=Performance; T1=Time 1; T2=Time 2; MAP=Mastery Approach Goal Orientation; PAP=Performance Approach Goal Orientation; NT=Numerical Test; VT=Verbal Test. **ρ<0.01 * ρ<0.05

	1	2	3	4	5	6	7	8	9	10	11	12	13	14	15	16
19. VT MAP T2	.14	.12	05	.29 [*]	.67**	.48**	.38**	.24 [*]	.72**	.44**	.39**	.32**	.26 [*]	.25*	.34**	10
20. VT MAV T2	<.01	.02	03	.24 [*]	.51**	.56**	.37**	.29 [*]	.63**	.51**	.48**	.33**	.11	.19	.27*	09
21. VT PAP T2	<.01	.06	.10	.22	.31**	.40**	.71**	.72**	.30 [*]	.26 [*]	.64**	.63**	.30 [*]	.36**	.16	.19
22. VT PAV T2	.04	.03	.04	.15	.22	.29*	.65**	.67**	.23	.22	.60**	.59**	.18	.25 [*]	.17	.13
23. NT MAP T2	<-01	.03	.02	.35**	.51**	.45**	.27*	.22	.68**	.49**	.42**	.28 [*]	.14	.40**	.16	<01
24. NT MAV T2	07	<01	.03	.24 [*]	.39**	.53**	.32**	.30 [*]	.58**	.52**	.42**	.31**	.19	.26 [*]	.15	.04
25. NT PAP T2	.05	.07	<.01	.21	.23	.30 [*]	.59**	.65**	.27*	.24 [*]	.65**	.60**	.13	.45**	.19	.07
26. NT PAV T2	06	02	.03	.14	.21	.38**	.67**	.71**	.23	.27*	.66**	.70**	.12	.40**	.12	.02
27. VT S. E. T2	22	30 [*]	.24*	.04	.13	.15	.25 [*]	.18	.12	.12	.12	.22	.63**	.36**	17	.23
28. NT S. E. T2	.06	02	.04	.39**	.35**	.49**	.39**	.33**	.45**	.37**	.58**	.45**	.14	.74**	.05	<.01
29. VT Practice	04	10	02	25 [*]	36**	13	21	22	23	17	15	11	15	04	07	.05
30. NT Practice	<01	04	06	24 [*]	41**	22	21	18	27 [*]	25 [*]	17	19	23	07	07	.03
31. VT T.E.	<01	.07	.19	.15	.17	.10	.09	.14	.13	.28 [*]	.08	.03	.21	.05	17	.25 [*]
32. NT T.E.	12	<01	.14	.30 [*]	.34**	.25 [*]	.27*	.22	.28 [*]	.23	.23 [*]	.23	.26 [*]	.33**	13	<01

Key: S.E.=Self-efficacy; M.E.=Mental Effort; T.E.=Task Experience; Perf.=Performance; T1=Time 1; T2=Time 2; MAP=Mastery Approach Goal Orientation; PAP=Performance Approach Goal Orientation; NT = Numerical Test; VT = Verbal Test. **ρ<0.01 * ρ<0.05

	17	18	19	20	21	22	23	24	25	26	27	28	29	30	31	32
17. NT M.E. T2	1															
18. NT Perf. T2	38**	1														
19. VT MAP T2	.06	04	1													
20. VT MAV T2	17	.16	.75**	1												
21. VT PAP T2	11	.30*	.42**	.50**	1											
22. VT PAV T2	13	.26*	.37**	.52**	.88**	1										
23. NT MAP T2	10	.34**	.66**	.66**	.40**	.28*	1									
24. NT MAV T2	08	.19	.68**	.80**	.50**	.43**	.76**	1								
25. NT PAP T2	02	.43**	.29 [*]	.43**	.82**	.75**	.49**	.45**	1							
26. NT PAV T2	15	.40**	.29*	.47**	.88**	.82**	.38**	.47**	.88**	1						
27. VT S. E. T2	21	.14	.15	.18	.32**	.30*	.20	.26*	.20	.24 [*]	1					
28. NT S. E. T2	26 [*]	.54**	.38**	.47**	.57**	.46**	.61**	.47**	.65**	.62**	.28 [*]	1				
29. VT Practice	.04	15	30 [*]	21	24 [*]	18	29 [*]	28 [*]	17	15	12	15	1			
30. NT Practice	.04	14	30 [*]	28 [*]	25 [*]	26 [*]	28 [*]	29 [*]	19	21	27 [*]	19	.83**	1		
31. VT T.E.	20	.17	.05	.06	.12	.13	.18	.15	.03	<.01	.31**	.16	53**	44**	1	
32. NT T.E.	13	.36**	.16	.13	.21	.19	.32**	.21	.22	.24 [*]	.34**	.43**	55**	57**	.56**	1

Key: S.E. = Self-efficacy; M.E. = Mental Effort; T.E. = Task Experience; Perf. = Performance; T1=Time 1; T2=Time 2; MAP=Mastery Approach Goal Orientation; PAP=Performance Approach Goal Orientation; PAV=Performance Goal Orientation; NT = Numerical Test; VT = Verbal Test. ** ρ <0.01 * ρ <0.05
	1. Gen MAP T1	2. Gen MAV T1	3. Gen PAP T1	4. Gen PAV T1	5. HYP MAP T1	6. HYP MAV T1	7. HYP PAP T1	8. HYP PAV T1	9. Gen MAP T2	10. Gen MAV T2	11. Gen PAP T2	12. Gen PAV T2	13. HYP MAP T2	14. HYP MAV T2	15. HYP PAP T2	16. HYP PAV T2
17. VT M.E. T1	.26*	.23	.29 [*]	.25*	.38**	.25*	.12	.11	.24	.12	.15	.21	.15	.23	.32*	.14
18. NT M.E. T1	.12	.02	.27*	.30 [*]	.15	.15	.13	.08	.07	.05	.09	<.01	.27	.10	.26	.15
19. VT Perf. T1	18	.10	06	.11	11	25 [*]	11	05	29 [*]	19	18	08	31 [*]	35 [*]	25	19
20. NT Perf. T1	<01	.03	.13	.10	.14	06	.02	.09	.08	09	.08	.11	.12	<01	.14	.11
21. VT MAP T1	.47**	.30 [*]	.26 [*]	<01	.33**	.26*	.16	.25 [*]	.59**	.45**	.33 [*]	.23	.59**	.51**	.32 [*]	.26
22. VT MAV T1	.38**	.27*	.28 [*]	05	<.01	.13	.13	.22	.33*	.34 [*]	.29 [*]	.20	.33 [*]	.31 [*]	.27	.22
23. VT PAP T1	.31 [*]	.30 [*]	.40**	.28*	.26*	.33**	.35**	.49**	.40**	.41**	.48**	.42**	.41**	.30 [*]	.47**	.41**
24. VT PAV T1	.16	.36**	.36**	.45**	.23	.36**	.39**	.53**	.33*	.45**	.59**	.57**	.34 [*]	.32 [*]	.62**	.53**
25. NT MAP T1	.34**	.26 [*]	.33**	.09	.24 [*]	.17	.16	.21	.34 [*]	.36 [*]	.30 [*]	.19	.50**	.38**	.30 [*]	.29*
26. NT MAV T1	.41**	.28*	.27*	05	.15	.22	.11	.15	.27	.28*	.16	.06	.33*	.27	.13	.14
27. NT PAP T1	.23	.15	.47**	.17	.07	.16	.39**	.38**	.23	.20	.50**	.37**	.20	.12	.53**	.31 [*]
28. NT PAV T1	.20	.34**	.39**	.30*	.23	.33**	.42**	.53**	.34*	.35*	.50**	.42**	.33*	.21	.56**	.43**
29. VT S. E. T1	.33**	.29*	10	21	02	.07	13	.02	.29*	.41**	.08	.19	.18	.36**	.06	.26
30. NT S. E. T1	.32**	.36**	.38**	.13	.07	.10	.39**	.36**	.22	.09	.19	.13	.08	02	.38**	.19
31. VT M.E. T2	.09	.16	.25*	.26*	.48**	.27*	.23	.21	.20	.02	.10	.06	.23	.10	.26	.11

 Table 8.45. Intercorrelational Analyses of Variables across Survey and Experiment (N=49-71)

	1. Gen MAP T1	2. Gen MAV T1	3. Gen PAP T1	4. Gen PAV T1	5. HYP MAP T1	6. HYP MAV T1	7. HYP PAP T1	8. HYP PAV T1	9. Gen MAP T2	10. Gen MAV T2	11. Gen PAP T2	12. Gen PAV T2	13. HYP MAP T2	14. HYP MAV T2	15. HYP PAP T2	16. HYP PAV T2
32. VT Perf. T2	05	.03	.04	.06	14	08	14	07	.10	.04	05	<01	05	04	16	02
33. NT M.E. T2	05	.03	.11	.25*	.17	03	.04	02	09	12	12	12	02	05	.02	03
34. NT Perf. T2	09	.07	.22	.09	13	18	.03	.10	.07	.02	.20	.22	04	11	.21	.12
35. VT MAP T2	.29 [*]	.23	.22	.03	.29 [*]	.21	.10	.25*	.31 [*]	.36 [*]	.18	.22	.39**	.42**	.20	.26
36. VT MAV T2	.15	.24 [*]	.32**	.08	.23	.18	.18	.27*	.24	.36 [*]	.31 [*]	.33*	.35*	.39**	.33*	.32*
37. VT PAP T2	.09	.22	.37**	.26 [*]	.14	.25 [*]	.31**	.50**	.36**	.41**	.54**	.55**	.38**	.35 [*]	.49**	.55***
38. VT PAV T2	.10	.16	.38**	.24 [*]	.18	.29 [*]	.33**	.47**	.30 [*]	.44**	.53**	.60**	.39**	.37**	.54**	.61**
39. NT MAP T2	.16	.31 [*]	.17	<.01	.27*	.09	.14	.14	.22	.27	.22	.17	.21	.15	.11	.05
40. NT MAV T2	.09	.24 [*]	.18	.05	.18	.07	.15	.17	.16	.31 [*]	.28	.27	.27	.34 [*]	.22	.27
41. NT PAP T2	.05	.23	.47**	.30 [*]	.13	.13	.43**	.43**	.27	.25	.57**	.50**	.28	.19	.57**	.41**
42. NT PAV T2	.03	.27*	.47**	.36**	.16	.22	.46**	.58**	.27	.34 [*]	.57**	.55**	.31 [*]	.22	.62**	.54**
43. VT S. E. T2	.24	.22	07	22	02	08	03	<.01	.09	.23	11	.03	.04	.04	<01	.07
44. NT S. E. T2	.20	.25 [*]	.49**	.10	.07	.13	.34**	.29*	.35*	.26	.39**	.24	.23	.11	.44**	.20
45. VT Practice	19	<01	15	.04	16	19	.09	10	35 [*]	22	03	04	24	28	05	10
46. NT Practice	20	.02	20	.07	07	07	.10	09	28 [*]	19	<.01	03	15	18	06	11
47. VT T.E.	.22	<01	05	20	16	<01	23	29 [*]	.28	.28	08	05	.08	.19	12	.02
48. NT T.E.	.02	<.01	.09	.03	.02	<.01	.05	.06	.31 [*]	.15	.09	06	.19	<01	.10	<.01

Key: Gen=General; HT=Hypothetical Task; S.E.=Self-efficacy; M.E.=Mental Effort; T.E.=Task Experience; Perf.=Performance; T1=Time 1; T2=Time 2; MAP=Mastery Approach Goal Orientation; PAP=Performance Approach Goal Orientation; PAV=Performance Avoidance Goal Orientation; NT = Numerical Test; VT = Verbal Test. **p<0.01 * p<0.05

The correlation matrices presented above provide the answers to a number of Research Questions and were the only tests required for a few of the Hypotheses. These Research Questions and Hypotheses will be addressed below. Moreover, as explained earlier, some additional Research Questions which have the same theme as the Research Questions and Hypotheses addressed by the correlational analyses will also be discussed below.

Before focusing on the relevant Research Questions and Hypotheses (relating to the correlation matrices) a few correlations of interest are addressed. These correlations of interest, which are not directly related to any of the Research Questions or Hypotheses, relate to three main themes. The main themes and reasons for examining each theme are presented below.

- a) Correlations between the four GOs. Examining the correlations between the four GOs will provide a better understanding of the structure of the 2x2 model of GOs and whether each of the four GOs in this model is differentiated enough to warrant measuring them using different scales (e.g. measuring MAP and MAV GOs separately as opposed to together).
- b) Correlations between task-experience at Time 1 and practice at Time 2 with selfefficacy, and performance. Task Experience and Practice were measured in this study since it was thought that these might influence the self-efficacy and performance of participants. Since no Research Questions or Hypotheses directly address these relationships it was thought reasonable to report them at this point in the study. Examining these correlations might provide a better basis for understanding the Research Questions and Hypotheses that focus on the relationships between GOs and self-efficacy and performance.
- c) Correlations between the GOs (General, Verbal test, and Numerical test) and Practice. Since Practice was included as a covariate in a number of analyses which investigated the relationships between GOs and other variables (including mental effort, self-efficacy and performance) it was thought appropriate to investigate the

relationships between Practice and GOs so as to have a better basis for understanding the analyses that include practice as a covariate.

Each of the themes described above will be discussed in Sections 8.6.1., 8.6.2. and 8.6.3., respectively.

8.6.1. Correlations between the four Goal Orientations

In Chapter 7 the correlations between the four GOs found in the study by Elliot and Murayama (2007) who developed the AGQ-R were presented (Section 7.1.5.1.). These indicated that the four GOs were low to moderately correlated. Moreover, correlations for GOs having the same definition but a different valence (e.g. MAP and MAV) were found to be higher than those for GOs having different definitions but the same valence (e.g. MAP and PAP). This was found to be the case in the present study too as indicated in Table 8.43. For example, for General GOs at Time 1, the correlation between MAP and MAV GOs was 0.54 (p<.01) and the correlation between PAP and PAV GOs was found to be 0.60 (ρ <.01). On the other hand, the correlation between MAP and PAP GOs (for the same sample) was 0.27 (ρ <.01) and the correlation between MAV and PAV GOs was 0.29 (ρ <.01). This pattern of correlations was also found for General GOs at Time 2. The correlations between the four GOs ranged from non-significant to 0.68 (ρ <.01) in the study by Elliot and Murayama (2008). The intercorrelations between the General GOs (both at Time 1 and at Time 2) in this study ranged between 0.13 (ρ <.01) and 0.69 (ρ <.01). Therefore, with respect to General GOs, the results in this study correspond with those obtained by Elliot & Murayama (2008).

In terms of the overall size of the correlations, the intercorrelations for the Hypothetical Task GOs (at both Time 1 and Time 2) are similar to those of General GOs. However, the Verbal and Numerical Test GO intercorrelations (at Time 1 and Time 2) are higher than the General and Hypothetical Task intercorrelations (the former range from non-significant to 0.88, ρ <.01, whilst the latter range from non-significant to 0.69, ρ <.01). Similarly to the study by Elliot and Murayama (2008) as well

as to the pattern observed for General GOs, Tables 8.43. and 8.44. show that taskspecific GO scales sharing the same definition (but having a different valence) had higher correlations than the scales sharing the same valence but (but having a different definition). However, the Hypothetical Task GOs at Time 2 was an exception since the correlation between MAV and PAV GOs was slightly higher than the correlation between MAV GOs (0.54 and 0.51, ρ <.01, respectively). Overall, the results obtained in this study correspond with those of Elliot and Murayama (2008) with respect to the intercorrelations between GOs. Since the correlations for General and Hypothetical Task GOs at Time 1 and Time 2 were all significant at ρ <.01, it is unlikely that any of these significant correlations are a result of Type I error.

8.6.2. Correlations between Verbal and Numerical test task-experience (at Time 1) and practice (at Time 2) with self-efficacy, and performance

Table 8.44. shows that the correlation between task experience and self-efficacy on the Verbal Test at Time 1 was not significant. However, Numerical Test task experience and self-efficacy at Time 1 were found to have a significant positive correlation of 0.33 (ρ <.01). This correlation indicates that having practised Numerical skills is positively related to a somewhat higher self-efficacy on the Numerical Test. Although it is possible that the significant correlation found on the Numerical Test may be due to Type I error, this is quite unlikely at a ρ <.01.

A possible explanation for the differences in the relationships between Verbal and Numerical tests and self-efficacy is that people may feel more strongly about their numeric ability than about their verbal ability. It might be the case that people feel either very confident or not at all confident about their numeracy skills but do not have these same strong feelings of confidence (or lack of confidence) about their verbal ability. If participants do not feel at all confident about their numeric ability and do not practise at all before the test, they might feel much less confident (and therefore have lower self-efficacy) about their performance on the test. On the other hand, a participant who feels confident about their Numeric ability because they practise this skill quite often may feel confident (and therefore have a higher selfefficacy) about their numeracy skills.

The correlational analyses also indicate that Verbal test task experience was not found to be significantly correlated with performance (r = -0.19, p > .05). However, Numerical test task experience was positively related to performance (r = 0.30, p < .05). Therefore, it seems as though practising numerical skills does in fact relate to higher performance. In this case, the 0.05 significance level must be kept in mind when interpreting the results due to the possibility of this significant correlation being a result of Type I error. However, since the results obtained here reflect the results obtained when assessing the relationships between the aptitude tests and self-efficacy there is a strong possibility that the relationship is in fact significant and not the result of Type I error.

No significant correlations between practice and self-efficacy were found on the Verbal and Numerical tests. There were also no significant correlations between practice and performance (on the Verbal and Numerical tests). This may be due to the fact that the majority of participants only had approximately two weeks between their first and second experimental sessions. Therefore, they had limited time to practise. Two weeks may not have been enough time for their confidence in their ability to change significantly or for their performance to significantly increase (assuming that most of the participants who practised did not practise intensively for two weeks).

8.6.3. Correlations between GOs (General, Verbal test, and Numerical test) and Practice

Although not directly related to any of the Research Questions or Hypotheses in the current study the correlations between GOs and practice on tests were thought to be of interest. There is a possibility that people adopting particular GOs (e.g. MAP) might be more likely to practise tasks in order to improve their learning or performance.

The results presented in Table 8.45. indicate that there were no significant correlations between General GOs (at Time 1) and practice. With respect to the Verbal and Numerical tests, there were significant negative correlations between Verbal and Numerical MAP GOs and practice (Verbal test: r = -0.36, $\rho < .01$; Numerical test: r = -0.27, ρ <.05). There was also a significant negative correlation between a MAV GO and practice on the Numerical test at Time 1 (r = -0.25, ρ <.05). These negative correlations are difficult to explain. One would expect a person with a MAP GO to practise more in order to increase their learning. Moreover, participants with a MAV GO would also be expected to practise the tasks more so as not to forget them or to avoid misunderstanding them. The possibility of Type I error further complicates interpretation of these correlation results. In the case of MAP GOs and practice, significant correlations were found for both tests (with one of the correlations being significant at ρ <.01). Consequently, it is quite unlikely that these significant results are a consequence of Type I error. In the case of the MAV GO, this result was only obtained on the Numerical Test and at ρ <.05. Therefore, it is possible that this significant correlation is a result of Type I error. Further investigation of these relationships is necessary in order to be able to provide an explanation for the findings of this study and to determine whether these results are accurate (and not the result of Type I error). This will be discussed further in the section on recommendations for future research in Chapter 11 (Section 11.3.).

As mentioned earlier, the correlational analyses results answer a number of Research Questions and test a few of the Hypotheses for the current study. These Hypotheses and Research Questions include Hypotheses 2a and 2b as well as Research Questions 6b, 6c, 7b, 8a, 8b, 8c and 8d. These Research Questions and Hypotheses relate to the relationships between GOs and self-efficacy, mental effort and performance. Research Questions 6a, 7a, and 9, also focus on the relationships between GOs and self-efficacy, mental effort and performance, respectively. Consequently, they will also be discussed in this chapter. Those Research Questions and Hypotheses which required considerable further analyses of data are addressed in Chapter 9.

8.7. Research Questions and Hypotheses addressed by the Correlation Matrices

Since the Research Questions and Hypotheses have been grouped according to theme they will not be presented in the numerical order assigned to them throughout the literature review (e.g. Hypothesis 1, Hypothesis 2, etc). For example, Hypotheses 2a and 2b and Research Questions 6a, 6b and 6c all focus on the relationships between GOs and self-efficacy. Consequently, they will be discussed as a group in the same subsection (8.7.1.). Research Questions 7a and 7b focus on the relationships between GOs and mental effort. The answers to these Research Questions are discussed in Section 8.7.2. Research Questions 8a, 8b, 8c, 8d and 9 address the relationships between GOs and performance and are discussed in sub-section 8.7.3.

Self-efficacy, mental effort and performance were only measured on the Verbal and Numerical Tests and not on the Hypothetical Tasks. Consequently, only the Verbal and Numerical Test GOs (and GO profiles) were used in the analyses to answer Research Questions and Hypotheses regarding the self-efficacy, mental effort or performance of 'task-specific GOs'. In order to understand the correlational analyses results better, the results of this study will be compared with those of previous studies. So as to compare like with like, the results of this study will be compared with those of studies investigating the *correlations* between GOs and self-efficacy, mental effort and performance. Moreover, since the 2x2 model of GOs was used for the non-profile perspective analyses, the results of these analyses will be compared to the results of studies using the three-factor model for MAP, PAP and PAV GOs and with the results of studies using the 2x2 model of GOs for MAP, MAV, PAP and PAV GOs. Given that the 3-factor model was used in the profile perspective analyses, the results of these analyses will be compared to the results of studies using the 3-factor model when assessing the relationships between GO profiles and performance.*

8.7.1. Analyses of the relationships between GOs and Self-efficacy

This sub-section focuses on the Research Questions and Hypotheses which address the relationships between GOs and self-efficacy. These include Hypotheses 2a and 2b as well as Research Questions 6a, 6b and 6c.

Hypothesis 2a. Task-specific mastery-approach goal orientations are expected to be significantly positively correlated with self-efficacy.

As indicated in Table 8.44. the MAP GO and self-efficacy are significantly positively correlated on the Numerical Tests at Time 1 and Time 2 (r=.41, ρ <.01, and r=.63, ρ <.01 respectively) and on the Verbal Test at Time 1 (r=.26, ρ <.05). However, no significant correlations were found between the MAP GO and self-efficacy on the Verbal Test at Time 2. Since the Verbal Test correlation at Time 1 was not significant and the one at Time 2 was found to be significant at ρ <.05, it is possible that the significant Time 2 correlation is a result of Type I error. Consequently, although overall, there seems to be significant support for Hypothesis 2a, the Verbal Test correlation results should be interpreted with caution. The results presented above indicate that a MAP GO seems to be more highly correlated with self-efficacy on the Numerical Tests than on the Verbal Tests.

^{*}Studies using the profile perspective only investigated the relationships between GO profiles and performance.

A number of previous studies found a MAP GO to be quite highly positively related to self-efficacy. For example, Bong (2009) found correlation coefficients ranging from 0.58 to 0.86 (ρ <.01). Sins et al. (2008) found a positive significant correlation of 0.59 (ρ <.01) whilst Bong (2001) found correlation coefficients ranging from 0.57 to 0.72 (ρ <.01). However, other studies found lower correlations between a MAP GO and self-efficacy. For example, Tanaka (2007) and Morris and Kavussanu (2008) found significant positive correlations of 0.27 (ρ <.01) and 0.31 (ρ <.01), respectively, between a MAP GO and self-efficacy.

The results of the Numerical Test correlations are similar in strength to those of the studies by Bong (2009), Bong (2001) and Sins et al. (2008). The correlation coefficient obtained for the Verbal Test at Time 1 seems to be quite similar in strength to those obtained in the studies by Tanaka (2007) and Morris and Kavussanu (2008). However, the non-significant relationship found for the Verbal Test at Time 2 does not correspond with the results of previous studies. Since the correlation coefficients obtained on the Verbal Tests are lower than those on the Numerical Tests it was thought appropriate to test whether there were any significant differences between the correlation coefficients on the Verbal and Numerical tests. In order to do this Fischer z was computed. The results are presented in Table 8.46. When the z-statistic is negative this shows that the first correlation input was smaller than the second. For example in the case of the Verbal and Numerical Test at Time 1, the negative z-statistic indicates that the Verbal Test correlation coefficient was smaller than the Numerical Test correlation coefficient.

Table 8.46. Testing for Significant Differences between Correlation Coefficients onVerbal and Numerical Tests at Time 1 and Time 2

Time Point	Comparis	on and Ns	z-statistic	p-value		
Time 1	VT NT		07	22		
Time I	(N=68)	(N=68)	97			
Time 2	VT	NT	2.20	< 01		
Time 2	(N=67)	(N=68)	-3.29	<.01		

Key: VT = Verbal Test; NT= Numerical Test

As shown in Table 8.46. there was no significant difference between Verbal and Numerical test correlations with self-efficacy at Time 1. However, significant differences between the correlation coefficients were found at Time 2 (z=-3.29, p<.01). It is possible that the type of task moderates the relationship between GOs and self-efficacy. However, this would need to be investigated further before any conclusions are drawn. This will be discussed further in Chapter 11 (Section 11.3.) when presenting the recommendations for future research.

Hypothesis 2b. Task-specific performance-approach goal orientations are expected to be significantly positively correlated with self-efficacy.

As indicated in Table 8.44. the PAP GO was significantly positively correlated with selfefficacy on the Verbal tests at Time 1 and Time 2 (r=.28, ρ <.01, and r=.32, ρ <.01 respectively) and on the Numerical tests at Time 1 and Time 2 (r=.47, ρ <.01, and r=.65, ρ <.01 respectively). Therefore, Hypothesis 2b was supported. Since all the correlations were found to be significant and at a ρ <.01, it is quite unlikely that these significant correlations are the result of Type I error.

Previous studies assessing the correlation coefficients between a PAP GO and selfefficacy obtained correlation coefficients ranging between 0.16 (ρ <.01) and 0.68 (ρ <.01). The correlations obtained for the Verbal Tests in this study were similar to those of studies on the lower end of the continuum such as Tanaka (2007) and Morris and Kavussanu (2008), who obtained correlation coefficients of 0.26 (ρ <.01) and 0.25 (ρ <.01), respectively. However, the correlation coefficients obtained on the Numerical Test were more similar to those of studies at the higher end of the continuum, for example Bong (2009) and Bong (2001) who obtained correlation coefficients ranging between 0.42 (ρ <.01) and 0.68 (ρ <.01) for different school subjects and participants of different ages.

Similarly to the results of Hypothesis 2a, self-efficacy is more highly correlated with PAP GO on the Numerical tests than on the Verbal tests. Consequently, the Fischer z statistic was computed to test for significant differences between the correlation

coefficients of Verbal and Numerical test PAP GOs and self-efficacy. The results of the Fischer z tests are presented in Table 8.47.

Table 8.47.	Testing	for S	Significant	Differences	between	Correlation	Coefficients	on
Verbal and	Numerica	al Te	sts at Time	and Time	2			

Time Point	Comparis	on and Ns	z-statistic	p-value		
Time 1	VT NT		1 0	10		
Time 1	(N=71)	(N=71)	1.5	.15		
Time 2	VT	NT	2 50	< 01		
rinte z	(N=71)	(N=70)	2.38	<.01		

Key: VT = Verbal Test; NT= Numerical Test

The results presented in Table 8.47. are similar to those obtained in Table 8.46., in that, no significant difference between correlation coefficients was found at Time 1 (z=-1.3, p>.05). However, there was a significant difference between the correlation coefficients at Time 2 (z=-2.58, p<.01). Therefore, these results also indicate a strong possibility that type of task might influence the relationship between a PAP GO and self-efficacy. This will be discussed in further detail in Chapter 10 Section 10.2.

Research Question 6b. How do task-specific mastery-avoidance goal orientations correlate with self-efficacy?

Table 8.44. shows that no significant correlations were found between a MAV GO and Verbal test self-efficacy at Time 1 or Time 2 (r=.20, p>.05; r=.18, p>.05). However, significant positive relationships were found on the Numerical Tests at Time 1 and Time 2 (r=.27, p<.05; r=.47, p<.01, respectively).

The results of this study do not correspond with the results of the studies by Radosevich et al. (2007), Lau et al. (2008) and Bong (2009, for Upper Elementary School students) since these found significant negative relationships between a MAV GO and self-efficacy. Similarly to the Verbal Test results of this study, Morris and Kavussanu (2008) and Bong (2009, for Middle Elementary and Middle School students) found no significant relationship between a MAV GO and self-efficacy. On the other hand, Bong (2009) found significant positive relationships between a MAV GO and selfefficacy for lower elementary school students. The results of this study are similar to those of Bong (2009) in that non-significant and positive relationships were found between a MAV GO and self-efficacy. However, in this study the differences found were NOT for participants of different ages but for the different tasks that participants completed.

Bong (2009), found correlation coefficients ranging from non-significant to 0.26 (p<.01) whilst in this study the correlation coefficients ranged between non-significant and 0.47 (p<.01). Therefore, some of the correlations obtained in this study are higher than those obtained in previous research. However, both in previous studies as well as in the current study, the strength of the relationship between a MAV GO and self-efficacy seems to be lower than for MAP and PAP GOs with self-efficacy. In order to test whether, for the results of this study, this difference is significant Fisher z tests were carried out. The results of these tests indicated that although it seems as though MAP and PAP GOs are more strongly related to self-efficacy than MAV GOs are, these differences are not significant.

Similarly to the results of Hypotheses 2a and 2b, the correlation coefficients between a MAV GO and self-efficacy were higher on the Numerical test than on the Verbal test at both Time 1 and Time 2 (0.20, p>.05; 0.18, p>.05; 0.27, p<.05; 0.47, p<.01 for the Verbal Test Time 1, Verbal Test Time 2, Numerical Test Time 1 and Numerical Test Time 2, respectively). Again, it seems as though the type of task might be influencing the relationship between MAV GOs and self-efficacy. In the case of the Numerical Test correlations, the results should be interpreted with caution due to the possibility of Type I error. However, although the Numerical Test correlation at Time 2 was found to be significant at p<.05, the result obtained reflects the result obtained at Time 1 and is consistent with the results of previous correlations. Consequently, it is unlikely to be the result of Type I error.

In order to test for significant differences between the correlation coefficients on Verbal and Numerical tests Fischer z was calculated. The results are presented in Table 8.48.

Table 8.48. Testing for Significant Differences between Correlation Coefficients onVerbal and Numerical Tests at Time 1 and Time 2

Time Point	Comparis	on and Ns	z-statistic	p-value	
Time 1	VT (N=71)	NT (N=70)	43	.67	
Time 2	VT (N=70)	NT (N=70)	-1.9	.06	

Key: VT=Verbal Test Goal Orientation; NT=Numerical Test Goal Orientation.

This time there were no significant differences between the correlation coefficients on the Verbal and Numerical tests at Time 1 or Time 2. However, at Time 2, the difference is very nearly significant. Overall, it seems as though a MAV GO is weakly to moderately positively correlated with self-efficacy, if at all.

Research Question 6c. How do task-specific performance-avoidance goal orientations correlate with self-efficacy?

The correlational analyses results presented in Table 8.44. indicate that a PAV GO is non-significantly or weakly positively related to self-efficacy on the Verbal test (r=.18, ρ >.05; r=.30, ρ <.05 at Time 1 and Time 2 respectively). However, the PAV GO and self-efficacy were more strongly positively related on the Numerical test (r=.39, ρ <.01; r=.62, ρ <.01 at Time 1 and Time 2 respectively). In this case, the Verbal Test correlations should be interpreted with caution due to the possibility of Type I error.

Previous GO studies assessing the relationship between a PAV GO and self-efficacy have found very inconsistent results. For example, Sins et al. (2008), Morris and Kavussanu (2008) and Tanaka (2007) found non-significant relationships between a PAV GO and self-efficacy. The results obtained for the Time 1 Verbal Test correspond with the results of these studies. Other studies including those by Radosevich et al. (2007), VandeWalle et al. (2001), Elliot and Church (1997), Liem et al. (2008) and Lau et al. (2008) found significant negative correlations between a PAV GO and self-efficacy. The results of this study are not consistent with the results of these previously mentioned studies. Bong (2009) found significant positive, significant negative and

non-significant relationships between PAV GOs and self-efficacy depending on the age of participants.

Moreover, Bong (2001) found significant positive and non-significant relationships between PAV GOs and self-efficacy depending on the school subject and age of participants. The results of Bong (2001) are very similar to the results obtained in this study in that non-significant and significant positive relationships were found across tasks. Similarly to the relationships between MAP, MAV and PAP GOs, the correlation coefficients obtained for the Numerical tests are much stronger than those on the Verbal Tests. Fischer z was computed to test for significant differences between these correlation coefficients. The results are presented in Table 8.49.

Table 8.49. Testing for Significant Differences between Correlation Coefficients atTime 1 and Time 2

Time Point	Comparison and Ns		z-statistic	p-value	
Time 1	VT NT (N-71) (N-71)		-1.34	.18	
Time 2	VT	NT	2.41	02	
Time 2	(N=70) (N=71)		-2.41	.02	

Key: VT=Verbal Test Goal Orientation; NT=Numerical Test Goal Orientation.

The results obtained here are identical to the results obtained for Hypotheses 2a and 2b, in that, no significant differences between the correlations were found at Time 1 but there were significant differences at Time 2. The results of the current study indicate that a PAV GO may be significantly positively related to self-efficacy. However, this relationship is possibly moderated by the type of task.

Similarly to the relationship between a MAV GO and self-efficacy, the correlations for a PAV GO (and self-efficacy) seem to be lower than those for the approach GOs. Again, Fisher z tests were carried out to test for significant differences in the strengths of these correlations. However, no significant differences were found in the strengths of the correlations for MAP, PAP and PAV GOs with self-efficacy. The results for Hypotheses 2a and 2b and Research Questions 6b and 6c indicate that, overall, it seems as though the four GOs are all positively related to self-efficacy to some extent.

Moreover, although it seems as though the approach GOs are more highly correlated with self-efficacy than the avoidance GOs these differences were not found to be significant. In addition, since it was found that for the MAP, PAP and PAV GOs the Numerical Test GOs were significantly more highly correlated with self-efficacy than the Verbal Test GOs it may be the case that task type (or participants' perceptions of the task) might moderate the relationships between the GOs and self-efficacy.

Research Question 6a. Do the different task-specific GO profiles score significantly differently on self-efficacy?

In order to answer this Research Question One-Way Analyses of Variance were carried out. The data collected at Time E1 were used in these analyses. It was thought appropriate to use the data at Time E1 because there was no induction at this time point. Consequently, the results obtained would **not** be a result of the interactions between the induced GOs and the GO profiles. This was also the case when examining the differences in the mental effort and performance of participants adopting different GO profiles (Sections 8.7.2. and 8.7.3., respectively).

Using the LCA results presented in Sections 8.5.3. and 8.5.4. participants were assigned to GO clusters using the posterior classification function in Latent Gold. This involves calculating participants' posterior class-membership probabilities. This computation is carried out by using individuals' estimated model parameters and their observed scores (Vermunt & Magidson, 2002). One-way Analyses of Variance were then carried out using GO clusters as the independent variable and self-efficacy as the dependent variable. The results of the analyses are presented in Tables 8.50. and 8.51. The small cluster sizes (34, 23 and 14 in Clusters 1, 2 and 3 respectively, on the Verbal Test and 40, 13, 13 and 5 in Clusters 1, 2, 3 and 4, respectively, on the Numerical Test) due to the small experimental sample size should be kept in mind when interpreting the following results (as well as the ANOVA results presented in Sections 8.7.2. and 8.7.3.). This limitation will be discussed further in Chapter 11 Sections 11.2. and 11.3.

In order to be able to understand the results presented in this Section better, it was thought appropriate to provide a brief description of the GO profiles obtained from the LCAs for Verbal and Numerical test GOs. The results presented in Sections 8.5.3. and 8.5.4. show the different types of GO profiles adopted by participants on the Verbal and Numerical tests, respectively. Three clusters were obtained as a result of the LCA for Verbal test GOs at Time 1. Participants in Cluster 1 were high on all three GOs. Participants in Cluster 2 were moderate to high on all three GOs. However, they had a very slightly higher MAP GO than PAP and PAV GOs. Participants in Cluster 3 had a high MAP GO and low PAP and PAV GOs. With respect to the results of the LCA for Numerical test GOs at Time 1, four types of GO profiles were obtained. Participants in Cluster 1 were found to be high on all three GOs. Participants in Cluster 2 were moderate on all three GOs. Cluster 3 was characterised by participants having a high MAP GO and low PAP and PAV GOs. Participants in Cluster 4 were found to have a low MAP GO, an even lower PAP GO and a high PAV GO. The clusters obtained from the LCAs for Verbal and Numerical test GOs were very similar to the clusters obtained from the LCA of the General GOs. These similarities are discussed in further detail in Chapter 9 Section 9.2.3.

Table 8.50. One-Way ANOVA to test for differences in self-efficacy as a result of Verbal Test GO profiles adopted (N=71)

Source of Variance	SS	Df	MS	F	Sig.
Between Groups	1.58	2	.79	2.23	.12
Within Groups	24.13	68	.36		
Total	25.71	70			

Table 8.51. One-	Way ANOVA	to test	t for	differences	in	self-efficacy	as a	result	of
Numerical Test G	O profiles ad	opted (N	V=71)					

Source of Variance	SS	Df	MS	F	Sig.
Between Groups	11.24	3	3.75	5.88	<.01
Within Groups	42.69	67	.64		
Total	53.93	70			

Table 8.50. indicates that there were no significant differences in the relationships between the different GO profiles and self-efficacy on the Verbal test (F=2.23; ρ >.05). However, as indicated in Table 8.51., significant differences were found between the

different GO profiles and self-efficacy on the Numerical test (F=5.88; p<.01). The LSD post hoc test indicated that there were significant differences between Cluster 1 and Cluster 2; Cluster 1 and Cluster 3; Cluster 1 and Cluster 4. Participants in Cluster 1 have higher levels of self-efficacy than participants in Clusters 2, 3, and 4 (see Table 8.52.). Therefore, on the Numerical test, participants in Cluster 1 had the highest levels of self-efficacy. Since participants in Cluster 1 are also high on all three GOs too it may seem that these participants are possibly answering 'strongly agree' on all measures irrespective of what is being measured. However, as will be presented in Section 8.7.3. these participants also had the highest levels of performance. This provides support against the theory that participants in Cluster 1 are simply agreeing with all statements on the measures.

The results of the relationships between GO profiles and self-efficacy seem to be consistent with the results obtained from the non-profile perspective, in that, the relationship between GOs and self-efficacy seems to be affected by the type of task at hand. This emerging pattern will be discussed further in Chapter 10 (Section 10.2.2.).

Table 8.52. Self-efficacy means on Verbal and Numerical Tests at Time 1 for participants in each GO cluster.

GO Cluster	Mean Self-efficacy Score on Verbal Test (Time 1)	Mean Self-efficacy Score on Numerical Test (Time 1)		
Cluster 1	3.76	3.89		
Cluster 2	3.71	2.99		
Cluster 3	3.37	3.18		
Cluster 4	N/A	3.14		

Key: Cluster 1=High Goal Orientations Cluster; Cluster 2=High Mastery, Moderate Performance Cluster; Cluster 3=High Mastery, Low Performance Cluster; Cluster 4:High Performance-Avoidance Cluster.

8.7.2. Analyses of the relationships between GOs and Mental Effort

This sub-section focuses on Research Question 7b, the results of which are presented in Table 8.44. It addresses the relationships between the four GOs and mental effort. In order maintain thematic consistency, Research Question 7a is also discussed in this sub-section since it focuses on the relationships between GOs and mental effort (from the profile perspective). **Research Question 7b.** How do the different task-specific goal orientations correlate with mental effort (if at all)?

The data used for answering this Research Question were obtained during the experimental study at Time E1 and Time E2. The results presented in Table 8.44. indicate that in all cases except for the Numerical test at Time 2 a MAP GO was significantly positively related to mental effort (r=.24, ρ <.05; r=.34, ρ <.01; r=.36, ρ <.01; r= -.10, ρ >.05 for Verbal test Time 1, Verbal Test Time 2, Numerical Test Time 1, and Numerical Test Time 2, respectively). It seems as though, overall, adopting a MAP GO relates to investing more effort when carrying out the tasks. Since the Verbal Test Time 1 correlation is significant at ρ <.05 it is possible that this is the result of Type I error. However, since the Time 2 correlation is significant at ρ <.01, there is a good possibility that the result is truly significant. Nevertheless, this should be kept in mind when interpreting the results.

For the Verbal Test at Time 2 and the Numerical Test at Time 1 a MAV GO was also significantly positively related to mental effort (r=.27, ρ <.05; r=.30, , ρ <.05, respectively). However, there were no significant relationships between a MAV GO and mental effort on the Verbal Test at Time 1 and the Numerical Test at Time 2. PAP and PAV GOs were **not** found to be significantly related to mental effort. Both significant correlations were found at ρ <.05. The inconsistency of these results and the fact that they were obtained at a significance level of ρ <.05 make it quite possible that these significant correlations are the result of Type I error. Further research into these relationships is required in order to make sound interpretations of the results.

Since correlational analyses were used in this study, the results obtained will only be compared with those studies in which the correlations between GOs and effort were presented. Consequently, the study by Phan (2009) was omitted from the comparison. As discussed in Chapter 5 (Section 5.3.) all the studies reviewed found significant positive relationships between a MGO and effort (r=.28, ρ <.01; r=.48, ρ <.01, r=.47, ρ <.05 for the studies by Elliot et al., 1999; Agbuga & Xiang, 2008; and Wolters, 2004, respectively). The results of this study (for the Verbal Tests as well as the Numerical

Test at Time 1) are consistent with those obtained in previous studies for a MAP GO and self-efficacy. The correlation strengths obtained in this study were quite similar to those obtained by Elliot et al. (1999) but slightly lower than those obtained by Agbuga and Xiang (2008) and Wolters (2004). In this study, the Numerical Test MAP GO at Time 2 was not found to be significantly related to mental effort. This result is not consistent with the other results obtained in this study and with the results of previous research studies. It is not clear why this non-significant relationship was found. However, there is a possibility that task characteristics might influence the relationship between GOs and effort. This will be discussed in further detail in Chapter 10 Section 10.3. It is also possible that, as mentioned earlier, the significant correlations were obtained as a result of Type Lerror.

Out of the studies reviewed in literature, none used the 2x2 model of GOs. Therefore, it was not possible to compare the results of this study with those obtained in previous studies with respect to the relationship between a MAV GO and effort. The fact that both significant and non-significant relationships were found between MAV GOs and mental effort on the Verbal and Numerical Tests indicates the need to further investigate these relationships. It also indicates the possibility that other variables such as task characteristics should be taken into account when investigating these relationships.

Previous studies found significant positive relationships between a PAP GO and effort whilst no significant relationships were found in this study. Therefore, the results of this study were not consistent with the results of previous studies for the relationships between a PAP GO and effort. Regarding the PAV GO, the results of the present study correspond with those of Elliot et al. (1999) who found no significant relationship between a PAV GO and effort. However, they are inconsistent with the results obtained by Wolters (2004) and Agbuga and Xiang (2008) who found significant negative and significant positive relationships, respectively, between PAV GOs and effort. The similarities and differences between the results of this study and those of previous studies as well as possible explanations for these will be discussed further in Chapter 10 Section 10.3. **Research Question 7a.** Do the different task-specific GO profiles score significantly differently on mental effort?

The experimental data collected at Time E1 were used in the following analyses. Similarly to Research Question 6a one-way Analyses of Variance were carried out in order to provide an answer to this Research Question. The same cluster classification of participants to GO profiles was used as in the analyses of Research Question 6a. Please refer to Sections 8.5.3. and 8.5.4. for descriptions of the Verbal and Numerical Test GO profiles (clusters) obtained. The results are presented in Table 8.53. and 8.54.

Table 8.53. One-Way ANOVA to test for differences in mental effort as a result of Verbal Test GO profiles adopted (N=71)

Source of Variance	SS	Df	MS	F	Sig.
Between Groups	8.99	2	4.49	3.15	.05
Within Groups	96.90	68	1.43		
Total	105.89	70			

Table 8.54. One-Way ANOVA to test for differences in mental effort as a result of Numerical Test GO profiles adopted (N=71)

Source of Variance	SS	Df	MS	F	Sig.
Between Groups	13.47	3	4.49	2.52	.07
Within Groups	119.27	67	1.78		
Total	132.73	70			

Table 8.53. shows that there were significant differences in mental effort depending on the type of GO profile adopted on the Verbal test (F=3.15; ρ =.05). The post hoc test (LSD) indicated that participants in Cluster 2 scored significantly lower than those in Clusters 1 and 3 (see Table 8.55.). As indicated in Table 8.54. there were no significant differences in mental effort as a result of the different GO profiles adopted on the Numerical test (F=2.52; ρ >.05). In response to Research Question 7a, different Verbal test GO profiles seem to relate differently to mental effort. However, different Numerical test GO profiles do not seem to relate differently to mental effort. Since the results of the latter were very nearly significant it may be wise to further investigate these relationships, especially due to the small sample size used in this study. The results obtained here also point towards the possibility that the relationships between GOs and mental effort may be moderated by task type. This emerging pattern of differences in the relationships between GOs and other variables depending on the task at hand might be important, especially if GOs are found to be task-specific rather than general stable traits. This will be discussed further in Chapter 10 Section 10.3.

Table 8.55. Mental Effort means on Verbal and Numerical Tests at Time 1 for participants in each GO cluster.

GO Cluster	Mean Mental Effort Score on Verbal Test (Time 1)	Mean Mental Effort Score on Numerical Test (Time 1)		
Cluster 1	6.56	6.93		
Cluster 2	5.83	5.92		
Cluster 3	6.64	6.69		
Cluster 4	N/A	5.80		

Key: Cluster 1=High Goal Orientations Cluster; Cluster 2=High Mastery, Moderate Performance Cluster; Cluster 3=High Mastery, Low Performance Cluster; Cluster 4:High Performance-Avoidance Cluster.

8.7.3. Analyses of the relationships between GOs and Performance

The Research Questions related to investigating the relationships between GOs and performance are discussed in this sub-section. The experimental data collected at Time E1 and Time E2 were used to answer these Research Questions. Each relevant Research Question is presented followed by a description and discussion of the results obtained.

Research Question 8a. How do task-specific mastery-approach goal orientations correlate with performance on tasks (if at all)?

The results presented in Table 8.44. indicate that a MAP GO was significantly positively related to performance on the Numerical test at Time 1 and at Time 2 (r=.28, ρ <.05; r=.34, ρ <.01, respectively). No significant correlations were found between a MAP GO and performance on the Verbal tests. Although there is the possibility of the Numerical Test Time 2 correlation being a result of Type I error this is quite unlikely

seeing that it is consistent with the result obtained on the Numerical Test at Time 1 (which had a significant correlation at ρ <.01).

Past research findings have been rather inconsistent with respect to the relationship between a mastery/MAP GO and performance. Some studies found non-significant relationships (e.g. Elliot & McGregor, 2001) whilst others found significant positive relationships (e.g. Elliot et al., 1999). Four of the five studies reviewed obtained different results for participants depending on the task, age of participants or the time at which they participated (e.g. Study 1 or Study 2). For example, Elliot et al. (1999) found a MGO to be significantly positively related to performance in Study 1 but not in study 2. Yeo et al. (2009) found a significant positive relationship between a MAP GO and performance on an air traffic control task but not on exam performance. The results of this study correspond with those of previous studies in that both nonsignificant and significant positive relationships were found on the Verbal and Numerical Tests, respectively. As was discussed previously, it is possible that the inconsistencies between the results of the different studies and those across tasks in the same study are a result of the relationship between GOs and performance being influenced by the type of task at hand. This possibility will be discussed in further detail in Chapter 10 Section 10.4.

Research Question 8b. How do task-specific mastery-avoidance goal orientations correlate with performance on tasks (if at all)?

The results presented in Table 8.44. indicate that no significant relationships were found between a MAV GO and performance. Three of the studies reviewed provided the correlational analyses results for a MAV GO with performance. Two out of these three studies (those by Elliot & McGregor, 2001, and Yeo et al., 2009) found a nonsignificant relationship between a MAV GO and performance. On the other hand, Bong (2009) found a significant negative relationship between a MAV GO and performance for Middle Elementary school students and non-significant relationships for Lower and Upper Elementary as well as Middle School students. Consequently, the results of this study are consistent with the results of Elliot and McGregor (2001), Yeo et al. (2009) and Bong (2009, for Lower and Upper Elementary as well as Middle School students).

Research Question 8c: How do task-specific performance-approach goal orientations correlate with performance on tasks (if at all)?

The results presented in Table 8.44. indicate that the only significant relationship found was that between the Numerical test PAP GO and performance at Time 2 (r=.43, ρ <.01). It is quite unlikely that a correlation found at ρ <.01 is a result of Type I error. Therefore, this is not a likely explanation for the single correlation found. The GO inductions introduced at Time 2 may be a contaminating factor since they may have contributed to changes in performance, mental effort, or self-efficacy (or any combination of these) on the tasks at Time 2. However, as will be discussed in Chapter 9 Section 9.1.2., the GO inductions were not successful. Consequently, it is highly unlikely that these influenced the performance of participants at Time 2. In answer to Research Question 8c, the Numerical Test PAP GO at Time 2 was found to be significantly related to performance. However, no other significant relationships were found between a PAP GO and performance.

Previous studies found significant positive or non-significant relationships between PAP GOs and performance. What is especially notable is that all the studies reviewed found inconsistent relationships across tasks, the time at which they participated (e.g. Study 1 or Study 2), or for participants of different ages, when investigating the relationships between PAP GOs and performance. For example, Yeo et al. (2009) found significant positive relationships on the air traffic control task and non-significant relationships for exam performance. On the other hand, Bong (2009) found significant positive relationships for Upper Elementary and Middle School students and nonsignificant relationships for Lower and Middle Elementary School students. The results of this study are consistent with those of previous research studies in that significant positive and non-significant relationships were found on different tasks (that is, the Verbal and Numerical Tests). Moreover, different relationships between a PAP GO and performance were found over time for the Numerical Test. These results strongly

222

point towards the relationships between a PAP GO and performance being moderated by task type. These results will be discussed in further detail in Chapter 10 Section 10.4.

Research Question 8d: How do task-specific performance- avoidance goal orientations correlate with performance on tasks (if at all)?

Similarly to the PAP GO results, Table 8.44. indicates that only the PAV GO on the Numerical Test at Time 2 was found to be significantly positively related to performance (r=.40, ρ <.01). Similarly to the Numerical Test PAP results, this correlation is unlikely to be the result of Type I error since it was found to be significant at a ρ <.01. In addition, again, there is the issue of the GO inductions at Time 2 being a contaminating factor. However, since these were not successful, they are probably not a major issue when investigating the relationships between GOs and performance at Time 2. The answer to Research Question 8d is that the Numerical Test PAV GO at Time 2 was found to be significantly positively related to performance. However, no other significant relationships were found between a PAV GO and performance.

With respect to previous research, Church et al. (2001), and Elliot et al. (1999) found a PAV GO to be significantly negatively related to performance. Church et al. (2001) found significant negative correlations with performance on both the chemistry exam and with the SAT scores (r=-0.27 and -0.19, respectively, ρ <.01) and Elliot et al. (1999) found significant negative correlations in both their studies (r=-0.27 and -0.30, ρ <.01). Again, three of the studies reviewed found both significant negative and non-significant relationships between a PAV GO and performance depending on the task, age of participants, or the time at which participants took part in the study (e.g. Study 1 or Study 2). For example, Elliot and McGregor (2001) found a non-significant negative relationship in Study 3. Moreover, Yeo et al. (2009) found a significant negative relationship for exam performance and a non-significant relationship on the air traffic control task.

Similarly to the relationships between a PAP GO and task performance, it is possible that the type of task influences the relationship between GOs and task performance. This might explain why a PAV GO was only found to be significantly related to performance on the Numerical task. However, the results of the current study are not consistent with the results of previous studies in that a significant *positive* relationship was found between a PAV GO and task performance in this study whilst in all other studies non-significant or negative relationships were found. Possible explanations for this and further discussion on the matter is provided in Chapter 10 Section 10.4.

Research Question 9 Do the different task-specific GO profiles score significantly differently on task performance?

One-Way Analyses of Variance were carried out in order to answer this Research Question. Similarly to the other analyses investigating the relationships between GO profiles and other variables (e.g. self-efficacy and mental effort), the cluster classification of participants to GO profiles was used as an independent variable in the analyses. Please refer to Sections 8.5.3. and 8.5.4. for descriptions of the Verbal and Numerical Test GO profiles obtained. Participants' performance on the Verbal and Numerical tests was included as the dependent variable. The results of the analysis are presented in Tables 8.56. and 8.57. below.

Table 8.56. One-Way ANOVA to differences in performance as a result of Verbal Tes	t
GO profiles adopted (N=71)	

Source of Variance	SS	Df	MS	F	Sig.
Between Groups	3.61	2	1.80	.15	.86
Within Groups	839.58	68	12.35		
Total	843.18	70			

Table 8.57. One-Way ANOVA to differences in performance as a result of NumericalTest GO profiles adopted (N=71)

Source of Variance	SS	Df	MS	F	Sig.
Between Groups	72.36	3	24.12	5.59	<.01
Within Groups	289.13	67	4.32		
Total	361.49	70			

The results presented in Table 8.56. indicate that there were no significant differences in the performance of participants having different GO profiles on the Verbal test (F=.15; ρ >.05). On the other hand, Table 8.57. shows that there were significant differences in the performance of participants having different GO profiles on the Numerical test. The post hoc tests (LSD) indicated that participants in Cluster 1 had higher performance than participants in Clusters 2 and 4 (refer to Table 8.58.).

There seems to be a pattern emerging throughout the analyses of the present study. As with a number of analyses carried out so far, the relationships between different GO profiles and performance seem to vary depending on the task at hand. This influence of task type will need to be looked into further since it has been quite persistent throughout the analyses. Further investigating this pattern may provide some insight into the inconsistent findings in literature (and in the current study) with respect to the relationships between GOs and variables such as self-efficacy, mental effort and performance. This will be discussed in further detail in Chapter 10 Section 10.4. and Chapter 11 Section 11.3.

Table 8.58. Mean Performance Score on Verbal and Numerical Tests at Time 1 for participants in each GO cluster.

GO Cluster	Mean Performance Score on Verbal Test (Time 1)	Mean Performance Score on Numerical Test (Time 1)
Cluster 1	15.44	6.85
Cluster 2	15.39	4.54
Cluster 3	14.86	3.18
Cluster 4	N/A	3.14

Key: Cluster 1=High Goal Orientations Cluster; Cluster 2=High Mastery, Moderate Performance Cluster; Cluster 3=High Mastery, Low Performance Cluster; Cluster 4:High Performance-Avoidance Cluster.

8.8. Synopsis

In this chapter the exploratory analyses results were presented. Moreover, decisions were made with respect to model choice (3-factor vs. 2x2) for the current study. In presenting the results of the exploratory analyses a number of Research Questions and Hypotheses were addressed. These related to the relationships between GOs and

self-efficacy, performance, and mental effort as well as one Research Question relating to the types of GO profiles (Research Question 1). The next chapter presents the results for those Hypotheses and Research Questions which were not directly answered by means of the exploratory analyses and required a substantial amount of further analyses. These include Research Questions and Hypotheses addressing the stability and task-specificity of GOs, as well as those focusing on the success and influence (on the performance of participants) of the GO inductions.

Chapter 9: Further Analyses

9.0. Introduction

The previous chapter focused on the exploratory analyses of data for this study and addressed the Research Questions and Hypotheses which were answered or tested as a result of these analyses. This chapter deals with those Research Questions and Hypotheses which required substantial further analyses in order to be answered or tested. The three themes addressed in this chapter include: the stability of GOs over time, task-specificity of GOs, and the interactions between state and trait GOs.

Similarly to the previous chapter, this one is organised according to theme so that Research Questions and Hypotheses having the same theme are grouped together. First, the issue of the stability of GOs is addressed (Section 9.1.). Following this, the questions regarding task-specificity (Section 9.2.) are discussed. Last but not least the effects of the interactions between Trait and State GOs on performance are assessed (Section 9.3.). In order to avoid having an unnecessarily long chapter any nonsignificant results are not presented but are reported as non-significant in the text.

9.1. Assessing the Stability of General and Task-Specific Goal Orientations over time

This section focuses on those Research Questions which address the stability of General and Task-Specific GOs (including Hypothetical Task, Verbal Test and Numerical Test GOs) over time. The data collected at Time Q1 and Time Q2 were used to assess the stability of General and Hypothetical Task GOs whilst the data collected at Time E1 and Time E2 were used to assess the stability of Verbal and Numerical test GOs over time.

The stability of GOs was examined from the profile and non-profile perspectives. The latter was assessed in terms of differential continuity and mean level change whilst the stability of GO profiles was assessed using LCAs and posterior classification (described in Section 8.7.1.). The first part of this section (sub-section 9.1.1.) focuses on the

stability of General GOs whilst the second part (sub-section 9.1.2.) addresses the stability of task-specific GOs.

9.1.1. Assessing the Stability of General GOs

As mentioned earlier the stability of General GOs was assessed from both the profile and non-profile perspectives. The results of the analyses for GO profiles are presented first. This is followed by the results of the analyses from the non-profile perspective.

Research Question 2a. Do individuals' General goal orientation profiles change over time?

In order to provide an answer to this Research Question the results of the LCAs for General GOs at Time 1 and Time 2 were used (please refer to Chapter 8 Section 8.5.). Participants were assigned to clusters using posterior classification. The GO profiles of participants from Time 1 to Time 2 were then compared. The results of this comparison are provided in Table 9.1. In interpreting the results of GO profile comparisons (throughout the study) it should be kept in mind that due to the exact nature of each cluster changing slightly (over time and across tasks) the analyses examining the number of participants staying in the same clusters and those changing clusters are inevitably a bit approximate.

Table 9.1. Frequency Table indicating the General Goal Orientation Profiles Adoptedby Participants at Time 1 and Time 2. (N=183)

		General GO Profile Cluster at Time 2								
0		1	2	3						
1 1 1 1 1 1 1 1 1 1 0 0	1	101	35	3	139					
eneral Profil Cluster Time	2	13	21	5	39					
	3	0	2	1	3					
0	4	0	2	0	2					
	Total	114	60	9	183					

Table 9.1. indicates the frequencies of participants in each Cluster at Time 1 and Time2 with respect to their General GO profiles. The numbers on the diagonal (those in

italics) indicate the frequencies of participants that adopted the same General GO profiles at Time 1 and Time 2. Therefore, overall, 123 participants (67%) adopted the same General GO profile at Time 1 and Time 2. On the other hand, 60 participants (33%) adopted different General GO profiles at Time 2 than they did at Time 1. In order to assess the stability of GO profiles over time in a more objective manner (by taking the sample size and number of clusters into account) a decision was made to compare the distribution of people to clusters from Time 1 to Time 2 presented in Table 9.1. to the distribution of people to clusters (from Time 1 to Time 2) by chance. A chi-square test could not be used with the raw data in this case for two main reasons. Firstly, our data consist of repeated measures which violates one of the assumptions of the chi-square. Secondly, examining whether the frequencies in all clusters were distributed equally was not of interest in this study. Therefore, before using the chisquare test, the number of people one would expect to find in clusters by chance, if, (given the total number of people in each cluster on each occasion) people had distributed themselves proportionately amongst clusters, was calculated. For example, if 60 out of 100 people were in Cluster 1 at Time 1 and 30 out of the same 100 people were in Cluster 1 at Time 2, the number of individuals one would expect to stay in Cluster 1 (from Time 1 to Time 2) by chance is: 60x30 = 18 people.

100

Since only the stability of the number of people in the diagonal is of interest for the purposes of this study (i.e. the number of people adopting the same GO profile at Time 1 and Time 2), the above calculation was only performed for people in the diagonal. The formula used to compute the proportions of people expected to fall into the top left category (of Table 9.1.) by chance is:

Number of People in Cluster 1 at Time 1 x Number of People in Cluster 1 at Time 2 Total Number of People

The same formula was repeated for participants falling into Cluster 2 at Time 1 and Time 2 as well as those falling into Cluster 3 at Time 1 and Time 2. Using the formula presented above the following frequencies were obtained:

Table 9.2. Observed Frequencies and Proportions Expected to Remain in the sameClusters (from Time 1 to Time 2) by Chance

	Cluster 1	Cluster 2	Cluster 3
Observed	101	21	1
Proportions by chance	87	13	0

Key: Cluster 1=High Goal Orientations Cluster; Cluster 2=High Mastery, Moderate Performance Cluster; Cluster 3=High Mastery, Low Performance Cluster.

Following from the results presented in Table 9.2., and keeping in mind that the total number of participants was 183, the total numbers of participants remaining in the same cluster and those changing cluster from Time 1 to Time 2 is presented in Table 9.3. for the Observed Frequencies and the Proportions by Chance.

Table	9.3.	Total	Numbers	of	Participants	remaining	in	the	Same	Cluster	and	Total
Numb	oer Cl	nangin	g Cluster f	ror	n Time 1 to T	'ime 2						

	Remain in Same Cluster from Time 1 to Time 2	Change Cluster from Time 1 to Time 2	Total	
Observed	123	60	183	
Proportions by chance	100	83	183	
Total	223	143	366	

A Chi-square test was then carried out using the frequencies presented in Table 9.3. The X^2_{obt} in this case was 6.07 whilst X^2_{crit} is 6.64 (ρ <.01) and 3.84 (ρ <.05). Therefore, the results indicate that at a significance level of ρ <.05, one could be confident that more participants remained in the same clusters over time than they would have by chance alone. These results point towards the possibility of General GO profiles showing some stability over time. The procedure presented above was also used to examine the stability of task-specific GO profiles over time (Section 9.1.2.5.) and the task-specificity of GOs (Section 9.2.1.) as well as to assess whether General GO profiles are different from task-specific GO profiles (Section 9.2.3.).

Research Question 2b. Do individuals' General goal orientations change significantly over time?

Changes in participants' General GOs over time from the non-profile perspective were assessed in terms of (a) mean level change and (b) differential continuity. In order to examine mean level change over time paired sample t-tests were carried out. The correlational analyses presented in Chapter 8 Section 8.6. were used to assess the differential continuity of General GOs.

Variables	Time 1 Mean	Time 1 SD	Time 2 Mean	Time 2 SD	Df	T-value	Sig.
GEN MAP	3.93	.57	3.89	.60	187	.84	.40
GEN MAV	3.77	.60	3.71	.63	183	1.24	.22
GEN PAP	3.94	.66	3.81	.71	186	2.74	<.01
GEN PAV	3.98	.72	3.81	.76	187	3.15	<.01

Table 9.4. Assessing Mean Level Change of General GOs from Time 1 to Time 2

Key: GEN MAP = General mastery-approach Goal Orientation; GEN MAV = General mastery-avoidance Goal Orientation GEN PAP = General performance-approach Goal Orientation; GEN PAV = General performance-avoidance Goal Orientation

The results of the paired sample t-test presented in Table 9.4. indicate that although all the mean General GO scores seem to decrease over time, only the mean General PAP and PAV GOs decreased significantly over time (t=2.74 and 3.15, ρ <.01 respectively). The results of the correlational analyses used to measure differential continuity were presented in Chapter 8 (Section 8.6.). These results indicate that the correlation coefficients between the four General GOs at Time 1 and Time 2 ranged between 0.47 (ρ <0.01) and 0.56 (ρ <0.01). Consequently, people's General GOs relative to each other seem to be somewhat stable over time. However, in order to more objectively assess the significance of the correlational analyses results, the correlation coefficients obtained in this study will be compared with stability correlation coefficients from other studies on GOs as well as from studies assessing the stability of personality traits (since these have been widely researched and generally assumed to be stable over time).

With respect to GO studies, Fryer and Elliot (2007) found correlations of 0.57 to 0.78 (ρ <0.001) and Elliot and McGregor (2001) found correlation coefficients ranging between 0.70 and 0.74 (ρ <0.01). The correlation coefficients obtained in these studies are higher than those obtained in the present study. However, the GOs measured in

the studies by Fryer and Elliot (2007) and Elliot and McGregor (2001) could be considered to be task-specific since the measures used focused on the GOs adopted on a particular college course. The lower correlation coefficients obtained in the present study may be a result of the fact that general measures of GOs were used. Therefore, the correlation coefficients obtained for General GOs as well as those obtained in other studies will be compared to those of the task-specific GOs (obtained in this study) so as to assess whether task-specific GOs are more stable than General GOs in terms of differential continuity.

Payne et al. (2007) calculated stability coefficient estimates for mastery, PAP and PAV GOs during their meta-analysis. Their results indicate stability coefficient estimates of 0.66 (k=20), 0.70 (k=16) and 0.73 (k=4), with 'k' being the number of studies included, for mastery, PAP and PAV GOs, respectively. These correlations are for the stability of GOs from one to fourteen weeks. These stability coefficients are higher than those obtained in the present study. However, it is difficult to tell whether the coefficients obtained from the study by Payne et al. (2007) are for General or task-specific GOs. Payne et al. (2007) also found that the relationships between the time interval (between GO measures) and the coefficients of stability were negative for all three GOs (r=-0.20, -0.29, and -0.74, for mastery, PAP and PAV GOs respectively). Therefore, the longer the time frames between the administrations, the smaller the coefficients of stability. Since the time intervals between the Time 1 and Time 2 General GO measures ranged from 5 to 51 weeks in this study, the lower coefficients obtained may have resulted from the longer time interval between the measures. In order to examine this possibility further, participants were divided into groups according to the time interval between their Time Q1 and Time Q2 questionnaires. Subsequently, correlational analyses were carried out in order to investigate the stability of General GOs over time depending on the time interval between the two measures.

Since the numbers of participants who completed the questionnaires at different points in time were not equal, the time intervals for analysing data were chosen in a way for there to be roughly equal numbers of participants in each group. For example, a substantial number of participants completed the second questionnaire 4-8 weeks after the first questionnaire (N=94). On the other hand the same number of

232

participants completed the second questionnaire between 17 to 51 weeks after the first questionnaire. Splitting participants (who completed the Time Q2 questionnaire 17 to 51 weeks subsequent to the Time Q1 questionnaire) into smaller groups would not have been feasible for statistical analyses. Consequently, participants were split up into two groups for the purpose of these analyses. One group consisted of participants who completed the Time Q2 questionnaire 4 to 8 weeks subsequent to the Time Q1 questionnaire and the second group was made up of those participants who completed the Time Q2 questionnaire 9 to 51 weeks after the Time Q1 questionnaire. The results of these analyses are presented in Tables 9.5. and 9.6. In both tables all the correlations of interest were found to be significant at ρ <.01. Consequently, the probability of these being the result of Type I error is highly unlikely.

Table 9.5. Correlations between General GOs at Time 1 and Time 2 for participants who completed Time Q2 questionnaire 4-8 weeks after Time Q1 Questionnaire (N≈93)

Variables	1	2	3	4	5	6	7	8
1. GEN MAP T1	1.00							
2. GEN MAV T1	.67**	1.00						
3. GEN PAP T1	.30**	.33**	1.00					
4. GEN PAV T1	.13	.29**	.50**	1.00				
5. GEN MAP T2	.63**	.44**	.30**	.25*	1.00			
6. GEN MAV T2	.32**	.38**	.04	.06	.53**	1.00		
7. GEN PAP T2	.09	.20	.69**	.52	.41**	.14	1.00	
8. GEN PAV T2	03	.14	.45**	.53**	.31**	.34	.64**	1.00

Key: GEN MAP = General Mastery-Approach Goal Orientation; GEN MAV = General Mastery-AvoidanceGoal Orientation GEN PAP = General Performance-Approach Goal Orientation; GEN PAV = GeneralPerformance-Avoidance Goal Orientation; T1 = Time 1; T2 = Time2* ρ <.05</td>

Table 9.6. Correlations between General GOs at Time 1 and Time 2 for participants who completed Time Q2 questionnaire 9-51 weeks after Time Q1 Questionnaire (N≈94)

Variables	1	2	3	4	5	6	7	8
1. GEN MAP T1	1.00							
2. GEN MAV T1	.37**	1.00						
3. GEN PAP T1	.25*	.30**	1.00					
4. GEN PAV T1	08	.36**	.66**	1.00				
5. GEN MAP T2	.50**	.30**	.18	04	1.00			
6. GEN MAV T2	.39**	.56**	.15	.14	.62**	1.00		
7. GEN PAP T2	.08	.23*	.38**	.40**	.31**	.37**	1.00	
8. GEN PAV T2	.02	.29**	.32**	.42**	.17	.45**	.75**	1.00

Key: GEN MAP = General Mastery-Approach Goal Orientation; GEN MAV = General Mastery-AvoidanceGoal Orientation GEN PAP = General Performance-Approach Goal Orientation; GEN PAV = GeneralPerformance-avoidance Goal Orientation; T1 = Time 1; T2 = Time2*p<.05

Similarly to the results obtained by Payne et al. (2007) the stability of the GOs over time seems to be decreasing in terms of correlations over time with the exception of the MAV GO which seems to be increasing over time. It is not clear why the stability of MAV GO seems to be increasing with a longer time interval. Further investigation is required in order to clarify this. So as to test whether the correlation coefficients presented in Table 9.5. are significantly different from those presented in Table 9.6. Fisher z tests were carried out. The results are presented in Table 9.7.

Table 9.7. Testing for Significant Differences between Correlation Coefficients for General GOs at Time 1 and Time 2 (for different Time intervals between Time Q1 and Time Q2 questionnaires).

GO	Comparis	on and Ns	z-statistic	p-value	
ΜΑΡ	4-8 weeks (N=94)	9-51 weeks (N=94)	1.3	.19	
MAV	4-8 weeks (N=92)	9-51 weeks (N=92)	-1.55	.12	
ΡΑΡ	4-8 weeks (N=94)	9-51 weeks (N=93)	3.01	<.01	
PAV	4-8 weeks (N=95)	9-51 weeks (N=93)	.96	.34	

The Fisher z test results indicate that only the stability of the PAP GO significantly decreases with a longer the time interval between the questionnaires. Moreover,
although the strength of the correlation coefficients for MAV GO seems to increase with a greater time interval, this increase is not statistically significant. The implications of these results are discussed in further detail in Chapter 10 Section 10.6.

Since General GOs are assumed to be general, stable traits, it was thought useful to compare the correlation coefficients obtained for General GOs over time with those for personality traits. This was considered to be useful since personality traits are also thought to be general stable traits and their stability has been investigated for far longer than that of GOs. Since hardly any studies, if at all, assessed the stability of **General** GOs over time it was thought to be helpful to use the stability coefficients of personality traits as a point for comparison. Rantanen et al. (2007) assessed the stability of the Big 5 personality traits over a nine year period. They found test-retest correlation coefficients ranging between 0.64 and 0.81 (p<.001) for men and between 0.55 and 0.81 (p<.001) for women. Additionally, Viswesvaran and Ones (2000) found mean coefficients of stability ranging between 0.69 to 0.76 for the Big Five personality traits over a period of one to two years. These correlation coefficients are higher than the ones obtained in the present study even though the time intervals in the studies by Rantanen et al. (2007) and Viswesvaran and Ones (2000) were much longer than the ones in the current study.

Overall it seems that, in terms of mean-level change, participants' mastery GOs tend to remain stable over time for this sample. However, participants' performance GOs change significantly from Time 1 to Time 2. Moreover, in terms of differential continuity, participants' GOs are not as stable as personality traits (especially those having a time interval of between 9 and 51 weeks). The results obtained in this study seem to challenge the assumption of stability of GOs from the non-profile perspective. Therefore, overall it seems as though from the non-profile perspective, General GOs are not as stable as they were initially thought to be. At first glance the results obtained from the profile and non-profile perspectives seem to be rather inconsistent with the former indicating stability and the latter indicating change over time. However, the profile perspective results show that stability is only better than chance at ρ <0.05. This is not enough evidence to support a claim that General GO profiles are stable over time. The implications of these results will be discussed in further detail in Chapter 10 Section 10.6. Research Questions 5a and 5b, which address the stability of task-specific GOs (that is, Hypothetical Task, Verbal Test and Numerical Test GOs) over time are presented next.

9.1.2. Assessing the Stability of Task-Specific GOs

The MAP and PAP GO inductions may have been a contaminating factor when assessing the stability of Verbal and Numerical Test GOs over time. However, using only the control group would have drastically reduced the sample size. As will become evident throughout this section, the inductions were not successful. Consequently, a decision was made to include the whole experimental sample when assessing the stability of Verbal and Numerical Test GOs over time. However, the possible presence of a contaminating factor will be kept in mind when interpreting the results. The results of the analyses assessing the stability of task-specific GOs over time from the non-profile perspective (and in doing so testing the success of the inductions) are presented first.

Research Question 5b. Are task-specific goal orientations stable over time?

The stability of task-specific GOs over time from the non-profile perspective was assessed in terms of mean-level change as well as differential continuity. Paired samples t-tests and correlational analyses were carried out in order to assess the stability of Hypothetical Task GOs over time. Repeated Measures Analyses of Variance and correlational analyses were carried out in order to assess the stability of Verbal and Numerical test GOs over time. The results of the analyses assessing the stability of Hypothetical Task GOs over time are presented first (sub-section 9.1.2.1.) and are followed by the presentation of results assessing the stability of Verbal Test GOs over time (sub-section 9.1.2.2.) and Numerical Tests over time (sub-section 9.1.2.3.).

9.1.2.1. Assessing the Stability of Hypothetical Task Goal Orientations over time

The data included in the analyses for examining the stability of Hypothetical Task GOs over time were collected at Time Q1 and Time Q2. The results obtained when assessing mean-level change in Hypothetical Task GOs over time are presented in Table 9.8.

Table 9.8. Assessing Mean Level Change for Hypothetical Task GOs at Time 1 and Time 2 (N≈186)

Variables	HYP GO	HYP GO	HYP GO	HYP GO	Df	T-value	Sig
Variables	T1 Mean	T1 SD	T2 Mean	T2 SD			
MAP	4.01	.65	3.93	.70	191	1.82	.07
MAV	3.70	.70	3.61	.77	189	1.56	.12
ΡΑΡ	3.74	.81	3.70	.85	191	.58	.56
PAV	3.83	.66	3.72	.75	191	1.88	.06

Key: HYP = Hypothetical Task Goal Orientation; T1 = Time 1; T2 = Time 2

In terms of mean-level change, there were no significant differences in Hypothetical Task GOs from Time 1 to Time 2. Although the values obtained for MAP and PAV GOs over time were very nearly significant, it seems as though mean levels of Hypothetical Task GOs seem to be rather stable over time. The results presented in Table 8.43. (Chapter 8) show the results for the correlational analyses of Hypothetical Task GOs at Time 1 and Time 2. These results indicate that the correlation coefficients for the four Hypothetical Task GOs at Time 1 and Time 2 range between 0.44 (p<0.01) and 0.63 (p<0.01). These correlations are of similar strength to those obtained for General GOs at Time 1 and Time Q2 varied quite a lot (from 4 to 51 weeks) similar analyses to those for General GOs were carried out in order to examine whether the stability of Hypothetical Task GOs decreased with an increasing time interval. The results of these analyses are presented in Tables 9.9. and 9.10. In both tables all the correlations of interest were found to be significant at p<.01. Consequently, the probability of these being the result of Type I error is highly unlikely.

Table 9.9. Correlations between Hypothetical Task GOs at Time 1 and Time 2 for participants who completed Time Q2 questionnaire 4-8 weeks after Time Q1 Questionnaire (N≈99)

Variables	1	2	3	4	5	6	7	8
1. HYP MAP T1	1.00							
2. HYP MAV T1	.52**	1.00						
3. HYP PAP T1	.38**	.38**	1.00					
4. HYP PAV T1	.36**	.55**	.62**	1.00				
5. HYP MAP T2	.53**	.18	.27**	.22*	1.00			
6. HYP MAV T2	.20	.46**	.14	.14	.47**	1.00		
7. HYP PAP T2	.10	.11	.54**	.37**	.39**	.25*	1.00	
8. HYP PAV T2	.12	.19	.37**	.34**	.57**	.51**	.76**	1.00

Key: HYP MAP = Hypothetical Task Mastery-Approach Goal Orientation; HYP MAV = Hypothetical Task Mastery-Avoidance Goal Orientation; HYP PAP = Hypothetical Task Performance-Approach Goal Orientation; HYP PAV = Hypothetical Task Performance-Avoidance Goal Orientation; T1 = Time 1; T2 = Time2 *p<.05 **p<.01

Table 9.10. Correlations between Hypothetical Task GOs at Time 1 and Time 2 for participants who completed Time Q2 questionnaire 9-51 weeks after Time Q1 Questionnaire (N≈98)

Variables	1	2	3	4	5	6	7	8
1. HYP MAP T1	1.00							
2. HYP MAV T1	.69**	1.00						
3. HYP PAP T1	.35**	.27**	1.00					
4. HYP PAV T1	.53**	.60**	.60**	1.00				
5. HYP MAP T2	.56**	.47	.16**	.34**	1.00			
6. HYP MAV T2	.51**	.57**	.07	.38**	.60*	1.00		
7. HYP PAP T2	.18	.19	.55**	.41**	.32**	.23*	1.00	
8. HYP PAV T2	.28**	.37**	.29**	.41**	.44**	.61**	.67**	1.00

Key: HYP MAP = Hypothetical Task Mastery-Approach Goal Orientation; HYP MAV = Hypothetical Task Mastery-Avoidance Goal Orientation; HYP PAP = Hypothetical Task Performance-Approach Goal Orientation; HYP PAV = Hypothetical Task Performance-Avoidance Goal Orientation; T1 = Time 1; T2 = Time2 p<.05 **p<.01

It was expected that the longer the time interval between Time Q1 and Time Q2 the lower the correlation coefficients would be. However, the results above seem to indicate that the longer the time interval between measures, the higher the correlation coefficients! However, when Fisher z tests were carried out these differences in the strengths of the correlation coefficients for the two different time intervals were not significant. In order to be able to assess the stability of Hypothetical Task GOs more objectively, the stability coefficients obtained are compared with those obtained in other studies. As mentioned earlier, Fryer and Elliot (2007) found correlations of 0.57 to 0.78 (p<0.001) whilst Elliot and McGregor (2001) found correlation coefficients ranging between 0.70 and 0.74 (p<0.01) for the four GOs over time. Moreover, as a result of their meta-analysis, Payne et al. (2007) obtained stability coefficient estimates of 0.66 (k=20), 0.70 (k=16) and 0.73 (k=4), with 'k' being the number of studies included, for mastery, PAP and PAV GOs, respectively. The correlations from the study by Payne et al. (2007) are for the stability of GOs from one to fourteen weeks. It seems as though the correlation coefficients obtained in previous studies are slightly higher than those obtained in the present study. However, overall, the results of this study nevertheless point towards Hypothetical task GOs being quite stable over time. These results will be discussed in further detail in Chapter 10 Section 10.7.

9.1.2.2. Assessing the Stability of Verbal Test Goal Orientations over time

The data included in the analyses for examining the stability of Verbal Test GOs over time were collected at Time E1 and Time E2. The mean-level change results are discussed first followed by the differential continuity results.

Analysis of Variance results indicated that there was no significant change in the mean level MAP GO on the Verbal test (for all three experimental groups) from Time 1 to Time 2 (F=.02; ρ =.89). Consequently, in terms of mean-level change, Verbal Test MAP GO seems to be stable over time. There was (only just!) a significant difference between the mean-level MAP GO scores of the different experimental groups (F=3.18; ρ =.05). However, the interaction effect of Time by Condition (which reflects the effect of the GO induction) was not significant (F=.02; ρ =.98). Therefore, it seems as though the MAP GO are across groups but NOT across groups over time. It may be that for some reason (other than the inductions) participants' in the different experimental groups adopted different levels of MAP GO. Possible reasons for this may be the small sample sizes in each experimental group and the fact that participants were not

completely randomly assigned to experimental conditions (for the reasons explained in the methodology, Chapter 7 Section 7.2.).

Similarly to the MAP Verbal Test GOs, there were no significant changes in participants' (in all experimental conditions) mean-level Verbal Test MAV GO from Time 1 to Time 2 (F=.86; ρ =.36). Also similarly to the Verbal Test MAV GO there was a significant difference in the mean-level MAV GO across experimental groups (F=4.59; ρ <.01). However, again, this was not a result of the inductions since the Time by Condition interaction effect was not significant (F=.17; ρ =.84). These results show that, in terms of mean-level change, the Verbal Test MAV GO seems to be stable over time.

No significant changes in the mean-level PAP GO on the Verbal Test (F=.01; ρ =.91) were found. Therefore, in terms of mean-level change, the Verbal Test PAP GO seems to be stable over time. The results also show that there were no significant differences in the mean-level PAP GO adopted by participants in different experimental groups (F=.43; ρ =.65). Moreover, there were no significant interaction effects between Time and Condition (F=.15; ρ =.86). Therefore, the mean-level PAP GO on the Verbal test did not change as a result of the inductions indicating that the PAP GO inductions on the Verbal Test were not successful.

There were no significant mean-level changes in participants' (in all experimental groups) Verbal Test PAV GO from Time 1 to Time 2 (F<.01, ρ =.92). Therefore, in terms of mean-level change, it seems as though all four Verbal Test GOs tend to remain stable over time. However, contrary to the Verbal Test MAP, MAV and PAP GOs there was a significant Time by Experimental Condition interaction (F=4.21, ρ =.02) for the Verbal Test PAV GO (as indicated in Table 9.11.). The mean-level PAV GO of participants in the PAP induction group seems to be decreasing from Time 1 to Time 2. It is possible that the focus on a PAP GO resulting from the induction caused participants' PAV GO to decrease. Therefore, although the PAP GO induction did not seem to be successful on the Verbal Test (since the Verbal Test PAP GO of participants in the PAP induction group did not increase significantly as a result of the inductions) it may have influenced participants' PAV GO. These results will be compared with those

obtained for the PAV Numerical Test GOs (Section 9.1.2.3.) in order to determine whether a pattern could be established.

Table 9.11.	Repeated	Measures	ANOVA	to	Examine	Mean-Level	Change	of	Verbal
Test Perform	nance-Avo	idance GO	s over Tir	ne	(N=71)				

Source of Variance	SS	Df	MS	F	Sig.
Between Subjects					
Intercept	63.85	1	63.85	43.76	<.01
Task Experience on Verbal Test	.04	1	.04	.03	.87
Practice on Verbal Test	3.29	1	3.29	2.25	.14
Experimental Condition	1.20	2	.60	.41	.67
Error	94.85	65	1.46		
Within Subject					
Time	<.01	1	<.01	<.01	.92
Time* Task Experience on Verbal Test	.07	1	.07	.25	.62
Time* Practice on Verbal Test	.11	1	.11	.39	.53
Time* Experimental Condition	2.36	2	1.18	4.21	.02
Error	18.26	65	.28		





Covariates appearing in the model are evaluated at the following values: Task Experience on Verbal Test Time 1 = 2.2628, Practice on Verbal Test between Time 1 and Time 2 = 1.5286

The stability of Verbal Test GOs was also assessed in terms of differential continuity. The results presented in Table 8.44. (Chapter 8) indicate the correlations for each of the four GOs on the Verbal Test from Time E1 to Time E2. These correlation coefficients range from 0.56 to 0.71 (ρ <0.01). Since they were all found to be significant at ρ <0.01, it is highly unlikely that they are the result of Type I error. These correlations are quite high and are similar in strength to the correlation coefficients obtained by Fryer and Elliot (2007) who obtained correlations of 0.57 to 0.78 (ρ <0.001). However, they are slightly lower than the coefficients obtained by Elliot and McGregor (2001). Overall, the correlation coefficients for Verbal Test GOs over time seem to indicate a good level of stability.

9.1.2.3. Assessing the Stability of Numerical Test Goal Orientations over time

The stability of Numerical Test GOs was also assessed in terms of mean-level change (using Analysis of Variance) and differential continuity. As was done in the previous section the mean-level change results are presented first.

No significant differences were found in the mean-level MAP GO adopted by participants (in all three experimental groups) on the Numerical tests from Time 1 to Time 2. Therefore, in terms of mean-level change, the Numerical Test MAP GO indicates stability over time. There were also no significant differences in the mean-level MAP GO across the different experimental groups (F=2.14; ρ =.13). Moreover, the interaction effects between Time and experimental condition were non-significant (F=.44; ρ =.65). Therefore, the mean-level MAP GO adopted by participants in different experimental groups did not change as a result of the inductions on the Numerical Test. Consequently, the MAP induction does not seem to have worked on either the Verbal or the Numerical tests.

No significant changes were found in the mean-level MAV GO of participants (in all experimental groups) on the Numerical test from Time 1 to Time 2 (F=.36; ρ =.55). There were also no statistically significant differences in the mean-level MAV GO for participants in different experimental groups.

The results obtained for the Numerical Test PAP GO over time are consistent with those for the MAP and MAV Numerical Test GOs over time. No significant changes were found in the mean-level PAP GO on the Numerical Test from Time 1 to Time 2. Moreover, there were no significant interaction effects between Time and experimental Condition (F=.84; ρ =.44). Consequently, it seems as though the PAP GO induction was not successful on the Numerical test either.

Similarly to the other Numerical Test GOs, there were no significant changes in the mean-level PAV GO of participants (in all experimental groups) on the Numerical Test from Time 1 to Time 2 (F=.24; ρ =.63). The differences in the mean-level PAV GO across experimental groups were also statistically non-significant. In addition, contrary to the results obtained on the Verbal Test, there was no significant time by condition interaction effect. Thus, the PAP induction did not influence the PAV GO of participants on the Numerical Test. Overall the results indicate that, in terms of mean-level change, participants Numerical Test GOs seem to remain stable from Time 1 to Time 2.

Unfortunately, the results indicate that neither the MAP nor the PAP GO inductions were successful in this experimental study. Although the PAP induction seems to have influenced the Verbal Test PAV GO, this was not found to be the case for the Numerical Test PAV GO. It is not clear why the PAP induction seemed to influence the PAV GO of participants on the Verbal Test but not on the Numerical Test. Again, it is possible that this relationship was influenced by task type. For example, if participants do not feel as confident about their numeracy skills as they do about their verbal skills, the focus on a PAP GO might not be enough to decrease their PAV GO. Further investigation of GO inductions is necessary in order to be able to understand these relationships better. This will be discussed in further detail in Sections 10.10. and 11.3.

When the LCAs of Verbal and Numerical Test GOs were carried out (Chapter 8 Sections 8.5.3. and 8.5.4.), it was found that 34 participants adopted the 'high GOs' profile on the Verbal Test at Time 1 and 40 participants adopted this GO profile on the Numerical Test (at Time 1). The inductions would not have had any influence on the GOs of these

participants since they already had high MAP and PAP GOs. Ideally, in order to test the success of the inductions, the analyses should only have included participants who had low to moderate MAP GOs (to test the MAP GO induction) and PAP GOs (to test the PAP GO induction). However, due to the small sample size this was not feasible as there would have been only about 12 participants in each experimental group if the participants having a 'high GOs' profile were not included in the analyses. The implications of this are discussed in further detail in Chapter 10 (Section 10.10) and Chapter 11 (Section 11.2.).

In terms of differential continuity for Numerical Test GOs over time, Table 8.44. (Chapter 8) shows the results obtained for the correlational analyses of participants' Numerical Test GOs over time. The correlation coefficients obtained range from 0.52 to 0.70 (p<0.01). Like the correlation coefficients obtained on the Verbal Tests, since they were all found to be significant at p<0.01, it is highly unlikely that they are the result of Type I error. Again, similarly to the Verbal Test correlations the Numerical Test correlations are quite similar to those obtained by Fryer and Elliot (2007). However, once again, they are slightly lower than those obtained by Elliot and McGregor (2001). Overall, these correlation coefficients seem to indicate a good level of stability for participants' Numerical Test GOs over time. The overall results of these analyses indicate that both in terms of mean-level change as well as in terms of differential continuity there seems to be evidence of stability for task-specific GOs over time. These results will be discussed in further detail in Chapter 10 Section 10.7.

Before assessing the stability of task-specific GO **profiles** over time (in answer to Research Question 5a) a sub-section of this Chapter is dedicated to comparing the stability of General GOs with that of task-specific GOs. This decision was made due to the fact that, from the results presented above, it seems as though General GOs are less stable over time than task-specific GOs. If this is the case, then GOs might be more accurately measured using task-specific measures as opposed to General measures. The comparative results are presented in Section 9.1.2.4.

9.1.2.4. Comparing the Stability of General Goal Orientations to that of Task-Specific Goal Orientations.

In order test whether there are significant differences between the stability coefficients of General GOs and Task-Specific GOs Fisher z tests were carried out. The stability coefficient of each General GO was compared with that of the corresponding Hypothetical Task, Verbal Test and Numerical Test GOs. In addition, since the Hypothetical Task GOs stability coefficients seemed to be lower than the Verbal and Numerical Test ones, the stability coefficient of each of the Hypothetical Task GO was compared with that of the corresponding Verbal and Numerical Test GOs.

For General and Hypothetical Task GOs the time intervals between Q1 and Q2 ranged from 5 to 51 weeks. On the other hand for Verbal and Numerical Test GOs the time intervals between Time E1 and Time E2 ranged from 1 to 31 weeks. In order to reduce variation it was decided that only those participants who had a time interval of up to 10 weeks between Time Q1 and Time Q2 (for General and Hypothetical Task GOs) and between Time E1 and Time E2 (for Verbal and Numerical Test GOs) should be included in the analyses. In this way, any significant differences in the stability would **not** be a result of the different time intervals. The results are presented in Table 9.12.

Table 9.12. Testing for Significant Differences between the Stability Coefficients of General and Task-specific GOs and between Hypothetical Task and Aptitude Test GOs

GO	Compariso	on and Ns	z-statistic	p-value
МАР	GEN (N=128)	HYP (N=127)	0.12	0.90
MAV	GEN (N=124)	HYP (N=127)	0.17	0.34
ΡΑΡ	GEN (N=128)	HYP (N=126)	0.45	0.90
PAV	GEN (N=128)	HYP (N=127)	0.67	0.50
ΜΑΡ	GEN (N=128)	VT (N=56)	-0.78	0.44
MAV	GEN (N=124)	VT (N=56)	-1.97	0.05
ΡΑΡ	GEN (N=128)	VT (N=56)	-1.31	0.19
PAV	GEN (N=128)	VT (N=56)	-2.26	0.02
МАР	GEN (N=128)	NT (N=56)	-1	0.30
MAV	GEN (N=124)	NT (N=54)	-1.66	0.10
ΡΑΡ	GEN (N=128)	NT (N=56)	-0.61	0.54
PAV	GEN (N=128)	NT (N=56)	-2.38	0.02
ΜΑΡ	HYP (N=127)	VT (N=56)	-0.87	0.38
MAV	HYP (N=127)	VT (N=56)	-1.23	0.22
ΡΑΡ	HYP (N=126)	VT (N=56)	-1.21	0.23
PAV	HYP (N=127)	VT (N=56)	-2.78	<0.01
МАР	HYP (N=127)	NT (N=56)	-1.08	0.28
MAV	HYP (N=127)	NT (N=54)	-0.93	0.35
РАР	HYP (N=126)	NT (N=56)	-0.51	0.61
PAV	HYP (N=127)	NT (N=56)	-2.9	<0.01

Key: MAP=Mastery-Approach Goal Orientation; MAV=Mastery-Avoidance Goal Orientation; PAP=Performance-Approach Goal Orientation; PAV=Performance-Avoidance Goal Orientation.

The results presented in Table 9.12. indicate that Verbal and Numerical Test PAV GOs are significantly more stable than General and Hypothetical Task PAV GOs. Moreover,

it seems as though a Verbal Test MAV GO is significantly more stable than a General MAV GO. Consequently, it seems as though even after the attempted inductions (which may have contaminated the stability), Verbal and Numerical Test GOs look to be quite stable over time. No other significant differences between the correlation coefficients were found. These results are discussed in further detail in Chapter 10 Section 10.7. The next sub-section presents the results of the analyses carried out in order to answer Research Question 5a.

9.1.2.5. Research Question 5a. Are task-specific goal orientation **profiles** stable over time?

In order to answer this Research Question the results from the LCAs of Hypothetical Task, Verbal Test and Numerical Test GOs at Time 1 and Time 2 were used (Sections 8.5.2., 8.5.3. and 8.5.4., respectively). The GO profiles adopted by participants at Time 1 and Time 2 on each task are compared. The comparison of Hypothetical Task GO profiles at Time 1 and Time 2 are presented first. This is followed by the analyses of the Verbal and Numerical Test GO profiles over time, respectively.

	Hypothetical Task GO Profile Cluster at Time 2						
_ e e		1	2	3			
rofi Tir	1	106	24	8	138		
1 1	2	19	15	4	38		
Hypo Isk G uste	3	0	7	2	9		
C 1 ^a +	4	1	0	1	2		
	Total	126	46	15	187		

Table 9.13. Frequency Table indicating the Hypothetical Task Goal OrientationProfiles Adopted by Participants at Time 1 and Time 2. (N=187)

Table 9.13. indicates that 123 participants (66%) adopted the same GO profile on the Hypothetical Task at Time 1 and Time 2. On the other hand, 64 participants (34%) adopted a different GO profile at Time 2 than they did at Time 1. These frequencies were used to calculate the proportion of participants that would be expected to remain in the same clusters by chance alone (Table 9.14.)

Table 9.14. Observed Frequencies and Proportions Expected to Remain in the sameClusters (from Time 1 to Time 2) by Chance

	Cluster 1	Cluster 2	Cluster 3
Observed	106	15	2
Proportions by chance	93	9	1

Following from the results presented in Table 9.14., and keeping in mind that the total number of participants was 187, the total numbers of participants remaining in the same cluster and those changing cluster from Time 1 to Time 2 is presented in Table 9.15. for the Observed Frequencies and the Proportions by Chance.

Table 9.15. Total Numbers of Participants remaining in the Same Cluster and TotalNumber Changing Cluster from Time 1 to Time 2

	Remain in Same Cluster from Time 1 to Time 2	Change Cluster from Time 1 to Time 2	Total
Observed	123	64	187
Proportions by chance	103	84	187
Total	226	148	374

A Chi-square test was then carried out using the frequencies presented in Table 9.15. The X^2_{obt} in this case was 4.46 whilst X^2_{crit} is 6.64 (ρ <.01) and 3.84 (ρ <.05). Therefore, the results indicate that more participants remained in the same clusters over time than they would have by chance alone at a significance level of ρ <.05. The results presented above will also be compared with those obtained for the Verbal and Numerical Test GO profiles over time. The results of the analyses for assessing stability of Verbal Test GO profiles over time are presented next.

The data collected at Time E1 and Time E2 were used in the analyses of Verbal Test GO profiles over time. The analyses of the stability of Verbal Test GO profiles over time are presented below.

Table 9.16. Frequency Table indicating the Verbal Test GO Profiles Adopted byParticipants at Time 1 and Time 2. (N=68)

	Ver	Total			
st t		1	2	3	
ll Te rofil er a	1	23	7	2	32
erba O Pl Ilust Tim	2	9	12	2	23
N O O	3	2	3	8	13
	Total	34	22	12	68

The frequency table shows that 43 participants (63%) adopted the same GO profile on the Verbal Test at Time 1 and Time 2 whilst 25 participants (37%) adopted different GO profiles on the Verbal Test at Time 1 and Time 2. In order to assess the stability of Verbal Test GO profiles further, the computations described in Section 9.1.1. were carried out and the results are presented in Tables 9.17. and 9.18.

Table 9.17. Observed Frequencies and Proportions Expected to Remain in the sameClusters (from Time 1 to Time 2) by Chance

	Cluster 1	Cluster 2	Cluster 3
Observed	23	12	8
Proportions by chance	16	7	2

Following from the results presented in Table 9.17., and keeping in mind that the total number of participants was 68, the total numbers of participants remaining in the same cluster and those changing cluster from Time 1 to Time 2 is presented in Table 9.18. for the Observed Frequencies and the Proportions by Chance.

Table 9.18. Total	Numbers	of Participants	remaining in	the Same	e Cluster	and	Total
Number Changing	; Cluster fr	om Time 1 to Ti	ime 2				

	Remain in Same Cluster from Time 1 to Time 2	Change Cluster from Time 1 to Time 2	Total
Observed	43	25	68
Proportions by chance	25	43	68
Total	68	68	136

A Chi-square test was carried out using the frequencies presented in Table 9.18. The X^2_{obt} in this case was 9.53 whilst X^2_{crit} is 6.64 (ρ <.01). Therefore, the results indicate that more participants remained in the same clusters over time than they would have by chance alone. These results seem to indicate a possibility of Verbal Test GO profiles being stable over time. The results of the analyses assessing the stability of Numerical Test GO profiles over time are presented next.

Table 9.19. Frequency Table indicating the Numerical Test Goal Orientation Profiles Adopted by Participants at Time 1 and Time 2. (N=71)

	Num	Гime 2	Total		
Vumerical Test GO Profile Cluster at Time 1		1	2	3	
	1	30	10	0	40
	2	4	5	4	13
	3	1	7	5	13
	4	2	3	0	5
	Total	37	25	9	71

Table 9.19. shows that 40 participants (56%) adopted the same GO profile on the Numerical Test at both Time 1 and Time 2. On the other hand, 31 participants (44%) adopted different GO profiles on the Numerical Test at Time 1 and Time 2. The same computations used to assess the stability of Verbal Test GO profiles over time were used here. The results are presented below.

Table 9.20. Observed Frequencies and Proportions Expected to Remain in the sameClusters (from Time 1 to Time 2) by Chance

	Cluster 1	Cluster 2	Cluster 3
Observed	30	5	5
Proportions by chance	21	5	2

Following from the results presented in Table 9.20., and keeping in mind that the total number of participants was 71, the total numbers of participants remaining in the same cluster and those changing cluster from Time 1 to Time 2 are presented in Table 9.21. for the Observed Frequencies and the Proportions by Chance.

Table 9.21. Total Numbers of Participants remaining in the Same Cluster and TotalNumber Changing Cluster from Time 1 to Time 2

	Remain in Same Cluster from Time 1 to Time 2	Change Cluster from Time 1 to Time 2	Total
Observed	40	31	71
Proportions by chance	28	43	71
Total	68	74	142

A Chi-square test was carried out using the frequencies presented in Table 9.21. The X^2_{obt} in this case was 4.07 whilst X^2_{crit} is 6.64 (ρ <.01) and 3.84 (ρ <.05). Therefore, the results indicate that more participants remained in the same clusters over time than they would have by chance alone at ρ <.05. These results point towards the possibility of Numerical Test GO profiles being stable over time.

Overall, the task-specific GO profile results seem to indicate that task-specific GO profiles may be stable over time. The results indicate that for Verbal Test GO profiles one can be confident at a significance level of 0.01 that more participants remained in the same clusters over time than they would have by chance alone. In contrast, for Hypothetical Task and Numerical Test GO profiles one can be confident at a significance level of 0.05 that that more participants remained in the same clusters over time than they explore alone. The results obtained in the same clusters over time than they would have by chance alone. The results obtained in this study do not provide any definite conclusions regarding the stability of task-specific GO profiles over time. They do, however, provide an initial indication. Recommendations for future research investigating the stability of task-specific GO profiles over time will be made in Chapter 11 Section 11.3. The next section of this chapter addresses the issue of Generality versus Task-Specificity of GOs.

9.2. Assessing Generality and/or Task-Specificity of Goal Orientations

This section focuses on examining whether GOs are general or whether they are more accurately measured using task-specific measures. The results presented in this section address Research Questions 4a, 4b, 4c and 4d. First the Hypothetical Task, Verbal Test and Numerical Test GOs are compared (from the profile and non-profile perspectives). Following this, participants' General GOs are compared with their Hypothetical Task, Verbal Test and Numerical Test GOs (from the profile and non-profile perspectives).

9.2.1. Assessing the Task-Specificity of Goal Orientation Profiles

This sub-section addresses Research Question 4a.

Research Question 4a. Do participants adopt different goal orientation profiles across tasks?

In order to answer this Research Question three comparisons were made. Firstly, Hypothetical Task GO profiles (at Time 1) were compared with Verbal Test GO profiles (at Time 1). Secondly, Hypothetical Task GO profiles (at Time 1) were compared with Numerical Test GO profiles (at Time 1) and thirdly, Verbal Test GO profiles (at Time 1) were compared with Numerical Test GO profiles (at Time 1). The results of the LCAs for GOs on the Hypothetical Task (Table 8.31., Table 8.32., Figure 8.3.) Verbal Test (Table 8.35., Table 8.36., Figure 8.5.) and Numerical Test (Table 8.39., Table 8.40., Figure 8.7.) at Time 1 were used for making these comparisons.

Before comparing the GO profiles on different tasks it was necessary to compare the clusters obtained from the LCAs of the different task-specific GOs since it would only be possible to compare like with like. For example, it was necessary to check whether the 3 GO profile clusters obtained from the LCA of Verbal Test GOs (at Time 1) were similar to 3 of the 4 GO profile clusters obtained from the LCA of Hypothetical Task GOs (at Time 1). The comparison of the clusters indicated that the GO profiles adopted on different tasks were very similar with the exception of Cluster 4 on the Hypothetical Task and Numerical Test. Consequently, (with the exception of Cluster 4 on the Hypothetical Task are assumed to have adopted the same GO profiles. Again, since the clusters are not *exactly* the same analyses of whether participants change clusters or stay in the same one is necessarily approximate.

The posterior classification results used in the analyses of data for answering Research Question 5a were used to draw up the frequency tables indicating the GO profiles adopted by participants across tasks. The results comparing the GO profiles adopted by participants on different tasks are presented in Tables 9.22. through to 9.30. Hypothetical Task and Verbal Test GO profiles will be compared first. Next, Hypothetical Task and Numerical Test GO profiles will be compared. Finally, a comparison of the GO profiles adopted by participants on the Verbal and Numerical Tests is presented. The computations described in Section 9.1.1. are used to assess the task-specificity of GO profiles.

Table 9.22. Frequency Table indicating the Hypothetical Task and Verbal Test Goal Orientation Profiles Adopted by Participants at Time 1 (N=67)

	١	Verbal Test GO Profile Cluster at Time 1						
al		1	2	3				
hetic GO Clust me 1	1	31	15	6	52			
poth Task ofile at Tin	2	1	5	5	11			
H Pr	3	1	1	2	4			
	Total	33	21	13	67			

The results in Table 9.22. show that 38 participants (57%) adopted the same GO profile on the Hypothetical Task and the Verbal Test whilst 29 participants (43%) adopted a different GO profile. Therefore, nearly half of the participants adopted a different GO profile on the different tasks. The observed frequencies of participants in each cluster as well as the proportions of participants expected to be in each cluster by chance are presented in Table 9.23.

Table 9.23. Observed Frequencies and Proportions Expected to Remain in the sam	۱e
Clusters across the Hypothetical Task and Verbal Test by Chance	

	Cluster 1	Cluster 2	Cluster 3
Observed	31	5	2
Proportions by chance	26	3	1

Following from the results presented in Table 9.23., and keeping in mind that the total number of participants was 67, the total numbers of participants remaining in the same cluster and those changing cluster across the Hypothetical Task and Verbal Test are presented in Table 9.24. for the Observed Frequencies and the Proportions by Chance.

Table 9.	24. Total	Numbers	of F	Participants	remaining i	n the	e Same	Cluster	and	Total
Number	Changing	g Cluster a	cros	s the Hypot	hetical Task	and	Verbal	Test		

	Remain in SameChange ClusterCluster across tasksacross tasks		Total
Observed	38	29	67
Proportions by chance	30	37	67
Total	68	66	134

A Chi-square test was carried out using the frequencies presented in Table 9.24. The X^{2}_{obt} in this case was 1.87 whilst X^{2}_{crit} is 6.64 (ρ <.01) and 3.84 (ρ <.05). Therefore, the results indicate that more participants would have remained in the same clusters across tasks by chance alone thus pointing towards participants adopting different Hypothetical Task and Verbal Test GO profiles. The implications of these findings will be discussed in Chapter 10 Section 10.8.1. A comparison of Hypothetical Task and Numerical Test GO profiles is presented next.

Goal Orient	ation Profiles Adopted by Participants at Time 1 (N=67)	
	Numerical Test GO Profile Cluster at Time 1	Total

Table 9.25. Frequency Table indicating the Hypothetical Task and Numerical Test

	Numerical Test GO Profile Cluster at Time 1					
al ier		1	2	3	4	
hetic GO Clust me 1	1	36	6	5	5	52
/potl Task ofile at Tii	2	2	4	5	0	11
H r	3	1	2	1	0	4
	Total	39	12	11	5	67

Table 9.25. shows that 41 participants (61%) adopted the same GO profile on the Hypothetical Task as they did on the Numerical Test. Once again the observed frequencies were compared with the frequencies one would expect to obtain by chance. The results of this comparison are presented in Tables 9.26. and 9.27.

Table 9.26. Observed Frequencies and Proportions Expected to Remain in the sameClusters across the Hypothetical Task and Numerical Test by Chance

	Cluster 1	Cluster 2	Cluster 3
Observed	36	4	1
Proportions by chance	30	2	1

Following from the results presented in Table 9.26., and keeping in mind that the total number of participants was 67, the total numbers of participants remaining in the same cluster and those changing cluster across the Hypothetical Task and Numerical Test are presented in Table 9.27. for the Observed Frequencies and the Proportions by Chance.

Table 9.27. Total Numbers of Participants remaining in the Same Cluster and TotalNumber Changing Cluster across the Hypothetical Task and Numerical Test

	Remain in Same Cluster across tasks	Change Cluster across tasks	Total
Observed	41	26	67
Proportions by chance	33	34	67
Total	74	60	134

The X^2_{obt} from the Chi-square test in this case was 1.93 whilst X^2_{crit} is 6.64 (ρ <.01) and 3.84 (ρ <.05). Therefore, the results indicate that more participants would have remained in the same clusters across tasks by chance alone. These results point towards the possibility that participants seem to adopt different Hypothetical Task and Numerical Test GO profiles. The implications of these findings will be discussed in Chapter 10 Section 10.8.1. and Chapter 11 Section 11.1. The comparison of Verbal and Numerical Tests GO profiles at Time 1 is presented next.

		Total				
me t		1	2	3	4	
l Tes rofilo at Ti	1	31	1	0	2	34
erba 30 P ster	2	7	10	3	3	23
Clu o <	3	2	2	10	0	14
	Total	40	13	13	5	71

Table 9.28. Frequency Table indicating the Verbal and Numerical Test Goal Orientation Profiles Adopted by Participants at Time 1 (N=71)

51 participants (72%) seem to have adopted the same GO profiles on the Verbal and Numerical Tests. Therefore, it seems as though more participants adopted the same GO profiles across the Verbal and Numerical Tests than they did across the Hypothetical Task and Verbal and Numerical Tests, respectively. These observed frequencies were compared with the frequencies one would expect to obtain by chance (refer to Tables 9.29. and 9.30.)

Table 9.29. Observed Frequencies and Proportions Expected to Remain in the sameClusters across the Verbal and Numerical Tests by Chance

	Cluster 1	Cluster 2	Cluster 3
Observed	31	10	10
Proportions by chance	19	4	3

The total numbers of participants remaining in the same cluster and those changing cluster across the Verbal and Numerical Tests are presented in Table 9.30.

Table 9	9.30.	Total	Numbers	of	Participants	remaining	in	the	Same	Cluster	and	Total
Numbe	er Cha	anging	Cluster a	cro	ss the Verba	l and Nume	eric	al Te	ests			

	Remain in Same Cluster across tasks	Change Cluster across tasks	Total	
Observed	51	20	71	
Proportions by chance	26	45	71	
Total	77	65	142	

The Chi-square test produced a X^2_{obt} of 17.7 (X^2_{crit} is 6.64, ρ <.01). Therefore, the results indicate that (many!) more participants remained in the same clusters across the Verbal and Numerical Tests than they would have by chance alone thus showing that participants seem to adopt very similar Verbal and Numerical Test GO profiles.

There are three possible explanations for more participants adopting different Hypothetical Task and Verbal (and Numerical) Test GO profiles but similar Verbal and Numerical Test GO profiles. Firstly, the Hypothetical Task GOs were measured at a different point in time than the Verbal and Numerical Test GOs. Consequently, the adoption of different GO profiles may be due to lack of stability over time rather than across tasks (from here on referred to as 'time differences'). Since Verbal and Numerical Test GOs were measured during the same experimental session (Time E1), this might have decreased the variation in GO profiles adopted on these two tasks.

Secondly, the Verbal and Numerical Test GOs were measured under very similar conditions whilst the Hypothetical Task was measured in a very different context. The Verbal and Numerical Test GOs were measured under experimental conditions (at Time E1) whilst the Hypothetical Task GOs were measured during the survey (at Time Q1). Consequently, this variation in the adoption of GO profiles may have resulted due to situational variance (from here on referred to as 'situational differences').

Thirdly, the adoption of different GO profiles may have resulted from task variation since the Verbal and Numerical Tests were more similar to each other than they were to the Hypothetical Task. It may be said that Verbal and Numerical Tests fall into the same domain, possibly the 'academic domain' whilst the Hypothetical Tasks cannot be classified in the same domain as the Verbal and Numerical Tests. Therefore, it is possible that GO profiles are domain-specific.

Unfortunately, due to these different possibilities, it is not possible to answer this Research Question in a straightforward manner. However, the results do seem to indicate a strong possibility that GO profiles are domain-specific. These findings will be

257

discussed in further detail in Chapter 10 Section 10.8.1. Next the task-specificity of GOs from the non-profile perspective is presented.

9.2.2. Assessing the Task-Specificity of Goal Orientations

It was thought necessary to investigate the task-specificity of GOs from the non-profile perspective since this might provide further insight into the results obtained with respect to the task-specificity of GO *profiles*. This section addresses Research Question 4c.

Research Question 4c. Do participants' adopt different goal orientations across tasks?

The data collected at Time Q1 and Time E1 were used in these analyses. In order to answer this Research Question participants' Hypothetical Task GOs were compared with their Verbal and Numerical Test GOs. Moreover, participants' Verbal and Numerical Test GOs were compared with each other. Task-specificity was assessed in terms of mean-level change as well as differential continuity. First the differential continuity results will be presented. Following this the mean-level change results will be presented.

The correlational analyses results for assessing the differential continuity of GOs across tasks were provided in Chapter 8 Tables 8.44 and 8.45. The correlation coefficients obtained for Hypothetical Task GOs and Verbal Test GOs ranged from non-significant to 0.61 (ρ <.01). Since there was only one non-significant correlation (for the MAV GOs) and the rest were significant at ρ <.01, it is quite unlikely that the significant correlations found were a result of Type I error. The correlations for Hypothetical Task GOs ranged between non-significant and 0.57 (ρ <.01). In this case, the MAP and MAV GO correlations should be interpreted with caution due to the possibility of Type I error. The PAP and PAV GO correlations were all significant at ρ <.01. The correlation coefficients for Verbal and Numerical Test GOs (at Time 1) ranged from 0.68 to 0.81 (ρ <.01). Since they were all significant at ρ <.01, it is highly unlikely that they are the result of Type I error. The first noticeable thing is that the

correlation coefficients for Verbal and Numerical Tests are much stronger than those for the Hypothetical Task and Verbal and Numerical Tests, respectively. In order to test whether there were any significant differences between the strengths of these correlations Fisher z tests were carried out. The results are presented in Table 9.31.

In order to make things slightly simpler, the correlations between the Hypothetical Task and Verbal Test GOs will be referred to as 'Hypothetical Verbal'. The correlations between the Hypothetical Task and the Numerical Test GOs will be referred to as 'Hypothetical Numerical' and the correlations between the Verbal and Numerical Test GOs will be referred to as 'Aptitude Test'.

Table 9.31. Testing for Significant Differences in the correlation coefficients of the Task-Specific GOs

GO	Compariso	on and Ns	z-statistic	p-value	
MAP	HYPV	APT	-1.58	.11	
	(N=69)	(N=71)			
ΝΛΛΥ	HYPV	APT	2 0 2	< 01	
IVIAV	(N=68)	(N=70)	-2.92	<.01	
DAD	HYPV	APT	2.67	< 01	
PAP	(N=68)	(N=71)	-2.67	<.01	
	HYPV	APT	2.1	< 01	
PAV	(N=68)	(N=71)	-3.1	<.01	
	HYPN	APT	2 22	02	
IVIAP	(N=69)	(N=71)	-2.32	.02	
NAA \/	HYPN	APT	2.10	< 01	
IVIAV	(N=67)	(N=70)	-3.10	<.01	
	HYPN	APT	2 21	02	
PAP	(N=68)	(N=71)	-2.21	.03	
DAV	HYPN	APT	2.95	<.01	
PAV	(N=68)	(N=71)	-3.85		

Key: HYPV=Correlation between Hypothetical Task and Verbal Test Goal Orientations; HYPN= Correlation between Hypothetical Task and Numerical Test Goal Orientations; APT= Correlation between Verbal and Numerical Test Goal Orientations.

The Fisher z test results indicate clearly that there are significant differences between the strengths of the correlation coefficients of GOs for the Hypothetical Verbal and Aptitude Test as well as for Hypothetical Numerical and Aptitude Tests. The only exception is for the Hypothetical Verbal and Aptitude Test MAP GO. It seems as though there is significantly more stability of GOs across the Aptitude Tests than there is across the Hypothetical Tasks and aptitude tests. This is consistent with the results obtained when assessing the task-specificity of GO profiles. These differential continuity results will be compared with the mean-level change results (presented below).

Repeated measures Analyses of Variance were carried out using the data collected at Time Q1 and Time E1 to examine the mean-level change of participants' GOs across tasks. The mean GO scores on each task were presented in Chapter 8 Section 8.2. Task Experience was included as a covariate in these analyses since it might have influenced the type of GOs that participants adopted on the Verbal and Numerical Tests. Practice was not included this time since only the data collected at Time 1 were used. Practice refers to the amount of practice on tests between Time 1 and Time 2 therefore it was not relevant in this case. This time the task was the independent variable and the GO being assessed was the dependent variable. The significant results are presented in Table 9.32. through to Table 9.38.

Table	9.32.	Assessing	Mean	Level	Change	of	Mastery-Approach	GOs	across	the
Hypot	hetica	l Task and '	Verbal	Test (N	l=69)					

Source of Variance	SS	Df	MS	F	Sig.
Between Subjects					
Intercept	258.84	1	258.84	457.90	<.01
Task Experience on Verbal Test	<.01	1	<.01	<.01	.95
Error	37.87	67	.57		
Within Subject					
Task	2.53	1	2.53	9.51	<.01
Task * Task Experience on Verbal Test	1.45	1	1.45	5.44	.02
Error	17.83	67	.27		

Table 9.33.	Assessing N	/lean Level	Change of	Performance	-Approach	GO	across	the
Hypothetica	I Task and V	/erbal Test	(N=68)					

Source of Variance	SS	Df	MS	F	Sig.
Between Subjects					
Intercept	237.10	1	237.10	264.80	<.01
Task Experience on Verbal Test	.32	1	.32	.36	.55
Error	59.10	66	.90		
Within Subject					
Task	4.53	1	4.53	11.03	<.01
Task * Task Experience on Verbal Test	2.00	1	2.00	4.85	.03
Error	27.07	66	.41		

Table 9.34. Assessing Mean Level Change of Performance-Avoidance GO across the

Source of Variance	SS	Df	MS	F	Sig.
Between Subjects					
Intercept	237.46	1	237.46	251.36	<.01
Task Experience on Verbal Test	.08	1	.08	.08	.78
Error	62.35	66	.95		
Within Subject					
Task	6.11	1	6.11	22.92	<.01
Task * Task Experience on Verbal Test	3.42	1	3.42	12.84	<.01
Error	17.59	66	.27		

Hypothetical Task and Verbal Test (N=68)

Table 9.35. Assessing Mean Level Change of Mastery-Approach GO across the

Source of Variance	SS	Df	MS	F	Sig.
Between Subjects					
Intercept	183.83	1	183.83	294.62	<.01
Task Experience on Numerical Test	1.66	1	1.66	2.65	.11
Error	41.81	67	.62		
Within Subject					
Task	3.34	1	3.34	8.74	<.01
Task * Task Experience on Numerical Test	1.30	1	1.30	3.41	.07
Error	25.58	67	.38		

Hypothetical Task and Numerical Test (N=69)

Table 9.36. Assessing Mean Level Change of Mastery-Avoidance GO across theHypothetical Task and Numerical Test (N=67)

Source of Variance	SS	Df	MS	F	Sig.
Between Subjects					
Intercept	156.87	1	156.87	244.56	<.01
Task Experience on Numerical Test	1.11	1	1.11	1.73	.19
Error	41.69	65	.64		
Within Subject					
Task	3.42	1	3.42	8.45	<.01
Task * Task Experience on Numerical Test	1.09	1	1.09	2.68	.11
Error	26.34	65	.41		

Table 9.37. Assessing Mean Level Change of Performance-Approach GO across the

Source of Variance	SS	Df	MS	F	Sig.
Between Subjects					
Intercept	149.80	1	149.80	141.62	<.01
Task Experience on Numerical Test	2.35	1	2.35	2.22	.14
Error	69.81	66	1.06		
Within Subject					
Task	4.00	1	4.00	8.36	<.01
Task * Task Experience on Numerical Test	1.25	1	1.25	2.61	.11
Error	31.58	66	.48		

Hypothetical Task and Numerical Test (N=68)

Table 9.38. Assessing Mean Level Change of Performance-Avoidance GO across the

Source of Variance	SS	Df	MS	F	Sig.
Between Subjects					
Intercept	153.87	1	153.87	149.67	<.01
Task Experience on Numerical Test	3.24	1	3.24	3.16	.08
Error	67.85	66	1.03		
Within Subject					
Task	3.93	1	3.93	11.03	<.01
Task * Task Experience on Numerical Test	1.77	1	1.77	4.97	.03
Error	23.52	66	.36		

Hypothetical Task and Numerical Test (N=68)

No significant differences in mean-levels across Verbal and Numerical Test GOs were found. However, participants' Hypothetical Task GOs were significantly higher than their Verbal and Numerical Test GOs in all cases except for the MAV GOs across the Hypothetical Task and Verbal Test. These results reflect the Fisher z test results presented in Table 9.12. The only difference is that for the Fisher z results there were no significant differences in the correlation strengths for the MAP Hypothetical Verbal and Aptitude Test and a significant difference in the correlation strengths for the MAV Hypothetical Verbal and Aptitude Test.

As discussed earlier, it may be the case that there were no significant differences between the Verbal and Numerical Test GOs because these are quite similar to each other, in that they are both academic tasks. On the other hand, the Hypothetical Tasks were very different from both the Verbal and the Numerical Tests. Consequently, in answer to Research Question 4c, yes, GOs do seem to change significantly across tasks. However, these differences only seem to occur when the tasks are very different from each other. In fact, it seems possible that GOs may be domain-specific as opposed to task-specific. This possibility will be discussed further in Chapter 10 Section 10.8.1. In order to eliminate the possibility that these observed differences were due to situational differences (survey vs. experiment) or time differences (Verbal and Numerical Test were measured on the same day whilst Hypothetical Task GOs were measured on a different day) further research is required. This will be discussed in Chapter 11 Section 11.3.

9.2.3. Examining Differences between General and Task-Specific Goal Orientation Profiles

In order to assess the utility of using task-specific versus general measures of GOs in research it was thought useful to examine whether there are any differences between the General GOs of participants and their task-specific GOs. The differences between General and Task-Specific GO *profiles* of participants are addressed first (in response to Research Question 4b) followed by the analyses of differences from the non-profile perspective (Section 9.2.4.).

Research Question 4b. Are participants' task-specific goal orientation profiles different from their General goal orientation profiles?

The results of the analyses presented in Sections 8.5.1., 8.5.2., 8.5.3. and 8.5.4. (which presented the results of the LCAs of General and Task-specific GOs) were used to answer this Research Question. The posterior classification of participants to clusters (carried out in order to answer Research Questions 2a and 5a) was used to assess whether participants General GO profiles were different from their task-specific GO profiles. The results of these comparisons are presented in Table 9.39. through to Table 9.47. The comparison between the General and Hypothetical Task GO profiles is presented first.

	Hypothetical Task GO Profile Cluster at Time 1						
0		1	2	3	4		
1 at le	1	394	63	12	2	471	
eral ofi ster me	2	78	41	10	0	129	
ene Clu Ti	3	1	4	3	1	9	
0	4	1	4	0	3	8	
	Total	474	112	25	6	617	

Table 9.39. Frequency Table indicating the General and Hypothetical Task Goal Orientation Profiles Adopted by Participants at Time 1 (N=617)

The results presented above show that 441 participants (71%) adopted the same General and Hypothetical Task GO profiles whilst 176 participants (29%) adopted different General and Hypothetical Task GO profiles. These results seem to indicate that there was not much variation in the General and Hypothetical Task GO profiles adopted by participants. However, before drawing any conclusions, the frequencies presented above will be compared to the frequencies expected by chance (using the methods described in Section 9.1.1.). The results are presented in Tables 9.40. and 9.41.

Table 9.40. Observed Frequencies and Proportions Expected to Remain in the sameGeneral and Hypothetical Task GO Clusters by Chance

	Cluster 1	Cluster 2	Cluster 3	Cluster 4
Observed	394	41	3	3
Proportions by chance	362	23	0	0

The total numbers of participants remaining in the same cluster and those changing cluster for General and Hypothetical Task GOs is presented in Table 9.41. for the Observed Frequencies and the Proportions by Chance.

Table 9.41	. Total	Numbers	of Partici	pants	remaining	in t	he S	Same	Cluster	and	Total
Number Cl	nanging	g Cluster G	eneral an	d Hype	othetical Ta	ask G	GOs				

	Remain in Same Cluster	Change Cluster	Total
Observed	441	176	617
Proportions by chance	385	232	617
Total	826	408	1234

A Chi-square test was carried out using the frequencies presented in Table 9.41. The X^2_{obt} in this case was 11.49 whilst X^2_{crit} is 6.64 (ρ <.01). The results indicate that more participants remained in the same clusters than they would have by chance alone (at ρ <.01). These results seem to point towards the possibility of participants adopting similar General and Hypothetical Task GO profiles. The comparison of the General and Verbal Test GO profiles adopted by participants is presented next.

Table 9.42. Frequency Table indicating the General and Verbal Test Goal Orientation
Profiles Adopted by Participants at Time 1 (N=67)

		Total			
0		1	2	3	
1 at e	1	30	8	6	44
eral ofi ster me	2	2	14	5	21
Ti Pr	3	0	1	0	1
5	4	0	0	1	1
	Total	32	23	12	67

The results presented in Table 9.42. show that 44 participants (66%) adopted the same Verbal Test and General GO profiles whilst 23 participants (34%) had a different Verbal Test and General GO profiles. The frequencies presented above were compared to those expected by chance (Tables 9.43. and 9.44.).

Table 9.43. Observed Frequencies	and Proportions	Expected to	Remain in	the sa	ame
General and Verbal Test GO Cluste	ers by Chance				

	Cluster 1	Cluster 2	Cluster 3
Observed	30	14	0
Proportions by chance	21	7	0

The total numbers of participants remaining in the same cluster and those changing cluster for General and Verbal Test GOs are presented in Table 9.44. for the Observed Frequencies and the Proportions by Chance.

Table 9.44. Total Numbers of Participants remaining in the Same Cluster and TotalNumber Changing Cluster General and Verbal Test GOs

	Remain in Same Cluster	Change Cluster	Total
Observed	44	23	67
Proportions by chance	28	39	67
Total	72	62	134

The Chi-square test carried out produced a X^2_{obt} of 7.69 (X^2_{crit} = 6.64, ρ <.01). Therefore, the results indicate that more participants remained in the same clusters than they would have by chance alone (at ρ <.01). Consequently, it seems as though participants may be adopting similar General and Verbal Test GO profiles. The comparison of participants' General and Numerical Test GO profiles is presented next.

Table 9.45. Frequency Table indicating the General and Numerical Test Goal Orientation Profiles Adopted by Participants at Time 1 (N=71)

	Numerical Test GO Profile Cluster at Time 1							
0		1	2	3	4			
1 1 1	1	30	5	7	2	44		
eral ofi ste	2	8	7	3	3	21		
Ti Clu Pi en	3	0	1	0	0	1		
G	4	0	0	1	0	1		
	Total	38	13	11	5	67		

Table 9.45. shows that 37 participants (55%) adopted the same General and Numerical Test GO profiles. However, 30 participants (45%) adopted different General and Numerical Test GO profiles. The analyses results to test whether more or less participants than would be expected by chance alone changed cluster (or not) are presented below (Tables 9.46. and 9.47.).

Table 9.46. Observed Frequencies and Proportions Expected to Remain in the sa	ame
General and Numerical Test GO Clusters by Chance	

	Cluster 1	Cluster 2	Cluster 3	Cluster 4
Observed	30	7	0	0
Proportions by chance	25	4	0	0

The total numbers of participants remaining in the same cluster and those changing cluster for General and Numerical Test GOs is presented in Table 9.47. for the Observed Frequencies and the Proportions by Chance.

Table 9.4	47. Total	Numbers	of Pa	rticipants	remaining	in	the	Same	Cluster	and	Total
Number	Changing	g Cluster G	enera	l and Nun	nerical Test	GC	Ds				

	Remain in Same Cluster	Change Cluster	Total
Observed	37	30	67
Proportions by chance	29	38	67
Total	66	68	134

The X^2_{obt} in this case was 1.91 whilst X^2_{crit} is 6.64 (ρ <.01) and 3.84 (ρ <.05). Therefore, the results indicate that more participants would have remained in the same clusters by chance alone. These results indicate that participants may be adopting different General and Numerical Test GO profiles.

Overall, the chi-square results point towards the possibility that General GO profiles are different to Numerical Test GO profiles but not necessarily different to Hypothetical Task and Verbal Test GO profiles. However, as described earlier, these results are not enough to draw any solid conclusions regarding the task-specificity of GO profiles. They do however, provide an indication that General GO profiles may be different to task-specific GO profiles (particularly Numerical Test GO profiles). Furthermore, it is not entirely clear why General GO profiles seem to be more different to Numerical Test GO profiles than they are to Hypothetical Task and Verbal Test GO profiles. Additional investigation of the differences between General and taskspecific GO profiles is definitely required before any conclusions may be drawn.

The answer to Research Question 4b is that at a significance level of ρ <.01 more participants would be expected to adopt the same General and Numerical Test GO profiles by chance alone. However, at a significance level of ρ <.05 more participants than would be expected by chance alone adopt similar General, Verbal Test and Hypothetical Task GO profiles. The next section takes this examination of the

similarities and/or differences between General and Task-specific GOs a step further by comparing the General and Task-specific GOs from the non-profile perspective.

9.2.4. Examining Differences between General and Task-Specific Goal Orientations

This section focuses on examining differences between General and Task-Specific GOs from the non-profile perspective and provides the results required to answer Research Question 4d.

Research Question 4d. Are participants' task-specific goal orientations significantly different from their General goal orientations?

The data collected at Time Q1 and Time E1 were used to answer this Research Question. Similarly to Section 9.2.2., changes were assessed in terms of differential continuity as well as mean-level change. The differential continuity results are presented first and are followed by the mean-level change results.

The correlational analyses results for General and Task-Specific GOs were presented in Chapter 8 Tables 8.43, 8.44. and 8.45. The results indicate that the correlation coefficients for participants' General GOs with the Hypothetical Task GOs at Time 1 ranged from 0.38 (ρ <0.01) to 0.52 (ρ <0.01). Those for participants' General and Verbal Test GOs at Time 1 ranged from 0.27 (ρ <0.01) to 0.47 (ρ <0.01) whilst the ones for participants' General and Numerical Test GOs ranged from 0.28 (ρ <0.01) to 0.47 (ρ <0.01). These correlation coefficients are all relatively low when compared to the correlation coefficients assessing stability over time (ranging from 0.44 to 0.71 at ρ <0.01 for General, Hypothetical Task, Verbal and Numerical Test GOs over time). Since they were all found to be significant at ρ <0.01 it is unlikely that there is any Type I error here.

Repeated Measures Analyses of Variance were carried out in order to assess the mean-level change across participants' General and Verbal Test GOs as well as across participants' General and Numerical Test GOs. Similarly to the previous section Task Experience was included as a covariate in these analyses. Task (that is, General GOs versus Verbal Test and Numerical Test GOs) was the independent variable whereas the relevant GO was the dependent variable. Paired samples t-tests were carried out to assess the mean-level change across participants' General and Hypothetical Task GOs (since these data were collected at Time 1 and there were no covariates to include in the analyses). The *significant* results are presented in Table 9.48. through to Table 9.54.

 Table 9.48. Assessing Mean Level Change for Performance-Approach across the

 General GOs measure and Verbal Test (N=69)

Source of Variance	SS	Df	MS	F	Sig.
Between Subjects					
Intercept	216.34	1	216.34	256.90	<.01
Task Experience on Verbal Test	.08	1	.08	.10	.76
Error	56.42	67	.84		
Within Subject					
PAP GO	1.58	1	1.58	4.31	.04
PAP GO * Task Experience on Verbal Test	.46	1	.46	1.26	.27
Error (PAP GO)	24.62	67	.37		

Table 9.49. Assessing Mean Level Change for Performance-Avoidance across theGeneral GOs measure and Verbal Test (N=67)

Source of Variance	SS	Df	MS	F	Sig.
Between Subjects					
Intercept	233.61	1	233.61	230.06	<.01
Task Experience on Verbal Test	<.01	1	<.01	<.01	.96
Error	66.00	65	1.02		
Within Subject					
PAV GO	4.88	1	4.88	13.61	<.01
PAV GO * Task Experience on Verbal Test	2.65	1	2.65	7.39	<.01
Error (PAV GO)	23.30	65	.36		

Table 9.50. Assessing Mean Level Change for Mastery-Approach across the General

Source of Variance	SS	Df	MS	F	Sig.
Between Subjects					
Intercept	174.04	1	174.04	270.75	<.01
Task Experience on Numerical Test	1.67	1	1.67	2.60	.11
Error	43.07	67	.64		
Within Subject					
MAP GO	2.13	1	2.13	6.64	<.01
MAP GO * Task Experience on Numerical Test	1.29	1	1.29	4.01	.05
Error (MAP GO)	21.51	67	.32		

GOs measure and Numerical Test (N=69)

Table 9.51. Assessing Mean Level Change for Mastery-Avoidance across the General

GOs measure and Numerical Test (N=68)

Source of Variance	SS	Df	MS	F	Sig.
Between Subjects					
Intercept	154.92	1	154.92	252.63	<.01
Task Experience on Numerical Test	1.18	1	1.18	1.92	.17
Error	40.47	66	.61		
Within Subject					
MAP GO	2.84	1	2.85	8.27	<.01
MAP GO*Task Experience on Numerical Test	1.01	1	1.01	2.94	.09
Error (MAP GO)	22.72	66	.34		

Table 9.52. Assessing Mean Level Change for Performance-Approach across theGeneral GOs measure and Numerical Test (N=69)

Source of Variance	SS	Df	MS	F	Sig.
Between Subjects					
Intercept	146.57	1	146.57	145.12	<.01
Task Experience on Numerical Test	3.16	1	3.16	3.13	.08
Error	67.67	67	1.01		
Within Subject					
PAP GO	3.40	1	3.40	8.46	<.01
PAP GO * Task Experience on Numerical Test	1.23	1	1.23	3.07	.08
Error (PAP GO)	26.92	67	.40		
Table 9.53. Assessing Mean Level Change for Performance-Avoidance across the

Source of Variance	SS	Df	MS	F	Sig.
Between Subjects					
Intercept	159.84	1	159.84	163.66	<.01
Task Experience on Numerical Test	2.62	1	2.62	2.69	.11
Error	63.48	65	.98		
Within Subject					
PAV GO	3.59	1	3.59	6.70	<.01
PAV GO * Task Experience on Numerical Test	1.80	1	1.80	3.36	.07
Error (PAV GO)	34.87	65	.54		

General GOs measure and Numerical Test (N=67)

Table 9.54. Assessing Mean Level Change in GOs across the General GO measures and the Hypothetical Task (N \approx 65)

Variables	GEN GO	GEN GO	HYP GO	HYP GO	Df	T-value	Sig
variables	Mean	SD	Mean	SD			
MAP	3.88	.56	3.96	.66	623	-2.87	<.01
MAV	3.76	.64	3.71	.69	620	1.70	.09
PAP	3.96	.67	3.80	.76	624	5.78	<.01
PAV	3.98	.74	3.87	.62	619	3.86	<.01

Key: GEN = General; HYP = Hypothetical Task

In terms of mean-level change, General and Verbal Test GOs were not found to be significantly different from each other for MAP and MAV GOs. However, the PAP and PAV General GOs were significantly higher than the corresponding Verbal Test GOs. Moreover, all four General GOs were significantly higher than the corresponding Numerical Test GOs. The MAP General GO was significantly lower than the MAP Hypothetical Task GO whilst the PAP and PAV General GOs were significantly higher than the PAP and PAV Hypothetical Task GOs, respectively. Therefore, in response to Research Question 4d, taken together, these results indicate that General GOs are significantly different from task-specific GOs though in absolute terms the differences are small. These findings will be discussed in further detail in Chapter 10 Section 10.8.2. The next section focuses on the effects of the interactions between Trait and Induced GOs on performance.

9.3. Examining the Effects (if any) of the Interactions between Trait and Induced GOs on Participant Performance

In Section 9.1.2. the results indicated that the inductions did not seem to be successful. Nevertheless it was thought necessary to investigate if there were any interaction effects (between the Trait and Induced GOs) on the performance of participants. This decision was made because there was a possibility that although participants did not consciously change their GOs, there might have still been an influence on their performance. If GOs were found to be general and stable the General GOs of participants would have been used as the 'trait' GOs in these analyses. However, since the results presented so far provide some evidence that participants' General GOs may be different from their task-specific GOs it was thought best to use the task-specific (Verbal and Numerical Test) GOs measured at Time 1 (when no inductions were present) as the trait GOs. This decision was made in order to ensure that any differences in the mean levels of GOs were not a result of different 'types' (e.g. General vs. Verbal Test) of GOs being measured but a result of the inductions. Therefore, in order to test the GO interactions on the Verbal Tests, the Verbal Test GOs at Time 1 were considered to be the 'trait GOs' whereas the Verbal Test GOs at Time 2 were considered to be the 'induced GOs'. Similarly, the Numerical Test GOs at Time 1 and Time 2 were considered to be the 'trait' and 'induced' GOs, respectively, when assessing the interactions on the Numerical Tests. The results of the analyses carried out in order to test these interaction effects address Hypothesis 1 and Research Questions 3a, 3b, 3c and 3d which are discussed below. The data used in these analyses were collected at Time E1 (Verbal and Numerical Test GOs and Performance) and Time E2 (Verbal and Numerical Test GOs and Performance).

Research Question 3a addresses the interaction effects between 'trait' Verbal and Numerical Test GO **profiles** (i.e. Verbal and Numerical Test GO profiles at Time E1) and induced MAP and PAP GOs on performance. On the other hand, Hypothesis 1 and Research Questions 3b, 3c and 3d focus on the interaction effects between trait and induced GOs on performance from the non-profile perspective. Research Question 3a is answered first followed by Hypothesis 1 and Research Questions 3b, 3c, and 3d. **9.3.1. Research Question 3a.** How do trait goal orientation profiles interact with induced mastery-approach and induced performance-approach goal orientations in order to influence the task performance of participants?

The experimental sample consisted of 71 participants at Time 2. The results presented in Chapter 8 Section 8.5.3. (Table 8.36.) indicated that at Time 1, 48%, 29%, 23% of participants (N=71) had Verbal Test GO profiles corresponding to Clusters 1, 2 and 3, respectively. The 29% and 23% of participants in Clusters 2 and 3 were assigned to the three experimental groups. Therefore, the sample size for participants having a Cluster 2 or Cluster 3 GO profile in each experimental group would not have been large enough to carry out statistical tests on. This resulted in Cluster 1 being the only cluster with a large enough sample to be tested. However, participants in this cluster were high on all the GOs. Consequently, the inductions would not have had much effect on participants in this cluster since they were designed to increase GO scores that were already near the top of the possible range. This was the same for Numerical Test GO profiles. Consequently, as a result of the small experimental sample size it was not possible to answer this Research Question.

9.3.2. Hypothesis 1. Individuals holding a trait mastery-approach goal orientation are expected to perform significantly better when a performance-approach goal orientation is induced as opposed to when a mastery-approach goal orientation is induced.

The focus of this Hypothesis was to test for differences in performance (depending on induction condition) for participants having a trait MAP GO. Therefore, only participants having a moderate to high Verbal Test MAP GO at Time 1 were included in these analyses. Ideally, only participants having a high Verbal Test MAP GO should have been included in the analyses. However, this would have made the sample size too small to analyse. For example, 51 participants had a high Verbal Test MAP GO at Time 1. However, at Time 2 these were assigned to three experimental groups, thus having only approximately 17 participants in each group. Consequently, participants having moderate Verbal Test MAP GO were also included in the analyses. Similarly,

only participants having a moderate to high Numerical Test MAP GO at Time 1 were included in the analyses for assessing mean-level change in performance on the Numerical Test (as a result of the inductions). Moreover, had the experimental sample size been larger, only participants having low to moderate Verbal (or Numerical) Test PAP GOs at Time 1 would have been included. As mentioned in Section 9.1.2. the inductions would not have had much influence on participants who already had considerably high PAP GOs. However, due to the small experimental sample size and because a considerable portion of the sample size (as discussed in Section 9.1.2.) had high Verbal Test (and Numerical Test) PAP GOs it was not possible to exclude these participants.

In order to test this Hypothesis the performance of participants having a moderate to high Verbal/Numerical Test MAP GO at Time E1 (when there were no inductions) and Time E2 (when the GO inductions were made) was assessed. Repeated measures Analyses of Variance were carried out. This time Test Performance was the dependent variable, Time was a within-subjects variable and the Experimental Condition (that is, whether participants were assigned to the Control, MAP or PAP induction groups) was the between-subjects variable. Participants in all three experimental conditions were included in the analyses. Moreover, task experience and practice on the respective tests were included as covariates since these may have influenced participants' performance.

No significant changes in the performance of participants (having a moderate to high Verbal Test MAP GOs at Time 1) as a result of the experimental manipulations were found since the Time by Experimental Condition interaction was not significant (F=0.55, p=0.58). Consequently, with respect to the Verbal Test, Hypothesis 1 is not supported.

The results presented in Table 9.55. and Figure 9.2. indicate that, for the Numerical Tests, there was a significant difference in the performance of participants in different experimental conditions (F=4.07, ρ =0.02). However, there were no significant differences in Numerical Test performance over time (F=0.85, ρ = 0.36). In addition,

there were no significant interaction effects (between Verbal Test MAP GO and MAP and PAP inductions) on performance (F=0.17, ρ = 0.85). Consequently, Hypothesis 1 was not supported for the Numerical Test either. The fact that there was a significant difference in the performance of participants in different experimental conditions (even at Time E1) indicates that more people with higher levels of verbal ability must have been assigned to the MAP induction group than the Control and PAP induction groups. This was an unfortunate happening which was probably magnified by the small sample size.

It is possible that the lack of support for Hypothesis 1 may have been caused by the inductions not being entirely successful. Moreover, as explained earlier, Hypothesis 1 may have been supported if only participants having a high Verbal (or Numerical) Test MAP GO and a low to moderate Verbal (or Numerical) Test PAP GO been included in the analyses. This will be discussed further in Chapter 10 Section 10.10.

Table 9.55. Repeated Measures ANOVA to test for changes in Numerical Test Performance from Time 1 to Time 2 depending on the Experimental Condition participants were assigned to (N=67)

Source of Variance	SS	Df	MS	F	Sig.
Between Subjects					
Intercept	29.42	1	29.42	4.68	.03
Task Experience on Numerical Test	65.31	1	65.31	10.39	<.01
Practice on Numerical Test	4.26	1	4.26	.68	.41
Experimental Condition	51.18	2	25.59	4.07	.02
Error	389.74	62	6.29		
Within Subject					
Time	2.77	1	2.77	.85	.36
Time*Task Experience on Numerical Test	5.45	1	5.45	1.67	.20
Time*Practice on Numerical Test	3.29	1	3.29	1.00	.32
Time*Experimental Condition	1.07	2	.54	.17	.85
Error	201.91	62	3.26		

Figure 9.2. Estimated Marginal Means of Numerical Test Performance from Time 1 to Time 2



Covariates appearing in the model are evaluated at the following values: TEXPNT1 = 2.1495, Practice on Numerical Test between Time 1 and Time 2 = 1.5672

9.3.3. Interaction Effects on Performance for Trait Mastery-Avoidance, Performance-Approach and Performance-Avoidance GOs and state Mastery-Approach and Performance-Approach GOs

Research Questions 3b, 3c and 3d focus on the interaction effects on performance of trait MAV GO (Research Questions 3b), PAP GO (Research Question 3c) and PAV GO (Research Question 3d) with state MAP and PAP GOs.

Since a considerable number of participants had a 'high GOs' profile on the Verbal and Numerical Tests (refer to Chapter 8 Sections 8.5.3. and 8.5.4.), many of the participants who were included in the analyses for Hypothesis 1 were also included in the analyses for these Research Questions. Consequently, the results of the analyses for Research Questions 3b, 3c and 3d are very similar to each other. Similarly to the results of Hypothesis 1, the results obtained indicated that there were no significant interaction effects on the (Verbal and Numerical Test) performance of participants having moderate to high trait MAV, PAP and PAV GOs when MAP and PAP GOs were induced.

9.4. Summary of Results

Before proceeding to the next chapter, a summary table of all the results obtained in this research study is provided in Table 9.56. The results are presented in the same order that they were presented in Chapters 8 and 9.

Table 9.56. Summary of Results

Hypothesis or Research Question	Result
Research Question 1. Using LCA as a method of clustering GOs, how many different types of GO profiles are there and what are the characteristics of each GO profile? Does the 2x2 model of GOs significantly improve on the 3-factor model in terms of identifying GO profiles?	 Four main types of GO profiles emerged: a) High GOs b) High Mastery, Moderate Performance c) High Mastery, Low Performance d) High Performance-Avoidance In this study, for GO profiles, the 2x2 model of GOs was not found to necessarily improve on the 3-factor model in terms of the statistical viability and interpretability of profiles.
Relationsh	ips between GOs and Self-Efficacy
<i>Hypothesis 2a.</i> Task-specific MAP GOs are expected to be significantly positively correlated with self-efficacy.	Supported (except for the Verbal Test at Time 2).
<i>Hypothesis 2b.</i> Task-specific PAP GOs are expected to be significantly positively correlated with self-efficacy.	Supported.
Research Question 6b. How do task-specific MAV GOs correlate with self-efficacy?	Verbal and Numerical Test MAV GOs are weakly related to self-efficacy, if at all.

Hypothesis or Research Question	Result
Research Question 6c. How do task-specific PAV GOs correlate with self-efficacy?	Verbal Test PAV GOs were found to be weakly related to self-efficacy (if at all) whilst Numerical Test PAV GOs were found to be strongly related to self-efficacy. The relationship between a PAV GO and self-efficacy seems to be influenced by the type of task at hand.
Research Question 6a. Do the different task-specific GO profiles score significantly differently on self-efficacy?	The different Verbal Test GO profiles did not relate differently to self-efficacy. However, there were significant differences between the relationships of different Numerical Test GO profiles and self-efficacy. For the latter, participants in Cluster 1 ('High GOs' profile) were found to have significantly higher levels of self-efficacy than participants in the other 3 clusters.

Hypothesis or Research Question	esearch Question Result		
Relationships between GOs and Mental Effort			
Research Question 7b. How do the different task-specific GOs correlate with mental effort (if at all)?	 a) MAP GOs were found to be significantly positively correlated with mental effort on the Verbal and Numerical Tests (except for the Numerical test at Time 2). b) MAV GOs were found to be significantly positively related to mental effort on the Verbal Test at Time E2 and on the Numerical Test at Time E1. No significant relationships were found on the Verbal Test at Time E1 and on the Numerical Test at Time E2. c) PAP GOs were not found to be significantly related to mental effort on the Verbal or Numerical Tests. d) PAV GOs were not found to be significantly related to mental effort on the Verbal or Numerical Tests. 		

Hypothesis or Research Question	Result
	There were significant differences in the mental effort scores of participants who
	adopted different GO profiles on the Verbal Test, but not on the Numerical Tests.
Research Question 7a. Do the different task-specific GO	For the former, participants adopting the 'High GOs' and the 'High Mastery, Low
profiles score significantly differently on mental effort?	Performance' GO profiles were found to invest significantly higher levels of
	mental effort than participants adopting the 'High Mastery, Moderate
	Performance' GO profile.
Relationsh	ips between GOs and Performance
Research Question 8a. How do task-specific MAP GOs	MAP GOs were found to be significantly positively related to performance on the
correlate with performance on tasks (if at all)?	Numerical Tests but not on the Verbal Tests.
Research Question 8b. How do task-specific MAV GOs	MAN GOs were not found to be significantly related to performance
correlate with performance on tasks (if at all)?	MAV GOS were not round to be significantly related to performance.
Research Question 8c: How do task-specific PAP GOs	Only the Numerical Test PAP GO at Time 2 was significantly positively correlated
correlate with performance on tasks (if at all)?	with performance.

Hypothesis or Research Question	Result
Research Question 8d: How do task-specific PAV GOs correlate with performance on tasks (if at all)?	The only significant correlation found was a positive one between a Numerical Test PAV GO at Time 2 and performance.
<i>Research Question 9.</i> Do the different task-specific GO profiles score significantly differently on task performance?	The different Verbal Test GO profiles were not significantly differently related to performance. However, there were significant differences in the Numerical Test performance of participants adopting different GO profiles. For the latter, participants adopting the 'High GOs' profile were found to have significantly higher levels of performance than participants adopting the 'High Mastery, Moderate Performance' and 'High Performance-Avoidance' profiles.
Assessing th	e Stability of General GOs over Time
<i>Research Question 2a.</i> Do individuals' General GO profiles change over time?	The comparison results of participants' General GO profiles at Time 1 and Time 2 indicate that 67% of participants adopted the same General GO profile at Time 2 as they did at Time 1. Moreover, more people than would have been expected by chance adopted the same General GO profile from Time 1 to Time 2 $X^2(1, N=183) = 6.07$, p<.05.

Hypothesis or Research Question	Result
Research Question 2b. Do individuals' General GOs change significantly over time?	 In terms of mean-level change it seems as though participants' MAP and MAV GOs tend to remain stable over time whilst their PAP and PAV GOs decrease significantly from Time 1 to Time 2. In terms of differential continuity participants' General GOs are not as stable as personality traits (reported in other studies). Moreover, the correlation strength for General PAP GOs was found to decrease significantly (z-statistic=3.01, ρ<0.01) the longer the time interval between questionnaires.
Assessing the S	Stability of Task-Specific GOs over Time
<i>Research Question 5b.</i> Are task-specific GOs stable over time?	 In terms of mean-level change no significant differences in the GO scores of participants from Time 1 to Time 2 (on the Hypothetical Task, Verbal Test and Numerical Test) were found. In terms of correlations over time, correlation coefficients ranging between 0.44 to 0.71 (p<0.01) were found for task-specific GOs over time. Hypothetical Task GOs were no less stable (as assessed by correlations over time) over a 9-51 weeks period than over a 4-8 week period. Overall, the results indicate that task-specific GOs are fairly stable over time.

Hypothesis or Research Question	Result
<i>Research Question 5a.</i> Are task-specific GO profiles stable over time?	The results show that 66%, 63% and 56% of participants adopted the same GO profile from Time 1 to Time 2 on the Hypothetical Task, Verbal Test and Numerical Test, respectively. Moreover, the Chi-square test results indicated that more participants than would have been expected by chance adopted the same Hypothetical Task, Verbal Test and Numerical Test GO profile from Time 1 to Time 2 [X ² (1, N=187) = 4.46, p<.05; X ² (1, N=68) = 9.53, p<.01; X2 (1, N=71) = 4.07, p<.05, respectively for Hypothetical Task, Verbal Test and Numerical Test GO profile sover time].
Assessing the G	enerality and/or Task-Specificity of GOs
Research Question 4a. Do participants adopt different GO profiles across tasks?	The Chi-square results indicated that more participants would have been expected to adopt the same GO profiles on the Hypothetical Task and Verbal Test $[X^2 (1, N=67) = 1.87, p>.05]$ as well as on the Hypothetical Task and Numerical Test $[X^2 (1, N=67) = 1.93, p>.05]$ by chance alone. However, more participants than would have been expected by chance alone adopted the same GO profiles on the Verbal and Numerical Tests $[X^2 (1, N=71) = 17.7, p<.01]$. These results seem to point towards the domain-specificity of GO profiles.

Hypothesis or Research Question	Result
	• Correlations for Hypothetical Task and Verbal Test GOs ranged from 0.31
	(ρ <0.05) to 0.59 (ρ <0.01). Those for Hypothetical Task and Numerical Test GOs
	ranged from 0.27 (p<0.05) to 0.53 (p<0.01) whilst the ones for Verbal and
	Numerical Test GOs ranged from 0.68 (p<0.01) to 0.81 (p<0.01). Fisher z tests
Recorder Question de De nortigioante' adapt different	indicated that, overall, Verbal and Numerical Test GOs were significantly more
<i>Research Question 4c.</i> Do participants' adopt different goal orientations across tasks?	highly correlated with each other than with Hypothetical Task GOs.
	• The mean-level change results reflected the correlation results in that there
	were no significant differences between Verbal and Numerical Tests GOs.
	However, overall, Hypothetical Task GOs were found to be significantly higher
	than Verbal and Numerical Test GOs.
	• These results also point towards the domain-specificity of GOs.
	The Chi-square tests indicated that more participants than would be expected by
	chance alone adopted the same General and Hypothetical Task GO profiles $[X^2 (1,$
Research Question 4b. Are participants' task-specific GO	N=617) = 11.49, p<.01] as well as the same General and Verbal Test GO profiles
profiles different from their General GO profiles?	$[X^2 (1, N=67) = 7.69, p<.01]$. However, more participants would be expected to
	adopt the same General and Numerical Test GO profiles $[X^2 (1, N=67) = 1.91,$
	p>.05] by chance alone.

Hypothesis or Research Question	Result
Hypothesis or Research Question <i>Research Question 4d.</i> Are participants' task-specific GOs significantly different from their General GOs?	 Result Correlations between General and Hypothetical Task GOs ranged from 0.38 to 0.52 (ρ<0.01). Those between General and Verbal Test GOs ranged from 0.27 to 0.47 (ρ<0.01) whilst the correlations between General and Numerical Test GOs ranged from 0.28 to 0.47 (ρ<0.01). These correlation coefficients are quite low especially when compared with those of stability of GOs over time (e.g. Research Question 5b). The results for mean-level change indicated that General PAP and General PAV GOs were significantly higher than the respective Verbal Test GOs. All four
	General GOs were significantly higher than the respective Numerical Test GOs. In addition, the MAP Hypothetical Task GO was found to be significantly higher than the respective General GO whilst the Hypothetical PAP and PAV GOs were found to be significantly lower than the respective General GOs. Taken together, these results provide a good indication of participants' task-specific GOs being significantly different from their General GOs.

Hypothesis or Research Question	Result			
Examining the Effects of the Interactions between General and Induced GOs on Participants' Performance				
Research Question 3a. How do trait GO profiles interact with induced MAP and induced PAP GOs in order to influence task performance of participants?	It was not possible to answer this Research Question due to the small experimental sample size.			
Hypothesis 1. Individuals holding a trait mastery-approach goal orientation are expected to perform significantly better when a performance-approach goal orientation is induced as opposed to when a mastery-approach goal orientation is induced.	No interaction effects (between trait MAP and induced MAP and PAP GOs) on performance were found. Consequently, this Hypothesis was not supported.			
Research Question 3b. What effects will the relationships between trait MAV GOs and induced MAP and induced PAP GOs have on the performance of participants?	No interaction effects (between trait MAV and induced MAP and PAP GOs) on performance were found.			
Research Question 3c. What effects will the relationships between trait PAP GOs and induced MAP and induced PAP GOs have on the performance of participants?	No interaction effects (between trait PAP and induced MAP and PAP GOs) on performance were found.			
Research Question 3d. What effects will the relationships between trait PAV GOs and induced MAP and induced PAP GOs have on the performance of participants?	No interaction effects (between trait PAV and induced MAP and PAP GOs) on performance were found.			

9.5. Synopsis

Following the presentation of the results in this chapter and the previous one (Chapter 8) a discussion of the findings of this study will be provided in the next chapter. The same order of themes as that in Chapters 8 and 9 is maintained.

Chapter 10: Discussion

10.0. Introduction

In this chapter the findings from this research study (presented in Chapters 8 and 9) are discussed and compared with the results of previous research (presented in Chapters 2 to 5). The same order of themes specified in Chapters 8 and 9 is maintained throughout this chapter. Therefore, the results of the LCA of General GOs are discussed first. Following this, the findings regarding the relationships between GOs and self-efficacy, mental effort and performance are described. Next, the results regarding the stability and task-specificity of GOs will be discussed. Finally, the results concerning the interaction effects between General and Induced GOs on performance are reviewed.

10.1. Latent Class Analysis of General Goal Orientations and choosing between the 2x2 and 3-factor models of Goal Orientations for the profile analyses

The main focus of this section is on Research Question 1 which examines the number and types of GO profiles emerging from the LCA of General GOs and questions whether the 2x2 model of GOs significantly improves on the 3-factor model in terms of identifying GO profiles. The latter is discussed first. Following this, the General GO profile results obtained are discussed and compared with GO profiles obtained in previous studies.

In order to determine whether the 2x2 model of GOs significantly improves on the 3factor model LCAs were carried out using both the 3-factor and the 2x2 models of GOs. The results indicated that there was inadequate model fit for General and Numerical Test GOs at Time 1 when using the 2x2 model of GOs. Consequently, for the purposes of the current study, the 2x2 model did not seem to significantly improve on the 3factor model in terms of GO profiles. As a result, it was decided that the 3-factor model of GOs should be used in all analyses involving GO profiles in the current study. With respect to the number and types of GO profiles that emerged from the analysis, the LCA results for General GOs at Time 1 revealed four types of GO profiles when using the 3-factor model of GOs. These included 'High GOs'; 'High Mastery, Moderate Performance'; 'High Mastery, Low Performance'; and 'High Performance-Avoidance' GO profiles. The LCAs of General GOs at Time 2 and Hypothetical Task, Verbal and Numerical Test GOs at Time 1 and Time 2 revealed remarkably similar GO profiles to those obtained for General GOs at Time 1. A number of times (e.g. for General and Hypothetical Task GOs at Time 2) only 3 out of the 4 profiles emerged. This was thought to be due to the smaller sample sizes.

Since the 3-factor model of GOs was used for examining GO profiles in this study it was thought appropriate to compare the GO profiles obtained from this study to those of other studies using the 3-factor model. Consequently, the study by Cano and Berben (2009) was excluded from the comparison since they used the 2x2 model of GOs in their study. In addition, the GO profiles obtained from the 3-factor model (NOT the ones obtained using the 2x2 model) in the study by Pastor et al. (2007) were used for comparison purposes. Table 10.1. presents the General GO profiles obtained in the current study as well as those from the studies by Pastor et al. (2007) and Fortunato and Goldblatt (2006).

Table 10.1. Comparison of GO profiles	obtained in	this study	with those	obtained in
previous studies				

	Pastor et al. (2007) N=1868	Fortunato & Goldblatt (2006) N=311	Present Study N=628
	(12%)	(32%)	(67%)
Chuster 1	MAP: High	MGO : Low	MGO : High
Cluster 1	PAP: High	PAP : Moderate	PAP : High
	PAV: Moderate	PAV : Moderate	PAV : High
	(9%)	(18%)	(29%)
Chuster 2	MAP: High	MGO : Moderate	MGO : High
Cluster 2	PAP: High	PAP : Low	PAP : Moderate
	PAV: Mod Low	PAV : Low	PAV : Moderate
	(25%)	(28%)	(2%)
Chuster 2	MAP: Mod High	MGO : Moderate	MGO : High
Cluster 3	PAP: Mod High	PAP : High	PAP : Low
	PAV: Moderate	PAV : High	PAV : Low
	(44%)	(22%)	(2%)
Chusten A	MAP: Mod High	MGO : High	MGO : Low
Cluster 4	PAP: Moderate	PAP : Moderate	PAP : Low
	PAV: Mod Low	PAV : Low	PAV : High
	(10%)		
Chusten F	MAP: Moderate		
Cluster 5	PAP: Low		
	PAV: Mod Low		

Key: MAP=Mastery-Approach Goal Orientation; MGO=Mastery Goal Orientation; PAP=Performance-Approach Goal Orientation; PAV=Performance-Avoidance Goal Orientation.

It is currently extremely difficult to compare GO profiles across studies. This is mainly due to the fact that it is tricky to tell whether what is considered to be a low, moderate, or high GO in this study would also be considered to be low, moderate or high (respectively) in the other studies examining GO profiles. For example, it may be the case that what was considered to be a high GO in this study would be considered to be moderate in the study by Pastor et al. (2007). It is not easy to compare mean GO scores for the 3 GOs in each profile across studies for two main reasons. Firstly, the three studies (being compared above) made use of different GO measures: the AGQ-R was used in this study, the AGQ was used in the study by Pastor et al. (2007). Secondly, although the AGQ and the AGQ-R may be argued to be reasonably similar, a 5-point Likert scale was used in the AGQ-R in this study whereas Pastor et al.

(2007) used a 7-point Likert scale in their study. As a result comparisons between the GO profiles across studies will not be exact.

Bearing the above in mind, Table 10.1. indicates that there are a number of similar clusters across the three studies. For example, Cluster 5 in the study by Pastor et al. (2007) and Cluster 2 obtained by Fortunato and Goldblatt (2006) are similar to the 'High Mastery, Low Performance' profile obtained in the present study (Cluster 3) in that the MAP GOs in these clusters are higher than the PAP and PAV GOs. However, the percentage of participants adopting this type of GO profile in the current study is much lower than the percentages in the other studies. A second notable commonality is that Clusters 3, 3 and 1 from the studies by Pastor et al (2007), Fortunato and Goldblatt (2006) and the present study, respectively, all seem to have moderate to high GOs for all three GOs in the profiles. However, this time the percentage of participants adopting this GO profile in the present study is much higher compared to the percentages in the other studies. Clusters 2 and 4 from the studies by Pastor et al (2007) and Fortunato and Goldblatt (2006) may be argued to be similar in that the PAV GOs in these profiles are lower than the MAP and PAP GOs. Consequently, participants adopting this particular GO profile seem to have high approach and low avoidance GOs. Again, there is a difference in the percentages of participants adopting this type of GO profile across studies. A higher percentage of participants adopt this type of GO profile in the study by Fortunato and Goldblatt (2006). Another similarity is between Cluster 1 from the study by Fortunato and Goldblatt (2006) and Cluster 4 (the High Performance Avoidance GO profile) of the present study. A closer look at Cluster 1 (of Fortunato and Goldblatt, 2006) showed that the PAP GO in this profile was lower than the PAV GO in the same profile. Consequently, it seems as though both profiles tend to have higher PAV GOs than MAP and PAP GOs. However, the percentage of participants adopting this type of GO profile in the present study was much lower when compared with Fortunato and Goldblatt (2006).

Although it may seem possible that Cluster 1 from the study by Pastor et al. (2007) is similar to Cluster 2 from the current study (the High Mastery, Moderate Performance profile), in the former study the PAP GO mean was actually higher than the MAP GO

mean. Consequently, the 2 clusters seem to be rather different from each other. Finally, Pastor et al. (2007) had an additional cluster (Cluster 4) consisting of moderate approach and low avoidance GOs. This cluster seems to be quite different from the clusters obtained in the present study and that by Fortunato and Goldblatt (2006).

Although a number of similar clusters were found, the percentages of participants adopting the different types of GO profiles seems to differ across studies. This is possibly due to the participant demographics and/or type of tasks that GOs were measured on. For example, in this study participants were asked to complete aptitude tests which would not influence the course grade of those participants who were students. Moreover, a number of participants were employed or retired. This may have greatly influenced the type of GO profiles adopted. In the study by Pastor et al. (2007) GOs were measured at an Assessment Day whilst Fortunato and Goldblatt (2006) measured GOs on a university course. Since the results of this study show some evidence of domain-specificity it is possible that the different percentages obtained were a result of different tasks GOs being measured in the different studies.

Since a number of similar profiles were obtained across studies using different GO measures, scales and tasks, it seems as though there is ground for future studies to investigate whether it is possible to obtain a comprehensive set of GO profiles. In order to achieve this it would be necessary to make a final decision regarding the dimensionality of GOs as well as agree on a GO measure. However, the advantages of this would be that researchers will be able to effectively compare GO profiles (as well as their antecedents and consequences) across studies. This will enable researchers to assess whether certain GO profiles are more beneficial than others and investigate whether it is possible to encourage employees to adopt the more beneficial GO profiles. The relevant recommendations for future research are made in Chapter 11 Section 11.3. The relationships between GOs (as well as GO profiles) and self-efficacy are discussed next.

10.2. The relationships between Goal Orientations and Self-Efficacy

Hypotheses 2a and 2b as well as Research Questions 6a, 6b and 6c focused on the relationships between GOs and self-efficacy. The relationships between the GOs from the non-profile perspective and self-efficacy are focused on first. Following this, the relationships between GO profiles and self-efficacy are discussed.

10.2.1. Relationships between Goal Orientations and Self-Efficacy from the Non-Profile Perspective

Since numerous studies investigated the relationships between GOs and self-efficacy it was thought useful to present a comparison table (Table 10.2. below) indicating similarities and differences between the results of this study and those of previous studies.

GO being Investigated	Results of Present Study	Significant Positive Relationship Found	Significant Negative Relationship Found	No Significant Relationship Found
Mastery/Mastery- Approach GO	• Significant Positive relationship in all cases except on Verbal Test at Time 2	 Elliot & Church (1997) Liem et al. (2008) Lau et al. (2008) Morris & Kavussanu (2008) Tanaka (2007) Linnenbrink (2005) Sins et al. (2008) Bong (2009) Radosevich et al. (2007) Bong (2001) Wolters et al. (1996) VandeWalle et al. (2001) 		
Mastery- Avoidance GO	 No significant relationships on Verbal Test. Significant positive relationships on Numerical Test 	• Bong (2009) ¹	 Lau et al. (2008) Radosevich et al. (2007) Bong (2009)² 	 Morris & Kavussanu (2008) Bong (2009)³

 Table 10.2. Comparison across studies for relationships between Goal Orientations and Self-Efficacy

Table	10.2.	continued	1.
-------	-------	-----------	----

GO being Investigated	Results of Present Study	Significant Positive Relationship Found	Significant Negative Relationship Found	No Significant Relationship Found
Performance- Approach GO	• Significant positive relationship.	 Liem et al. (2008) Lau et al. (2008) Morris & Kavussanu (2008) Elliot & Church (1997) Tanaka (2007) Linnenbrink (2005) Bong (2009) Radosevich et al. (2007) Bong (2001) 		• VandeWalle et al. (2001)
Performance- Avoidance GO	 No significant relationship for Verbal Test at Time 1 Weak positive (significant) relationship found for Verbal Test at Time 2. Significant positive relationships on Numerical Test (at Time 1 and Time 2) 	 Bong (2009)¹ Bong (2001)⁶ 	 Liem et al. (2008) Lau et al. (2008) Elliot & Church (1997) Radosevich et al. (2007) Bong (2009)⁴ VandeWalle et al. (2001) 	 Morris & Kavussanu (2008) Tanaka (2007) Sins et al. (2008) Bong (2009)⁵ Bong (2001)⁷

¹ For Lower Elementary; ² For Upper Elementary; ³ For Middle Elementary and Middle School; ⁴ For Middle School; ⁵ For Middle and Upper Elementary; ⁶For English, Maths and Science in Middle School and Korean, English and Maths in High School.

Table 10.2. shows that all the studies reviewed found a MGO/MAP GO to be significantly positively correlated with self-efficacy. The results obtained in this study are consistent with those of previous research with the exception of the Verbal Test at Time 2. Moreover, as discussed in Chapter 8, it is possible that the correlation found for the MAP GO at Time 1 might be the result of Type I error, in which case Hypothesis 2a would not be entirely supported. This may be considered to be a limitation since it is not possible to draw clear conclusions regarding the support or lack of support for Hypothesis 2a. Further research is necessary in order to be able to draw conclusions as to the relationships between a MAP GO and self-efficacy on different tasks.

The findings of the present study also indicate that the Numerical Test MAP GO was found to be significantly more strongly correlated with self-efficacy than the Verbal Test MAP GO at Time 2 but not at Time 1. This suggests that the relationships between GOs and self-efficacy may be moderated by task type and the person's prior experience of it. Task difficulty is a possible characteristic of tasks that influences the relationships between GOs and self-efficacy. For example, the Numerical Test may be perceived as being more difficult than the Verbal Test, consequently, participants who are confident enough to adopt a strong MAP GO on the Numerical Test would probably also have high self-efficacy on the test. On the other hand, if the Verbal Test is not perceived to be difficult, participants might not feel the need to adopt a strong MAP GO on this test since it involves skills that people tend to practice on a day-to-day basis. Consequently, although they may have high self-efficacy scores on the Verbal Test they may not necessarily have high MAP GO scores too.

However, this does not explain why there were significant differences in the correlations strengths (for a MAP GO and self-efficacy on the Verbal and Numerical Tests) at Time 2 and not at Time 1. Since there were GO inductions at Time 2, one might assume that these differences occurred as a result of the inductions. However, since the inductions were not found to be successful (Chapter 9 Section 9.1.2.) this is highly unlikely. A possible explanation for these differences is that at Time 2 the participants already had a good idea of what the tests were like and their level of difficulty. As a result, the effects of task difficulty may have been more pronounced at

Time 2 than they were at Time 1. Moreover, participants completed the self-efficacy measures following the example questions but prior to completing each test. Consequently, they would not have had as good an idea about the test difficulty at Time 1 as they did at Time 2. These differences in the relationships between GOs and self-efficacy depending on task type may account for the differences in findings across studies. Overall however, it seems as though a MAP GO is significantly positively related to self-efficacy. A plausible explanation for this relationship is that the focus of a MAP GO is on improving one's understanding of a task. Consequently, one would probably need to feel quite confident about their ability in order to adopt such a GO. The opposite may also be the case. One may choose to adopt a MAP because he/she feels confident about his/her ability to improve understanding.

The relationship between a MAV GO and self-efficacy is much less straightforward than that for a MAP GO and self-efficacy. As indicated in Table 10.2., previous studies found significant positive, negative, as well as non-significant relationships between a MAV GO and self-efficacy. In this study non-significant relationships were found between a MAV GO and self-efficacy on the Verbal Test. However, significant positive relationships were found on the Numerical Test. Therefore, the results for the Numerical Test (at Time 1 and Time 2) are consistent with those of Bong (2009, for Lower Elementary School students). On the other hand, the results for the Verbal Test (at Time 1 and Time 2) are consistent with the results of Morris and Kavussanu (2008) and Bong (2009, for Middle Elementary and Middle School students). These differences obtained across tasks may be explained by the possibility that the relationships between a MAV GO and self-efficacy are moderated by task type. Similarly to the results for the MAP GO, the Numerical Test MAV GO was found to be significantly more strongly correlated with self-efficacy than the Verbal Test MAV GO. Furthermore, again similarly to the MAP GO, these significant differences were only found at Time 2 and not at Time 1. Consequently, it seems as though previous experience of the task and possibly some task characteristics may be influencing the relationship between a MAV GO and self-efficacy. It is not easy to explain why and how a MAV GO is related to self-efficacy. For example, a focus on not forgetting or not misunderstanding a task could be interpreted as being a negative approach to tasks

298

(hence the term 'avoidance') because a person may be worried that they will not be able to understand the new task. In such a case the negative relationship between a MAV GO and self-efficacy could be explained by the individual trying not to misunderstand/forget a task because they do not feel confident about their ability. However, it may be the case that a person does feel confident about their ability and chooses to adopt both MAP and MAV GOs because they want to make sure that they learn the new task properly and do not misunderstand it. In such a case the MAV GO may be positively correlated with self-efficacy. In order to understand this relationship better it may be useful to investigate the *reasons* for adopting a MAV GO as well as if (and how) task characteristics such as task difficulty moderate this relationship. Recommendations for such research will be made in Chapter 11 (Section 11.3).

The relationships between a PAP GO and self-efficacy have been quite consistent across studies (including the present one) since all the studies reviewed found a significant positive relationship between a PAP GO and self-efficacy. The only exception is the study by VandeWalle et al. (2001) who found a non-significant relationship. Similarly to the MAP GO it is understandable how a PAP GO and selfefficacy would be positively related to each other. When a person adopts a PAP GO their focus is on performing better than others. One would probably need to feel quite confident about their ability if their focus is on being one of the best. The opposite may also be the case. One may choose to adopt a PAP GO because he/she feels confident about his/her ability to do better than others. Overall, the results of this study and previous ones suggest that adopting a PAP GO on tasks may be quite beneficial since self-efficacy on tasks has been associated with a number of positive outcomes such as increased performance (e.g. Pajares, 1997; Bouffard-Bouchard, 1990). However, research investigating the direction of causality needs to be carried out prior to making recommendations regarding 'beneficial' GOs. Similarly to the relationships between MAP and MAV GOs and self-efficacy, the Numerical Test PAP GO was found to be significantly more highly correlated to self-efficacy than the Verbal Test PAP GO. Again, these significant differences were found at Time 2 but not at Time 1.

Previous studies have not found consistent relationships between a PAV GO and selfefficacy. As indicated in Table 10.2. some studies found significant negative relationships and other studies found non-significant relationships. Moreover, Bong (2001) found significant positive, negative and non-significant relationships between a PAV GO and self-efficacy depending on the subject and age of participants and Bong (2009) found significant positive, negative and non-significant relationships depending on the age of participants. Similarly to the relationship between a MAV GO and selfefficacy, it is not easy to explain the relationships between a PAV GO and self-efficacy. One may expect a person with a high PAV GO to have low levels of self-efficacy since it may be assumed that if you aim to try and not be the worst you cannot be feeling very confident about your skills on the task. However, it may be argued that if a person adopts a high PAV GO they may have high self-efficacy because they are aiming not to be the worst on a task *as opposed to* simply giving up or not trying to do better than anyone else. For example, if a group of people are learning how to use a new machine at work an employee might adopt a high PAV GO because he/she feels confident that he/she will not be the worst at using this new machine for the first time. Other employees who are not confident about their ability to use the new machine might choose not to compare their performance with that of others. On the other hand, in a different situation a person may choose to adopt a PAV GO because they are worried that they might look silly if they are the worst performer on a task. In such a case the person may not be feeling confident about their ability. Consequently, (similarly to the relationship between a MAV GO and self-efficacy) it is possible that the relationships between a PAV GO and self-efficacy are influenced by the reasons for adopting such a GO. Task characteristics, such as task difficulty, may also be influencing the relationships between a PAV GO and self-efficacy as will be described shortly.

The results of the present study are very similar to those obtained by Bong (2001) in that a non-significant relationship was found between a Verbal Test PAV GO and selfefficacy at Time 1 and a weak positive relationship was found at Time 2. However, the Numerical Test PAV GO was found to be significantly positively correlated with selfefficacy at both Time 1 and Time 2. The inconsistencies found (in the relationships between a PAV GO and self-efficacy) across tasks in the study by Bong (2001) provide further support for the possibility that task type may moderate the relationships between GOs and self-efficacy. Similarly to the relationships between the other three GOs and self-efficacy, the Numerical Test PAV GO was found to be more strongly correlated with self-efficacy than the Verbal Test PAV GO. However, again similarly to the other three GOs, these significant differences were only found at Time 2 and not at Time 1.

Taken together, the results obtained in this study as well as those in previous studies indicate that, first and foremost, MAP and PAP GOs seem to be quite consistently significantly positively related to self-efficacy. Consequently, they seem to be beneficial GOs to adopt on tasks. The results are not as clear-cut with respect to the relationships between the avoidance GOs (MAV and PAV GOs) and self-efficacy. Moreover, there also seems to be a distinct possibility that these relationships (as well as those for the approach GOs and self-efficacy) are moderated by task type (possibly task difficulty or perceived task difficulty). Previous research provides some evidence for this possibility. Mangos and Steele-Johnson (2001) found that subjective task complexity (perceived task difficulty) completely mediated the effects of GOs on self-efficacy. The implications of these findings as well as recommendations for future research will be made in Chapter 11 (Sections 11.1. and 11.3., respectively). Before discussing the results obtained with respect to GOs and mental effort, the relationships between GO *profiles* and self-efficacy are discussed.

10.2.2. The Relationship between Goal Orientation Profiles and Self-Efficacy

The results presented in Chapter 8 (Tables 8.50. and 8.51.) show that there were no significant differences in the self-efficacy of participants adopting different GO profiles on the Verbal Test. However, there were significant differences in the self-efficacy of participants adopting different GO profiles on the Numerical Test. The results presented in Table 8.52. show that participants in the 'High GOs' cluster had the highest levels of self-efficacy whilst participants in the 'High Mastery, Moderate Performance' cluster had the lowest levels of self-efficacy. Since the individual GO results indicated that MAP and PAP GOs are strongly related to self-efficacy whilst

MAV and PAV GOs are not as strongly related, it seems as though the lower PAP GO in the 'High Mastery, Moderate Performance' cluster might be responsible for the significantly lower levels of self-efficacy of participants in this cluster.

Since there were no significant differences in the self-efficacy of participants adopting different GO profiles on the Verbal Test but significant differences for the Numerical Test this might be a further indication that the relationships between GOs and selfefficacy are in fact moderated by task type. Since participants adopting different GO profiles were found to have different levels of self-efficacy it might also be worth investigating which profiles might be more beneficial than others. For example, in this case, participants adopting the 'High GOs' profile were found to have the highest levels of self-efficacy on the Numerical Test whilst participants adopting the 'High Mastery, Moderate Performance' GO profile were found to have the lowest levels of self-efficacy (on the Numerical Test). Consequently, adopting the former GO profile may have more beneficial effects than adopting the latter. However, prior to investigating this any further it would be necessary to investigate the direction of causality since it might be the self-efficacy that causes the adoption of particular GO profiles. It is also possible that the different results obtained on the Verbal and Numerical Tests are a result of the small experimental sample size (resulting in small cluster sizes). Therefore, it is possible that either Type I or Type II error occurred due to the small cluster sizes. The small cluster sizes should be taken into serious consideration when interpreting these results. Further research is required before any clear conclusions can be drawn regarding the relationships between GO profiles and self-efficacy on different tasks. Suggestions for future research regarding the relationships between GO profiles and self-efficacy will be made in Chapter 11 Section 11.3. The next section focuses on the results obtained when assessing the relationships between GOs and mental effort.

10.3. The relationships between Goal Orientations and Mental Effort

As was discussed in Chapter 5, only three of the studies reviewed examined the correlations between GOs and mental effort. These all used the 3-factor model. Since

it was found that MGO scales generally include only MAP items (as was discussed in Chapter 5 Section 5.2.1.) the MAP GO scale in this study is compared to the MGO scales in the previously mentioned studies. In order to make the comparisons across studies as clear as possible a table comparing the results obtained in this study with those obtained in previous studies was drawn up (Table 10.3. below).

Table 10.3. Comparison across studies for relationships between Goal Orientations and Mental Effor
--

GO being Investigated	Results of Present Study	Significant Positive Relationship Found	Significant Negative Relationship Found	No Significant Relationship Found
Mastery/Mastery- Approach GO	• Significant positive relationship except for Numerical Test at Time 2	 Wolters (2004) Agbuga and Xiang (2008) Elliot et al. (1999) 		
Mastery-Avoidance GO	 Significant positive relationship for Numerical Test at Time 1 and Verbal Test at Time 2 No significant relationship for Numerical Test at Time 2 and Verbal Test at Time 1 			
Performance- Approach GO	 No significant relationships 	 Agbuga and Xiang (2008) Elliot et al. (1999) Wolters (2004) 		
Performance- Avoidance GO	 No significant relationships 	 Agbuga and Xiang (2008) 	• Wolters (2004)	• Elliot et al. (1999)

Table 10.3. shows that previous GO studies all found significant positive relationships between a MAP GO and effort. Overall, the results of this study support previous research findings with the exception of the Numerical Test MAP GO at Time 2 which was not found to be significantly positively related to mental effort. A possible explanation for this may be that again, the relationships between GOs and mental effort are moderated by task type. It may be the case that if participants perceive a task as being too difficult this reduces the amount of mental effort invested by participants as both goal-setting and expectancy theories would predict. Another possible explanation is that participants may have adopted a MAP GO and viewed the experimental study as a learning opportunity but then did not invest too much mental effort when they came across difficulties during the Numerical Task (possibly because they did not value the Numerical task highly and/or because they knew the tests were not for "real"). These explanations are only possibilities and the relationships between a MAP GO and mental effort definitely need to be investigated further before any reliable conclusions may be drawn. However, overall, the results of this study and those of previous studies do indicate that a MAP GO seems to be significantly positively related to mental effort on tasks.

Unfortunately, none of the studies reviewed examined the relationships between a MAV GO and effort. Consequently, there is no comparison point for the results of this study. The results obtained are actually quite confusing since significant positive relationships were found between a MAV GO and mental effort on the Verbal Test at Time 2 and on the Numerical Test at Time 1. However, no significant relationships were found on the Verbal Test at Time 1 and the Numerical Test at Time 2. Overall, the relationships between a MAV GO and mental effort are quite weak (when present). This indicates that at times adopting a MAV GO **might** be related to investing more mental effort on tasks (or vice versa). However, the relationships between a MAV GO and mental effort are significant on different tasks at different time points and non-significant at other times. Once again task characteristics (such as task value and/or task difficulty) may possibly be influencing the relationships between a MAV GO and mental effort. Alternatively, as

305

discussed in Chapter 8, it is quite possible that these significant results were due to type I error. Consequently, it is not possible to offer a clear conclusion for the relationships between MAV GOs and mental effort. The large number of variables included in this study (creating a very large correlation matrix) may be considered to be a limitation since it increases the probability of type I error. Consequently, this should be kept in mind when interpreting the results.

The results of previous studies found a PAP GO to be significantly positively related to effort. However, this study did not find any significant relationships between a PAP GO and mental effort. This was quite a surprising finding since it was expected that a participant who is interested in performing better than others would invest more mental effort on tasks. It is possible that the non-significant findings in the current study are a result of the types of tasks used or perhaps the task value. For example, since participants' performance on these tasks did not have any significant influence on their university or work performance they may have not valued the tasks highly. Consequently, although they may have aimed to do better than others on the tasks (and adopted a PAP GO) they may not have invested as much mental effort as they would have done if these tasks were to influence their university or work performance (e.g. their final university grade or their chances of being promoted). With respect to previous studies, Elliot et al. (1999) assessed the relationships between GOs and mental effort for students' exams. Consequently, participants adopting a PAP GO in this study may have been more motivated to invest effort in order to do better than others on their exams. Wolters (2004) used mathematics grades and participants' GOs with respect to their mathematics classes. Agbuga and Xiang (2008) measured GOs during physical education classes. Since physical education classes may foster a competitive environment, it is possible that the significant positive relationship found between a PAP GO and effort in the study by Agbuga and Xiang (2008) was down to the task chosen too. The inconsistent findings obtained in this study when compared with previous research studies need to be investigated further in order to clarify the relationships between a PAP GO and effort. This study seems to indicate that adopting a high PAP GO may not increase the level of mental effort invested. However, previous studies indicate otherwise. Consequently, no clear-cut conclusions may be drawn from
the results of the present study with respect to the relationships between a PAP GO and mental effort apart from the fact that task value (or some other task characteristic) may possibly be influencing these relationships.

The relationships between a PAV GO and effort have been very inconsistent in the past with studies finding positive, negative and non-significant relationships (Agbuga & Xiang, 2008; Wolters, 2004; and Elliot et al., 1999, respectively). The results of the present study correspond with those obtained by Elliot et al. (1999) in that no significant relationships were found between a PAV GO and mental effort. It may be argued that a person who does not want to perform worst at a task (i.e. a person adopting a PAV GO) would tend to invest more mental effort than a person who is not interested in how their performance compares with that of others. However, only the results obtained by Agbuga and Xiang (2008) support this line of reasoning. Clearly, further investigation of this relationship is required. Since it is possible that task type (e.g. task difficulty and/or task value) influences the relationships between GOs and mental effort it may be the case that task characteristics could account for the differences in results across studies. Recommendations for future research, in order to clarify the relationships between GOs and mental effort, are made in Chapter 11 Section 11.3. The results obtained for the relationships between GO profiles and mental effort are discussed next.

The GO profile results indicated that participants adopting different GO profiles had significantly different mean-level mental effort scores on the Verbal Test but not on the Numerical Test. Participants in the 'High Mastery, Moderate Performance' Cluster had the lowest mean-levels of mental effort on the Verbal Test whilst participants in the 'High GOs' cluster had the highest mean-levels of mental effort. As described in the previous section, participants in the former cluster also had the lowest levels of self-efficacy. Consequently, it seems as though adopting the 'High Mastery, Moderate Performance' GO profile might not be very beneficial. A general discussion regarding which GOs and GO profiles might be considered to be more beneficial than others is provided in Section 10.5. The results obtained indicate that adopting a high PAP GO (and/or PAV GO) seems to make a difference with respect to the amount of mental

effort invested since participants adopting a lower PAP GO (and/or PAV GO) seem to invest significantly less mental effort on tasks than those adopting high PAP and PAV GOs. The GO profile results seem to somewhat contradict the non-profile perspective results since, as was previously discussed, PAP and PAV GOs were not found to be significantly related to mental effort. This finding does however fit in nicely with the research findings of the studies reviewed which found a PAP GO to be significantly positively correlated with mental effort. Again, the different findings across the Verbal and Numerical Tests seem to indicate that the relationships between GO profiles and mental effort may be moderated by the type of task (similarly to self-efficacy). Due to the nature of the tasks, it is possible that task difficulty influenced the amount of mental effort invested by participants on the different tests. However, as described in Section 10.2.2. it is possible that the different results obtained on the Verbal and Numerical Tests are a result of the small experimental sample size (resulting in small cluster sizes). Therefore, it is possible that either Type I or Type II error occurred due to the small cluster sizes. The small cluster sizes should be taken into serious consideration when interpreting these results. The small cluster sizes could also be responsible for the similarities in the relationships between GO profiles and other variables (mental effort and self-efficacy) e.g. the finding that participants having a 'high GOs' profile were found to have both higher self-efficacy and mental effort scores than participants adopting other profiles. Consequently, these similarities in the relationships between GOs and other variables should also be interpreted with caution.

Taken together, the results of the current study and those of previous studies seem to indicate that mental effort is mostly related to the approach GOs but not as much to the avoidance GOs. The results also indicate that participants adopting the 'High GOs' profile seem to invest higher levels of mental effort than those adopting the 'High Mastery, Moderate Performance' profile. Moreover, the GO profile results provide an indication of the relationships between mental effort and GOs being moderated by task type. However, due to the small sample size issue further research is required before making any sweeping conclusions and practical recommendations regarding promoting a MAP GO in order to increase mental effort on tasks. Recommendations for such research are made in Chapter 11 Section 11.3. Next, the relationships between GOs (and GO profiles) and performance are discussed.

10.4. The relationships between Goal Orientations and Performance

The relationships between GOs and performance are of extreme interest in a number of areas of psychology. Consequently, they have been widely researched. In Chapter 8 (Section 8.7.3.) the results of this study from the non-profile perspective were compared with the correlational analyses results of previous studies. In order to maintain clarity it was thought appropriate to present a comparison table (Table 10.4.) showing the results obtained in the current study and those of previous studies prior to discussing the results.

Table 10.4. Comparison across studies for relationships	s between Goal Orientations and Performance
---	---

GO being Investigated	Results of Present Study	Significant Positive Relationship Found	Significant Negative Relationship Found	No Significant Relationship Found
Mastery GO/Mastery- Approach GO	 Significant positive relationships for Numerical Tests. Non-significant relationships for Verbal Tests. 	 Bong (2009)¹ Yeo et al. (2009)³ Church et al. (2001)⁵ Elliot et al. (1999)⁷ 		 Elliot et al. (1999)⁸ Yeo et al. (2009)⁴ Bong (2009)² Elliot & McGregor (2001) Church et al. (2001)⁶
Mastery Avoidance GO	Non-significant		• Bong (2009) ¹⁰	 Elliot & McGregor (2001) Bong (2009)¹¹ Yeo et al. (2009)⁹
Performance- Approach GO	 Significant positive relationship for Numerical Test at Time 2. Non-significant relationships for Numerical Test at Time 1 and Verbal Tests. 	 Elliot et al. (1999)¹² Church et al. (2001)¹⁴ Elliot & McGregor (2001)¹⁶ Yeo et al. (2009)¹⁸ Bong (2009)²⁰ 		 Elliot et al. (1999)¹³ Church et al. (2001)¹⁵ Elliot & McGregor (2001)¹⁷ Yeo et al. (2009)¹⁹ Bong (2009)²¹
Performance- Avoidance GO	 Significant positive relationship for Numerical Test at Time 2. Non-significant relationships for Numerical Test at Time 1 and Verbal Tests. 		 Elliot et al. (1999) Church et al. (2001) Elliot & McGregor (2001)¹⁷ Bong (2009)²² Yeo et al. (2009)¹⁹ 	 Elliot & McGregor (2001)¹⁶ Bong (2009)²³ Yeo et al. (2009)¹⁸

¹ For Middle and Upper Elementary and Middle School; ² For Lower Elementary; ³On air traffic control task; ⁴On exam; ⁵For Graded Performance; ⁶For SAT scores; ⁷In Study 1; ⁸In Study 2; ⁹On exam; ¹⁰For Middle Elementary; ¹¹For Lower and Upper Elementary and Middle School; ¹²In Study 1; ¹³In Study 2; ¹⁴For Graded Performance; ¹⁵For SAT scores; ¹⁶For Study 2; ¹⁷For Study 3; ¹⁸On air traffic control task; ¹⁹On exam; ²⁰ For Upper Elementary and Middle School; ²¹ For Lower and Middle Elementary; ²²For Middle and Upper Elementary and Middle School; ²¹ For Lower and Middle School; ²³For Lower Elementary.

The results presented in Table 10.4. indicate that a few of the studies reviewed found a significant positive relationship between a MGO/MAP GO and performance whilst the rest did not find any significant relationships. All the studies reviewed, except that by Elliot and McGregor (2001) found different relationships between a MGO/MAP GO and performance depending on the task type, time at which participants took part in the study, or age of participants. For example, Bong (2009) found a significant positive relationship between a MAP GO and performance for Middle and Upper Elementary and Middle School students and a non-significant relationship for lower elementary school students. Church et al. (2001) found significant positive relationships for graded performance but not for SAT scores. It is possible that the differences occurring within studies are due to general GO measures being used or participants being quite young. However, these differences (e.g. Yeo et al., 2009) may also be a result of differences across tasks.

The results of the present study are similar to those of previous studies in that different relationships between a MAP GO and performance were found on different tasks. Significant positive relationships were found between a MAP GO and performance on the Numerical Tests but non-significant relationships were found on the Verbal Tests. Consequently, it seems as though some task characteristic/s might be moderating the relationships between a MAP GO and performance. This possible moderation effect was also found for the relationships between GOs and self-efficacy as well as mental effort. Although it was quite an unexpected finding, it is also an extremely important one. If task characteristics moderate the relationships between GOs and other variables, this may explain the inconsistencies in results across studies as well as the inconsistencies for GOs on different tasks within studies. Recommendations for future research based on these findings are presented in Chapter 11 Section 11.3.

The relationships between a MAV GO and performance in this study were found to be consistent with the results obtained by Elliot and McGregor (2001), Bong (2009, for Lower and Upper Elementary and Middle School students) and Yeo et al. (2009, for exam performance). Bong (2009) also found significant negative relationships between

a MAV GO and performance for Middle Elementary School students. Overall, these results indicate that a MAV GO does not seem to be beneficial to performance.

The results of studies assessing the relationships between a PAP GO and performance are not straightforward. Table 10.4. indicates that some studies found significant positive relationships and others found non-significant relationships. All the studies reviewed obtained inconsistent results across tasks, participant ages, or time at which participants took part in the study. The results of this study are very similar to those of previous studies since a significant positive relationship was found between a PAP GO and performance on the Numerical Test at Time 2 but non-significant relationships were found on the Numerical Test at Time 1 as well as on the Verbal Tests. The inconsistencies found here (and in previous studies) point towards the possibility that some task characteristics are moderating the relationship between GOs and performance.

Finally, it seems as though previous studies found significant negative and/or nonsignificant relationships between a PAV GO and performance. Again, most of the studies reviewed found different relationships between a PAV GO and performance (within the same research study) depending on the task type, study, or age of participants. The results of this study correspond with those of previous studies in that different relationships were found depending on task type. However, they are not consistent with previous research results in that a significant **positive** relationship was found for the Numerical Test PAV GO at Time 2 (as opposed to significant negative or non-significant relationships). Once again, it seems as though task type may be moderating the relationships between GOs and performance.

The possibility of task characteristics moderating the relationships between GOs and performance was investigated by three studies reviewed. Steele-Johnson et al. (2000) investigated differences in performance as a result of GOs adopted on simple and difficult tasks. They hypothesised that when a task is simple, rehearsing strategies (associated with performance GOs) as opposed to elaborate strategies (associated with mastery GOs) may have beneficial effects on performance. On the other hand,

when the task is difficult, dedicating cognitive resources to mastering the task (associated with mastery GOs) may be more beneficial than rehearsing known but not ideal task strategies (associated with performance GOs). They found partial support for their hypothesis in that "individuals with a performance goal orientation outperformed those with a learning orientation on a simple task, but no goal orientation effects were observed in the difficult task condition" (Steele-Johnson et al., 2000:730). Steele-Johnson et al. (2000) did not actually measure task difficulty but rather they assigned simple and difficult tasks to participants. However, according to Mangos and Steele-Johnson (2001) it is important to examine the effects of *subjective* task complexity (or task difficulty) as well as objective task complexity on the relationships between GOs and performance. According to Mangos and Steele-Johnson (2001:171) subjective task complexity refers to "perceptions of task complexity" whilst objective task complexity refers to "complexity related to task characteristics or behavioural demands of the task". The study by Steele-Johnson et al. (2000) investigated the effects of objective task complexity on the relationships between GOs and performance. Mangos and Steele-Johnson (2001) investigated how subjective task complexity mediates the relationships between induced GOs and performance. They induced mastery and performance GOs and found that GO effects on performance were mediated by subjective task complexity. Their results indicated that participants in the induced mastery GO condition reported higher levels of subjective task complexity than participants in the induced performance GO condition. They also found that subjective task complexity had a significant negative effect on performance and participants in the induced performance GO condition performed better than participants in the induced mastery GO condition. The authors' explanation for these findings are that participants having high mastery GOs look for higher levels of challenge which causes them to invest more effort on tasks. This increased effort in turn leads to an increase in subjective task complexity which has a negative influence on performance. This explanation is not consistent with that of Steele-Johnson et al. (2000) who suggest that mastery GOs are associated with the use of better cognitive strategies on difficult tasks (which in turn should lead to higher performance). Having said that, Steele-Johnson et al. (2000) did not find support for participants having high mastery GOs performing better on more complex tasks. Yeo

et al. (2009) also investigated the effect of task difficulty (which they refer to as 'task demands') on the relationships between GOs and performance. They found that the relationship between a PAP GO and performance switched from positive to negative as task demands increased. Although further research is definitely required in order to understand these relationships better, these studies, along with the results of the present study indicate that it is important to take into consideration task characteristics such as task difficulty in order to better understand the relationships between GOs and performance.

The results of this study are also important because they provide evidence that a PAV GO is not necessarily detrimental to performance. Consequently, although the approach GOs seem to be more consistently significantly positively related to performance, the avoidance GOs may also be significantly positively related, or at least not significantly negatively related. However, before making any decisions regarding which GOs should be promoted (if it is possible for people to change their GOs) research must be carried out in order to investigate which task characteristics might be moderating the relationships between GOs and performance and how. Research examining the direction of causality of the relationships between GOs and performance is also necessary. These recommendations for future research will be discussed in further detail in Chapter 11 Section 11.3.

When the relationships between GOs and performance were examined in terms of GO profiles it was found that participants adopting different GO profiles on the Numerical Test had significantly different levels of performance. Participants adopting the 'High GOs' profile were found to have significantly higher levels of performance than participants adopting the 'High Mastery, Moderate Performance' profile and the 'High Performance-Avoidance' profile. These results are consistent with those obtained when examining the relationships between GO profiles and self-efficacy and mental effort. As discussed in Sections 10.2.2. and Section 10.3. the small cluster sizes should be kept in mind when interpreting the results obtained.

Pastor et al. (2007) and Fortunato and Goldblatt (2006) also investigated the relationships between GO profiles and performance in their studies. A comparison of the results of this study (for Numerical Test GO profiles) and those of the studies by Pastor et al. (2007) and Fortunato and Goldblatt (2006) revealed some interesting findings. Previously (in Section 10.1.), the GO profiles obtained from the three studies were compared and similarities in the profiles obtained across studies were discussed. In order to maintain clarity, a table (Table 10.5.) was drawn up so as to indicate the similar GO profiles obtained across studies. The performance of participants in each cluster, compared with that of participants in different clusters, is indicated in brackets for each of the studies.

Type of GO Profile	Present Study [Numerical Test Performance]	Pastor et al. (2007) [Semester GPA]	Fortunato & Goldblatt (2006) [Exam Performance]
'High Mastery, Low Performance'	3 (low)	5 (low)	2 (low)
'High GOs'	1 (high)	3 (average)	3 (high)
'High Approach, Low Avoidance'		2 (high)	4 (high)
'High Performance- Avoidance'	4 (low)		1 (low)*

Table 10.5. Relationships between Goal Orientation Profiles and Performance

Key: * On exams 1 and 2.

The Numerical Test performance of participants in Cluster 3 (in this study) was not significantly lower than that of participants in Cluster 1. However, the mean performance scores on the Numerical Test (see Table 8.58.) of participants in Cluster 3 (mean = 3.18) do seem to be quite low compared to the mean performance scores of participants in Cluster 1 (mean=6.85). Consequently, the performance of participants in Cluster 3 was included as 'low' in Table 10.5. rather than average.

Table 10.5. shows that the results across studies with respect to the relationships between GO profiles and performance seem to be quite consistent with participants adopting 'High GOs' and 'High Approach, Low Avoidance' profiles performing better than participants adopting 'High Mastery, Low Performance' and 'High Performance Avoidance' GO profiles. Consequently, it seems as though some GO profiles may be more beneficial than others in terms of performance. This will be discussed in further detail in Section 10.5.

When analyses were carried out in order to test for significant differences in performance (depending on the GO profile adopted) on the Verbal Test no significant differences were found. These results provide further support for the possibility that the relationships between GOs and performance may be moderated by task type. Understanding how the relationships between GOs and performance are moderated by the type of task may provide invaluable insight into which GOs are most beneficial in which circumstances. For example, it may be the case that if a person is already good at selling a particular computer software program, adopting a MAP GO will not significantly increase their sales performance because they already understand how to sell the computer software program and will only make small increases in learning and performance. On the other hand, if this person adopts a PAP GO they may perform significantly better because of the different standards (performing better than the other salespeople) set by adopting the PAP GO. Another person might not fully understand the techniques required to sell the same computer software, in which case adopting a MAP GO might significantly improve this person's sales. Task characteristics such as task difficulty and task value may also influence the relationships between GOs and performance. Recommendations for future research based on the findings of this study are made in Chapter 11 Section 11.3. Prior to discussing the stability and taskspecificity of GOs the next section addresses the question regarding whether a particular GO or GO profile is better than the others.

10.5. Is one Goal Orientation or Goal Orientation Profile better than the others?

Since GOs were found to be related to variables such as mental effort and performance researchers have been searching to discover whether there is a 'best' GO which will improve people's performance, mental effort, self-efficacy, and use of better cognitive strategies, amongst others. The discovery of a 'best' GO would make a huge difference to organisations, education and sports (amongst others). It may lead

to the development of strategies to encourage people to adopt this 'best' GO. As a result they may perform better, invest more effort and feel more confident about their abilities. Overall, the results of this study and those of previous studies, point towards the approach GOs being significantly positively related to self-efficacy, mental effort and performance. In contrast, the relationships between the avoidance GOs and self-efficacy, mental effort and performance have not been entirely consistent. Therefore, it seems as though approach GOs may be more beneficial than avoidance GOs (if high levels of self-efficacy, mental effort and performance are considered desirable).

A recurring finding throughout this study is that the relationships between GOs and other variables seem to be moderated by task type. As discussed earlier, a number of other studies (e.g. Bong, 2001; Yeo et al., 2009; and Church et al., 2001) also found GOs to be related differently to variables on different tasks. Consequently, it may be time to stop attempting to find the 'best' GO and focus instead on determining which task characteristics cause GOs to be more or less related to other variables. Such research may be able to explain the inconsistencies found across studies (and within the same studies across tasks) for the relationships between GOs and other variables. Recommendations for future research of this kind are made in Chapter 11 Section 11.3.

In terms of GO profiles, the results seem to indicate that the 'High GOs' profile might be a beneficial profile to adopt since it is more strongly related to self-efficacy, mental effort and performance than other GO profiles. On the other hand, the 'High Mastery, Moderate Performance' profile does not seem to be quite so beneficial since participants adopting this profile seemed to have the lowest levels of self-efficacy and mental effort. Participants adopting this profile were also found to have significantly lower levels of performance than participants adopting the 'High GOs' profile. Participants adopting the 'High Performance-Avoidance' profile were also found to have significantly lower levels of performance than those adopting the 'High GOs' profile. Therefore, this profile too might not be such a beneficial one to adopt. As described in Section 10.4., the results obtained in this study were consistent with those obtained by Pastor et al. (2007) and Fortunato and Goldblatt (2006), with

respect to the relationships between GO profiles and performance. Consequently, there seems to be a growing body of evidence that certain GO profiles seem to be more beneficial than others. More specifically, adopting 'High GOs' or 'High Approach, Low Avoidance' GO profiles may be more beneficial than adopting 'High Performance-Avoidance' or 'High Mastery, Moderate Performance' GO profiles. The main difference between the two sets of profiles (those that are more strongly correlated vs. those that are less strongly correlated with performance) is that the former have a high PAP GO whilst the latter have a moderate to low PAP GO. This indicates that a PAP GO may be an important GO to adopt. This finding contradicts the previously prevailing belief that mastery GOs are adaptive whilst performance GOs are non-adaptive (e.g. Meece & Holt, 1993; Schraw et al., 1995). Instead, it seems as though adopting both MAP and PAP GOs may be advantageous since the 'more beneficial' profiles were found to have high MAP and PAP GOs. The profile results reflect the results obtained using the non-profile perspective in which the approach GOs were found to be related to self-efficacy, mental effort and performance.

As one would expect from the names of the concepts themselves, the 'approach' GOs seem to be more positive and constructive than the 'avoidance' GOs. However, this is not to say that adopting avoidance GOs would be disadvantageous or reduce the beneficial effects of the approach GOs. In fact, participants who scored highly on all four GOs ('High GOs' profile) seemed to have high self-efficacy, mental effort and performance on tasks. As discussed earlier (Section 10.2.1.) it is possible that the *reasons* for adopting the avoidance GOs may influence the outcome (thus explaining the inconsistent results obtained across studies). Since the relationships between GO **profiles** and performance have been quite consistent across studies so far, this may be an indication that it is better to use profile perspective as opposed to the non-profile perspective when investigating these relationships.

The results of this study also point towards the relationships between GO profiles and self-efficacy, mental effort and performance being moderated by task type. This was evident since no significant differences were found in the self-efficacy and performance of participants adopting different GO profiles on the Verbal Test but

significant differences were found on the Numerical Test. Moreover, no significant differences in mental effort were found for participants having different GO profiles on the Numerical Test but significant differences were found on the Verbal Test. Once again, suggestions for future research based on these findings will be made in the next Chapter (Section 11.3.). Moving away from the relationships between GOs and other variables, the next section focuses on the stability of General GOs over time.

10.6. Examining the Stability of General Goal Orientations over Time

Research Questions 2a and 2b focused on the stability of General GOs over time (from the profile and non-profile perspectives). The non-profile perspective results will be discussed first since they might provide more insight into the profile perspective results. The stability of GOs over time from the non-profile perspective was assessed in terms of correlations over time and in terms of mean-level change. The mean-level change results indicated that General PAP and PAV GOs decreased significantly from Time Q1 to Time Q2. It is not easy to explain why General PAP and PAV GOs changed significantly over time but General MAP and MAV GOs did not. General GOs are assumed to be stable dispositions. However, the changes observed in this study suggest that this is not necessarily the case. Since the majority of participants in this study were students, it is possible that the stability of General MAP and MAV GOs over time is a result of the constant focus on learning encouraged at university. On the other hand, General PAP and PAV GOs may have been more susceptible to change as a result of changing situational factors. For example, if participants had been working on group assignments the strong emphasis on teamwork (as opposed to competition) may have influenced their performance GOs.

The correlations over time were found to be lower than those of other GO studies. This may be because the GO studies reviewed assessed correlations over time using task-specific GO measures. If General GO measures are less accurate than task-specific GO measures this may have been the cause of the low correlations for General GOs over time. This is possibly the case since General GOs were found to be significantly different from task-specific GOs (as will be discussed in Section 10.8. below). Since General GOs are assumed to be general, stable traits it was thought useful to compare their stability over time with those of other general, stable traits. Personality traits were chosen for comparison since these have been widely researched and found to be stable traits. The correlation coefficients for General GOs over time were found to be lower than those of personality traits thus indicating less stability over time. These findings contradict the assumption that GOs are stable dispositions.

Since the time interval between the General GO measures at Time 1 and Time 2 varied from between 4 to 51 weeks, correlation coefficients for participants having a 4-8 week time interval and those having a 9-51 week time interval were drawn up. The correlation coefficients for MAP, PAP and PAV GOs seemed to be lower for participants having a longer time interval between the Time 1 and Time 2 questionnaires. The MAV GO however, seemed to be an exception, with the stability increasing with a longer time interval. This finding was quite unexpected and it is still not clear why the stability of MAV GOs would increase with a longer time interval. However, when Fisher z tests were carried out, the changes in stability for MAP, MAV and PAV GOs were not found to be significant. The PAP General GO however, was found to have significantly lower stability the longer the time interval between Time 1 and Time 2 measures. This finding was consistent with Payne et al. (2007), who found that stability over time decreased when the time intervals between measures were longer. However, Payne et al. (2007), who used a 3-factor model of GOs in their study, also found this decrease in stability for MAP and PAV GOs.

Clearly, the results of this study along with those of Payne et al. (2007) indicate that further research into the stability of General GOs over time is necessary since there is a growing repertoire of evidence that General GOs are not quite as stable as they were initially thought to be. These findings are not entirely surprising because although a number of researchers assume that GOs are general, stable traits (as discussed in Chapter 4 Section 4.1.) most studies tend to measure GOs using task-specific measures. The use of task-specific measures and the assumption that GOs are general, stable traits are rather contradictory (which is partially why this issue was addressed in the current study). General GOs not being stable would mean that they are more state-like rather than trait-like. Consequently, it might be easier to encourage individuals to adopt GOs which may be more beneficial for them in different circumstances than if they were trait-like. For example, if MAP GOs are found to be highly beneficial during training whilst PAP GOs are found to be more beneficial on the job it might be possible to develop ways to help employees to adapt their GOs depending on the situation e.g. to adopt a MAP GO during training and a PAP GO on the job. Moreover, if General GOs are not stable over time, then the whole idea of trait GOs interacting with state GOs might not be feasible. Rather than assessing interaction effects it may be possible to manipulate GOs and examine the effects of these manipulations without having to worry about possible interactions.

The comparison of participants' GO profiles from Time Q1 to Time Q2 indicated that 67% of participants adopted the same General GO profile over time. Moreover, it was found that more participants adopted the same General GO profile from Time 1 to Time 2 than would have been expected by chance alone (at ρ <0.05). This tendency towards no cluster change seems to point towards the possibility of General GO profiles being stable over time. However, the results of this study alone are not enough to support such a claim. It is possible that this significant result is due to type I error. If GO profiles were found to be stable over time the profile results would contradict the non-profile results with respect to the stability of General GOs over time. A potential explanation for these (possibly) contradictory results is that although there may be fluctuations in participants' GOs, overall, participants are still adopting the same GO profiles over time. For example, the mean-level General PAP GO of a participant adopting a 'High Mastery, Moderate Performance' GO profile may decrease significantly from a certain point in time to another. Nevertheless, the lower mean General PAP GO score may still be considered to be within the 'moderate' range when compared to that of participants adopting a 'High Performance Avoidance' GO profile. However, as described earlier, the results of this study do not provide enough evidence to make a claim that General GO profiles are stable over time. Hence, before taking the explanation of these 'possible contradictions' any further additional research into the stability of General GO profiles over time is required.

Recommendations for such research are made in Chapter 11 Section 11.3. The next section addresses the stability of task-specific GOs over time.

10.7. Examining the Stability of Task-Specific Goal Orientations over Time

The results relating to Research Questions 5a and 5b focus on determining the stability of task-specific GOs over time (from the profile and non-profile perspectives). The task-specific GOs included Verbal and Numerical Test GOs as well as Hypothetical Task GOs. In keeping with the same order as that of Chapter 9, Research Question 5b is discussed first.

The stability of task-specific GOs over time was assessed in terms of mean-level change as well as correlations over time. The results presented in Chapter 9 (Section 9.1.2.) show that there were no significant changes in the mean scores of participants over time for all four GOs on the Hypothetical Task, as well as the Verbal and Numerical Tests. This indicates good levels of stability over time for the task-specific GOs. The Verbal and Numerical Test GOs were therefore found to show stability over time notwithstanding the induction attempts. Although there are other possible reasons for the inductions not being entirely successful (as will be discussed in Section 10.9. below) this lack of change in the Verbal and Numerical Test GOs over time provides further support of their stability. Nearly significant differences were found when assessing the mean-level change of Hypothetical Task MAP and PAV GOs over time. Although it is possible that these were due to the longer time intervals between measures, the correlational analyses do not quite support this explanation since no significant differences in the strengths of the correlation coefficients were found for participants having different time intervals between the Time Q1 and Time Q2 questionnaires. Possible explanations for these nearly significant differences are provided (next) when discussing the correlational analyses.

The results for the correlations over time indicated high levels of stability for Verbal and Numerical Test GOs and slightly lower levels of stability for the Hypothetical Task GOs. The correlation coefficients for Verbal and Numerical Test were very similar to those obtained by Fryer and Elliot (2007) but slightly lower than those found by Elliot and McGregor (2001). When the correlations over time for Verbal and Numerical Tests were compared with those of the Hypothetical Task GOs (Chapter 9 Section 9.1.2.) the Hypothetical Task PAV GO was found to have significantly lower stability than the respective Verbal and Numerical Test GOs. Therefore, it seems as though Hypothetical Task GOs may not be as stable as Verbal and Numerical Test GOs. A possible reason for this is that participants were asked to imagine the GOs they would adopt on the Hypothetical Task whereas they had actually completed the Verbal and Numerical Tests. Consequently, the measures for Hypothetical Task GOs may not have been as accurate as those for the Verbal and Numerical Tests resulting in more variation in the responses. This explanation may also account for the lower stability of General GOs over time. Additional research into the stability of task-specific GOs over time is required in order to further clarify whether these are in fact stable over time (especially longer periods of time). Recommendations for future research on this matter are presented in Chapter 11 Section 11.3. The results regarding the stability of task-specific GO profiles over time are discussed next.

The GO profile results indicated that 66%, 63% and 56% of participants adopted the same GO profile at Time 1 and Time 2 on the Hypothetical Task, Verbal Test, and Numerical Test, respectively. When the observed frequencies of participants to clusters (GO profiles) were compared with those that would have been expected by chance the Chi-square test results obtained (see Tables 9.13., 9.16., and 9.19.) indicate that more people than would have been expected by chance alone at p<.01 adopted the same Verbal Test GO profile from Time E1 to Time E2. This was also the case for Numerical and Hypothetical Task GO profiles but at p<.05. The results show that fewer participants changed clusters on the Verbal Test from Time 1 to Time 2 than they did for the Numerical Test and Hypothetical Task. The higher changes across clusters for the Hypothetical Task than the Verbal Test may have been due to the fact that participants did not actually complete these tasks and were therefore not as certain (and more susceptible to change) than they would have been had they actually completed the task. However, it may also be a result of the much smaller sample size in the experiment than in the survey. With respect to the changes in Numerical Test

GO profiles over time, it is possible that task difficulty may have contributed towards these changes in clusters. Although these results indicate that GO profiles are more stable than what would have been expected by chance further research investigating the stability of task-specific GOs and GO profiles over time is required before any concrete conclusions are drawn.

Taken together, the results obtained when assessing the stability of General and Taskspecific GOs over time indicate that, in terms of the non-profile perspective, taskspecific GOs seem to have higher levels of stability over time than General GOs. In fact, General GOs do not seem to be as stable as they were initially thought to be. In terms of GO profiles, more participants would have been expected to change clusters over time by chance alone for both General and Task-specific GOs. These results seem to point towards the stability of General and Task-specific GO profiles over time. However, the results of this study alone are not enough to make such a claim and further investigation of the stability of GO profiles (General and Task-Specific) over time is definitely required.

10.8. Assessing the Generality and/or Task-Specificity of Goal Orientations

In order to determine whether GOs are General or task-specific the stability of participants' GOs *across tasks* was examined. Moreover, participants' General GOs were compared with their task-specific GOs. These comparisons were made using both the profile and non-profile perspectives. In order to maintain the same order as Chapter 9 the comparison of GOs across tasks (from the profile and non-profile perspectives) is discussed first.

10.8.1. Assessing the Task-Specificity of Goal Orientations

The GO profile results indicated that a higher percentage of participants adopted the same GO profile on the Verbal and Numerical Tests (72%) than they did on the Hypothetical Task and Verbal Test (57%) and on the Hypothetical Task and Numerical Test (61%). The Chi-square test results (presented in Section 9.2.1.) showed that more

participants would be expected to adopt the same Hypothetical Task and Verbal Test GO profiles by chance alone. This was also the case for the Hypothetical Task and Numerical Test GO profiles adopted by participants. However, more participants than would have been expected by chance alone adopted the same Verbal and Numerical Test GO profiles (at ρ <0.01). Before discussing these differences any further, the non-profile perspective results will be described.

The non-profile perspective results reflected those from the profile perspective. Overall, the mean-level change results indicated that Verbal and Numerical Test GOs were not significantly different from each other. However, participants seemed to have significantly higher Hypothetical Task GOs than Verbal and Numerical Test GOs (Chapter 9 Section 9.2.2). Moreover, the Fisher z test results showed that the Verbal and Numerical Test GOs were found to be significantly more strongly correlated with each other than with the Hypothetical Task GOs. Duda and Nicholls (1992) did not find any differences in participants' GOs across domains. Bong (2001) and Magson et al. (2008) found performance GOs to generalise across tasks and MGOs to be taskspecific. On the contrary in this study, overall, significant changes were found across the Hypothetical Task and Verbal Test as well as across the Hypothetical Task and Numerical Test. These findings were consistent with those of Anderman and Midgley (1997). The type of tasks chosen in the different studies may account for the different results obtained across studies.

As discussed in Chapter 9 Section 9.2.1., as opposed to task-specificity, the results of this study seem to provide more support for domain-specificity of GOs. Verbal and Numerical Tests may be considered to fall within the same domain, that is, the academic domain. No significant differences in GOs or GO profiles were observed across Verbal and Numerical Tests. On the other hand, the Hypothetical Tasks would fall into different domains (e.g. sales and training) than the Verbal and Numerical Tests. Significant differences were found between Hypothetical Task and Verbal Test GOs (and GO profiles) as well as between Hypothetical Task and Numerical Test GOs (and GO profiles). Consequently, the results obtained in this study seem to indicate domain-specificity as opposed to task-specificity. It is possible that different studies

obtained different results when assessing changes in GOs across tasks due to the nature of the tasks chosen. The non-significant differences found may be a result of tasks measured being in the same domain. On the other hand, any differences found may have resulted from domain-specificity.

The findings presented above imply that it might be better to use task- or domainspecific measures of GOs as opposed to General measures because one may not be measuring the GOs of interest when using General measures. For example, if a researcher would like to measure participants' GOs during training and simply uses a General GO measure, the participants' responses on the General GO measure might not necessarily be an accurate reflection of their GOs during training. On the other hand, if the researcher adapts a GO measure to the training task at hand, this will be more likely to provide an accurate representation of the GOs being adopted on the this task. However, before drawing any conclusions regarding the Generality, Domainor Task-specificity of GOs, the results assessing the similarities and differences between General and Task-Specific GOs are discussed.

10.8.2. Comparing Participants' General Goal Orientations with their Task-Specific Goal Orientations

In-keeping with the same structure as the results, the profile perspective results will be discussed first followed by the non-profile perspective results. The results presented in Chapter 9 (Section 9.2.3.) indicate that 71% of participants adopted the same General and Hypothetical Task GO profiles. Moreover, 66% of the experimental participants adopted the same General and Verbal Test GO profiles and 55% adopted the same General and Numerical Test GO profiles. The Chi-square test results indicated that more participants than would have been expected by chance alone adopted the same General and Hypothetical Task GO profiles. This was also found to be the case for General and Verbal Test GO profiles. However, more participants would have been expected to adopt the same General and Numerical Test GO profiles. However, more participants by chance alone (at p<0.01). The results obtained in this study are not enough to come to a definite conclusion regarding General and Numerical Test GO profiles being

significantly different from each other. However, they raise a flag regarding the utility of measuring General GOs as opposed to task-specific (or domain-specific) GO. Before discussing this any further, the results comparing General and Task-specific GOs from the non-profile perspective are reviewed.

Tables 8.43. and 8.45. presented the correlation coefficients for General GOs with Hypothetical Task, as well as with Verbal and Numerical Test GOs. These correlations were on the low side (ranging between 0.28 and 0.52, ρ <0.01) compared with the correlations for General and Task-Specific GOs over time (ranging between 0.44 and 0.78, ρ <0.01). This points towards participants' General GOs being different from their task-specific GOs.

The mean-level change results showed that participants' General GOs were found to be significantly higher than their Verbal (PAP and PAV) and Numerical Test GOs. In addition, participants General PAP and PAV GOs were found to be significantly higher than their respective Hypothetical Task GOs whilst participants' General MAP GO was found to be significantly lower than their respective Hypothetical Task GO. With the exception of the Hypothetical Task MAP GO, the participants' General GOs were found to be significantly higher than their task-specific GOs. This may be due to General measures of GOs not being specific enough. For example, it is possible that when given a General measure of GOs participants may think back to times when they adopted the different GOs and therefore tend to score more highly on each of the GOs than they would if they were given a task-specific measure. This explanation is based on the assumption that GOs are not stable traits and that participants tend to adopt different GOs at different times. However, the results obtained when assessing the stability of GOs do reflect this possibility. The finding that a MAP Hypothetical Task GO was significantly higher than the respective General GO is not consistent with this explanation. However, it is possible, as mentioned earlier that the Hypothetical Task GO measures were not as accurate as the Verbal and Numerical Test GO measures since participants did not actually complete the tasks. Taken together, the results of this study seem to indicate that there is a strong possibility of General GOs being different from Task-specific GOs.

When assessing the extent to which General GOs are similar to Task-Specific GOs the non-profile perspective analyses did not totally correspond with the profile analyses. The non-profile results showed General GOs to be significantly different from Hypothetical Task, Verbal and Numerical Test GOs overall. However, the profile results showed that only for General and Numerical Test GO profiles were more participants expected to adopt the same GO profiles by chance alone (ρ <0.01). These inconsistencies across the findings may be due to the GO profiles not being as sensitive to changes as GOs from the non-profile perspective (as discussed in Section 10.6.). The implications of these findings are discussed in Section 11.1.

Overall, the results point towards General GOs being different from Task-specific GOs. Although there is not enough evidence as yet to conclude that General GO profiles are different from task-specific GO profiles, this study provides a good indication of this possibility. Therefore, if General measures of GOs are used they may not be accurately measuring the actual GOs adopted by participants on the task at hand. Consequently, the results of this study strongly suggest that it might be better to use domain- or taskspecific measures of GOs as opposed to General measures. Although the results of this study point towards domain-specificity it may still be useful to use *task*-specific measures of GOs as opposed to domain-specific measures. This is because as discussed in Sections 10.3., 10.4. and 10.5. there is a strong possibility that task characteristics may influence the relationships between GOs and other variables.

10.9. Is it best to use Goal Orientations or Goal Orientation profiles in research and practice?

On a number of occasions throughout this study the GO profile results were not entirely consistent with the results obtained when examining GOs from the non-profile perspective (e.g. when assessing whether participants adopt different General and Task-Specific GOs). This leads to the question of whether it is best to examine GOs in terms of the profile or non-profile perspective.

As mentioned earlier (Section 10.6.), it may be the case that even though mean-level changes in GOs are statistically significant, these changes might not be large enough to change the GO profile adopted by the individual. If this is the case, then would it be more useful to use the profile perspective or would this not provide sufficient information regarding participants' GOs? The differences in results obtained when using the profile and non-profile perspectives may also provide some insight into the inconsistent findings across different studies examining GOs. Although as yet it is not possible to provide a straightforward answer to the question posed above, it seems as though it might be useful to make use of the different perspectives depending on the aims one would like to achieve. For example, if a researcher is interested in investigating how and why GOs change over time and across tasks, it might be useful to use the non-profile perspective since this will provide more detail. On the other hand, if a researcher is more concerned about discovering patterns in the relationships between GOs and other variables on different tasks, then the profile perspective might be more appropriate for his/her purposes. However, prior to making any recommendations regarding which perspective should be used further research into the utility and accuracy of measuring GOs and GO profiles is required. Recommendations for such research are made Chapter 11 Section 11.3. The next section focuses on the interaction effects of General and Induced GOs on the performance of participants.

10.10. Examining the Effects of the Interactions between Trait and Induced Goal Orientations on Participants' Performance

In Chapter 4 studies investigating the interactions between state and trait GOs were discussed. The three studies discussed, those by Chen and Mathieu (2008), Gerhardt and Luzadis (2009) and Jagacinski et al. (2001), all used a 2-factor model of GOs. Consequently, this study attempted to further knowledge on these interactions by using the 2x2 model of GOs as well as examining the interaction effects using the GO profile approach. Unfortunately, due to the small sample size it was not possible to investigate the interaction effects from the profile perspective.

Prior to assessing the interaction effects of trait and induced GOs on performance it was found that the inductions were not successful. This greatly reduced the chances of there being any interaction effects on performance. In fact, when tested, no significant interaction effects on performance were found. Great attention was paid to the way that successful inductions were carried out in previous studies and the inductions of the current study were based on these (e.g. Chen & Mathieu, 2008; Kozlowski et al., 2001). There is a strong possibility that the lack of success of the inductions may not be down to the way the inductions were carried out but a result of the fact that a large number of participants who were in the MAP and PAP induction conditions already had high MAP and PAP GOs, respectively. Consequently, the inductions might not have made much of a difference to these participants' GOs. If only participants who had low MAP and PAP GOs had been included in the respective induction conditions the inductions might have been successful and the possibility of interaction effects on performance would have increased greatly. However, unfortunately, due to the small experimental sample size it was not possible to do this. The finding in Chapter 9 (Section 9.1.2.) that the Verbal Test PAP induction seemed to decrease participants' PAV GOs indicates that there is a possibility of the inductions having influenced GOs. Another possible reason for the inductions not being successful is that in quite a few cases there was not a big time lag between the first and second experimental sessions. Consequently, participants may have answered the Time E2 GO questionnaires similarly to the Time E1 GO questionnaires. Therefore, the lack of a longer time lag between the Time E1 and Time E2 experiments may be considered a limitation. Consequently, GO inductions should be definitely be investigated further. Such recommendations for future research will be made in Chapter 11 (Section 11.3.). As discussed in the previous sections, the results of this study seem to strongly suggest that GOs are domain-specific. Consequently, investigating the interactions between state and trait GOs might not necessarily be appropriate or useful. The finding that GOs are domain-specific is inconsistent with the assumption that GOs are general dispositions (traits). This brings into question the utility of investigating the interactions between trait and state GOs. However, this does not mean that GOs should not be induced or that GO inductions should not be investigated further. On the contrary, GO manipulations may be extremely useful if certain GOs are found to be

more 'beneficial' than others. In such a case investigating how to promote the 'beneficial' GOs might be of great benefit to organisations. As discussed earlier, the fact that the majority of the sample in this study had a 'High GOs' profile may have been the cause of the unsuccessful induction attempts. The implications of these results as well as recommendations for future research are discussed in Chapter 11 (Sections 11.1. and 11.3.).

10.11. Synopsis

In this chapter the results of this study were discussed and compared with results of previous GO studies. This revealed a number of important findings which are summarised below.

- For the purposes of this study the 2x2 model of GOs did not seem to improve on the 3-factor model in terms of GO profiles.
- A comparison of GO profiles found in this study and those obtained in previous studies revealed a number of common GO profiles (refer to Section 10.1.).
- Overall, MAP and PAP GOs were found to be more strongly related to self-efficacy, mental effort and performance than MAV and PAV GOs were.
- The relationships between GOs (and GO profiles) and other variables (that is, selfefficacy, mental effort, and performance) seems to be moderated by task type (possibly task difficulty and/or task value).
- The present study and previous studies indicate that the 'High GOs' and 'High Approach, Low Avoidance' GO profiles may be more beneficial profiles to adopt than the 'High Performance-Avoidance', 'High Mastery, Low Performance' and 'High Mastery, Moderate Performance' GO profiles since they were found to be more strongly related to self-efficacy, mental effort and performance.
- General GOs do not seem to be quite so stable over time. However, more participants would have been expected to change clusters from Time 1 to Time 2 by chance alone (at ρ<0.01). Therefore there is a possibility that General GO profiles may stable over time.
- Task-specific GOs and GO profiles seem to be stable over time.
- Participants' GOs (and GO profiles) seem to be domain-specific.

 Overall participants were found to have significantly higher General GOs than Taskspecific GOs. Moreover, at a significance level of 0.01, more participants would have been expected to adopt the same General and Numerical Test GO profiles by chance alone.

The theoretical and practical implications, limitations of the current study as well as recommendations for future research are discussed in the next chapter.

Chapter 11: Conclusion

11.0. Introduction

In the previous chapter the research findings of this study were discussed and compared with those of previous studies. In this chapter the theoretical and practical implications of these research findings are discussed. This is followed by a description of the limitations of the current study and a section proposing recommendations for future research.

11.1. Theoretical and Practical Implications

The findings of this study give rise to a number of implications regarding theory and practice related to the concept of GOs. The theoretical implications are discussed first and will be followed by a description of the practical implications of this study.

11.1.1. Theoretical Implications

As things stand, it is extremely difficult to compare the results of GO studies since these use different models and measures of GOs. Moreover, there seems to be an issue of results being inconsistent even when similar models and measures are used. The results of this study indicate that these inconsistencies may be resulting from (a) GOs being examined in terms of the non-profile perspective as opposed to the profile perspective, (b) assumptions being made about the stability and generality of GOs, and (c) the relationships between GOs and other variables being moderated by task type. Consequently, a number of points need to be taken into consideration when carrying out GO research in the future.

Firstly, it may be useful to use GO profiles as opposed to analysing GOs from the nonprofile perspective. Since people can and do adopt multiple GOs this might provide a more holistic picture of the GOs that people adopt. It is possible that GO profiles may not be as sensitive to fluctuations as the individual GO scales are. However, if these differences are not large enough to make a difference to the overall profile that an individual adopts, are they worth pursuing? It may be the case that by assessing GOs from the *non-profile* perspective researchers may end up focusing on details and lose sight of the bigger picture. For example, it is possible (as mentioned in Chapter 10) that when GOs are analysed from the non-profile perspective, a person's masteryapproach GO decreases significantly over time. If this person adopts the 'High GOs' profile, this decrease might be trivial because they may still have a high masteryapproach GO compared to individuals adopting the 'High Performance-Avoidance' GO profile. Consequently, using the profile perspective may provide a better overall understanding compared to the non-profile perspective.

Secondly, since the GO profiles obtained in this study were similar to GO profiles obtained in previous studies, this might be an indication that it is possible to obtain a comprehensive set of GO profiles. This may open new doors with respect to theory and research on GOs. For example, having a common set of GO profiles might make it easier to compare results across studies and allow for more generalisation. It might also reduce variation across studies if a common set of GO profiles are used in research.

A third implication of the findings is that using General measures might not be an accurate way of measuring GOs. General GOs were found to be significantly different from the Hypothetical Task, as well as Verbal and Numerical Test GOs. Consequently, using General measures of GOs and assuming that participants adopt these GOs irrespective of the task or situation may well provide inaccurate results. These results along with those obtained when assessing the stability of General GOs (from the non-profile perspective) challenge the assumption that GOs are general stable traits. As a result it seems as though using more specific measures (task- or domain-specific) of GOs in future research may be more appropriate. Moreover, investigating the interactions between General GOs (trait) and state GOs might not be suitable. However, since task-specific GOs were found to be stable over time it may be possible to investigate the interactions between task-specific GOs and induced GOs.

Finally, an unexpected and yet extremely important finding from this study is that it seems as though the relationships between GOs and variables such as self-efficacy, mental effort and performance are moderated by task type (possibly task difficulty and/or task value). As described in Chapter 10 (Sections 10.2. and 10.4.) a number of studies (e.g. Mangos & Steele-Johnson, 2001; Steele-Johnson et al., 2000; Yeo et al., 2009) found that task difficulty/complexity does in fact influence the relationships between GOs and other variables (such as self-efficacy and performance). This finding may have a major influence on future GO research since it may be the reason underlying inconsistencies in previous research results (with respect to the relationships between GOs and other variables). If this is the case, it could be possible to determine what characteristics of tasks influence these relationships. This, in turn, may allow researchers to modify tasks and develop strategies in order to obtain the most beneficial effects from the relationships between GOs and other variables. For example, if task value is found to increase the strength of the relationships between GOs and performance, then creating a better understanding of the importance of a task and promoting the desirable GOs may increase self-efficacy and performance on the task.

11.1.2. Practical Implications

The findings of this study have four main practical implications. Firstly, since General GOs were found to be different from task-specific GOs it would probably not be wise to use GOs in selection because they seem to be quite changeable. If GOs were to be measured during selection the results may not accurately reflect the GOs that recruits would adopt on different tasks or in other situations. An additional issue with the use of GOs in selection is that prospective employees may not answer GO measures entirely truthfully if they perceived certain items (the mastery- and performance-avoidance items) as being less desirable to the employer than others.

Secondly, since common GO profiles were obtained in three studies (the present study, that by Pastor et al., 2007; and that by Fortunato and Goldblatt, 2006) it might be possible to develop a comprehensive set of GO profiles. This may be more useful to

organisations than the non-profile perspective since it might provide a more holistic understanding of the GOs being adopted by employees.

Thirdly, the GO profile results indicate that the 'High GOs' and 'High Approach, Low Avoidance' GO profiles may be more beneficial profiles to adopt than the 'High Performance-Avoidance', 'High Mastery, Low Performance' and 'High Mastery, Moderate Performance' GO profiles. Since the 'more beneficial' GO profiles include having high mastery-approach and performance-approach GOs it might be advantageous for organisations to promote these two types of GOs (or the 'High GOs' and 'High Approach, Low Avoidance' GO profiles). This could have a positive effect on the self-efficacy, motivation and performance of employees. Promoting both masteryapproach and performance-approach GOs within an organisation might prove difficult. The performance-approach GO focuses on doing better than others (thus introducing competition) whilst the mastery-approach GO focuses on learning (which is not always easy in a competitive environment). For example, in a sales company a salesperson may choose to use the one sales strategy that they are good at in order to increase their sales performance and not 'waste time' on learning new sales strategies because of the stiff competition in the organisation. Learning a new sales strategy may reduce short-term performance until the salesperson masters it. However, the new technique may enhance performance in the long-term when the salesperson has a larger repertoire of selling techniques which may be beneficial in different circumstances. Therefore, it is probably best for organisations to promote a balance of Mastery-Approach and Performance-Approach GOs for example, by praising and rewarding employees who perform well as well as fostering a learning environment by creating training opportunities and not punishing mistakes. However, it is possible that one cannot simply 'promote' certain GOs in organisations if the organisational culture is inconsistent with these GOs. As mentioned in Chapter 1 there was a strong emphasis on performance-approach GOs in the call centre I used to work in. This was part of the call-centre culture. Employees knew that it was all about competition and that whoever got the most sales and the best call times got bonuses and got promoted. There was no time for learning and mistakes were punished harshly. It would probably not be possible to 'promote' a mastery-approach GO in such an organisation unless

there is a change in the organisational culture. The prospect of promoting GOs in organisations and how this links in with organisational culture should definitely be looked into once more knowledge regarding the nature of GOs and the ways in which they relate to other variables is established.

Finally, the possibility that task characteristics seem to moderate the relationships between GOs (and GO profiles) and other variables (e.g. performance) should also be taken into account by organisations. They may need to examine job characteristics in order to make the most out of the relationships between GOs and variables such as mental effort and performance. For example, if selling a particular product is particularly difficult, adopting high mastery-approach and performance-approach GOs might not be enough to ensure high sales. Employees may have high masteryapproach and performance-approach GOs but not the adequate skills to do well at selling this product. However, these employees may do better than others if the company were to provide a training course on how to sell this product and emphasises the importance of selling it. These employees would also probably have higher performance than they would have without attending the training course.

11.2. Limitations of the Current Study

The main limitation of this study was the small experimental sample size. This was found to be a limitation for three main reasons. Firstly, it resulted in some of the analyses not being carried out since it would not have been statistically feasible to do so (e.g. Research Question 3a). Secondly, the inductions could not be properly tested since the experimental sample size was too small to be able to include only participants having low Mastery-Approach and Performance-Approach GOs in the respective experimental induction groups. Thirdly, although analyses were carried out to investigate whether participants adopting different GO profiles had different levels of performance, self-efficacy and mental effort these should be interpreted with caution due to the small sample sizes in each GO profile cluster. Since a number of participants were employed and retired individuals, promoting a mastery-approach GO on the aptitude tests was not reasonable because they had no reason to improve their learning on the tests. Consequently, these participants were not randomly assigned to experimental groups but rather they were allocated to the control and performance-approach induction groups. This may have influenced the effectiveness of the tests of the inductions.

The aim was to make the study as work-related as possible by using aptitude tests and inviting employed participants to participate. However, the fact that the study was not carried out in an organisational setting means that it was a little removed from GOs in day to day work. Consequently, as will be discussed in Section 11.3., it would be useful to carry out further research on the nature of GOs in an organisational setting.

The model choice and item inclusion in the scales may also be considered a limitation. With respect to model choice, although the CFA results indicated that the 2x2 model provided the best fit to the data, this was still not an excellent fit. Moreover, the 2x2 model did not provide adequate fit when examining GOs in terms of profiles. The different factor structures obtained for General and Hypothetical Task GOs when carrying out the EFAs also attest to the unclarity in the factor structure. If the 2x2 model is in fact the optimal factor structure, which may make the clusters harder to replicate. However, since the 2x2 model did not provide adequate fit on the use the 2x2 model for the analyses of GO profiles.

Item inclusion on scales may be considered a limitation. The reliability analyses as well as the factor analyses results (EFA and CFA) indicated that some items (particularly items 1, 8, 10 and 11) seemed to be problematic in some instances but not in others. The fact that they were only problematic when measuring some types of GOs (e.g. General GOs) but not others (e.g. Verbal Test GOs) made it difficult to decide which items to include on the scales. For the sake of consistency and comparability it was essential to keep the same items on the scales measuring the different GOs. Consequently, a decision was made to include all the items on all the scales since removal of the problem items would have caused significant decreases in reliability on the scales overall. This could be considered a limitation due to the problem items causing some scales to have lower reliability than others.

The modifications made to GO items measuring different types of GOs (e.g. General and Verbal Test GOs) may also be considered a limitation since it introduced some variation across the GO scales. However, as discussed in Chapter 7 (Sections 7.1.5.1. and 7.2.5.1.) these modifications were kept to a bare minimum in order to maintain as much consistency as possible across the GO scales.

As mentioned at various points throughout Chapters 8 and 10 (e.g. Sections 8.6., 8.6.2., 8.7.2., 10.2.2., 10.3.), the large correlation matrices increase the probability of Type I error. In order to take this into account without increasing the probability of Type II error whenever the correlational analysis results were used in answering research questions the issue of Type I error was addressed. In most instances the probability of Type 1 error was quite low (based on the high significance levels obtained and the consistency across results). However, there were instances when there was a possibility that the results were a consequence of Type 1 error. As discussed at various points throughout Chapter 10 (e.g. Section 10.2.1. when examining the relationships between a MAP GO and self-efficacy) this was a limitation due to the fact that in these instances it was not always possible to draw clear-cut conclusions from the results.

An additional limitation was caused by the time intervals between the two questionnaires, the two experiments and the questionnaires and experiments varying across the groups of participants. For example, due to student holidays and exams as well as problems gaining access to the aptitude tests there were not always (at least) four weeks between the two experimental sessions and between the questionnaires and experimental sessions as planned. Consequently, the time intervals for examining the stability of Verbal and Numerical Test GOs over time was not very long in most cases. Additionally, the time interval between the Time Q1 and Time Q2 questionnaires ranged from 4 to 51 weeks. This was mainly due to the geographical location of some experimental participants (e.g. participants from Malta and those away from Loughborough) which made it difficult to carry out the experiments at exact 4 week intervals. It was also a result of taking longer than expected to gain access to the aptitude tests. For example, a number of participants completed the Time Q1 questionnaire in March 2009 and the Time Q2 questionnaire after participating in the experiments, which took place in October through to December 2009 (since they had left for their summer vacations by the time I gained access to the aptitude tests). This variation in time intervals between questionnaires and experiments may have introduced bias.

As mentioned above, the time interval between the two experiments was not always as long as initially planned. This shorter time interval may have influenced the success of the inductions. With a short time interval between experimental sessions, participants may have remembered their answers on the Time E1 questionnaires and answered in a similar way at Time 2. This would have resulted in the inductions being unsuccessful. Consequently, although the inductions were not successful in the current study they definitely need to be investigated further.

The complexity of this study (caused by the various types of GOs measured, the number of time points at which they were measured and the additional variables assessed) made it quite difficult to portray a clear, comprehensive picture regarding the nature of GOs and their relationships with variables of interest. This, compounded by the fact that these were all examined from the profile and non-profile perspectives, led to the attainment of a number of inconsistent results (e.g. the stability of General GOs from the profile and non-profile perspective). This could be considered a limitation in that although some questions were answered, new ones were created and drawing conclusions from the results was not always possible. The complexity of this study and the inconsistent results obtained (e.g. with respect to the factor structure, the stability and specificity of GOs) may be considered a limitation. However, they may also be considered a strength since they show how essential it is to discover more about the (intricate) nature of GOs before moving on to assess complex

relationships with other variables. Bearing in mind the research findings, implications and limitations of the current study, recommendations for future research are presented below.

11.3. Recommendations for Future Research

In this study choosing between the 2x2 and 3-factor model was found to be quite problematic. In fact, the 2x2 model was used in the non-profile perspective analyses whilst the 3-factor model was used in the profile perspective analyses. It is therefore recommended that further research into the dimensionality of GOs is carried out in order to assess whether the 2x2 model of GOs significantly improves on the 3-factor model. The main difference between these two models is the mastery-avoidance GO. This concept was developed and is mostly used in an academic environment (e.g. with college and university students). Its focus is on not forgetting or misunderstanding what one is learning. Consequently, this GO might not be quite as useful in an organisational setting as it is in an educational setting. As a result, it may be useful to assess the utility of the mastery-avoidance GO in organisations. Therefore, research carried out (from the profile and non-profile perspectives) having employed participants might be especially useful in determining the utility of the 2x2 model over the 3-factor model in organisations.

Since some GO profiles obtained in this study were similar to those obtained in other studies it is suggested that research should be carried out in order to determine whether a comprehensive set of GO profiles could be obtained. It is also recommended that further research (with large sample sizes, if possible) using the Achievement Goal Questionnaire-Revised should be carried out since this has been used in a number of research studies and has been shown to have good reliability and validity (as described in Chapter 7 Section 7.1.5.1.). It is essential to use the same measure and scales in future research so that any differences in GO profiles found would not be a result of the measures and/or scales being different. Furthermore, it would be much easier to compare GO scores or means across studies in order to determine whether the GOs obtained are low, moderate or high compared with those

in other studies. If these research studies find similar profiles then it should be possible to create a comprehensive set of GO profiles. This will enable researchers to compare their results with those of other studies and integrate findings on GOs more easily. Moreover, further research using the profile perspective is required in order to find out more about the relationships between GO profiles and variables such as selfefficacy, mental effort and performance. Since these relationships were only assessed on Verbal and Numerical Tests in this study, it would be useful to see how GO profiles are related to these variables on other tasks too. This is especially important since the results of this study point towards the relationships between GOs and other variables being moderated by task type. This finding indicates that research investigating what task characteristics influence the relationships between GOs and other variables (and how) may be extremely beneficial. Once such research is carried out it might become feasible to investigate the direction of causality between GOs and variables such as self-efficacy, performance and mental effort.

With respect to stability of GOs and GO profiles over time, further research is necessary to establish whether General and Task-Specific GOs (and GO profiles) are in fact stable over time. Ideally, longer time intervals between measures should be used to assess long-term stability. It is also considered useful to examine why the profile and non-profile perspective results were not entirely consistent and assess which perspective should be used in future GO research and practice. Relating to the issue of generality versus task-specificity of GOs, it is recommended that additional research is carried out in order to assess the task-specificity of GOs. This study provided some evidence of domain-specificity, thus pointing towards more specific measures of GOs being more accurate than General measures. However, further research (especially with respect to the specificity of GO **profiles**) is definitely required before any such conclusions may be reliably drawn.

The correlations for the Verbal and Numerical Tests indicated that mastery-approach GOs were negatively related to practice. Moreover, the mastery-avoidance GO on the Numerical Test was also found to be significantly negatively related to practice. These results are quite difficult to interpret since, if participants are keen to learn more
(mastery-approach GO) and/or not forget or misunderstand what they have learnt (mastery-avoidance GO) one would expect them to practise more. Since it is unclear why these negative correlations were obtained it is suggested that further research investigating the relationships between GOs and practice should be carried out. Moreover, since the inductions were not successful in this study further research regarding the interaction effects of trait and state GOs is recommended. The finding that the performance-approach GO induction influenced the performance-avoidance GO of participants on the Verbal Test indicates the possibility that GO inductions may well be successful (especially with a larger sample size). It might prove useful to use task-specific measures as the 'trait' measures since GOs were found to be domainspecific. Moreover, it would be considered ideal if interaction effects are measured on different tasks.

11.4. Concluding Remarks

The findings of this study provided some support for previous research findings and created new directions for future research. There is still much GO research necessary before this concept can start to be used reliably in organisations. However, it is strongly believed that the concept of GOs has great potential for use organisations and will ultimately benefit both employers and employees since it has been found to consistently relate to variables such as self-efficacy, mental effort and performance.

I started this PhD strongly believing that I could radically improve knowledge on GOs and that my experiments would enable me to make straightforward and clear cut conclusions regarding the stability and dimensionality of GOs. Little did I know what lay ahead of me! Although I did not find any definite answers regarding GOs I have obtained some interesting results. However, more importantly, I have gained invaluable knowledge about the research process which will no doubt provide me with an excellent platform from which to embark on my research career.

References

ABLARD, K.E. and LIPSCHULTZ, R.E., 1998. Self-Regulated Learning in High-Achieving Students: Relations to Advanced Reasoning, Achievement Goals, and Gender. *Journal of Educational Psychology*, **90**(1), pp. 94-101.

ABRAHAMSEN, F.E., ROBERTS, G.C., PENSGAARD, A.M. and RONGLAN, L.T., 2008. Perceived ability and social support as mediators of achievement motivation and performance anxiety. *Scandinavian journal of medicine & science in sports*, **18**(6), pp. 810-821.

AGBUGA, B. and XIANG, P., 2008. Achievement Goals and Their Relations to Self-Reported Persistence/Effort in Secondary Physical Education: A Trichotomous Achievement Goal Framework. *Journal of Teaching in Physical Education*, **27**(2), pp. 179-191.

ALKHARUSI, H., 2008. Effects of Classroom Assessment Practices on Students' Achievement Goals. *Educational Assessment*, **13**, pp. 243-266.

AMES, C. and ARCHER, J., 1988. Achievement Goals In The Classroom - Students Learning Strategies And Motivation Processes. *Journal of Educational Psychology*, **80**(3), pp. 260-267.

AMES, C., 1992. Classrooms: Goals, Structures, and Student Motivation. *Journal of Applied Psychology*, **84**(3), pp. 261-271.

ANDERMAN, E.M. and MIDGLEY, C., 1997. Changes in Achievement Goal Orientations, Perceived Academic Competence, and Grades across the Transition to Middle-Level Schools. *Contemporary educational psychology*, **22**(3), pp. 269-298.

ANDERSON, D.M. and DIXON, A.W., 2009. Winning Isn't Everything: Goal Orientation and Gender Differences in University Leisure-Skills Classes. *Recreational Sports Journal*, **33**(1), pp. 54-64.

ARNOLD, J., COOPER, C.L. and ROBERTSON, I.T., 1998. *Work Psychology: Understanding Human Behaviour in the Workplace*. Third edn. Harlow, England: Pearson Education Limited.

BANDALOS, D.L., FINNEY, S.J. and GESKE, J.A., 2003. A Model of Statistics Performance Based on Achievement Goal Theory. *Journal of Educational Psychology*, **95**(3), pp. 604-616.

BANDURA, A., 1988. Organisational Applications of Social Cognitive Theory. *Australian Journal of Management*, **13**(2), pp. 275-302.

BANDURA, A., 1982. Self-Efficacy Mechanism in Human Agency. *American Psychologist*, **37**(2), pp. 122-147.

BANDURA, M. and DWECK, C.S., 1981. *Children's theories of intelligence as predictors of achievement goals.* Unpublished manuscript edn. Harvard University: .

BARANIK, L.E., BARRON, K.E. and FINNEY, S.J., 2010. Examining Specific Versus General Measures of Achievement Goals. *Human Performance*, **23**(2), pp. 155-172.

BARANIK, L.E., STANLEY, L.J., BYNUM, B.H. and LANCE, C.E., 2010. Examining the Construct Validity of Mastery-Avoidance Achievement Goals: A Meta-Analysis. *Human Performance*, **23**(3), pp. 265-282.

BARKER, K.L., MCINERNEY, D.M. and DOWNSON, M., 2002. Performance-Approach, Performance-Avoidance and Depth of Information Processing: A fresh look at relations between students' academic motivation and cognition. *Educational Psychology*, **22**(5), pp. 571-589.

BARRICK, M.R., PIOTROWSKI, M. and STEWART, G., L., 2002. Personality and Job Performance: Test of the Mediating Effects of Motivation Among Sales Representatives. *Journal of Applied Psychology*, **87**(1), pp. 43.

BARRON, K.E. and HARACKIEWICZ, J.M., 2001. Achievement Goals and Optimal Motivation: Testing Multiple Goal Models. *Journal of Personality and Social Psychology*, **80**(5), pp. 706-722.

BARRON, K.E. and HARACKIEWICZ, J.M., 2000. Achievement Goals and Optimal Motivation: A Multiple Goals Approach. In: C. SANSONE and J.M. HARACKIEWICZ, eds, *Intrinsic and Extrinsic Motivation: The Search for Optimal Motivation and Performance.* San Diego: California: Academic Press, pp. 229-254.

BARTELS, J.M. and MAGUN-JACKSON, S., 2009. Approach–avoidance motivation and metacognitive self-regulation: The role of need for achievement and fear of failure. *Learning and Individual Differences*, **19**(4), pp. 459-463.

BECKMANN, N., BECKMANN, J.F. and ELLIOTT, J.G., 2009. Self-confidence and performance goal orientation interactively predict performance in a reasoning test with accuracy feedback. *Learning and Individual Differences*, **19**(2), pp. 277-282.

BELL, B.S. and KOZLOWSKI, S.W.J., 2002. Goal Orientation and Ability: Interactive Effects on Self-Efficacy, Performance, and Knowledge. *Journal of Applied Psychology*, **87**(3), pp. 497-505.

BEREBY-MEYER, Y., MORAN, S. and SATTLER, L., 2010. The Effects of Achievement Motivational Goals and of Debriefing on the Transfer of Skills in Integrative Negotiations. *Negotiation and Conflict Management Research*, **3**(1), pp. 64-86.

BERNARDO, A.B.I., 2008. Individual and social dimensions of Filipino students' achievement goals. *International Journal of Psychology*, **43**(5), pp. 886-891.

BETTENCOURT, L.A., 2004. Change-oriented organizational citizenship behaviors: the direct and moderating influence of goal orientation. *Journal of Retailing*, **80**(3), pp. 165-180.

BIEMOND, J.M. and VAN YPEREN, N.W., 2001. Generalized achievement goal orientation: Development and validation of a questionnaire, *WAOP Conference* 2001.

BIPP, T., STEINMAYR, R. and SPINATH, B., 2008. Personality and achievement motivation: Relationship among Big Five domain and facet scales, achievement goals, and intelligence. *Personality and Individual Differences*, **44**(7), pp. 1454-1464.

BJORNEBEKK, G., 2009. Mediators and Moderators of Approach Performance and Avoidance Performance Relationships in Children. In: M. WOSNITZA, S.A. KARABENICK, A. EFKLIDES and P. NENNINGER, eds, *Contemporary Motivation Research: From Global to Local Perspectives.* 1 edn. Cambridge: Massachusetts: Hogrefe & Huber Publishing., pp. 185-205.

BONG, M., 2009. Age-Related Differences in Achievement Goal Differentiation. *Journal of Educational Psychology*, **101**(4), pp. 879-896.

BONG, M., 2001. Between- and Within-Domain Relations of Academic Motivation Among Middle and High School Students: Self-Efficacy, Task-Value, and Achievement Goals. *Journal of Educational Psychology*, **93**(1), pp. 23-34.

BONG, M. and HOCEVAR, D., 2002. Measuring Self-Efficacy:Multitrait-Multimethod Comparison of Scaling Procedures. *Applied Measurement in Education*, **15**(2), pp. 143-171.

BORKOWSKI, T., 2007. *Managerial and Professional Series - User Guide and Technical Manual.* 103. eras Ltd.

BOUFFARD-BOUCHARD, T., 1990. Influence of self-efficacy on performance in a cognitive task. *The Journal of Social Psychology*, **130**(3), pp. 353-363.

BRELAND, B.T. and DONOVAN, J.J., 2005. The Role of State Goal Orientation in the Goal Establishment Process. *Human Performance*, **18**(1), pp. 23-53.

BRETT, J.F. and ATWATER, L.E., 2001. 360⁰ Feedback: Accuracy, Reactions, and Perceptions of Usefulness. *Journal of Applied Psychology*, **86**(5), pp. 930-942.

BUTTON, S.B., MATHIEU, J.E. and ZAJAC, D.M., 1996. Goal Orientation in Organizational Research: A Conceptual and Empirical Foundation. *Organizational Behavior and Human Decision Processes*, **67**(1), pp. 26-48.

CANO, F. and BERBEN, A.B.G., 2009. University students' achievement goals and approaches to learning in mathematics. *British Journal of Educational Psychology*, **79**, pp. 131-153.

CAVRINI, G., GALIMBERTI, G. and SOFFRITTI, G., 2009. Evaluating patient satisfaction through latent class factor analysis. *Health & place*, **15**(1), pp. 210-218.

CHEN, G., GULLY, S.M., WHITEMAN, J. and KILCULLEN, R.N., 2000. Examination of Relationships Among Trait-Like Individual Differences, State-Like Individual Differences, and Learning Performance. *Journal of Applied Psychology*, **85**(6), pp. 835-847.

CHEN, G. and MATHIEU, J.E., 2008. Goal orientation dispositions and performance trajectories: The roles of supplementary and complementary situational inducements. *Organizational behavior and human decision processes*, **106**(1), pp. 21-38.

CHIAN, L.K.Z. and WANG, C.K.J., 2008. Motivational Profiles of Junior College Athletes: A Cluster Analysis. *Journal of Applied Sport Psychology*, **20**, pp. 137-156.

CHURCH, M.A., ELLIOT, A.J. and GABLE, S.L., 2001. Perceptions of Classroom environment, achievement goals, and achievement outcomes. *Journal of Educational Psychology*, **93**(1), pp. 43-54.

COLQUITT, J.A. and SIMMERING, M.J., 1998. Conscientiousness, Goal Orientation, and Motivation to Learn During the Learning Process: A Longitudinal Study. *Journal of Applied Psychology*, **83**(4), pp. 654-665.

CONROY, D.E., ELLIOT, A.J. and HOFER, S.M., 2003. A 2 x 2 Achievement Goals Questionnaire for Sport: evidence for factorial invariance, temporal stability, and external validity. *Journal of Sport & Exercise Psychology*, **25**(4), pp. 456-476.

CONROY, D.E., CASSIDY, C.M. and ELLIOT, A.J., 2008. Prospective Relations between 2x2 Achievement Goals and the Quality of Sport Training. *Revue Internationale De Psychologie Sociale*, **21**, pp. 139-134.

COSTA JR, P.T. and MCCRAE, R.R., 1988. Personality in Adulthood: A Six-Year Longitudinal Study of Self-Reports and Spouse Ratings on the NEO Personality Inventory. *Journal of Personality and Social Psychology*, **54**(5), pp. 853-863.

CRANDALL, V.J., KATKOVSKY, W. and PRESTON, A., 1960. A conceptual formulation for some research on children's achievement development. *Child Development*, **31**(4), pp. 787-797.

CRIPPEN, K.J., BIESINGER, K.D., MUIS, K.R. and ORGILL, M., 2009. The Role of Goal Orientation and Self-Efficacy in Learning from Web-Based Work Examples. *Journal of Interactive Learning Research*, **20**(4), pp. 385-403.

CUNNINGHAM, G.B. and XIANG, P., 2008. Testing the Mediating Role of Perceived Motivational Climate in the Relationship Between Achievement Goals and Satisfaction: Are These Relationships Invariant Across Sex? *Journal of Teaching in Physical Education*, **27**(2), pp. 192-204. DANIELS, L.M., HAYNES, T.L., STUPNISKY, R.H., PERRY, R.P., NEWALL, N.E. and PEKRUN, R., 2008. Individual differences in achievement goals: A longitudinal study of cognitive, emotional, and achievement outcomes. *Contemporary educational psychology*, **33**(4), pp. 584-608.

DARNON, C., DOMPNIER, B., GILLIERON, O. and BUTERA, F., 2010. The Interplay of Mastery and Performance Goals in Social Comparison: A Multiple-Goal Perspective. *Journal of Educational Psychology*, **102**(1), pp. 212-222.

DAY, E.A., RADOSEVICH, D.J. and CHASTEEN, C.S., 2003. Construct- and criterionrelated validity of four commonly used goal orientation instruments. *Contemporary Educational Psychology*, **28**, pp. 434-464.

DE CROOK, M.B.M. and VAN MERRIENBOER, J.J.G., 2007. Paradoxical effects of information presentation formats and contextual interference on transfer of a complex cognitive skill. *Computers and Human Behavior*, **23**, pp. 1740-1761.

DE CROOK, M.B.M., VAN MERRIENBOER, J.J.G. and PAAS, F.G.W.C., 1998. High versus low contextual interference in simulation based training of troubleshooting skills: Effects on transfer performance and invested mental effort. *Computers and Human Behavior*, **14**, pp. 249-267.

DEKKER, S. and FISCHER, R., 2008. Cultural Differences in Academic Motivation Goals: A Meta-Analysis Across 13 Societies. *The Journal of Educational Research*, **102**(2), pp. 99-110.

DESHON, R.P. and GILLESPIE, J.Z., 2005. A Motivated Action Theory Account of Goal Orientation. *Journal of Applied Psychology*, **90**(6), pp. 1096-1127.

DIAMANTOPOULOS, A. and SIGUAW, J. (2000). *Introducing LISREL. A Guide for the Uninitiated.* Sage: London.

DIEFENDORFF, J.M., 2004. Examination of the Roles of Action-State Orientation and Goal Orientation in the Goal-Setting and Performance Process. *Human Performance*, **17**(4), pp. 375-395.

DIENER, C.I. and DWECK, C.S., 1978. An Analysis of Learned Helplessness: Continuous Changes in Performance, Strategy, and Achievement Cognitions Following Failure. *Journal of Personality and Social Psychology*, **36**(5), pp. 451-462.

DRAGONI, L., 2005. Understanding the Emergence of State Goal Orientation in Organizational Work Groups: The Role of Leadership and Multilevel Climate Perceptions. *Journal of Applied Psychology*, **90**(6), pp. 1084-1095.

DUDA, J. and NICHOLLS, J.G., 1992. Dimensions of Achievement Motivation in Schoolwork and Sport. *Journal of Educational Psychology*, **84**(3), pp. 290-299.

DURIK, A.M., LOVEJOY, C.M. and JOHNSON, S.J., 2009. A longitudinal study of achievement goals for college in general: Predicting cumulative GPA and diversity in course selection. *Contemporary educational psychology*, **34**(2), pp. 113-119.

DWECK, C.S., 1986. Motivational Processes Affecting Learning. *American Psychologist*, **41**(10), pp. 1040-1048.

DWECK, C.S., 1975. The Role of Expectations and Attributions in the Alleviation of Learned Helplessness. *Journal of Personality and Social Psychology*, **31**(4), pp. 674-685.

DWECK, C.S. and & LEGGETT, E.L., 1988. A Social-Cognitive Approach to Motivation and Personality. *Psychological Review*, **95**(2), pp. 256-273.

DWECK, C.S. and ELLIOTT, E.S., 1983. Achievement Motivation. In: P.H. MUSSEN, ed, Handbook of child psychology. 4th edn. New York: Wiley, pp. 643-691.

DWECK, C.S. and REPPUCCI, N.D., 1973. Learned Helplessness And Reinforcement Responsibility In Children. *Journal of Personality and Social Psychology*, **25**(1), pp. 109-116.

DWECK, C.S. and ELLIOTT, E.S., 1981. A model of achievement motivation, a theory of *its origins, and a framework for motivational development*. Unpublished manuscript edn. Harvard University: .

ECCLES, J.S. and WIGFIELD, A., 2002. Motivational Beliefs, Values and Goals. *Annual Review of Psychology*, **53**, pp. 109-132.

ELLIOT, A.J., 2005. A conceptual history of the achievement goal construct. In: A.J. ELLIOT and C.S. DWECK, eds, *Handbook of competence motivation*. New York: Guildford Press, pp. 52-72.

ELLIOT, A.J. and HARACKIEWICZ, J.M., 1996. Approach and avoidance achievement goals and intrinsic motivation: A mediational analysis. *Journal of Personality and Social Psychology*, **70**(3), pp. 461-475.

ELLIOT, A.J. and MCGREGOR, H.A., 2001. A 2x2 achievement goal framework. *Journal of Personality and Social Psychology*, **80**(3), pp. 501-519.

ELLIOT, A.J., MCGREGOR, H.A. and GABLE, S., 1999. Achievement Goals, Study Strategies, and Exam Performance: A Mediational Analysis. *Journal of Educational Psychology*, **91**(3), pp. 549-563.

ELLIOT, A.J. and THRASH, T.M., 2001. Achievement Goals and the Hierarchical Model of Achievement Motivation. *Educational Psychology Review*, **13**(2), pp. 139-156.

ELLIOT, A.J., 1999. Approach and Avoidance Motivation and Achievement Goals. *Educational Psychologist*, **34**(3), pp. 169-189.

ELLIOT, A.J. and CHURCH, M.A., 1997. A Hierarchical Model of Approach and Avoidance Achievement Motivation. *Journal of Personality and Social Psychology*, **72**(1), pp. 218-232.

ELLIOT, A.J. and COVINGTON, M.V., 2001. Approach and Avoidance Motivation. *Educational Psychology Review*, **13**(2), pp. 73-92.

ELLIOT, A.J. and MCGREGOR, H.A., 1999. Test Anxiety and the Hierarchical Model of Approach and Avoidance Achievement Motivation. *Journal of Personality and Social Psychology*, **76**(4), pp. 628-644.

ELLIOT, A.J. and MURAYAMA, K., 2008. On the Measurement of Achievement Goals: Critique, Illustration, and Application. *Journal of Educational Psychology*, **100**(3), pp. 613-628.

ELLIOT, A.J. and SHELDON, K.M., 1997. Avoidance Achievement Motivation: A Personal Goals Analysis. *Journal of Personality and Social Psychology*, **73**(1), pp. 171-185.

ELLIOT, A.J., SHELL, M.M., HENRY, K.B. and MAIER, M.A., 2005. Achievement goals, performance contingencies, and performance attainment: An experimental test. *Journal of Educational Psychology*, **97**(4), pp. 630-640.

ELLIOT, A.J., CHIRKOV, V.I., KIM, Y. and SHELDON, K.M., 2001. A Cross-cultural Analysis of Avoidance (Relative to Approach) Personal Goals. *Psychological Science (Wiley-Blackwell)*, **12**(6), pp. 505.

ELLIOTT, E.S. and DWECK, C.S., 1988. Goals: An Approach to Motivation and Achievement. *Journal of Personality and Social Psychology*, **54**(1), pp. 5-12.

EREZ, A. and JUDGE, T.A., 2001. Relationship of Core Self-Evaluations to Goal Setting, Motivation, and Performance. *Journal of Applied Psychology*, **86**(6), pp. 1270.

FAN, J., MENG, H., BILLINGS, R.S., LITCHFIELD, R.C. and KAPLAN, I., 2008. On the Role of Goal Orientation Traits and Self-Efficacy in the Goal-Setting Process: Distinctions That Make a Difference. *Human Performance*, **21**(4), pp. 354-382.

FARR, J.L., HOFMANN, D.A. and RINGENBACH, K.L., 1993. Goal Orientation And Action Control Theory: Implications For Industrial And Organizational Psychology. In: C.L. COOPER and I.T. ROBERTSON, eds, *International Review of Industrial and Organizational Psychology.* Chichester: John Wiley & Sons, pp. 193-232.

FINNEY, S.J., PIEPER, S.L. and BARRON, K.E., 2004. Examining the Psychometric Properties of the Achievement Goal Questionnaire in a General Academic Context. *Educational and Psychological Measurement*, **64**, pp. 365-382.

FISHER, S.L. and FORD, J.K., 1998. Differential Effects of Learner Effort and Goal Orientation on Two Learning Outcomes. *Personnel Psychology*, **51**(2), pp. 397-420.

FLORES, J., SALGUERO, A. and MÁRQUEZ, S., 2008. Goal orientations and perceptions of the motivational climate in physical education classes among Colombian students. *Teaching and Teacher Education*, **24**(6), pp. 1441-1449.

FORD, K.J., SMITH, E.M., WEISSBEIN, D.A., GULLY, S.M. and SALAS, E., 1998. Relationships of Goal Orientation, Metacognitive Activity, and Practice Strategies With Learning Outcomes and Transfer. *Journal of Applied Psychology*, **83**(2), pp. 218-233.

FORTUNATO, V.J. and GOLDBLATT, A.M., 2006. An Examination of Goal Orientation Profiles Using Cluster Analysis and Their Relationships With Dispositional Characteristics and Motivational Response Patterns. *Journal of Applied Social Psychology*, **36**(9), pp. 2150-2183.

FOULADCHANG, M., MARZOOGHI, R. and SHEMSHIRI, B., 2009. The Effect of Gender and Grade Level Differences on Achievement Goal Orientations of Iranian Undergraduate Students. *Journal of Applied Sciences*, **9**(5), pp. 968-972.

FRYER, J.W. and ELLIOT, A.J., 2007. Stability and Change in Achievement Goals. *Journal of Educational Psychology*, **99**(4), pp. 700-714.

FU, F.Q., RICHARDS, K.A. and JONES, E., 2009. The Motivation Hub: Effects of Goal Setting and Self-Efficacy on Effort and New Product Sales. *Journal of Personal Selling and Sales Management*, **29**(3), pp. 277-292.

FURNHAM, A., FORDE, L. and FERRARI, K., 1999. Personality and work motivation. *Personality and Individual Differences*, **26**, pp. 1035.

GERHARDT, M.W. and LUZADIS, R.A., 2009. The Importance of Perceived Task Difficulty in Goal Orientation-Assigned Goal Alignment. *Journal of Leadership and Organizational Studies*, **16**(2), pp. 167-174.

GIST, M.E., 1987. Self-Efficacy: Implications for Organizational Behavior and Human Resource Management. *The Academy of Management Review*, **12**(3), pp. 472-485.

GIST, M.E., SCHWOERER, C. and ROSEN, B., 1989. Effects of Alternative Training Methods on Self-Efficacy and Performance in Computer Software Training. *Journal of Applied Psychology*, **74**(6), pp. 884-891.

GONG, Y., 2003. Goal orientations and cross-cultural adjustment: an exploratory study. *International Journal of Intercultural Relations*, **27**(3), pp. 297-305.

GRANT, H. and DWECK, C.S., 2003. Clarifying Achievement Goals and Their Impact. *Journal of Personality and Social Psychology*, **85**(3), pp. 541-553.

GREEN, J., MARTIN, A.J. and MARSH, H.W., 2005. Academic Motivation and Engagement: A Domain Specific Approach, P.L. JEFFERY, ed. In: *Australian Association for Research in Education Conference 27*, 27 Nov-1 Dec 2005 2005, Australian Association for Research in Education. GREEN, J., MARTIN, A.J. and MARSH, H.W., 2007. Motivation and engagement in English, mathematics and science high school subjects: Towards an understanding of multidimensional domain specificity. *Learning and Individual Differences*, **17**(3), pp. 269-279.

GREENE, B.A. and MILLER, R.B., 1996. Influences on Achievement: Goals, Perceived Ability, and Cognitive Engagement. *Contemporary educational psychology*, **21**(2), pp. 181-192.

HAGENAARS, J.A. and MCCUTCHEON, A.L., 1997. Comparative Social Research with Multi-Sample Latent Class Models. In: R. LANGEHEINE and J. ROST, eds, *Applications of Latent Trait and Latent Class Models in the Social Sciences*. New York: Waxmann, pp. 266-277.

HANRAHAN, S.J. and CERIN, E., 2009. Gender, level of participation, and type of sport: Differences in achievement goal orientation and attributional style. *Journal of Science and Medicine in Sport*, **12**(4), pp. 508-512.

HARACKIEWICZ, J.M. and ELLIOT, A.J., 1993. Achievement Goals and Intrinsic Motivation. *Journal of Personality and Social Psychology*, **65**(5), pp. 904-915.

HARACKIEWICZ, J.M., BARRON, K.E., CARTER, S.M., LEHTO, A.T. and ELLIOT, A.J., 1997. Predictors and Consequences of Achievement Goals in the College Classroom: Maintaining Interest and Making the Grade. *Journal of Personality and Social Psychology*, **73**(6), pp. 1284-1295.

HARACKIEWICZ, J.M., BARRON, K.E., TAUER, J.M., CARTER, S.M. and ELLIOT, A.J., 2000. Short-Term and Long-Term Consequences of Achievement Goals: Predicting Interest and Performance Over Time. *Journal of Educational Psychology*, **92**(2), pp. 316-330.

HARACKIEWICZ, J.M., BARRON, K.E., TAUER, J.M. and ELLIOT, A.J., 2002. Predicting Success in College: A Longitudinal Study of Achievement Goals and Ability Measures as Predictors of Interest and Performance From Freshman Year Through Graduation. *Journal of Educational Psychology*, **94**(3), pp. 562-575.

HARRIS, A., YUILL, N. and LUCKIN, R., 2008. The influence of context-specific and dispositional achievement goals on children's paired collaborative interaction. *British Journal of Educational Psychology*, **78**, pp. 355-374.

HARRIS, E.G., MOWEN, J.C. and BROWN, T.J., 2005. Re-examining salesperson goal orientations: Personality influencers, customer orientation, and work satisfaction. *Journal of the Academy of Marketing Science*, **33**(1), pp. 19-35.

HARWOOD, C.G. and SWAIN, A.B., 1998. Antecedents of pre-competition achievement goals in elite junior tennis players. *Journal of sports sciences*, **16**(4), pp. 357-371.

HECKHAUSEN, H., 1981. The development of achievement motivation. In: W.W. HARTUP, ed, *Review of Child Development Research*. Chicago: University of Chicago Press, pp. 600-668.

HERTENSTEIN, E.J., 2001. Goal Orientation and Practice Condition as Predictors of Training Results. *Human Resource Development Quarterly*, **12**(4), pp. 403-419.

HORVATH, M., HERLEMAN, H.A. and MCKIE, R.L., 2006. Goal Orientation, Task Difficulty, and Task Interest: A Multilevel Analysis. *Motivation and Emotion*, **30**, pp. 171-178.

HU, L. and BENTLER, P. (1999). Cutoff criteria for fit indexes in covariance structure analysis: Conventional criteria versus new alternatives. *Structural Equation Modeling: A Multidisciplinary Journal*, **6**(1), pp.1-55.

HULLEMAN, C.S., BODMANN, S.M., HARACKIEWICZ, J.M. and SCHRAGER, S.M., 2010. A Meta-Analytic Review of Achievement Goal Measures: Different Labels for the Same Constructs or Different Constructs With Similar Labels? *Psychological Bulletin*, **136**(3), pp. 422-449.

JAGACINSKI, C.M. and NICHOLLS, J.G., 1984. Conceptions of Ability and Related Affects in Task Involvement and Ego Involvement. *Journal of Educational Psychology*, **76**(5), pp. 909-919.

JAGACINSKI, C.M., MADDEN, J.L. and REIDER, M.H., 2001. The impact of situational and dispositional achievement goals on performance. / L ' impact des buts d ' accomplissement dispositionnels et situationnels sur la performance. *Human Performance*, **14**(4), pp. 321-337.

JAGACINSKI, C.M. and STRICKLAND, O.J., 2000. Task and ego orientation: The role of goal orientations in anticipated affective reactions to achievement outcomes. *Learning and Individual Differences*, **12**(2), pp. 189-208.

JANSSEN, O. and PRINS, J., 2007. Goal orientations and the seeking of different types of feedback information. *Journal of Occupational and Organizational Psychology*, **82**, pp. 235-249.

JÖRESKOG, K.G. (1969). A General Approach to Confirmatory Maximum Likelihood Factor Analysis. *Psychometrika*, **34**(2), pp. 183-202.

JUDGE, T.A., LOCKE, E.A. and DURHAM, C.C., 1997. The dispositional causes of job satisfaction: A core evaluation approach. *Research in Organizational Behaviour*, **19**, pp. 151.

JUDGE, T.A., LOCKE, E.A., DURHAM, C.C. and KLUGER, A.N., 1998. Dispositional Effects on Job and Life Satisfaction: The Role of Core Evaluations. *Journal of Applied Psychology*, **83**(1), pp. 17.

JUDGE, T., A and ILIES, R., 2002. Relationship of Personality to Performance Motivation: A Meta-Analytic Review. *Journal of Applied Psychology*, **87**(4), pp. 797.

KAPLAN, A. and MIDGLEY, C., 1997. The Effect of Achievement Goals: Does Level of Perceived Academic Competence Make a Difference? *Contemporary educational psychology*, **22**(4), pp. 415-435.

KARABENICK, S.A., 2004. Perceived Achievement Goal Sturcture and College Student Help Seeking. *Journal of Educational Psychology*, **96**(3), pp. 569-581.

KOHLI, A.K., SHERVANI, T.A. and CHALLAGALLA, G.N., 1998. Learning and Performance Orientation of Salespeople: The Role of Supervisors. *Journal of Marketing Research*, **35**(2), pp. 263-274.

KOZLOWSKI, S.W.J., GULLY, S.M., BROWN, K.G., SALAS, E., SMITH, E.M. and NASON, E.R., 2001. Effects of Training Goals and Goal Orientation Traits on Multidimensional Training Outcomes and Performance Adaptability. *Organizational Behaviour and Human Decision Processes*, **85**(1), pp. 1-31.

KRISHNAN, B., C., NETEMEYER, R.G. and BOLES, J.S., 2002. Self-Efficacy, Competitiveness, and Effort as Antecedents of Salesperson Performance. *Journal of Personal Selling and Sales Management*, **22**(4), pp. 285-295.

LANGEHEINE, R., PANNEKOEK, J. and VAN DE POL, F., 1996. Bootstrapping Goodnessof-Fit Measures in Categorical Data Analysis. *Sociological Methods and Research*, **24**(4), pp. 492-516.

LANZA, S.T. and COLLINS, L.M., 2010. *Latent class and Latent transition analysis with applications in the social, behavioral, and health sciences.* Hoboken, New Jersey: John Wiley & Sons Inc.

LAU, S., LIEM, A.D. and NIE, Y., 2008. Task- and self-related pathways to deep learning: The mediating role of achievement goals, classroom attentiveness, and group participation. *British Journal of Educational Psychology*, **78**, pp. 639-662.

LEE, C., TINSLEY, C. and BOBKO, P., 2003. Cross-cultural Variance in Goal Orientations and their effects. *Applied Psychology: An International Review*, **52**(2), pp. 272-297.

LEE, C., HUI, C., TINSLEY, C.H. and NIU, X., 2006. Goal Orientations and performance: role of temporal norms. *Journal of International Business Studies*, **37**, pp. 484-498.

LIEM, A.D., LAU, S. and NIE, Y., 2008. The role of self-efficacy, task value, and achievement goals in predicting learning strategies, task disengagement, peer relationship, and achievement outcome. *Contemporary educational psychology*, **33**(4), pp. 486-512.

LIN, C., HUNG, P., LIN, S., LIN, B. and LIN, F., 2009. The Power of Learning Goal Orientation In Predicting Student Mathematics Achievement. *International Journal of Science and Mathematics Education*, **7**, pp. 551-573.

LINNENBRINK, E.A., 2005. The Dilemma of Performance-Approach Goals: The Use of Multiple Goal Contexts to Promote Students' Motivation and Learning. *Journal of Educational Psychology*, **97**(2), pp. 197-213.

LINNENBRINK-GARCIA, L., TYSON, D.F. and PATALL, E.A., 2008. When are Achievement Goal Orientations Beneficial for Academic Achievement? A Closer Look at Main Effects and Moderating Factors. *Revue Internationale De Psychologie Sociale*, **21**(1), pp. 19-70.

LIU, W.C., WANG, C.K.J., TAN, O.S., EE, J. and KOH, C., 2009. Understanding students' motivation in project work: A 2x2 achievement goal approach. *British Journal of Educational Psychology*, **79**, pp. 87-106.

LORAAS, T. and DIAZ, M.C., 2009. Learning new uses of technology: Situational goal orientation matters. *International Journal of Human-Computer Studies*, **67**, pp. 50-61.

LYKE, J.A. and KELAHER YOUNG, A.J., 2006. Cognition in Context: Students' Perceptions of Classroom Goal Structures and Reported Cognitive Strategy Use in the College Classroom. *Research in Higher Education*, **47**(4), pp. 477-490.

MAEHR, M.L. and BRASKAMP, L.A., 1986. *The motivation factor: A theory of personal investment*. Lexington, Massachusetts: Lexington Books.

MAGIDSON, J. and VERMUNT, J.K., 2004. Latent Class Models. In: D. KAPLAN, ed, *The Sage Handbook of Quantitative Methodology for the Social Sciences*. Thousand Oakes, California: Sage Publications, pp. 175-198.

MAGSON, N.R., CRAVEN, R.G., NELSON, G.F. and SEESHING YEUNG, A., 2008. Domain-Specific School Motivation, P.L. JEFFERY, ed. In: *Changing Climates: Education for Sustainable Futures.*, 30th November - 4th December 2008 2008, The Australian Association for Research in Education.

MANGOS, P.M. and STEELE-JOHNSON, D., 2001. The Role of Subjective Task Complexity in Goal Orientation, Self-Efficacy, and Performance Relations. *Human Performance*, **14**(2), pp. 169-186.

MCCLELLAND, D., ATKINSON, J., CLARK, R. and LOWELL, E., 1953. *The Achievement Motive*. New York: Irvington Publishers.

MCGREGOR, H.A. and ELLIOT, A.J., 2002. Achievement Goals as Predictors of Achievement-Relevant Processes Prior to Task Engagement. *Journal of Educational Psychology*, **94**(2), pp. 381-395.

MEECE, J.L. and HOLT, K., 1993. A Pattern Analysis of Students' Achievement Goals. *Journal of Educational Psychology*, **85**(4), pp. 582-590.

MEHTA, A., FEILD, H., ARMENAKIS, A. and MEHTA, N., 2009. Team Goal Orientation and Team Performance: The Mediating Role of Team Planning. *Journal of Management*, **35**(4), pp. 1026-1046.

MIDGLEY, C., KAPLAN, A. and MIDDLETON, M., 2001. Preformance-Approach Goals: Good For What, For Whom, Under What Circumstances, and At What Cost? *Journal of Educational Psychology*, **93**(1), pp. 77-86.

MIDGLEY, C., KAPLAN, A., MIDDLETON, M., MAEHR, M.L., URDAN, T., ANDERMAN, L.H., ANDERMAN, E. and ROESER, R., 1998. The Development and Validation of Scales Assessing Students' Achievement Goal Orientations, *Contemporary educational psychology*, **23**(2), pp. 113-131.

MILLER, R.B., BEHRENS, J.T., GREENE, B.A. and NEWMAN, D., 1993. Goals and Perceived Ability: Impact on Student Valuing, Self-Regulation, and Persistence. *Contemporary educational psychology*, **18**(1), pp. 2-14.

MORRIS, R.L. and KAVUSSANU, M., 2008. Antecedents of approach-avoidance goals in sport. *Journal of sports sciences*, **26**(5), pp. 465-476.

MUIS, K.R. and EDWARDS, O., 2009. Examining the stability of achievement goal orientation. *Contemporary educational psychology*, **34**(4), pp. 265-277.

NARAYAN, A. and STEELE-JOHNSON, D., 2007. Relationships between prior experience of training, gender, goal orientation and training attitudes. *International Journal of Training & Development*, **11**(3), pp. 166-180.

NG, C.C., 2009. Profiling Learners' achievement goals when completing academic essays. *Educational Psychology*, **29**(3), pp. 279-295.

NICHOLLS, J.G., 1984. Achievement Motivation: Conceptions of Ability, Subjective Experience, Task Choice, and Performance. *Psychological Review*, **91**(3), pp. 328-346.

NICHOLLS, J.G., 1978. The Development of the Concepts of Effort and Ability, Perception of Academic Attainment, and the Understanding That Difficult Tasks Require More Ability. *Child Development*, **49**(3), pp. 800-814.

NICHOLLS, J.G., CHEUNG, P.C., LAUER, J. and PATASHNIK, M., 1989. Individual Differences In Academic Motivation: Perceived Ability, Goals, Beliefs, and Values. *Learning and Individual Differences*, **1**(1), pp. 63-84.

NICHOLLS, J.G. and DWECK, C.S., 1979. *A definition of achievement motivation*. Unpublished manuscript edn. University of Illinois: .

NICHOLLS, J.G., PATASHNICK, M. and NOLEN, S.B., 1985. Adolescents' theories of education. *Journal of Educational Psychology*, **77**(6), pp. 683-692.

NICHOLLS, J.G., COBB, P., WOOD, T., YACKEL, E. and PATASHNICK, M., 1990. Assessing Students' Theories of Success in Mathematics: Individual and Classroom Differences. *Journal for Research in Mathematics Education*, **21**(2), pp. 109-122.

NOLEN, S.B., 1988. Reasons for Studying: Motivational Orientations and Study Strategies. *Cognition and Instruction*, **5**(4), pp. 269-287.

Oxford English Dictionary1989-last update [Homepage of Oxford University Press], [Online]. Available:

http://dictionary.oed.com/cgi/entry/50247445?query_type=word&queryword=task&f irst=1&max_to_show=10&sort_type=alpha&result_place=1&search_id=w5f5-51M2MT-17501&hilite=50247445 [04/07, 2010].

PAAS, F.G.W.C., 1992. Training Strategies for Attaining Transfer of Problem-Solving Skill in Statistics: A Cognitive-Load Approach. *Journal of Educational Psychology*, **84**(4), pp. 429-434.

PAAS, F.G.W.C., VAN MERRIENBOER, J.J.G. and ADAM, J.J., 1994. Measurement of Cognitive Load In Instructional Research. *Perceptual and Motor Skills*, **79**, pp. 419-430.

PAJARES, F., 1997. Current Directions in Self-efficacy Research. In: M. MAEHR and P.R. PINTRICH, eds, *Advances in Motivation and Achievement*. Greenwich, CT: JAI Press, pp. 1-49.

PASTOR, D.A., BARRON, K.E., MILLER, B.J. and DAVIS, S.L., 2007. A latent profile analysis of college students' achievement goal orientation. *Contemporary educational psychology*, **32**(1), pp. 8-47.

PAYNE, S.C., YOUNGCOURT, S.S. and BEAUBIEN, J.M., 2007. A meta-analytic examination of the goal orientation nomological net. *Journal of Applied Psychology*, **92**(1), pp. 128-150.

PERNEGER, T.V. (1998). What's wrong with Bonferroni adjustments. *British Medical Journal*, **316**(7139), pp1236-1238.

PERVIN, L.A., 1989. Persons, Situations, Interactions: The History of a Controversy and a Discussion of Theoretical Models. *Academy of Management Review*, **14**(3), pp. 350-360.

PHAN, H.P., 2009. Amalgamation of future time orientation, epistemological beliefs, achievement goals and study strategies: Empirical evidence established. *British Journal of Educational Psychology*, **79**, pp. 155-173.

PHILLIPS, J.M. and GULLY, S.M., 1997. Role of goal orientation, ability, need for achievement, and locus of control in the self-efficacy and goal-setting process. *Journal of Applied Psychology*, **82**(5), pp. 792-802.

PINTRICH, P.R. and GARCIA, T., 1991. Student goal orientation and self-regulation in the college classroom. In: M.L. MAEHR and P.R. PINTRICH, eds, *Advances in motivation and achievement*. Greenwich: JAI Press, pp. 371-402.

PINTRICH, P.R. and SCHUNK, D.H., 2002. *Motivation in Education: theory, research, and applications.* 2nd edn. New Jersey: Prentice-Hall International.

PINTRICH, P.R., 2000. Multiple Goals, Multiple Pathways: The Role of Goal Orientation in Learning and Achievement. *Journal of Educational Psychology*, **92**(3), pp. 544-555.

PINTRICH, P.R., SMITH, D.A.F., GARCIA, T. and MCKEACHIE, W.J., 1993. Reliability and Predictive Validity of the Motivated Strategies for Learning Questionnaire (MSLQ). *Educational and Psychological Measurement*, **53**(3), pp. 801-814.

PINTRICH, P.R., 2000. An Achievement Goal Theory Perspective on Issues in Motivation Terminology, Theory, and Research. *Contemporary educational psychology*, **25**(1), pp. 92-104.

PINTRICH, P.R., CONLEY, A.M. and KEMPLER, T.M., 2003. Current issues in achievement goal theory and research. *International Journal of Educational Research*, **39**(4-5), pp. 319-337.

PUTWAIN, D.W. and DANIELS, R.A., 2010. Is the relationship between competence beliefs and test anxiety influenced by goal orientation? *Learning and Individual Differences*, **20**(1), pp. 8-13.

RADOSEVICH, D.J., ALLYN, M.R. and YUN, S., 2007. Predictive validity of goal orientation: four-factor versus three-factor model. *Review of Business Research*, **7**(4),.

RANTANEN, J., METSAPELTO, R., FELDT, T., PULKKINEN, L. and KOKKO, K., 2007. Long Term Stability in the Big Five Personality traits in adulthood. *Scandinavian Journal of Psychology*, **48**, pp. 511-518.

REINARD, J.C. (2006). Communication Research Statistics. SAGE: London.

RHEINBERG, F., VOLLMEYER, R. and ROLLETT, W., 2000. Motivation and Action in Self-Regulated Learning. In: M. BOEKAERTS, P.R. PINTRICH and M. ZEIDNER, eds, *Handbook of Self-Regulation*. 1 edn. London: Academic Press, pp. 503-531.

ROBERSON, L. and ALSUA, C.J., 2002. Moderating Effects of Goal Orientation on the Negative Consequences of Gender-Based Preferential Selection. *Organizational behavior and human decision processes*, **87**(1), pp. 103-135.

ROESER, R.W., MIDGLEY, C. and URDAN, T., 1996. Perceptions of the School Psychological Environment and Early Adolescents' Psychological and Behavioral Functioning in School: The Mediating Role of Goals and Belonging. *Journal of Educational Psychology*, **88**(3), pp. 408-422.

ROGERS, A. and SPITZMUELLER, C., 2009. Individualism-collectivism and the role of goal orientation in organizational training. *International Journal of Training and Development*, **13**(3), pp. 185-201.

RYAN, R., M and DECI, E., L., 2000. Intrinsic and Extrinsic Motivations: Classic Definitions and New Directions. *Contemporary Educational Psychology*, **25**, pp. 54.

SCHANTZ, L.H. and CONROY, D.E., 2009. Achievement motivation and intraindividual affective variability during competence pursuits: A round of golf as a multilevel data structure. *Journal of Research in Personality*, **43**(3), pp. 472-481.

SCHRAW, G., HORN, C., THORNDIKE-CHRIST, T. and BRUNING, R., 1995. Academic Goal Orientations and Student Classroom Achievement. *Contemporary educational psychology*, **20**(3), pp. 359-368.

SEIJTS, G.H., LATHAM, G.P., TASA, K. and LATHAM, B.W., 2004. Goal setting and goal orientation: An integration of two different yet related literatures. *Academy of Management Journal*, **47**(2), pp. 227-239.

SENKO, C. and HARACKIEWICZ, J.M., 2005. Achievement Goals, Task Performance, and Interest: Why Perceived Goal Difficulty Matters. *Personality and Social Psychology Bulletin*, **31**(12), pp. 1739-1753.

SINS, P.H.M., VAN JOOLINGEN, W.R., SAVELSBERGH, E.R. and VAN HOUT-WOLTERS, B., 2008. Motivation and performance within a collaborative computer-based modeling task: Relations between students' achievement goal orientation, self-efficacy, cognitive processing, and achievement. *Contemporary educational psychology*, **33**(1), pp. 58-77.

SKAALVIK, E.M., 1997. Self-Enhancing and Self-Defeating Ego Orientation: Relations With Task and Avoidance Orientation, Achievement, Self-Perceptions, and Anxiety. *Journal of Educational Psychology*, **89**(1), pp. 71-81.

SMITH, M., DUDA, J., ALLEN, J. and HALL, H., 2002. Contemporary measures of approach and avoidance goal orientations: Similarities and differences. *British Journal of Educational Psychology*, **72**, pp. 155-190.

SMITH, M., DUDA, J., ALLEN, J. and HALL, H., 2002. Contemporary measures pf approach and avoidance goal orientations: Similarities and differences. *British Journal of Educational Psychology*, **72**, pp. 155-190.

STEELE-JOHNSON, D., BEAUREGARD, R.S., HOOVER, P.B. and SCHMIDT, A.M., 2000. Goal Orientation and Task Demand Effects on Motivation, Affect, and Performance. *Journal of Applied Psychology*, **85**(5), pp. 724-738.

STEELE-JOHNSON, D., HEINTZ JR, P. and MILLER, C.E., 2008. Examining Situationally Induced State Goal Orientation Effects on Task Perceptions, Performance, and Satisfaction: A Two-Dimensional Conceptualization. *Journal of Applied Social Psychology*, **38**(2), pp. 334-365.

SUJAN, H., WEITZ, B.A. and KUMAR, N., 1994. Learning Orientation, Working Smart, and Effective Selling. *Journal of Marketing*, **58**(3), pp. 39-52.

TABACHNIK, B.G. and FIDELL, L.S., 1996. Using Multivariate Statistics. Third edn. New York: HarperCollins.

TANAKA, K., 2007. Relations between general goal orientations and task-specific selfappraisals. *Japanese Psychological Research*, **49**(4), pp. 235-247.

THORKILDSEN, T.A., 1988. Theories of education among academically able adolescents. *Contemporary educational psychology*, **13**(4), pp. 323-330.

UEBERSAX, J.S., 10th August 2000, 2000-last update, A Brief Study of Local Maximum Solutions in Latent Class Analysis [Homepage of John Uebersax Enterprises LLC], [Online]. Available: <u>http://www.john-uebersax.com/stat/local.htm</u> [03/08, 2011].

URDAN, T.C., 1997. Achievement Goal Theory: Past Results, Future Directions. In: M.L. MAEHR and P.R. PINTRICH, eds, *Advances in Motivation and Achievement*. Greenwich: Connecticutt: JAI Press, pp. 99-141.

URDAN, T.C. and MAEHR, M.L., 1995. Beyond a Two-Goal Theory of Motivation and Achievement: A Case for Social Goals. *Review of Educational Research*, **65**(3), pp. 213-243.

VAN HOOFT, E.A.J. and NOORDZIJ, G., 2009. The Effects of Goal Orientation on Job Search and Reemployment: A Field Experiment Among Unemployed Job Seekers. *Journal of Applied Psychology*, **6**, pp. 1581-1590.

VAN YPEREN, N.W., 2003. Task Interest and Actual Performance: The Moderating Effects of Assigned and Adopted Purpose Goals. *Journal of Personality and Social Psychology*, **85**(6), pp. 1006-1015.

VAN YPEREN, N.W., ELLIOT, A.J. and ANSEEL, F., 2009. The influence of masteryavoidance goals on performance improvement. *European Journal of Social Psychology*, **39**(6), pp. 932-943.

VAN YPEREN, N.W. and RENKEMA, L.J., 2008. Performing great and the purpose of performing better than others: on the recursiveness of the achievement goal adoption process. *European Journal of Social Psychology*, **38**(2), pp. 260-271.

VAN YPEREN, N.W., 2006. A Novel Approach to Assessing Achievement Goals in the Context of the 2x 2 Framework: Identifying Distinct Profiles of Individuals With Different Dominant Achievement Goals. *Personality and Social Psychology Bulletin*, **32**(11), pp. 1432-1445.

VANDEWALLE, D., 1997. Development and validation of a work domain goal orientation instrument. *Educational and Psychological Measurement.*, **8**(6), pp. 995-1015.

VANDEWALLE, D., BROWN, S.P., CRON, W.L. and SLOCUM JR., J.W., 1999. The Influence of Goal Orientation and Self-Regulation Tactics on Sales Performance: A Longitudinal Field Test. *Journal of Applied Psychology*, **84**(2), pp. 249-259.

VANDEWALLE, D., CRON, W., L. and SLOCUM JR., J., W., 2001. The Role of Goal Orientation Following Performance Feedback. *Journal of Applied Psychology*, **86**(4), pp. 629-640.

VANDEWALLE, D., 2003. A goal orientation model of feedback-seeking behavior. *Human Resource Management Review*, **13**(4), pp. 581-604.

VANDEWALLE, D. and CUMMINGS, L.L., 1997. A Test of the Influence of Goal Orientation on the Feedback-Seeking Process. *Journal of Applied Psychology*, **82**(3), pp. 390-400.

VERMUNT, J.K. and MAGIDSON, J., 2005. *Latent GOLD 4.0 User's Guide*. Belmont, Massachusetts: Statistical Innovations Inc.

VERMUNT, J.K. and MAGIDSON, K., 2002. Latent Class Cluster Analysis. In: J.A. HAGENAARS and A.L. MCCUTCHEON, eds, *Applied Latent Class Analysis*. Cambridge: Cambridge University Press, pp. 89-106.

VISWESVARAN, C. and ONES, D.S., 2000. Measurement Error in "Big Five Factors" Personality Assessment: Reliability Generalization across Studies and Measures. *Educational and Psychological Measurement*, **60**(2), pp. 224-235.

VRUGT, A., FRANS, J.O. and ZEEBERG, C., 2002. Goal orientations, perceived selfefficacy and study results amongst beginners and advanced students. *British Journal of Educational Psychology*, **72**, pp. 385.

WANG, M. and ERDHEIM, J., 2007. **Does the five-factor model of personality relate to** goal orientation?. *Personality and Individual Differences*, **43**, pp. 1493.

WARBURTON, V.E. and SPRAY, C.M., 2009. Antecedents of Approach-Avoidance Achievement Goal Adoption in Physical Education: A Longitudinal Perspective. *Journal of Teaching in Physical Education*, **28**(2), pp. 214-232.

WIGFIELD, A., 1997. Reading Motivation: A Domain-Specific Approach to Motivation. *Educational Psychologist*, **32**(2), pp. 59-68.

WIGFIELD, A. and CAMBRIA, J., 2010. Students' achievement values, goal orientations, and interest: Definitions, development, and relations to achievement outcomes. *Developmental Review*, **30**(1), pp. 1-35.

WOLTERS, C.A., 2004. Advancing Achievement Goal Theory: Using Goal Structures and Goal Orientations to Predict Students' Motivation, Cognition, and Achievement. *Journal of Educational Psychology*, **96**(2), pp. 236-250.

WOLTERS, C.A., YU, S.L. and PINTRICH, P.R., 1996. The Relation Between Goal Orientation And Students' Motivational Beliefs and Self-Regulated Learning. *Learning and Individual Differences*, **8**(3), pp. 211-238.

WOOD, R. and BANDURA, A., 1989. Impact of Conceptions of Ability on Self-Regulatory Mechanisms and Complex Decision Making. *Journal of Personality and Social Psychology*, **56**(3), pp. 407-415.

XIANG, P. and LEE, A., 2002. Achievement goals, perceived motivational climate, and students' self-reported mastery behaviors. *Research Quarterly For Excercise And Sport*, **73**(1), pp. 58-65.

YEO, G., LOFT, S., XIAO, T. and KIEWITZ, C., 2009. Goal Orientations and Performance: Differential Relationships Across Levels of Analysis and as a Function of Task Demands. *Journal of Applied Psychology*, **94**(3), pp. 710-726.

YEO, G.B. and NEAL, A., 2004. A Multilevel Analysis of Effort, Practice, and Performance: Effects of Ability, Conscientiousness, and Goal Orientation. *Journal of Applied Psychology*, **89**(2), pp. 231-247.

YEO, G.B., SORBELLO, T., KOY, A. and SMILLIE, L.D., 2008. Goal orientation profiles and task performance growth trajectories. *Motivation and Emotion*, **32**, pp. 296-309.

YPEREN, N.W.V. and JANSSEN, O., 2002. Fatigued and Dissatisfied or Fatigued but Satisfied? Goal Orientations and Responses to High Job Demands. *The Academy of Management Journal*, **45**(6), pp. 1161-1171.

ZWEIG, D. and WEBSTER, J., 2004. What are we measuring? An examination of the relationships between the big-five personality traits, goal orientation, and performance intentions. *Personality and Individual Differences*, **36**(7), pp. 1693-1708.

ZWEIG, D. and WEBSTER, J., 2004. What are we measuring? An examination of the relationships between the big-five personality traits, goal orientation, and performance intentions. *Personality and Individual Differences*, **36**(7), pp. 1693-1708.

Appendices

APPENDIX A TIME Q1 QUESTIONNAIRES



Dear Participant,

My name is Debbie Naudi and I am currently completing a PhD at the Loughborough University Business School. My PhD focuses on the ways in which people approach activities. I have compiled three short questionnaires which I am asking you to kindly complete. In the first questionnaire I ask about the ways in which you approach life activities in general. In the second questionnaire I ask about the ways in which you approach a particular task whilst the third questionnaire is a personality questionnaire. Please note that participation in this study is completely voluntary and you may withdraw from this study at any point in time without any negative consequences whatsoever. The information in these questionnaires shall be kept in **very strict confidentiality** throughout the study and will only be used for this research project. If you would like any further information regarding the nature of this study or the results obtained please feel free to contact me on the following e-mail address: <u>D.Naudi@lboro.ac.uk</u>

Your help is very much appreciated!

Kind Regards

Debbie Naudi

About You

In a few months time I will be asking you to complete another (shorter) questionnaire. Therefore, it is important for me to take your contact details in order to be able to send out the second questionnaire. Your personal details will **ONLY** be used for the purpose of this study and will be kept in **very strict confidentiality.** Findings in general will be reported in research papers; however, no individual will be identifiable.

Please use BLOCK capitals throughout

Name:		
Age: Years Months	Gender: Male	Female
E-mail address:		
How would you describe your ethnicity? .		
What course are you in?		

<u>General Activities.</u> The following items refer to activities in life in general e.g. fixing something; doing a sport in your free time; cooking; working on the computer etc. Please provide answers to the following items by keeping in mind the approach that you would GENERALLY adopt when pursuing such activities.

Some of the questions may look similar to each other but please do not worry about appearing to be consistent. Answer each question on its own merits.

Please circle **ONE** answer for each of the items below.

		Strongly Disagree	Disagree	Neither Agree Nor Disagree	Agree	Strongly Agree
1	My aim is to completely master everything I do.	1	2	3	4	5
2	I strive to do well compared with others in everything I do.	1	2	3	4	5
3	My goal is to learn as much as possible about the things I do.	1	2	3	4	5
4	My aim is to perform well relative to others in everything I do.	1	2	3	4	5
5	My aim is to ensure I do not miss out on new learning opportunities.	1	2	3	4	5
6	My goal is to avoid performing poorly compared to others in everything that I do.		2	3	4	5
7	I strive to understand the content of the things I do as much as possible.	1	2	3	4	5
8	My goal is to perform better than others in everything I do.	1	2	3	4	5
9	My goal is to avoid missing opportunities to fully understand an activity.	1	2	3	4	5
10	I strive to avoid performing worse than others in everything that I do.	1	2	3	4	5
11	I strive to avoid an incomplete understanding of the things I do.	1	2	3	4	5
12	My aim is to avoid doing worse than others in everything that I do.	1	2	3	4	5

Task Scenario. Please read the work scenario provided below and complete the questionnaire that follows. Kindly keep in mind how you would approach the task described below when providing answers to the questionnaire.

Some of the questions may look similar to each other (and to the previous questionnaire) but please do not worry about appearing to be consistent. Answer each question on its own merits.

Scenario: You are a sales person in a company and have been asked to sell a new range of carpets that have just appeared on the market. These carpets are harder to sell than the other carpets since they are higher in price. Your boss does not set any sales target for you for the time being since this is a new product.

Please answer the questions below keeping in mind the approach you will adopt when carrying out this task.

		Strongly Disagree	Disagree	Neither Agree Nor Disagree	Agree	Strongly Agree
1	I will aim to completely master selling the new carpets.	1	2	3	4	5
2	I will strive to do better at selling the new carpets compared to my colleagues.	1	2	3	4	5
3	My goal will be to learn as much as possible regarding how to sell these new carpets.	1	2	3	4	5
4	My aim will be to sell more new carpets relative to my colleagues.	1	2	3	4	5
5	My aim will be to avoid learning less than I possibly could about selling these new carpets.	1	2	3	4	5
6	My goal will be to avoid selling fewer new carpets compared to my colleagues.	1	2	3	4	5
7	I will be striving to understand how to sell these new carpets as thoroughly as possible.	1	2	3	4	5
8	My goal will be to perform better than my colleagues at selling these new carpets.	1	2	3	4	5
9	My goal will be to avoid learning less than it is possible to learn about selling new carpets.	1	2	3	4	5
10	My goal will be to avoid missing opportunities to fully understand how to sell the new carpets.	1	2	3	4	5
11	I will be striving to avoid an incomplete understanding about how to sell the new carpets.	1	2	3	4	5
12	My aim will be to avoid doing worse than my colleagues when selling the new carpets.	1	2	3	4	5

How I am in general. Here are a number of characteristics that may or may not apply to you. For example, do you agree that you are someone who likes to spend time with others? Please write a number next to each statement to indicate the extent to which you agree or disagree with that statement. Please note that there are no right or wrong answers and that the answers will be used **SOLELY** for the purpose of this research study.

						1		
	1	2	3	4	5			
	Disagree Strongly	Disagree a little	Neither agree nor disagree	Agree a little	Agree strongly			
l am se	omeone who							
1	Is talkative			23 Tends to	be lazy			
2	Tends to find fau	ult with others		24 ls	s emotionally stable, n	ot easily upset		
3	Does a thorough	ı job		25 Is invent	ive			
4	Is depressed, blu	Je		26 Has an a	ssertive personality			
5	Is original, come	s up with new ideas		27 Can be c	old and aloof			
6	Is reserved			28 Persever	es until the task is finis	shed		
7	Is helpful and un	selfish with others		29 Can be n	noody			
8	Can be somewha	at careless		30 Values artistic, aesthetic experiences				
9	9 Is relaxed, handles stress well.			31 Is sometimes shy, inhibited				
10	10 Is curious about many different things			32 Is considerate and kind to almost everyone				
11	Is full of energy			33 Does things efficiently				
12	Starts quarrels w	vith others		34 Remains	calm in tense situation	ns		
13	Is a reliable work	ker		35 Prefers v	work that is routine			
14	Can be tense			36 Is outgoi	ng, sociable			
15	Is ingenious, a de	eep thinker		37 Is somet	imes rude to others			
16	Generates a lot o	of enthusiasm		38 Makes p	lans and follows throu	gh with them		
17	Has a forgiving n	ature		39 Gets ner	vous easily			
18	Tends to be diso	rganized		40 Likes to r	reflect, play with ideas			
19	Worries a lot			41 Has few	artistic interests			
20	Has an active im	agination		42 Likes to a	cooperate with others			
21	Tends to be quie	et		43 Is easily	distracted			
22	Is generally trust	ting		44 Is sophis	ticated in art, music, o	r literature		

Optional Participation in Experiment:

Enhance your chances of being chosen for a job!

My study also requires that I carry out experiments. During these sessions you will be given the opportunity to complete aptitude tests. Employers very often ask job applicants to complete aptitude tests as part of their selection process. Therefore, apart from helping me in my research, your participation in these experiments will benefit you as you get the chance to practice and excel in aptitude tests used for job selection. You will also be given confidential feedback regarding your performance on these tests as well as advice regarding how to improve on them! In addition, you will be entered into a draw with the chance to win **one of three £50 cash prizes**.

If you agree to be contacted in the near future in order to participate in the experiment (as well as get the chance to win £50 cash!) please tick 'yes' below.



Thank you!



Dear Participant,

My name is Debbie Naudi and I am currently completing a PhD at the Loughborough University Business School. My PhD focuses on the ways in which people approach activities. I have compiled three short questionnaires which I am asking you to kindly complete. In the first questionnaire I ask about the ways in which you approach life activities in general. In the second questionnaire I ask about the ways in which you approach a particular task whilst the third questionnaire is a personality questionnaire. Please note that participation in this study is completely voluntary and you may withdraw from this study at any point in time without any negative consequences whatsoever. The information in these questionnaires shall be kept in **very strict confidentiality** throughout the study and will only be used for this research project. If you would like any further information regarding the nature of this study or the results obtained please feel free to contact me on the following e-mail address: <u>D.Naudi@lboro.ac.uk</u>

Your help is very much appreciated!

Kind Regards

Debbie Naudi

<u>About You</u>

In a few months time I will be asking you to complete another (shorter) questionnaire. Therefore, it is important for me to take your contact details in order to be able to send out the second questionnaire. Your personal details will **ONLY** be used for the purpose of this study and will be kept in **very strict confidentiality.** Findings in general will be reported in research papers; however, no individual will be identifiable.

Please use BLOCK capitals throughout

Name:		
Age: Years Months	Gender: Male	Female
E-mail address:		
How would you describe your ethnicity? .		
What course are you in?		

<u>General Activities.</u> The following items refer to activities in life in general e.g. fixing something; doing a sport in your free time; cooking; working on the computer etc. Please provide answers to the following items by keeping in mind the approach that you would GENERALLY adopt when pursuing such activities.

Some of the questions may look similar to each other but please do not worry about appearing to be consistent. Answer each question on its own merits.

Please circle **ONE** answer for each of the items below.

		Strongly Disagree	Disagree	Neither Agree Nor Disagree	Agree	Strongly Agree
1	My aim is to completely master everything I do.	1	2	3	4	5
2	I strive to do well compared with others in everything I do.	1	2	3	4	5
3	My goal is to learn as much as possible about the things I do.	1	2	3	4	5
4	My aim is to perform well relative to others in everything I do.	1	2	3	4	5
5	My aim is to ensure I do not miss out on new learning opportunities.	1	2	3	4	5
6	My goal is to avoid performing poorly compared to others in everything that I do.	1	2	3	4	5
7	I strive to understand the content of the things I do as much as possible.	1	2	3	4	5
8	My goal is to perform better than others in everything I do.	1	2	3	4	5
9	My goal is to avoid missing opportunities to fully understand an activity.	1	2	3	4	5
10	I strive to avoid performing worse than others in everything that I do.	1	2	3	4	5
11	I strive to avoid an incomplete understanding of the things I do.	1	2	3	4	5
12	My aim is to avoid doing worse than others in everything that I do.	1	2	3	4	5

Task Scenario. Please read the work scenario provided below and complete the questionnaire that follows. Kindly keep in mind how you would approach the task described below when providing answers to the questionnaire.

Some of the questions may look similar to each other (and to the previous questionnaire) but please do not worry about appearing to be consistent. Answer each question on its own merits.

Scenario: Your department has been asked to attend a training course in order to learn how to use a new computer program that will be used on the job.

Please answer the questions below keeping in mind the approach you will adopt when carrying out this task.

		Strongly Disagree	Disagree	Neither Agree Nor Disagree	Agree	Strongly Agree
1	My aim will be to completely master the new computer program.	1	2	3	4	5
2	I will strive to do better in training compared to my colleagues.	1	2	3	4	5
3	My goal will be to learn as much as possible about this new computer program.	1	2	3	4	5
4	My aim will be to perform better during training relative to my colleagues.	1	2	3	4	5
5	My aim will be to avoid learning less than I possibly could about this new computer program.	1	2	3	4	5
6	My goal will be to avoid performing poorly during training compared to my colleagues.	1	2	3	4	5
7	I will be striving to understand the content of this computer program as thoroughly as possible.	1	2	3	4	5
8	My goal will be to perform better than my colleagues during training.	1	2	3	4	5
9	My goal will be to avoid learning less than it is possible to learn about this new computer program.	1	2	3	4	5
10	My goal will be to avoid missing opportunities to fully understand the new computer program.	1	2	3	4	5
11	I will be striving to avoid an incomplete understanding of this new computer program.	1	2	3	4	5
12	My aim will be to avoid doing worse than my colleagues during training.	1	2	3	4	5

<u>How I am in general.</u> Here are a number of characteristics that may or may not apply to you. For example, do you agree that you are someone who *likes to spend time with others*? Please write a number next to each statement to indicate the extent to which <u>you agree or disagree with that statement</u>. Please note that there are no right or wrong answers and that the answers will be used **SOLELY** for the purpose of this research study.

	1	2	3		4	5			
	Disagree Strongly	Disagree a little	Neither agree nor disagree		Agree a little	Agree strongly			
l am se	omeone who	·							
1.	Is talkativ	e		23.	Tends	to be lazy			
2.	Tends to f	find fault with others		24 Is emotionally stable, not ea					
3.	Does a the	orough job		25.	Is inve	ntive			
4.	Is depress	sed, blue		26.	Has an				
5.	Is original	, comes up with new i	deas	27.	Can be				
6.	ls reserve	d		28.	Persev	eres until the task is fi	nished		
7.	Is helpful	and unselfish with oth	iers	29.	Can be	Can be moody			
8.	Can be so	mewhat careless		30.	Values	Values artistic, aesthetic expe			
9.	Is relaxed,	handles stress well.		31.	Is som	Is sometimes shy, inhibited			
10.	Is curious	about many different	things	32.	Is cons	iderate and kind to alr	nost everyone		
11.	Is full of e	energy		33.	Does t	hings efficiently			
12.	Starts qu	arrels with others		34.	Remai	ns calm in tense situat	ions		
13.	Is a reliab	ole worker		35.	Prefers	s work that is routine			
14.	Can be te	ense		36.	Is outg	oing, sociable			
15.	Is ingenic	ous, a deep thinker		37.	Is som	etimes rude to others			
16.	Generate	es a lot of enthusiasm		38.	Makes	plans and follows thro	ough with them		
17.	Has a for	giving nature		39.	Gets n	ervous easily			
18.	Tends to	be disorganized		40.	Likes to	o reflect, play with ide	as		
19.	Worries a	a lot		41.	Has fe	w artistic interests			
20.	Has an ac	ctive imagination		42.	Likes to	o cooperate with othe	rs		
21.	Tends to	be quiet		43.	Is easil	y distracted			
22.	Is genera	Illy trusting		44.	Is soph	isticated in art, music,	or literature		

Optional Participation in Experiment:

Enhance your chances of being chosen for a job!

My study also requires that I carry out experiments. During these sessions you will be given the opportunity to complete aptitude tests. Employers very often ask job applicants to complete aptitude tests as part of their selection process. Therefore, apart from helping me in my research, your participation in these experiments will benefit you as you get the chance to practice and excel in aptitude tests used for job selection. You will also be given confidential feedback regarding your performance on these tests as well as advice regarding how to improve on them! In addition, you will be entered into a draw with the chance to win **one of three £50 cash prizes**.

If you agree to be contacted in the near future in order to participate in the experiment (as well as get the chance to win £50 cash!) please tick 'yes' below.



Thank you!

APPENDIX B TIME Q2 QUESTIONNAIRES



Dear Participant,

As you may remember, my name is Debbie Naudi and I am currently reading for a PhD at the Loughborough University Business School. My PhD focuses on the ways in which people approach activities. I have compiled a follow up questionnaire for you to kindly complete. This questionnaire consists of two sections. The first section is about the ways in which you approach life activities in general whilst the second section focuses on the ways in which you approach a particular task. Please note that participation in this study is completely voluntary and you may withdraw from this study at any point in time without any negative consequences whatsoever. The information in these questionnaires shall be kept in **very strict confidentiality** throughout the study and will only be used for this research project. If you would like any further information regarding the nature of this study or the results obtained please feel free to contact me on the following e-mail address: <u>D.Naudi@lboro.ac.uk</u>

Your help is very much appreciated!

Kind Regards

Debbie Naudi
About You

Since I would like to examine approaches to goals over time, I am required to take your name, age, and course in order to be able to match this questionnaire to the first questionnaire you completed. Your personal details will **ONLY** be used for the purpose of this study and will be kept in **very strict confidentiality.** Findings in general will be reported in research papers; however, no individual will be identifiable.

Please use BLOCK capitals throughout

Name: _____ Age: Years ____ Months _____

Students only: What course are you in? _____

<u>General Activities.</u> The following items refer to activities in life in general e.g. fixing something; doing a sport in your free time; cooking; working on the computer etc. Please provide answers to the following items by keeping in mind the approach that you would GENERALLY adopt when pursuing such activities.

Some of the questions may look similar to each other but please do not worry about appearing to be consistent. Answer each question on its own merits.

Neither Agree Strongly Strongly Disagree Agree Disagree Nor Agree Disagree My aim is to completely master everything I do. I strive to do well compared with others in everything I do. My goal is to learn as much as possible about the things I do. My aim is to perform well relative to others in everything I do. My aim is to ensure I do not miss out on new learning opportunities. My goal is to avoid performing poorly compared to others in everything that I do. I strive to understand the content of the things I do as much as possible. My goal is to perform better than others in everything I do. My goal is to avoid missing opportunities to fully understand an activity. I strive to avoid performing worse than others in everything that I do. I strive to avoid an incomplete understanding of the things I do. My aim is to avoid doing worse than others in everything that I do.

Please circle **ONE** answer for each of the items below.

Task Scenario. Please read the work scenario provided below and complete the questionnaire that follows. Kindly keep in mind how you would approach the task described below when providing answers to the questionnaire.

Some of the questions may look similar to each other (and to the previous questionnaire) but please do not worry about appearing to be consistent. Answer each question on its own merits.

Scenario: You are a sales person in a company and have been asked to sell a new range of carpets that have just appeared on the market. These carpets are harder to sell than the other carpets since they are higher in price. Your boss does not set any sales target for you for the time being since this is a new product.

Please answer the questions below keeping in mind the approach you will adopt when carrying out this task.

		Strongly Disagree	Disagree	Neither Agree Nor Disagree	Agree	Strongly Agree
1	I will aim to completely master selling the new carpets.	1	2	3	4	5
2	I will strive to do better at selling the new carpets compared to my colleagues.	1	2	3	4	5
3	My goal will be to learn as much as possible regarding how to sell these new carpets.	1	2	3	4	5
4	My aim will be to sell more new carpets relative to my colleagues.	1	2	3	4	5
5	My aim will be to avoid learning less than I possibly could about selling these new carpets.	1	2	3	4	5
6	My goal will be to avoid selling fewer new carpets compared to my colleagues.	1	2	3	4	5
7	I will be striving to understand how to sell these new carpets as thoroughly as possible.	1	2	3	4	5
8	My goal will be to perform better than my colleagues at selling these new carpets.	1	2	3	4	5
9	My goal will be to avoid learning less than it is possible to learn about selling new carpets.	1	2	3	4	5
10	My goal will be to avoid missing opportunities to fully understand how to sell the new carpets.	1	2	3	4	5
11	I will be striving to avoid an incomplete understanding about how to sell the new carpets.	1	2	3	4	5
12	My aim will be to avoid doing worse than my colleagues when selling the new carpets.	1	2	3	4	5

Thank you!

Task Scenario 2 Questionnaire



Dear Participant,

As you may remember, my name is Debbie Naudi and I am currently reading for a PhD at the Loughborough University Business School. My PhD focuses on the ways in which people approach activities. I have compiled a follow up questionnaire for you to kindly complete. This questionnaire consists of two sections. The first section is about the ways in which you approach life activities in general whilst the second section focuses on the ways in which you approach a particular task. Please note that participation in this study is completely voluntary and you may withdraw from this study at any point in time without any negative consequences whatsoever. The information in these questionnaires shall be kept in **very strict confidentiality** throughout the study and will only be used for this research project. If you would like any further information regarding the nature of this study or the results obtained please feel free to contact me on the following e-mail address: <u>D.Naudi@lboro.ac.uk</u>

Your help is very much appreciated!

Kind Regards

Debbie Naudi

About You

Since I would like to examine approaches to goals over time, I am required to take your name, age, and course in order to be able to match this questionnaire to the first questionnaire you completed. Your personal details will **ONLY** be used for the purpose of this study and will be kept in **very strict confidentiality.** Findings in general will be reported in research papers; however, no individual will be identifiable.

Please use BLOCK capitals throughout

Name: _____ Age: Years ____ Months _____

Students only: What course are you in? _____

<u>General Activities.</u> The following items refer to activities in life in general e.g. fixing something; doing a sport in your free time; cooking; working on the computer etc. Please provide answers to the following items by keeping in mind the approach that you would GENERALLY adopt when pursuing such activities.

Some of the questions may look similar to each other but please do not worry about appearing to be consistent. Answer each question on its own merits.

Neither Agree Strongly Strongly Disagree Agree Disagree Nor Agree Disagree My aim is to completely master everything I do. I strive to do well compared with others in everything I do. My goal is to learn as much as possible about the things I do. My aim is to perform well relative to others in everything I do. My aim is to ensure I do not miss out on new learning opportunities. My goal is to avoid performing poorly compared to others in everything that I do. I strive to understand the content of the things I do as much as possible. My goal is to perform better than others in everything I do. My goal is to avoid missing opportunities to fully understand an activity. I strive to avoid performing worse than others in everything that I do. I strive to avoid an incomplete understanding of the things I do. My aim is to avoid doing worse than others in everything that I do.

Please circle **ONE** answer for each of the items below.

Task Scenario. Please read the work scenario provided below and complete the questionnaire that follows. Kindly keep in mind how you would approach the task described below when providing answers to the questionnaire.

Some of the questions may look similar to each other (and to the previous questionnaire) but please do not worry about appearing to be consistent. Answer each question on its own merits.

Scenario: Your department has been asked to attend a training course in order to learn how to use a new computer program that will be used on the job.

Please answer the questions below keeping in mind the approach you will adopt when carrying out this task.

		Strongly Disagree	Disagree	Neither Agree Nor Disagree	Agree	Strongly Agree
1	My aim will be to completely master the new computer program.	1	2	3	4	5
2	I will strive to do better in training compared to my colleagues.	1	2	3	4	5
3	My goal will be to learn as much as possible about this new computer program.	1	2	3	4	5
4	My aim will be to perform better during training relative to my colleagues.	1	2	3	4	5
5	My aim will be to avoid learning less than I possibly could about this new computer program.	1	2	3	4	5
6	My goal will be to avoid performing poorly during training compared to my colleagues.	1	2	3	4	5
7	I will be striving to understand the content of this computer program as thoroughly as possible.	1	2	3	4	5
8	My goal will be to perform better than my colleagues during training.	1	2	3	4	5
9	My goal will be to avoid learning less than it is possible to learn about this new computer program.	1	2	3	4	5
10	My goal will be to avoid missing opportunities to fully understand the new computer program.	1	2	3	4	5
11	I will be striving to avoid an incomplete understanding of this new computer program.	1	2	3	4	5
12	My aim will be to avoid doing worse than my colleagues during training.	1	2	3	4	5

Thank you!

APPENDIX C E-MAIL WITH TIME Q2 QUESTIONNAIRE (Non-Experimental Participants)

Dear {name of participant}

My name is Debbie Naudi, I'm a PhD student at the Loughborough University Business School. About 2 months ago I handed out a questionnaire during your {name of module} lecture. I have a follow-up questionnaire (attached) which only takes 3 minutes to complete, however, it is extremely important for my research. I would really really appreciate it if you could kindly complete this questionnaire and send it back to me.

Thank you so much for your help and support and good luck in your exams!

Kind Regards

Debbie

APPENDIX D E-MAIL SENT PRIOR TO TIME 1 EXPERIMENT

Dear {name of participant}

My name is Debbie Naudi, I'm a PhD student at the Loughborough University Business School. A few weeks ago you kindly completed a questionnaire for my research study (during one of your lectures) and you indicated that you were interested in participating in my research experiments. By participating in these experiments you will be helping me greatly with my research. However, these experiments will also benefit you.

During the experimental sessions you will be given the opportunity to complete aptitude tests. Employers very often ask job applicants to complete aptitude tests as part of their selection process. Therefore, apart from helping me in my research, these experiments will provide you with the chance to practise and excel in aptitude tests used for job selection. The tests used in this experimental study are not practice tests found on websites, they are proper tests that have been purchased from a testing company. Therefore, you will gain a competitive edge over other job applicants when applying for a job in the future.

You will also be given confidential feedback regarding your performance on these tests as well as advice regarding how to improve on them! In addition, you will be entered into a draw with the chance to win **one of three £50 cash prizes**.

The experimental session will last 1 hour and you will be asked to attend two sessions in total. Following the second session I shall be providing you with feedback on the aptitude tests as well as feedback regarding the personality test and goal orientations.

The following is a list of dates when experiments will be taking place. If you are interested in taking part please choose a date and time from the list below and reply to this e-mail with the chosen date and time. If you are not able to attend any of the sessions below but would still like to participate, please send me an e-mail and I will arrange an extra session.

There will be a maximum of 10 participants per experimental session. I will provide you with further details regarding the experimental session closer the time.

Thank you for taking the time to help me with my research, it is very much appreciated!

Please choose **one** session for which it is preferable for you to attend from the list below.

Date Time

{List of dates} {List of Times}

APPENDIX E INFORMATION SHEET AND CONSENT FORM



Information Sheet for Participants

Dear Participant,

My name is Debbie Naudi and I am currently completing a PhD at the Loughborough University Business School. The focus of my PhD is about the ways in which people approach different activities. During this session I shall be asking you to carry out two different tasks: a verbal task and a numerical task. Before each task as well as on completion of each task I shall be asking you to complete a short questionnaire. This experiment should take approximately 1 hour and I shall be asking you to kindly attend another experiment after a period of 1 month. Feedback will be provided after the second session and you will be given a small item as a token of my appreciation. By attending the second experiment you will automatically be entered into a prize draw with the chance of winning one of three £50 cash prizes.

Please note that participation in this study is completely voluntary and you may withdraw from this study at any point in time without any negative consequences whatsoever. The answers to the questionnaires as well as the experiments shall be kept in very strict confidentiality throughout the study. If you would like any further information regarding the nature of this study or the results obtained please feel free to contact me (at any point in time) on the following e-mail address: <u>D.Naudi@lboro.ac.uk</u>

INFORMED CONSENT FORM

The purpose and details of this study have been explained to me. I understand that this study is designed to further scientific knowledge and that all procedures have been approved by the Loughborough University Ethical Advisory Committee.

I have read and understood the information sheet and this consent form.

I have had an opportunity to ask questions about my participation.

I understand that I am under no obligation to take part in the study.

I understand that I have the right to withdraw from this study at any stage for any reason, and that I will not be required to explain my reasons for withdrawing.

Chapter 12I understand that all the information I provide will be treated in strict confidence and will be kept anonymous and confidential to the researchers unless (under the statutory obligations of the agencies which the researchers are working with), it is judged that confidentiality will have to be breached for the safety of the participant or others.

I agree to participate in this study.

Your name	
Your signature	
Signature of investigator	
Date	

APPENDIX F Self-Efficacy Questionnaire

NAME:

<u>My feelings about this task:</u> The items below ask about your feelings regarding the task you are about to perform. Please circle **ONE** answer for each of the items below.

		Strongly Disagree	Disagree	Neither Agree Nor Disagree	Agree	Strongly Agree
1	I'm certain I can understand the most difficult items presented on this task.	1	2	3	4	5
2	I'm confident I can learn the basic concepts of this task.	1	2	3	4	5
3	I'm confident I can understand the most complex items presented on this task.	1	2	3	4	5
4	I'm confident I can do an excellent job on this task.	1	2	3	4	5
5	I expect to do well on this task.	1	2	3	4	5
6	I'm certain I can master the skills required for this task.	1	2	3	4	5
7	Considering the difficulty of this task and my skills, I think I will do well on this task.	1	2	3	4	5

APPENDIX G TIME E1 EXPERIMENT INSTRUCTIONS

Neutral Task Instructions

Prepare an information sheet and consent form on each desk before start of session. Also put 2 pencils, an eraser, a calculator, and some rough paper on each desk.

First of all I would like to thank you for attending this session. The session will take 1 hour. Could you all switch off you mobile phones please?

First I would like you to read the paper with the title 'information sheet' found on your desks. This sheet is for you to keep.

Do you have any questions?

You will also find an informed consent form on your desk. Could you kindly read and sign this form please? I am required to provide this information and ask you to sign a consent form for ethical purposes. This is required by the Loughborough University Ethical Advisory Committee. Once you have completed the forms could you all pass them forward please?

Count number of forms and check that all of them have been completed appropriately.

Hand out the answer sheets.

Please print your name in the space provided at the top of your answer sheet, surname first, then your first names.

Now fill in today's date which is the Please ignore the reference number. Please do not use pens on the answer sheets, only use the pencils provided.

Hand out the verbal test booklets. Please don't open these booklets until I tell you to do so.

When all booklets have been handed out: Now please turn to pages 2 and 3 of the booklet and follow the instructions while I read them aloud.

In this test you will be asked to reason with written information drawn from the world of work. The test consists of a series of passages, each of which is followed by several statements. Your task is to assess whether each statement is either "true", "false", or "cannot say" in relation to the passage which preceded it.

So having read a passage and the statements which follow, you would;

fill in *circle A* if you think the statement is *true* in relation to the passage, fill in *circle B* if you think the statement is *false* in relation to the passage, fill in *circle C* if you *cannot say* whether the statement is true or false in relation to the passage. You should assess each statement *only in relation to the passage which preceded it* and not in relation to any personal views or knowledge you might hold.

Do not write anything on this question booklet, but indicate your answer by completely filling in the appropriate circle on your answer sheet.

Before you start there are some example questions to complete. Look at the passage on the opposite page. Now, in your own time, read the passage and assess the statements which follow, according to the rules above. Choose what you think is the best answer for each statement, A, B, or C and mark your choice in the Examples Section on your answer sheet.

Hand out self-efficacy questionnaires. When everyone has finished the examples:

The answer to Example 1 is B, the statement is false given the information in the passage.

The answer to Example 2 is A, the statement is true given the information in the passage.

The answer to Example 3 is C, you cannot say from the information in the passage.

The answer to Example 4 is A, the statement is true given the information in the passage.

Are there any questions? If you are not clear about what you have to do, ask me now as I can't answer any questions once the test has started.

I have handed out a short questionnaire which I would like you to complete. Please don't forget to print your name at the top of this questionnaire.

Ask if everyone's ready and when they are...

Could you all pass your questionnaires forward please?

Count questionnaires and make sure that everyone has written their names on it.

Now could you look back at page 2 and follow whilst I read out the reminder points please?

- The test consists of 24 questions and you will have 14 minutes in which to do them.
- Try to work quickly but accurately.
- Fill in completely the appropriate circles on the answer sheet.
- Do not make any marks on this booklet.
- Fully erase any answer you wish to change.
- If you are not sure of an answer mark your best choice, but avoid simply guessing.

Are there any final questions? Now turn over to page 5 and begin.

Start stopwatch on the word 'begin', and note the start time on a piece of paper. Walk around after 2 minutes and again after 10 minutes. After exactly 15 minutes say:

Please stop now. Pencils down. Close your booklets.

Collect in all the materials, booklets, answer sheets, pencils and erasers. Check that there are 10 booklets and 10 answer sheets, and check that all answer sheets have names on them.

Could you kindly complete this short questionnaire I'm handing out please?

Hand out post-test questionnaires.

Please don't forget to write your names down on the first page of the questionnaires.

Has everyone completed the questionnaire? Can you all pass your questionnaires forward please?

So that was the first half of the session. We're going to go through exactly the same process for the second aptitude test.

Hand out the answer sheets.

Please print your name in the space provided at the top of your answer sheet, surname first, then your first names.

Now fill in today's date which is theOnce again, please ignore the reference number.

Hand out the numerical test booklets. Please don't open these booklets until I tell you to do so.

When all booklets have been handed out: Now please turn to pages 2 and 3 of the booklet and follow the instructions while I read them aloud.

In this test you will be using facts and figures taken from the world of work, to assess your ability to interpret and evaluate numerical information. All the information you need will be provided for you in the form of various tables and charts.

For each question you are given five answers to choose from. One, and only one, of the answers is correct in each case.

Do not write anything on this question booklet, but indicate your answer by completely filling in the appropriate circle on your answer sheet.

You may wish to use a calculator for this test. Rough paper is also provided to help you with any working out.

Before you start there are some example questions to complete. Look at these example questions below. Now, in your own time and using just the statistical information provided on the opposite page, choose what you think is the best answer for each question from the five alternatives given. Mark your choices in the Examples Section on your answer sheet.

When all have finished the examples, say:

The answer to Example 1 is B, 36 Older Adults.

The answer to Example 2 is C, 200, 000 tyres.

The answer to example 3 is D, 53.7%.

Are there any questions? If you are not clear about what you have to do, ask me now as I can't answer any questions once the test has started.

Now, I have handed out a short questionnaire which I would like you to complete. Please don't forget to print your name at the top of this questionnaire.

Ask if everyone's ready and when they are...

Could you all pass your questionnaires forward please?

Count questionnaires and make sure that everyone has written their names on it.

Now could you proceed to page 5 and follow whilst I read out the reminder points please?

- The test consists of 12 questions and you will have 15 minutes in which to do them.
- Try to work quickly but accurately.
- Fill in completely the appropriate circles on the answer sheet.
- Do not make any marks on this booklet.
- Use the rough paper provided for your working out.
- Fully erase any answer you wish to change.
- If you are not sure of an answer mark your best choice, but avoid simply guessing.

Are there any final questions?

Now turn over to page 6 and begin.

Start stopwatch on the word 'begin', and note the start time on a piece of paper. Walk around after 2 minutes and again after 10 minutes. After exactly 17 minutes say:

Please stop now. Pencils down. Close your booklets.

Collect in all the booklets and answer sheets. Check that there are 10 booklets and 10 answer sheets, and check that all answer sheets have names on them.

Could you kindly complete this short questionnaire I'm handing out please?

Hand out post-test questionnaires.

Please don't forget to write your names down on the first page of the questionnaires.

Collect calculators and place a BS pen on each desk.

Has everyone completed the questionnaire? Can you all pass your questionnaires forward please? Collect questionnaires, pencils and erasers. Check that they all wrote their names on the questionnaire.

Have you got any questions?

I will be asking you to come in again in approximately one month's time to complete the tests again and then I will give you personal feedback regarding your performance on the tests, your goal orientations, personality and advice on how to improve on these tests. That's it for today. I have placed a pen on each of your desks. That is for you to keep as a token of my appreciation.

Thank you so much for your participation, I really appreciate it!!

Neutral Task Instructions (Numerical Test First)

Prepare an information sheet and consent form on each desk before start of session. Also put 2 pencils, an eraser, a calculator, and some rough paper on each desk.

First of all I would like to thank you for attending this session. The session will take 1 hour. Could you all switch off you mobile phones please?

First I would like you to read the paper with the title 'information sheet' found on your desks. This sheet is for you to keep.

Do you have any questions?

You will also find an informed consent form on your desk. Could you kindly read and sign this form please? I am required to provide this information and ask you to sign a consent form for ethical purposes. This is required by the Loughborough University Ethical Advisory Committee. Once you have completed the forms could you all pass them forward please?

Count number of forms and check that all of them have been completed appropriately.

Hand out the answer sheets.

Please print your name in the space provided at the top of your answer sheet, surname first, then your first names.

Now fill in today's date which is the Please ignore the reference number. Please do not use pens on the answer sheets, only use the pencils provided.

Hand out the numerical test booklets. Please don't open these booklets until I tell you to do so.

When all booklets have been handed out: Now please turn to pages 2 and 3 of the booklet and follow the instructions while I read them aloud.

In this test you will be using facts and figures taken from the world of work, to assess your ability to interpret and evaluate numerical information. All the information you need will be provided for you in the form of various tables and charts.

For each question you are given five answers to choose from. One, and only one, of the answers is correct in each case.

Do not write anything on this question booklet, but indicate your answer by completely filling in the appropriate circle on your answer sheet.

You may wish to use a calculator for this test. Rough paper is also provided to help you with any working out.

Before you start there are some example questions to complete. Look at these example questions below. Now, in your own time and using just the statistical information provided on the opposite page, choose what you think is the best answer for each question from the five alternatives given. Mark your choices in the Examples Section on your answer sheet.

When all have finished the examples, say:

The answer to Example 1 is B, 36 Older Adults.

The answer to Example 2 is C, 200, 000 tyres.

The answer to example 3 is D, 53.7%.

Are there any questions? If you are not clear about what you have to do, ask me now as I can't answer any questions once the test has started.

Now, I have handed out a short questionnaire which I would like you to complete. Please don't forget to print your name at the top of this questionnaire.

Ask if everyone's ready and when they are...

Could you all pass your questionnaires forward please?

Count questionnaires and make sure that everyone has written their names on it.

Now could you proceed to page 5 and follow whilst I read out the reminder points please?

- The test consists of 12 questions and you will have 15 minutes in which to do them.
- Try to work quickly but accurately.
- Fill in completely the appropriate circles on the answer sheet.
- Do not make any marks on this booklet.
- Use the rough paper provided for your working out.
- Fully erase any answer you wish to change.
- If you are not sure of an answer mark your best choice, but avoid simply guessing.

Are there any final questions?

Now turn over to page 6 and begin.

Start stopwatch on the word 'begin', and note the start time on a piece of paper. Walk around after 2 minutes and again after 10 minutes. After exactly 17 minutes say:

Please stop now. Pencils down. Close your booklets.

Collect in all the booklets and answer sheets. Check that there are 10 booklets and 10 answer sheets, and check that all answer sheets have names on them.

Could you kindly complete this short questionnaire I'm handing out please?

Hand out post-test questionnaires.

Please don't forget to write your names down on the first page of the questionnaires.

Has everyone completed the questionnaire? Can you all pass your questionnaires forward please?

Have you got any questions?

So that was the first half of the session. We're going to go through exactly the same process

Hand out the answer sheets.

for the second aptitude test.

Please print your name in the space provided at the top of your answer sheet, surname first, then your first names.

Now fill in today's date which is theOnce again, please ignore the reference number.

Hand out the verbal test booklets. Please don't open these booklets until I tell you to do so.

When all booklets have been handed out: Now please turn to pages 2 and 3 of the booklet and follow the instructions while I read them aloud.

In this test you will be asked to reason with written information drawn from the world of work. The test consists of a series of passages, each of which is followed by several statements. Your task is to assess whether each statement is either "true", "false", or "cannot say" in relation to the passage which preceded it.

So having read a passage and the statements which follow, you would;

fill in circle A if you think the statement is true in relation to the passage,

fill in *circle B* if you think the statement is *false* in relation to the passage, fill in *circle C* if you *cannot say* whether the statement is true or false in relation to the passage.

You should assess each statement *only in relation to the passage which preceded it* and not in relation to any personal views or knowledge you might hold.

Do not write anything on this question booklet, but indicate your answer by completely filling in the appropriate circle on your answer sheet.

Before you start there are some example questions to complete. Look at the passage on the opposite page. Now, in your own time, read the passage and assess the statements which follow, according to the rules above. Choose what you think is the best answer for each statement, A, B, or C and mark your choice in the Examples Section on your answer sheet.

Hand out self-efficacy questionnaires. When everyone has finished the examples:

The answer to Example 1 is B; the statement is false given the information in the passage.

The answer to Example 2 is A, the statement is true given the information in the passage.

The answer to Example 3 is C, you cannot say from the information in the passage.

The answer to Example 4 is A, the statement is true given the information in the passage.

Are there any questions? If you are not clear about what you have to do, ask me now as I can't answer any questions once the test has started.

I have handed out a short questionnaire which I would like you to complete. Please don't forget to print your name at the top of this questionnaire.

Ask if everyone's ready and when they are...

Could you all pass your questionnaires forward please?

Count questionnaires and make sure that everyone has written their names on it.

Now could you look back at page 2 and follow whilst I read out the reminder points please?

- The test consists of 24 questions and you will have 14 minutes in which to do them.
- Try to work quickly but accurately.

- Fill in completely the appropriate circles on the answer sheet.
- Do not make any marks on this booklet.
- Fully erase any answer you wish to change.
- If you are not sure of an answer mark your best choice, but avoid simply guessing.

Are there any final questions? Now turn over to page 5 and begin.

Start stopwatch on the word 'begin', and note the start time on a piece of paper. Walk around after 2 minutes and again after 10 minutes. After exactly 15 minutes say:

Please stop now. Pencils down. Close your booklets.

Collect in all the booklets and answer sheets. Check that there are 10 booklets and 10 answer sheets, and check that all answer sheets have names on them.

Could you kindly complete this short questionnaire I'm handing out please?

Hand out post-test questionnaires.

Please don't forget to write your names down on the first page of the questionnaires.

Collect calculators and place a BS pen on each desk.

Has everyone completed the questionnaire? Can you all pass your questionnaires forward please? Collect questionnaires, pencils and erasers. Check that they all wrote their names on the questionnaire.

Have you got any questions?

I will be asking you to come in again in approximately one month's time to complete the tests again and then I will give you personal feedback regarding your performance on the tests, your goal orientations, personality and advice on how to improve on these tests. That's it for today. I have placed a pen on each of your desks. That is for you to keep as a token of my appreciation.

Thank you so much for your participation, I really appreciate it!!

APPENDIX H TIME E1 POST-TEST QUESTIONNAIRE

NAME:							
Your experience on this	s kind of task						
1.) Have you ever completed a numerical aptitude test?							
Yes	No Do not remember						
2.) Does your work invo just completed?	lve using the skills requi	ired to perform well on t	he test				
Frequently 🗌	Sometimes 🗌	Rarely	Never 🗌				
3.) Do you practice this kind of task often?							
Frequently Sometimes Rarely Never							

How much effort did I invest in the task? Please circle the most appropriate answer keeping in mind the task that you have just completed.

In solving the preceding task I invested:

- 1: very, very low mental effort
- 2: very low mental effort
- 3: low mental effort
- **4:** rather low mental effort
- **5**: neither low nor high mental effort
- 6: rather high mental effort
- 7: high mental effort
- 8: very high mental effort
- **9**: very, very high mental effort

How I approached the task: The following items refer to the task that you have just carried out. Please answer these items keeping in mind the approach that you used for the task that you have just completed.

Some of the questions may look similar to each other but please do not worry about appearing to be consistent. Answer each question on its own merits.

Please circle **ONE** answer for each of the items below.

		Strongly Disagree	Disagree	Neither Agree Nor Disagree	Agree	Strongly Agree
1	My aim was to completely master this task.	1	2	3	4	5
2	I was striving to do well on this task compared to the other participants.	1	2	3	4	5
3	My goal was to learn as much as possible.	1	2	3	4	5
4	My aim was to perform well on this task relative to the other participants.	1	2	3	4	5
5	My aim was to avoid learning less than the maximum possible about this task.	1	2	3	4	5
6	My goal was to avoid performing poorly compared to others on this task.	1	2	3	4	5
7	I was striving to understand the content of this task as thoroughly as possible.	1	2	3	4	5
8	My goal was to perform better on this task than other participants.	1	2	3	4	5
9	My goal was to avoid learning less than it was possible to learn about this task.	1	2	3	4	5
10	I was striving to avoid performing worse than others on this task.	1	2	3	4	5
11	I was striving to avoid an incomplete understanding of this task.	1	2	3	4	5
12	My aim was to avoid doing worse than the other participants on this task.	1	2	3	4	5

NAME:							
Your experience on this kind of task							
1.) Have you ever comp	bleted a verbal aptitude	test?					
Yes	No	Do not remer	nber 🗌				
2.) Does your work involve using the skills required to perform well on the test just completed?							
Frequently	Sometimes 🗌	Rarely	Never				
3.) Do you practice this kind of task often?							
Frequently Sometimes Rarely Never							

How much effort did I invest in the task? Please circle the most appropriate answer keeping in mind the task that you have just completed.

In solving the preceding task I invested:

- 1: very, very low mental effort
- 2: very low mental effort
- 3: low mental effort
- **4:** rather low mental effort
- **5:** neither low nor high mental effort
- **6:** rather high mental effort
- 7: high mental effort
- 8: very high mental effort
- 9: very, very high mental effort

How I approached the task: The following items refer to the task that you have just carried out. Please answer these items keeping in mind the approach that you used for the task that you have just completed.

Some of the questions may look similar to each other but please do not worry about appearing to be consistent. Answer each question on its own merits.

Please circle **ONE** answer for each of the items below.

		Strongly Disagree	Disagree	Neither Agree Nor Disagree	Agree	Strongly Agree
1	My aim was to completely master this task.	1	2	3	4	5
2	I was striving to do well on this task compared to the other participants.	1	2	3	4	5
3	My goal was to learn as much as possible.	1	2	3	4	5
4	My aim was to perform well on this task relative to the other participants.	1	2	3	4	5
5	My aim was to avoid learning less than the maximum possible about this task.	1	2	3	4	5
6	My goal was to avoid performing poorly compared to others on this task.	1	2	3	4	5
7	I was striving to understand the content of this task as thoroughly as possible.	1	2	3	4	5
8	My goal was to perform better on this task than other participants.	1	2	3	4	5
9	My goal was to avoid learning less than it was possible to learn about this task.	1	2	3	4	5
10	I was striving to avoid performing worse than others on this task.	1	2	3	4	5
11	I was striving to avoid an incomplete understanding of this task.	1	2	3	4	5
12	My aim was to avoid doing worse than the other participants on this task.	1	2	3	4	5

APPENDIX I TIME E2 EXPERIMENT INSTRUCTIONS

Neutral Task Instructions – Time 2

Prepare an information sheet and consent form on each desk before start of session. Also put 2 pencils, an eraser, a calculator, and some rough paper on each desk.

First of all I would like to thank you for attending this session. The session will take 1 hour. Could you all switch off you mobile phones please?

First I would like you to read the paper with the title 'information sheet' found on your desks. This sheet is for you to keep.

Do you have any questions?

You will also find an informed consent form on your desk. Could you kindly read and sign this form please? I am required to provide this information and ask you to sign a consent form for ethical purposes. This is required by the Loughborough University Ethical Advisory Committee. Once you have completed the forms could you all pass them forward please?

Count number of forms and check that all of them have been completed appropriately.

Hand out the answer sheets.

Hand out the verbal test booklets. Please don't open these booklets until I tell you to do so.

When all booklets have been handed out: Now please turn to page 2 of the booklet and follow the instructions while I read them aloud.

In this test you will be asked to reason with written information drawn from the world of work. The test consists of a series of passages, each of which is followed by several statements. Your task is to assess whether each statement is either "true", "false", or "cannot say" in relation to the passage which preceded it.

So having read a passage and the statements which follow, you would;

fill in circle A if you think the statement is true in relation to the passage,

fill in *circle B* if you think the statement is *false* in relation to the passage,

fill in *circle C* if you *cannot say* whether the statement is true or false in relation to the passage.

You should assess each statement *only in relation to the passage which preceded it* and not in relation to any personal views or knowledge you might hold.

Do not write anything on this question booklet, but indicate your answer by completely filling in the appropriate circle on your answer sheet.

Before you start there are some example questions to complete. Look at the passage page 11. Now, in your own time, read the passage and assess the statements which follow, according to the rules above. Choose what you think is the best answer for each statement, A, B, or C and mark your choice in the Examples Section on your answer sheet.

Hand out self-efficacy questionnaires. When everyone has finished the examples:
The answer to Example 1 is A, the statement is true given the information in the passage.

The answer to Example 2 is C, you cannot say from the information in the passage.

The answer to Example 3 is A, the statement is true given the information in the passage.

The answer to Example 4 is C, you cannot say from the information in the passage.

Are there any questions? If you are not clear about what you have to do, ask me now as I can't answer any questions once the test has started.

I have handed out a short questionnaire which I would like you to complete. Please don't forget to print your name at the top of this questionnaire.

Ask if everyone's ready and when they are...

Could you all pass your questionnaires forward please?

Count questionnaires and make sure that everyone has written their names on it.

Now could you look back at page 2 and follow whilst I read out the reminder points please?

- The test consists of 24 questions and you will have 14 minutes in which to do them.
- Try to work quickly but accurately.
- Fill in completely the appropriate circles on the answer sheet.
- Do not make any marks on this booklet.
- Fully erase any answer you wish to change.
- If you are not sure of an answer mark your best choice, but avoid simply guessing.

Are there any final questions?

Please note that this time the 1st test question is Question 29. So please start using your answer sheets from Question 29.

Now turn over to page 12 and begin.

Start stopwatch on the word 'begin', and note the start time on a piece of paper. Walk around after 2 minutes and again after 10 minutes. After exactly 14 minutes say:

Please stop now. Pencils down. Close your booklets.

Collect in all the materials, booklets, answer sheets, pencils and erasers. Check that there are 10 booklets and 10 answer sheets, and check that all answer sheets have names on them.

Could you kindly complete this short questionnaire I'm handing out please?

Hand out post-test questionnaires.

Please don't forget to write your names down on the first page of the questionnaires.

Has everyone completed the questionnaire? Can you all pass your questionnaires forward please?

So that was the first half of the session. We're going to go through exactly the same process for the second aptitude test.

Hand out the answer sheets.

Hand out the numerical test booklets. Please don't open these booklets until I tell you to do so.

When all booklets have been handed out: Now please turn to page 2 of the booklet and follow the instructions while I read them aloud.

In this test you will be using facts and figures taken from the world of work, to assess your ability to interpret and evaluate numerical information. All the information you need will be provided for you in the form of various tables and charts.

For each question you are given five answers to choose from. One, and only one, of the answers is correct in each case.

Do not write anything on this question booklet, but indicate your answer by completely filling in the appropriate circle on your answer sheet.

You may wish to use a calculator for this test. Rough paper is also provided to help you with any working out.

Before you start there are some example questions to complete. Look at the example questions on page 8. Now, in your own time and using just the statistical information provided at the top of the page, choose what you think is the best answer for each question from the five alternatives given. Mark your choices in the Examples Section on your answer sheet.

When all have finished the examples, say:

The answer to Example 1 is A, Year 1.

The answer to Example 2 is D, £144, 000.

The answer to example 3 is B, £5, 497, 000.

Are there any questions? If you are not clear about what you have to do, ask me now as I can't answer any questions once the test has started.

Now, I have handed out a short questionnaire which I would like you to complete. Please don't forget to print your name at the top of this questionnaire.

Ask if everyone's ready and when they are...

Could you all pass your questionnaires forward please?

Count questionnaires and make sure that everyone has written their names on it.

Now could you look back at page 2 and follow whilst I read out the reminder points please?

- The test consists of 12 questions and you will have 15 minutes in which to do them.
- Try to work quickly but accurately.
- Fill in completely the appropriate circles on the answer sheet.
- Do not make any marks on this booklet.
- Use the rough paper provided for your working out.
- Fully erase any answer you wish to change.
- If you are not sure of an answer mark your best choice, but avoid simply guessing.

Are there any final questions? Please note that this time the first test question is Question 19. So please start using your answer sheets from Question 19.

Now turn over to page 9 and begin.

Start stopwatch on the word 'begin', and note the start time on a piece of paper. Walk around after 2 minutes and again after 10 minutes. After exactly 15 minutes say:

Please stop now. Pencils down. Close your booklets.

Collect in all the booklets and answer sheets. Check that there are 10 booklets and 10 answer sheets, and check that all answer sheets have names on them.

Could you kindly complete this short questionnaire I'm handing out please?

Hand out post-test questionnaires.

Please don't forget to write your names down on the first page of the questionnaires.

Collect calculators.

Has everyone completed the questionnaire? Can you all pass your questionnaires forward please? Collect questionnaires, pencils and erasers. Check that they all wrote their names on the questionnaire.

Have you got any questions?

Once I've scored all the tests and questionnaires I will send you all a feedback pack by email. I would be very happy to provide you with a personal feedback session if you like. So if you would like to attend a personal feedback session could you kindly drop me an e-mail and I'll make sure that I set one up for you. I will also be sending you a final short questionnaire by e-mail in about 1 month's time and I would really really appreciate it of you could complete it and send it back by e-mail. It's very very short, but it is also very important for the research that I am doing.

Also, the draw for the three £50 cash prizes will take place in December or January and you'll obviously be informed if you've won. Please note that your name will be put into the draw once you've completed the final short questionnaire that I will be sending you in 1 month's time.

Once again, thank you so much for your participation, I really appreciate it !!

Neutral Task Instructions – Time 2

Prepare an information sheet and consent form on each desk before start of session. Also put 2 pencils, an eraser, a calculator, and some rough paper on each desk.

First of all I would like to thank you for attending this session. The session will take 1 hour. Could you all switch off you mobile phones please?

First I would like you to read the paper with the title 'information sheet' found on your desks. This sheet is for you to keep.

Do you have any questions?

You will also find an informed consent form on your desk. Could you kindly read and sign this form please? I am required to provide this information and ask you to sign a consent form for ethical purposes. This is required by the Loughborough University Ethical Advisory Committee. Once you have completed the forms could you all pass them forward please?

Count number of forms and check that all of them have been completed appropriately.

Hand out the answer sheets.

Hand out the numerical test booklets. Please don't open these booklets until I tell you to do so.

When all booklets have been handed out: Now please turn to page 2 of the booklet and follow the instructions while I read them aloud.

In this test you will be using facts and figures taken from the world of work, to assess your ability to interpret and evaluate numerical information. All the information you need will be provided for you in the form of various tables and charts.

For each question you are given five answers to choose from. One, and only one, of the answers is correct in each case.

Do not write anything on this question booklet, but indicate your answer by completely filling in the appropriate circle on your answer sheet.

You may wish to use a calculator for this test. Rough paper is also provided to help you with any working out.

Before you start there are some example questions to complete. Look at the example questions on page 8. Now, in your own time and using just the statistical information provided at the top of the page, choose what you think is the best answer for each question from the five alternatives given. Mark your choices in the Examples Section on your answer sheet.

When all have finished the examples, say:

The answer to Example 1 is A, Year 1.

The answer to Example 2 is D, £144, 000.

The answer to example 3 is B, £5, 497, 000.

Are there any questions? If you are not clear about what you have to do, ask me now as I can't answer any questions once the test has started.

Now, I have handed out a short questionnaire which I would like you to complete. Please don't forget to print your name at the top of this questionnaire.

Ask if everyone's ready and when they are...

Could you all pass your questionnaires forward please?

Count questionnaires and make sure that everyone has written their names on it.

Now could you look back at page 2 and follow whilst I read out the reminder points please?

- The test consists of 12 questions and you will have 15 minutes in which to do them.
- Try to work quickly but accurately.
- Fill in completely the appropriate circles on the answer sheet.
- Do not make any marks on this booklet.
- Use the rough paper provided for your working out.
- Fully erase any answer you wish to change.
- If you are not sure of an answer mark your best choice, but avoid simply guessing.

Are there any final questions? Please note that this time the first test question is Question 19. So please start using your answer sheets from Question 19.

Now turn over to page 9 and begin.

Start stopwatch on the word 'begin', and note the start time on a piece of paper. Walk around after 2 minutes and again after 10 minutes. After exactly 15 minutes say:

Please stop now. Pencils down. Close your booklets.

Collect in all the booklets and answer sheets. Check that there are 10 booklets and 10 answer sheets, and check that all answer sheets have names on them.

Could you kindly complete this short questionnaire I'm handing out please?

Hand out post-test questionnaires.

Please don't forget to write your names down on the first page of the questionnaires.

Has everyone completed the questionnaire? Can you all pass your questionnaires forward please?

So that was the first half of the session. We're going to go through exactly the same process for the second aptitude test.

Hand out the answer sheets.

Hand out the verbal test booklets. Please don't open these booklets until I tell you to do so.

When all booklets have been handed out: Now please turn to page 2 of the booklet and follow the instructions while I read them aloud.

In this test you will be asked to reason with written information drawn from the world of work. The test consists of a series of passages, each of which is followed by several statements. Your task is to assess whether each statement is either "true", "false", or "cannot say" in relation to the passage which preceded it.

So having read a passage and the statements which follow, you would;

fill in *circle A* if you think the statement is *true* in relation to the passage, fill in *circle B* if you think the statement is *false* in relation to the passage, fill in *circle C* if you *cannot say* whether the statement is true or false in relation to the passage.

You should assess each statement *only in relation to the passage which preceded it* and not in relation to any personal views or knowledge you might hold.

Do not write anything on this question booklet, but indicate your answer by completely filling in the appropriate circle on your answer sheet.

Before you start there are some example questions to complete. Look at the passage page 11. Now, in your own time, read the passage and assess the statements which follow, according to the rules above. Choose what you think is the best answer for each statement, A, B, or C and mark your choice in the Examples Section on your answer sheet.

Hand out self-efficacy questionnaires. When everyone has finished the examples:

The answer to Example 1 is A, the statement is true given the information in the passage.

The answer to Example 2 is C, you cannot say from the information in the passage.

The answer to Example 3 is A, the statement is true given the information in the passage.

The answer to Example 4 is C, you cannot say from the information in the passage.

Are there any questions? If you are not clear about what you have to do, ask me now as I can't answer any questions once the test has started.

I have handed out a short questionnaire which I would like you to complete. Please don't forget to print your name at the top of this questionnaire.

Ask if everyone's ready and when they are...

Could you all pass your questionnaires forward please?

Count questionnaires and make sure that everyone has written their names on it.

Now could you look back at page 2 and follow whilst I read out the reminder points please?

- The test consists of 24 questions and you will have 14 minutes in which to do them.
- Try to work quickly but accurately.
- Fill in completely the appropriate circles on the answer sheet.
- Do not make any marks on this booklet.
- Fully erase any answer you wish to change.
- If you are not sure of an answer mark your best choice, but avoid simply guessing.

Are there any final questions?

Please note that this time the 1st test question is Question 29. So please start using your answer sheets from Question 29.

Now turn over to page 12 and begin.

Start stopwatch on the word 'begin', and note the start time on a piece of paper. Walk around after 2 minutes and again after 10 minutes. After exactly 14 minutes say:

Please stop now. Pencils down. Close your booklets.

Collect in all the materials, booklets, answer sheets, pencils and erasers. Check that there are 10 booklets and 10 answer sheets, and check that all answer sheets have names on them.

Could you kindly complete this short questionnaire I'm handing out please?

Hand out post-test questionnaires.

Please don't forget to write your names down on the first page of the questionnaires.

Collect calculators.

Has everyone completed the questionnaire? Can you all pass your questionnaires forward please? Collect questionnaires, pencils and erasers. Check that they all wrote their names on the questionnaire.

Have you got any questions?

Once I've scored all the tests and questionnaires I will send you all a feedback pack by email. I would be very happy to provide you with a personal feedback session if you like. So if you would like to attend a personal feedback session could you kindly drop me an e-mail and I'll make sure that I set one up for you. I will also be sending you a final short questionnaire by e-mail in about 1 month's time and I would really really appreciate it of you could complete it and send it back by e-mail. It's very very short, but it is also very important for the research that I am doing.

Also, the draw for the three £50 cash prizes will take place in December or January and you'll obviously be informed if you've won. Please note that your name will be put into the draw once you've completed the final short questionnaire that I will be sending you in 1 month's time.

Once again, thank you so much for your participation, I really appreciate it!!

Learning Induction Task Instructions – Time 2

Prepare an information sheet and consent form on each desk before start of session. Also put 2 pencils, an eraser, a calculator, and some rough paper on each desk.

First of all I would like to thank you for attending this session. The session will take 1 hour. Could you all switch off you mobile phones please?

First I would like you to read the paper with the title 'information sheet' found on your desks. This sheet is for you to keep.

Do you have any questions?

You will also find an informed consent form on your desk. Could you kindly read and sign this form please? I am required to provide this information and ask you to sign a consent form for ethical purposes. This is required by the Loughborough University Ethical Advisory Committee. Once you have completed the forms could you all pass them forward please?

Count number of forms and check that all of them have been completed appropriately.

Hand out the answer sheets.

Hand out the verbal test booklets. Please don't open these booklets until I tell you to do so.

When all booklets have been handed out: Now please turn to page 2 of the booklet and follow the instructions while I read them aloud.

In this test you will be asked to reason with written information drawn from the world of work. The test consists of a series of passages, each of which is followed by several statements. Your task is to assess whether each statement is either "true", "false", or "cannot say" in relation to the passage which preceded it.

So having read a passage and the statements which follow, you would;

fill in *circle A* if you think the statement is *true* in relation to the passage, fill in *circle B* if you think the statement is *false* in relation to the passage, fill in *circle C* if you *cannot say* whether the statement is true or false in relation to the passage.

You should assess each statement *only in relation to the passage which preceded it* and not in relation to any personal views or knowledge you might hold.

Do not write anything on this question booklet, but indicate your answer by completely filling in the appropriate circle on your answer sheet.

Before you start there are some example questions to complete. Look at the passage on page 11. Now, in your own time, read the passage and assess the statements which follow, according to the rules above. Choose what you think is the best answer for each statement, A, B, or C and mark your choice in the Examples Section on your answer sheet.

Hand out self-efficacy questionnaires. When everyone has finished the examples:

The answer to Example 1 is A, the statement is true given the information in the passage.

The answer to Example 2 is C, you cannot say from the information in the passage.

The answer to Example 3 is A, the statement is true given the information in the passage.

The answer to Example 4 is C, you cannot say from the information in the passage.

Are there any questions? If you are not clear about what you have to do, ask me now as I can't answer any questions once the test has started.

I have handed out a short questionnaire which I would like you to complete. Please don't forget to print your name at the top of this questionnaire.

Ask if everyone's ready and when they are...

Could you all pass your questionnaires forward please?

Count questionnaires and make sure that everyone has written their names on it.

Now could you look back at page 2 and follow whilst I read out the reminder points please?

- The test consists of 24 questions and you will have 14 minutes in which to do them.
- Try to work quickly but accurately.
- Fill in completely the appropriate circles on the answer sheet.
- Do not make any marks on this booklet.
- Fully erase any answer you wish to change.
- If you are not sure of an answer mark your best choice, but avoid simply guessing.

Are there any final questions?

Please note that this time the 1st test question is Question 29. So please start using your answer sheets from Question 29.

It is important that you keep in mind that this session is a chance for you to practice aptitude tests for when you apply for employment. So it is important that you use this opportunity to learn more about what it is like to take a verbal/numerical aptitude test. The results on these tests are not as important as the opportunity to learn more about aptitude tests.

Now turn over to page 12 and begin.

Start stopwatch on the word 'begin', and note the start time on a piece of paper. Walk around after 2 minutes and again after 10 minutes. After exactly 14 minutes say:

Please stop now. Pencils down. Close your booklets.

Collect in all the materials, booklets, answer sheets, pencils and erasers. Check that there are 10 booklets and 10 answer sheets, and check that all answer sheets have names on them.

Could you kindly complete this short questionnaire I'm handing out please?

Hand out post-test questionnaires.

Please don't forget to write your names down on the first page of the questionnaires.

Has everyone completed the questionnaire? Can you all pass your questionnaires forward please?

So that was the first half of the session. We're going to go through exactly the same process for the second aptitude test.

Hand out the answer sheets.

Hand out the numerical test booklets. Please don't open these booklets until I tell you to do so.

When all booklets have been handed out: Now please turn to pages 2 of the booklet and follow the instructions while I read them aloud.

In this test you will be using facts and figures taken from the world of work, to assess your ability to interpret and evaluate numerical information. All the information you need will be provided for you in the form of various tables and charts.

For each question you are given five answers to choose from. One, and only one, of the answers is correct in each case.

Do not write anything on this question booklet, but indicate your answer by completely filling in the appropriate circle on your answer sheet.

You may wish to use a calculator for this test. Rough paper is also provided to help you with any working out.

Before you start there are some example questions to complete. Look at the example questions on page 8. Now, in your own time and using just the statistical information provided at the top of the page, choose what you think is the best answer for each question from the five alternatives given. Mark your choices in the Examples Section on your answer sheet.

When all have finished the examples, say:

The answer to Example 1 is A, Year 1.

The answer to Example 2 is D, £144, 000.

The answer to example 3 is B, £5, 497, 000.

Are there any questions? If you are not clear about what you have to do, ask me now as I can't answer any questions once the test has started.

Now, I have handed out a short questionnaire which I would like you to complete. Please don't forget to print your name at the top of this questionnaire.

Ask if everyone's ready and when they are...

Could you all pass your questionnaires forward please?

Count questionnaires and make sure that everyone has written their names on it.

Now could you look back at page 2 and follow whilst I read out the reminder points please?

- The test consists of 12 questions and you will have 15 minutes in which to do them.
- Try to work quickly but accurately.
- Fill in completely the appropriate circles on the answer sheet.
- Do not make any marks on this booklet.
- Use the rough paper provided for your working out.
- Fully erase any answer you wish to change.
- If you are not sure of an answer mark your best choice, but avoid simply guessing.

Are there any final questions?

Please note that this time the first test question is Question 19. So please start using your answer sheets from Question 19.

Again, I would like to remind you that it is important that you keep in mind that this session is a chance for you to practice aptitude tests for when you apply for employment. So it is important that you use this opportunity to learn more about what it is like to take a verbal/numerical aptitude test. The results on these tests are not as important as the opportunity to learn more about aptitude tests.

Now turn over to page 9 and begin.

Start stopwatch on the word 'begin', and note the start time on a piece of paper. Walk around after 2 minutes and again after 10 minutes. After exactly 15 minutes say:

Please stop now. Pencils down. Close your booklets.

Collect in all the booklets and answer sheets. Check that there are 10 booklets and 10 answer sheets, and check that all answer sheets have names on them.

Could you kindly complete this short questionnaire I'm handing out please?

Hand out post-test questionnaires.

Please don't forget to write your names down on the first page of the questionnaires.

Collect calculators.

Has everyone completed the questionnaire? Can you all pass your questionnaires forward please? Collect questionnaires, pencils and erasers. Check that they all wrote their names on the questionnaire.

Have you got any questions?

Once I've scored all the tests and questionnaires I will send you all a feedback pack by email. I would be very happy to provide you with a personal feedback session if you like. So if you would like to attend a personal feedback session could you kindly drop me an e-mail and I'll make sure that I set one up for you. I will also be sending you a final short questionnaire by e-mail in about 1 month's time and I would really really appreciate it of you could complete it and send it back by e-mail. It's very very short, but it is also very important for the research that I am doing.

Also, the draw for the three £50 cash prizes will take place in December or January and you'll obviously be informed if you've won. Please note that your name will be put into the draw once you've completed the final short questionnaire that I will be sending you in 1 month's time.

Once again, thank you so much for your participation, I really appreciate it!!

Learning Induction Task Instructions – Time 2

Prepare an information sheet and consent form on each desk before start of session. Also put 2 pencils, an eraser, a calculator, and some rough paper on each desk.

First of all I would like to thank you for attending this session. The session will take 1 hour. Could you all switch off you mobile phones please?

First I would like you to read the paper with the title 'information sheet' found on your desks. This sheet is for you to keep.

Do you have any questions?

You will also find an informed consent form on your desk. Could you kindly read and sign this form please? I am required to provide this information and ask you to sign a consent form for ethical purposes. This is required by the Loughborough University Ethical Advisory Committee. Once you have completed the forms could you all pass them forward please?

Count number of forms and check that all of them have been completed appropriately.

Hand out the answer sheets.

Hand out the numerical test booklets. Please don't open these booklets until I tell you to do so.

When all booklets have been handed out: Now please turn to page 2 of the booklet and follow the instructions while I read them aloud.

In this test you will be using facts and figures taken from the world of work, to assess your ability to interpret and evaluate numerical information. All the information you need will be provided for you in the form of various tables and charts.

For each question you are given five answers to choose from. One, and only one, of the answers is correct in each case.

Do not write anything on this question booklet, but indicate your answer by completely filling in the appropriate circle on your answer sheet.

You may wish to use a calculator for this test. Rough paper is also provided to help you with any working out.

Before you start there are some example questions to complete. Look at the example questions on page 8. Now, in your own time and using just the statistical information provided at the top of the page, choose what you think is the best answer for each question from the five alternatives given. Mark your choices in the Examples Section on your answer sheet.

When all have finished the examples, say:

The answer to Example 1 is A, Year 1.

The answer to Example 2 is D, £144, 000.

The answer to example 3 is B, £5, 497, 000.

Are there any questions? If you are not clear about what you have to do, ask me now as I can't answer any questions once the test has started.

Now, I have handed out a short questionnaire which I would like you to complete. Please don't forget to print your name at the top of this questionnaire.

Ask if everyone's ready and when they are...

Could you all pass your questionnaires forward please?

Count questionnaires and make sure that everyone has written their names on it.

Now could you look back at page 2 and follow whilst I read out the reminder points please?

- The test consists of 12 questions and you will have 15 minutes in which to do them.
- Try to work quickly but accurately.
- Fill in completely the appropriate circles on the answer sheet.
- Do not make any marks on this booklet.
- Use the rough paper provided for your working out.
- Fully erase any answer you wish to change.
- If you are not sure of an answer mark your best choice, but avoid simply guessing.

Are there any final questions? Please note that this time the first test question is Question 19. So please start using your answer sheets from Question 19.

It is important that you keep in mind that this session is a chance for you to practice aptitude tests for when you apply for employment. So it is important that you use this opportunity to learn more about what it is like to take a verbal/numerical aptitude test. The results on these tests are not as important as the opportunity to learn more about aptitude tests.

Now turn over to page 9 and begin.

Start stopwatch on the word 'begin', and note the start time on a piece of paper. Walk around after 2 minutes and again after 10 minutes. After exactly 15 minutes say:

Please stop now. Pencils down. Close your booklets.

Collect in all the booklets and answer sheets. Check that there are 10 booklets and 10 answer sheets, and check that all answer sheets have names on them.

Could you kindly complete this short questionnaire I'm handing out please?

Hand out post-test questionnaires.

Please don't forget to write your names down on the first page of the questionnaires.

Has everyone completed the questionnaire? Can you all pass your questionnaires forward please?

So that was the first half of the session. We're going to go through exactly the same process for the second aptitude test.

Hand out the answer sheets.

Hand out the verbal test booklets. Please don't open these booklets until I tell you to do so.

When all booklets have been handed out: Now please turn to page 2 of the booklet and follow the instructions while I read them aloud.

In this test you will be asked to reason with written information drawn from the world of work. The test consists of a series of passages, each of which is followed by several statements. Your task is to assess whether each statement is either "true", "false", or "cannot say" in relation to the passage which preceded it.

So having read a passage and the statements which follow, you would;

fill in circle A if you think the statement is true in relation to the passage,

fill in *circle B* if you think the statement is *false* in relation to the passage,

fill in *circle C* if you *cannot say* whether the statement is true or false in relation to the passage.

You should assess each statement *only in relation to the passage which preceded it* and not in relation to any personal views or knowledge you might hold.

Do not write anything on this question booklet, but indicate your answer by completely filling in the appropriate circle on your answer sheet.

Before you start there are some example questions to complete. Look at the passage page 11. Now, in your own time, read the passage and assess the statements which follow, according to the rules above. Choose what you think is the best answer for each statement, A, B, or C and mark your choice in the Examples Section on your answer sheet.

Hand out self-efficacy questionnaires. When everyone has finished the examples:

The answer to Example 1 is A, the statement is true given the information in the passage.

The answer to Example 2 is C, you cannot say from the information in the passage.

The answer to Example 3 is A, the statement is true given the information in the passage.

The answer to Example 4 is C, you cannot say from the information in the passage.

Are there any questions? If you are not clear about what you have to do, ask me now as I can't answer any questions once the test has started.

I have handed out a short questionnaire which I would like you to complete. Please don't forget to print your name at the top of this questionnaire.

Ask if everyone's ready and when they are...

Could you all pass your questionnaires forward please?

Count questionnaires and make sure that everyone has written their names on it.

Now could you look back at page 2 and follow whilst I read out the reminder points please?

- The test consists of 24 questions and you will have 14 minutes in which to do them.
- Try to work quickly but accurately.
- Fill in completely the appropriate circles on the answer sheet.
- Do not make any marks on this booklet.
- Fully erase any answer you wish to change.
- If you are not sure of an answer mark your best choice, but avoid simply guessing.

Are there any final questions?

Please note that this time the 1st test question is Question 29. So please start using your answer sheets from Question 29.

Again, I would like to remind you that it is important to keep in mind that this session is a chance for you to practice aptitude tests for when you apply for employment. So it is important that you use this opportunity to learn more about what it is like to take a verbal/numerical aptitude test. The results on these tests are not as important as the opportunity to learn more about aptitude tests.

Now turn over to page 12 and begin.

Start stopwatch on the word 'begin', and note the start time on a piece of paper. Walk around after 2 minutes and again after 10 minutes. After exactly 14 minutes say:

Please stop now. Pencils down. Close your booklets.

Collect in all the materials, booklets, answer sheets, pencils and erasers. Check that there are 10 booklets and 10 answer sheets, and check that all answer sheets have names on them.

Could you kindly complete this short questionnaire I'm handing out please?

Hand out post-test questionnaires.

Please don't forget to write your names down on the first page of the questionnaires.

Collect calculators.

Has everyone completed the questionnaire? Can you all pass your questionnaires forward please?

Collect questionnaires, pencils and erasers. Check that they all wrote their names on the questionnaire.

Have you got any questions?

Once I've scored all the tests and questionnaires I will send you all a feedback pack by email. I would be very happy to provide you with a personal feedback session if you like. So if you would like to attend a personal feedback session could you kindly drop me an e-mail and I'll make sure that I set one up for you. I will also be sending you a final short questionnaire by e-mail in about 1 month's time and I would really really appreciate it of you could complete it and send it back by e-mail. It's very very short, but it is also very important for the research that I am doing.

Also, the draw for the three £50 cash prizes will take place in December or January and you'll obviously be informed if you've won. Please note that your name will be put into the draw once you've completed the final short questionnaire that I will be sending you in 1 month's time.

Once again, thank you so much for your participation, I really appreciate it !!

Performance-Approach Inducing Task Instructions – Time 2

Prepare an information sheet and consent form on each desk before start of session. Also put 2 pencils, an eraser, a calculator, and some rough paper on each desk.

First of all I would like to thank you for attending this session. The session will take 1 hour. Could you all switch off you mobile phones please?

First I would like you to read the paper with the title 'information sheet' found on your desks. This sheet is for you to keep.

Do you have any questions?

You will also find an informed consent form on your desk. Could you kindly read and sign this form please? I am required to provide this information and ask you to sign a consent form for ethical purposes. This is required by the Loughborough University Ethical Advisory Committee. Once you have completed the forms could you all pass them forward please?

Count number of forms and check that all of them have been completed appropriately.

Hand out the answer sheets.

Hand out the verbal test booklets. Please don't open these booklets until I tell you to do so.

When all booklets have been handed out: Now please turn to page 2 of the booklet and follow the instructions while I read them aloud.

In this test you will be asked to reason with written information drawn from the world of work. The test consists of a series of passages, each of which is followed by several statements. Your task is to assess whether each statement is either "true", "false", or "cannot say" in relation to the passage which preceded it.

So having read a passage and the statements which follow, you would;

fill in circle A if you think the statement is true in relation to the passage,

fill in *circle B* if you think the statement is *false* in relation to the passage,

fill in *circle C* if you *cannot say* whether the statement is true or false in relation to the passage.

You should assess each statement *only in relation to the passage which preceded it* and not in relation to any personal views or knowledge you might hold.

Do not write anything on this question booklet, but indicate your answer by completely filling in the appropriate circle on your answer sheet.

Before you start there are some example questions to complete. Look at the passage page 11. Now, in your own time, read the passage and assess the statements which follow, according to the rules above. Choose what you think is the best answer for each statement, A, B, or C and mark your choice in the Examples Section on your answer sheet.

Hand out self-efficacy questionnaires. When everyone has finished the examples:

The answer to Example 1 is A, the statement is true given the information in the passage.

The answer to Example 2 is C, you cannot say from the information in the passage.

The answer to Example 3 is A, the statement is true given the information in the passage.

The answer to Example 4 is C, you cannot say from the information in the passage.

Are there any questions? If you are not clear about what you have to do, ask me now as I can't answer any questions once the test has started.

I have handed out a short questionnaire which I would like you to complete. Please don't forget to print your name at the top of this questionnaire.

Ask if everyone's ready and when they are...

Could you all pass your questionnaires forward please?

Count questionnaires and make sure that everyone has written their names on it.

Now could you look back at page 2 and follow whilst I read out the reminder points please?

- The test consists of 24 questions and you will have 14 minutes in which to do them.
- Try to work quickly but accurately.
- Fill in completely the appropriate circles on the answer sheet.
- Do not make any marks on this booklet.
- Fully erase any answer you wish to change.
- If you are not sure of an answer mark your best choice, but avoid simply guessing.

Are there any final questions?

Please note that this time the 1st test question is Question 29. So please start using your answer sheets from Question 29.

This time I shall be comparing your performance on the test to the test norms and I shall also be comparing your performance on this test with the performance of other participants of my research. I would like to remind you that the results of this test will ONLY be used for the purpose of this research and will not have any influence whatsoever on your course since the results will be kept in very strict confidentiality.

Now turn over to page 12 and begin.

Start stopwatch on the word 'begin', and note the start time on a piece of paper. Walk around after 2 minutes and again after 10 minutes. After exactly 14 minutes say:

Please stop now. Pencils down. Close your booklets.

Collect in all the materials, booklets, answer sheets, pencils and erasers. Check that there are 10 booklets and 10 answer sheets, and check that all answer sheets have names on them.

Could you kindly complete this short questionnaire I'm handing out please?

Hand out post-test questionnaires.

Please don't forget to write your names down on the first page of the questionnaires.

Has everyone completed the questionnaire? Can you all pass your questionnaires forward please?

So that was the first half of the session. We're going to go through exactly the same process for the second aptitude test.

Hand out the answer sheets.

Hand out the numerical test booklets. Please don't open these booklets until I tell you to do so.

When all booklets have been handed out: Now please turn to pages 2 of the booklet and follow the instructions while I read them aloud.

In this test you will be using facts and figures taken from the world of work, to assess your ability to interpret and evaluate numerical information. All the information you need will be provided for you in the form of various tables and charts.

For each question you are given five answers to choose from. One, and only one, of the answers is correct in each case.

Do not write anything on this question booklet, but indicate your answer by completely filling in the appropriate circle on your answer sheet.

You may wish to use a calculator for this test. Rough paper is also provided to help you with any working out.

Before you start there are some example questions to complete. Look at the example questions on page 8. Now, in your own time and using just the statistical information provided at the top of the page, choose what you think is the best answer for each question from the five alternatives given. Mark your choices in the Examples Section on your answer sheet.

When all have finished the examples, say:

The answer to Example 1 is A, Year 1.

The answer to Example 2 is D, £144, 000.

The answer to example 3 is B, £5, 497, 000.

Are there any questions? If you are not clear about what you have to do, ask me now as I can't answer any questions once the test has started.

Now, I have handed out a short questionnaire which I would like you to complete. Please don't forget to print your name at the top of this questionnaire.

Ask if everyone's ready and when they are...

Could you all pass your questionnaires forward please?

Count questionnaires and make sure that everyone has written their names on it.

Now could you look back at page 2 and follow whilst I read out the reminder points please?

- The test consists of 12 questions and you will have 15 minutes in which to do them.
- Try to work quickly but accurately.
- Fill in completely the appropriate circles on the answer sheet.
- Do not make any marks on this booklet.
- Use the rough paper provided for your working out.
- Fully erase any answer you wish to change.
- If you are not sure of an answer mark your best choice, but avoid simply guessing.

Are there any final questions? Please note that this time the first test question is Question 19. So please start using your answer sheets from Question 19.

Again, I would like you to keep in mind that this time I shall be comparing your performance on the test to the test norms and I shall also be comparing your performance on this test with the performance of other participants of my research. I would like to remind you that the results of this test will ONLY be used for the purpose of this research and will not have any influence whatsoever on your course since the results will be kept in very strict confidentiality.

Now turn over to page 9 and begin.

Start stopwatch on the word 'begin', and note the start time on a piece of paper. Walk around after 2 minutes and again after 10 minutes. After exactly 15 minutes say:

Please stop now. Pencils down. Close your booklets.

Collect in all the booklets and answer sheets. Check that there are 10 booklets and 10 answer sheets, and check that all answer sheets have names on them.

Could you kindly complete this short questionnaire I'm handing out please?

Hand out post-test questionnaires.

Please don't forget to write your names down on the first page of the questionnaires.

Collect calculators.

Has everyone completed the questionnaire? Can you all pass your questionnaires forward please? Collect questionnaires, pencils and erasers. Check that they all wrote their names on the questionnaire.

Have you got any questions?

Once I've scored all the tests and questionnaires I will send you all a feedback pack by email. I would be very happy to provide you with a personal feedback session if you like. So if you would like to attend a personal feedback session could you kindly drop me an e-mail and I'll make sure that I set one up for you. I will also be sending you a final short questionnaire by e-mail in about 1 month's time and I would really really appreciate it of you could complete it and send it back by e-mail. It's very very short, but it is also very important for the research that I am doing.

Also, the draw for the three £50 cash prizes will take place in December or January and you'll obviously be informed if you've won. Please note that your name will be put into the draw once you've completed the final short questionnaire that I will be sending you in 1 month's time.

Once again, thank you so much for your participation, I really appreciate it!!

Performance-Approach Inducing Task Instructions – Time 2

Prepare an information sheet and consent form on each desk before start of session. Also put 2 pencils, an eraser, a calculator, and some rough paper on each desk.

First of all I would like to thank you for attending this session. The session will take 1 hour. Could you all switch off you mobile phones please?

First I would like you to read the paper with the title 'information sheet' found on your desks. This sheet is for you to keep.

Do you have any questions?

You will also find an informed consent form on your desk. Could you kindly read and sign this form please? I am required to provide this information and ask you to sign a consent form for ethical purposes. This is required by the Loughborough University Ethical Advisory Committee. Once you have completed the forms could you all pass them forward please?

Count number of forms and check that all of them have been completed appropriately.

Hand out the answer sheets.

Hand out the numerical test booklets. Please don't open these booklets until I tell you to do so.

When all booklets have been handed out: Now please turn to page 2 of the booklet and follow the instructions while I read them aloud.

In this test you will be using facts and figures taken from the world of work, to assess your ability to interpret and evaluate numerical information. All the information you need will be provided for you in the form of various tables and charts.

For each question you are given five answers to choose from. One, and only one, of the answers is correct in each case.

Do not write anything on this question booklet, but indicate your answer by completely filling in the appropriate circle on your answer sheet.

You may wish to use a calculator for this test. Rough paper is also provided to help you with any working out.

Before you start there are some example questions to complete. Look at the example questions on page 8. Now, in your own time and using just the statistical information provided at the top of the page, choose what you think is the best answer for each question from the five alternatives given. Mark your choices in the Examples Section on your answer sheet.

When all have finished the examples, say:

The answer to Example 1 is A, Year 1.

The answer to Example 2 is D, £144, 000.

The answer to example 3 is B, £5, 497, 000.

Are there any questions? If you are not clear about what you have to do, ask me now as I can't answer any questions once the test has started.

Now, I have handed out a short questionnaire which I would like you to complete. Please don't forget to print your name at the top of this questionnaire.

Ask if everyone's ready and when they are...

Could you all pass your questionnaires forward please?

Count questionnaires and make sure that everyone has written their names on it.

Now could you look back at page 2 and follow whilst I read out the reminder points please?

- The test consists of 12 questions and you will have 15 minutes in which to do them.
- Try to work quickly but accurately.
- Fill in completely the appropriate circles on the answer sheet.
- Do not make any marks on this booklet.
- Use the rough paper provided for your working out.
- Fully erase any answer you wish to change.
- If you are not sure of an answer mark your best choice, but avoid simply guessing.

Are there any final questions? Please note that this time the first test question is Question 19. So please start using your answer sheets from Question 19.

This time I shall be comparing your performance on the test to the test norms and I shall also be comparing your performance on this test with the performance of other participants of my research. I would like to remind you that the results of this test will ONLY be used for the purpose of this research and will not have any influence whatsoever on your course since the results will be kept in very strict confidentiality.

Now turn over to page 9 and begin.

Start stopwatch on the word 'begin', and note the start time on a piece of paper. Walk around after 2 minutes and again after 10 minutes. After exactly 15 minutes say:

Please stop now. Pencils down. Close your booklets.

Collect in all the materials, booklets, answer sheets, pencils and erasers. Check that there are 10 booklets and 10 answer sheets, and check that all answer sheets have names on them.

Could you kindly complete this short questionnaire I'm handing out please?

Hand out post-test questionnaires.

Please don't forget to write your names down on the first page of the questionnaires.

Has everyone completed the questionnaire? Can you all pass your questionnaires forward please?

So that was the first half of the session. We're going to go through exactly the same process for the second aptitude test.

Hand out the answer sheets.

Hand out the verbal test booklets. Please don't open these booklets until I tell you to do so.

When all booklets have been handed out: Now please turn to pages 2 of the booklet and follow the instructions while I read them aloud.

In this test you will be asked to reason with written information drawn from the world of work. The test consists of a series of passages, each of which is followed by several statements. Your task is to assess whether each statement is either "true", "false", or "cannot say" in relation to the passage which preceded it.

So having read a passage and the statements which follow, you would;

fill in *circle A* if you think the statement is *true* in relation to the passage,

fill in *circle B* if you think the statement is *false* in relation to the passage,

fill in *circle C* if you *cannot say* whether the statement is true or false in relation to the passage.

You should assess each statement *only in relation to the passage which preceded it* and not in relation to any personal views or knowledge you might hold.

Do not write anything on this question booklet, but indicate your answer by completely filling in the appropriate circle on your answer sheet.

Before you start there are some example questions to complete. Look at the passage page 11. Now, in your own time, read the passage and assess the statements which follow, according to the rules above. Choose what you think is the best answer for each statement, A, B, or C and mark your choice in the Examples Section on your answer sheet.

Hand out self-efficacy questionnaires. When everyone has finished the examples:

The answer to Example 1 is A, the statement is true given the information in the passage.

The answer to Example 2 is C, you cannot say from the information in the passage.

The answer to Example 3 is A, the statement is true given the information in the passage.

The answer to Example 4 is C, you cannot say from the information in the passage.

Are there any questions? If you are not clear about what you have to do, ask me now as I can't answer any questions once the test has started.

I have handed out a short questionnaire which I would like you to complete. Please don't forget to print your name at the top of this questionnaire.

Ask if everyone's ready and when they are...

Could you all pass your questionnaires forward please?

Count questionnaires and make sure that everyone has written their names on it.

Now could you look back at page 2 and follow whilst I read out the reminder points please?

- The test consists of 24 questions and you will have 14 minutes in which to do them.
- Try to work quickly but accurately.
- Fill in completely the appropriate circles on the answer sheet.
- Do not make any marks on this booklet.
- Fully erase any answer you wish to change.
- If you are not sure of an answer mark your best choice, but avoid simply guessing.

Are there any final questions?

Please note that this time the 1st test question is Question 29. So please start using your answer sheets from Question 29.

Again, I would like you to keep in mind that this time I shall be comparing your performance on the test to the test norms and I shall also be comparing your performance on this test with the performance of other participants of my research. I would like to remind you that the results of this test will ONLY be used for the purpose of this research and will not have any influence whatsoever on your course since the results will be kept in very strict confidentiality.

Now turn over to page 12 and begin.

Start stopwatch on the word 'begin', and note the start time on a piece of paper. Walk around after 2 minutes and again after 10 minutes. After exactly 14 minutes say:

Please stop now. Pencils down. Close your booklets.

Collect in all the booklets and answer sheets. Check that there are 10 booklets and 10 answer sheets, and check that all answer sheets have names on them.

Could you kindly complete this short questionnaire I'm handing out please?

Hand out post-test questionnaires.

Please don't forget to write your names down on the first page of the questionnaires.

Collect calculators.

Has everyone completed the questionnaire? Can you all pass your questionnaires forward please?

Collect questionnaires, pencils and erasers. Check that they all wrote their names on the questionnaire.

Have you got any questions?

Once I've scored all the tests and questionnaires I will send you all a feedback pack by email. I would be very happy to provide you with a personal feedback session if you like. So if you would like to attend a personal feedback session could you kindly drop me an e-mail and I'll make sure that I set one up for you. I will also be sending you a final short questionnaire by e-mail in about 1 month's time and I would really really appreciate it of you could complete it and send it back by e-mail. It's very very short, but it is also very important for the research that I am doing.

Also, the draw for the three £50 cash prizes will take place in December or January and you'll obviously be informed if you've won. Please note that your name will be put into the draw once you've completed the final short questionnaire that I will be sending you in 1 month's time.

Once again, thank you so much for your participation, I really appreciate it!!

APPENDIX J E-MAIL WITH TIME Q2 QUESTIONNAIRE (Experimental Participants) Hi {name of participant}

How are you? I hope you had a fantastic Christmas and New Year! I've attached the 2nd questionnaire as I had mentioned in the experiment. I would really really appreciate it if could complete it and send it back to me, it should only take you about 3 minutes to complete! I'll be getting back to you about the feedback in a few weeks time if that's ok.

Thanks {and good luck in your exams - if participant was a student}!

Debbie

APPENDIX K E-MAIL SENT PRIOR TO TIME 2 EXPERIMENT

Hi {name of participant}

How are you? I hope all is well. I'm e-mailing you regarding part 2 of the experiment. I've put a list of times and dates when I will be carrying out experimental sessions next week. If you could kindly choose a slot and send me an e-mail with the time and day you'd like to attend I would really appreciate it!

This session will be exactly the same as the first session, only this time you'll be completing the second half of the verbal and numerical tests. I will then score the tests over the holidays and after your exams I'll provide you with feedback on the tests and give you tips on how to excel in these tests. I can also provide you with feedback on the questionnaires if you'd like that.

Date Time

{list of dates} {list of times}

If you cannot attend on any of the dates mentioned above please don't hesitate to contact me and I'll be very happy to organise an extra session.

Once again, thank you so much for your help and support, I really appreciate it!

I look forward to hearing from you.

Kind Regards

Debbie

APPENDIX L TIME E2 POST-TEST QUESTIONNAIRE

NAME:								
Your experience on this kind of task								
2.) Have you completed any other verbal aptitude test (other than that in the first part of this experiment)?								
Yes	No	Do not remember						
2.) Does your work involve using the skills required to perform well on the test just completed?								
Frequently 🗌	Sometimes 🗌	Rarely	Never 🗌					
3.) Have you practiced this kind of task since the first experiment?								
Frequently 🗌	Sometimes 🗌	Rarely	Never 🗌					

How much effort did I invest in the task? Please circle the most appropriate answer keeping in mind the task that you have just completed.

In solving the preceding task I invested:

- 1: very, very low mental effort
- 2: very low mental effort
- 3: low mental effort
- **4:** rather low mental effort
- **5**: neither low nor high mental effort
- 6: rather high mental effort
- 7: high mental effort
- 8: very high mental effort
- **9:** very, very high mental effort

How I approached the task: The following items refer to the task that you have just carried out. Please answer these items keeping in mind the approach that you used for the task that you have just completed.

Some of the questions may look similar to each other but please do not worry about appearing to be consistent. Answer each question on its own merits.

Please circle **ONE** answer for each of the items below.

		Strongly Agree	Agree	Neither Agree Nor Disagree	Disagree	Strongly Disagree		
1	My aim was to completely master this task.	1	2	3	4	5		
2	I was striving to do well on this task compared to the other participants.	1	2	3	4	5		
3	My goal was to learn as much as possible.	1	2	3	4	5		
4	My aim was to perform well on this task relative to the other participants.	1	2	3	4	5		
5	My aim was to avoid learning less than the maximum possible about this task.	1	2	3	4	5		
6	My goal was to avoid performing poorly compared to others on this task.	1	2	3	4	5		
7	I was striving to understand the content of this task as thoroughly as possible.	1	2	3	4	5		
8	My goal was to perform better on this task than other participants.	1	2	3	4	5		
9	My goal was to avoid learning less than it was possible to learn about this task.	1	2	3	4	5		
10	I was striving to avoid performing worse than others on this task.	1	2	3	4	5		
11	I was striving to avoid an incomplete understanding of this task.		2	3	4	5		
12	My aim was to avoid doing worse than the other participants on this task.	1	2	3	4	5		
NAME:								
--	-------------	-------------	---------	--	--	--	--	--
Your experience on this kind of task								
3.) Have you completed any other numerical aptitude test (other than that in the first part of this experiment)?								
Yes	No	Do not reme	mber					
2.) Does your work involve using the skills required to perform well on the test just completed?								
Frequently 🗌	Sometimes 🗌	Rarely	Never 🗌					
3.) Have you practiced this kind of task since the first experiment?								
Frequently 🗌	Sometimes 🗌	Rarely	Never					

How much effort did I invest in the task? Please circle the most appropriate answer keeping in mind the task that you have just completed.

In solving the preceding task I invested:

- 1: very, very low mental effort
- 2: very low mental effort
- 3: low mental effort
- **4:** rather low mental effort
- **5:** neither low nor high mental effort
- 6: rather high mental effort
- 7: high mental effort
- 8: very high mental effort
- **9:** very, very high mental effort

How I approached the task: The following items refer to the task that you have just carried out. Please answer these items keeping in mind the approach that you used for the task that you have just completed.

Some of the questions may look similar to each other but please do not worry about appearing to be consistent. Answer each question on its own merits.

Please circle **ONE** answer for each of the items below.

		Strongly Agree	Agree	Neither Agree Nor Disagree	Disagree	Strongly Disagree
1	My aim was to completely master this task.	1	2	3	4	5
2	I was striving to do well on this task compared to the other participants.	1	2	3	4	5
3	My goal was to learn as much as possible.	1	2	3	4	5
4	My aim was to perform well on this task relative to the other participants.	1	2	3	4	5
5	My aim was to avoid learning less than the maximum possible about this task.	1	2	3	4	5
6	My goal was to avoid performing poorly compared to others on this task.	1	2	3	4	5
7	I was striving to understand the content of this task as thoroughly as possible.	1	2	3	4	5
8	My goal was to perform better on this task than other participants.	1	2	3	4	5
9	My goal was to avoid learning less than it was possible to learn about this task.	1	2	3	4	5
10	I was striving to avoid performing worse than others on this task.	1	2	3	4	5
11	I was striving to avoid an incomplete understanding of this task.	1	2	3	4	5
12	My aim was to avoid doing worse than the other participants on this task.	1	2	3	4	5

APPENDIX M EXPERIMENTAL FEEDBACK FORM TEMPLATE

Name: Test Scores:

Numerical Test Time 1: x/12Numerical Test Time 2: x/12

Verbal Test Time 1: x/24Verbal Test Time 2: x/24

Well done!! Keep practicing...it will boost your confidence!!

The average score on the verbal test at Time 1 was 15 and at Time 2 was 14 (the maximum score was 21 at Time 1 and 20 at Time 2).

The average score on the numerical test at Time 1 was 6 and at Time 2 was 7 (the maximum score was 11 at Time 1 and 11 at Time 2).

Irrespectively of how well you fared on these tests there's always room for improvement! Keep in mind that even a 1 mark difference in your score can make the difference between getting a job or not!

With aptitude tests it is extremely important to practice. Practice can make an **immense** difference to your score and greatly increase your chances of getting a job!

In order to do well on aptitude tests

- a) Make sure you know what you have to do before you start. If you do not understand the instructions ask the test administrator.
- b) Read the instructions carefully before the test starts. Reading through them quickly might make you overlook important instructions and cause silly mistakes (e.g. if you do not mark the answer as they ask you to your answers will be counted as wrong or invalid).
- c) Do not assume that instructions are the same as the last time you looked at them.
- d) Work as quickly and accurately as you can once you start the test. Every question you leave unanswered is a missed scoring opportunity.
- e) Check frequently to make sure that the question you are answering matches the corresponding space on the answer sheet.
- f) If you find a question difficult, leave it and go back to it if you have time.
- g) If you are uncertain about an answer enter your most reasoned choice (try not to simply guess it).
- h) If you have spare time go back and check your answers.
- i) Work as hard as you can throughout the test and go to the test with a positive attitude even if you've not done well on previous tests. Do not allow yourself to start off negatively since this will impact the way you work and therefore your results.
- j) Give your full concentration, you cannot afford to be distracted by anything or anyone!

(Tolley & Thomas, 1996).

When taking a real (as opposed to a practice) test

- a) Find out about as much as possible about the test in advance e.g. ask if you can get any example questions, whether it's power or speed test*, which types of test verbal, numerical, spatial.
- b) Make sure you get to the place of the test in good time so as not to get anxious.
- c) Inform the employer or organisation about any disability you may have so that they may make any necessary arrangements for you.
- d) Always take a watch!
- e) Listen very carefully to all instructions given by the test administrator.
- f) Do EXACTLY as you are told.
- g) Work carefully through any practise questions provided.
- h) When given the go ahead to begin read each test question carefully before answering it.
- i) Keep an eye on the time.
- j) Don't make assumptions especially during verbal tests.
- k) Do not hesitate to ask questions before a test in order to make sure you have understood correctly.
- I) Stop working immediately when told to do so.

*Speed tests normally consist of fairly straightforward questions and you need to try and answer as many questions as possible in the allocated time. Power tests on the other hand consist of a small number of complex questions. The latter are normally given at the professional or managerial level.

Practice makes perfect!! The most important thing to do with aptitude tests is to get lots of practice. Every little improvement greatly enhances your chance of getting the job!

Excelling on Numeracy Tests

In order to excel on numeracy tests make sure that you know the following very well since they frequently turn up in numeracy tests:

- a) weights and measures
- b) units of time
- c) addition, subtraction, multiplication, and division of fractions
- d) calculating areas of shapes
- e) calculating averages
- f) calculating percentages
- g) extracting information from line graphs, bar graphs, pie charts and statistical tables.

With numeracy tests, the more you can work things out without using a calculator the better it is because you learn to work much faster and many times speed is a very important factor in tests. So, for example, know your multiplication tables, learn how to compute averages and get comfortable working without a calculator as much as possible.

Some good websites to take a look at include:

- http://www.shldirect.com/practice_tests.html
- <u>http://www.kent.ac.uk/careers/psychotests.htm</u>
- <u>http://www.psychometric-success.com/aptitude-tests/aptitude-tests-introduction.htm</u>
- <u>http://www.assessmentday.co.uk/</u>
- <u>http://www.prospects.ac.uk/cms/ShowPage/Home_page/Applications_CVs_and_intervie_ws/Tests_and_exercises/Psychometric_tests/pleLagFgF</u>
- <u>http://www.practicetests.co.uk/</u>
- <u>http://www.aptitudeonline.co.uk/</u>

Also, if you're asked to attend an assessment day, you should consider taking a look at what 'in-tray tests/exercises' are and practising some of these. There are many practice tests on the internet, some useful websites include:

- <u>http://www.careers.ed.ac.uk/STUDENTS/Applications_Interviews/AssessmentCentres/in-tray%20exercise/intray_exercise.htm</u>
- <u>http://www.psychometric-success.com/assessment-centers/assessment-center-in-tray-exercise.htm</u>
- <u>http://www.kent.ac.uk/careers/interviews/intray.htm</u>

